



Canadian Journal of Cardiology 36 (2020) 1847-1948

Society Guidelines

The 2020 Canadian Cardiovascular Society/Canadian Heart Rhythm Society Comprehensive Guidelines for the Management of Atrial Fibrillation

Jason G. Andrade, MD, (Co-chair),^{a,b} Martin Aguilar, MD, PhD,^b Clare Atzema, MD, MSc,^c Alan Bell, MD,^c John A. Cairns, OBC, MD,^a Christopher C. Cheung, MD, MPH,^a
Jafna L. Cox, MD,^d Paul Dorian, MD, MSc,^c David J. Gladstone, MD, PhD,^c Jeff S. Healey, MD,^e Paul Khairy, MD, PhD,^b Kori Leblanc, ACPR, PharmD,^c M. Sean McMurtry, MD, PhD,^f
L. Brent Mitchell, MD,^g Girish M. Nair, MBBS, MSc,^h Stanley Nattel, MD,^b Ratika Parkash, MD,^d Louise Pilote, MD, MPH, PhD,ⁱ Roopinder K. Sandhu, MD,^f Jean-François Sarrazin, MD,^j
Mukul Sharma, MD, MSc,^k Allan C. Skanes, MD,¹ Mario Talajic, MD,^m Teresa S.M. Tsang, MD,^a Atul Verma, MD,ⁿ Subodh Verma, MD, PhD,^c Richard Whitlock, MD, PhD,^e D. George Wyse, MD, PhD,^g and Laurent Macle, MD, (Co-chair);^b

and Members of the Secondary Panel*

^a University of British Columbia, Vancouver, British Columbia, Canada; ^b Institut de Cardiologie de Montréal, Université de Montréal, Montréal, Québec, Canada; ^c University of Toronto, Toronto, Ontario, Canada; ^d Dalhousie University, Halifax, Nova Scotia, Canada; ^e McMaster University, Hamilton, Ontario, Canada; ^f University of Alberta, Edmonton, Alberta, Canada; ^g University of Calgary, Calgary, Alberta, Canada; ^h University of Ottawa Heart Institute, Ottawa, Ontario, Canada; ⁱ McGill University, Montreal, Quebec, Canada; ^j Institut universitaire de cardiologie et de pneumologie de Québec, Université Laval, Québec, Québec, Canada; ^k McMaster University, Population Health Research Institute, Hamilton, Ontario, Canada; ^l Western University, London, Ontario, Canada; ^m Montreal Heart Institute, University of Montreal, Montréal, Quebec, Canada; ^a Southlake Regional Health Centre, University of Toronto, Toronto, Ontario, Canada

ABSTRACT

The Canadian Cardiovascular Society (CCS) atrial fibrillation (AF) guidelines program was developed to aid clinicians in the management of these complex patients, as well as to provide direction to policy makers and health care systems regarding related issues. The most recent comprehensive CCS AF guidelines update was published

RÉSUMÉ

Le programme de lignes directrices de la Société canadienne de cardiologie (SCC) en matière de fibrillation auriculaire (FA) a été élaboré pour aider les cliniciens à prendre en charge ces patients complexes, ainsi que pour orienter les décideurs politiques et les systèmes de soins de santé sur des questions connexes. La dernière édition

experts on this topic with a mandate to formulate disease-specific recommendations. These recommendations are aimed to provide a reasonable and practical approach to care for specialists and allied health professionals obliged with the duty of bestowing optimal care to patients and families, and can be subject to change as scientific knowledge and technology advance and as practice patterns evolve. The statement is not intended to be a substitute for physicians using their individual judgement in managing clinical care in consultation with the patient, with appropriate regard to all the individual circumstances of the patient, diagnostic and treatment options available and available resources. Adherence to these recommendations will not necessarily produce successful outcomes in every case.

Received for publication August 21, 2020. Accepted September 5, 2020.

^{*}For a full listing of primary and secondary panel members, see Supplemental Appendix S1.

Corresponding author: Dr Jason G. Andrade, 2775 Laurel St, Vancouver, British Columbia V5Z 1M9, Canada. Tel.: +1-604-875-5069; fax: +1-604-875-5874.

E-mail: jason.andrade@vch.ca

The disclosure information of the authors and reviewers is available from the CCS on their guidelines library at www.ccs.ca.

This statement was developed following a thorough consideration of medical literature and the best available evidence and clinical experience. It represents the consensus of a Canadian panel comprised of multidisciplinary

in 2010. Since then, periodic updates were published dealing with rapidly changing areas. However, since 2010 a large number of developments had accumulated in a wide range of areas, motivating the committee to complete a thorough guideline review. The 2020 iteration of the CCS AF guidelines represents a comprehensive renewal that integrates, updates, and replaces the past decade of guidelines, recommendations, and practical tips. It is intended to be used by practicing clinicians across all disciplines who care for patients with AF. The Grading of Recommendations, Assessment, Development and Evaluations (GRADE) system was used to evaluate recommendation strength and the quality of evidence. Areas of focus include: AF classification and definitions, epidemiology, pathophysiology, clinical evaluation, screening and opportunistic AF detection, detection and management of modifiable risk factors, integrated approach to AF management, stroke prevention, arrhythmia management, sex differences, and AF in special populations. Extensive use is made of tables and figures to synthesize important material and present key concepts. This document should be an important aid for knowledge translation and a tool to help improve clinical management of this important and challenging arrhythmia.

complète des lignes directrices de la SCC en matière de FA a été publiée en 2010. Depuis lors, des mises à jour périodiques ont été publiées, traitant de domaines en évolution rapide. Cependant, en 2020, un grand nombre de développements s'y étaient ajoutés, couvrant un large éventail de domaines, ce qui a motivé le comité à créer une refonte complète des lignes directrices. L'édition 2020 des lignes directrices de la SCC en matière de FA représente un renouvellement complet qui intègre, met à jour et remplace les lignes directrices, les recommandations et les conseils pratiques des dix dernières années. Elle est destinée à être utilisée par les cliniciens praticiens de toutes les disciplines qui s'occupent de patients souffrant de FA. L'approche GRADE (Gradation des Recommandations, de l'Appréciation, du Développement et des Évaluations) a été utilisée pour évaluer la pertinence des recommandations et la qualité des résultats. Les domaines d'intérêt incluent : la classification et les définitions de la FA, son épidémiologie, sa physiopathologie, l'évaluation clinique, le dépistage de la FA, la détection et la gestion des facteurs de risque modifiables, l'approche intégrée de la gestion de la FA, la prévention des accidents vasculaires cérébraux, la gestion de l'arythmie, les différences entre les sexes et la FA dans des populations particulières. Des tableaux et figures ont été largement utilisés pour synthétiser les éléments importants et présenter les concepts clés. Ce document devrait représenter une aide importante pour l'intégration des connaissances et un outil pour aider à améliorer la gestion clinique de cette arythmie importante et difficile à traiter.

Atrial fibrillation (AF), the most common sustained cardiac arrhythmia, is associated with reduced quality of life (QOL), functional status, cardiac performance, and survival. The contemporary management of AF is centred on symptomatic improvement, diminution in morbidity and mortality (particularly the prevention of cardiomyopathy, and stroke/ systemic embolism), and reduction in AF-related emergency department (ED) visits or hospitalizations (Fig. 1).

The Canadian Cardiovascular Society (CCS) AF guidelines program was developed to aid clinicians in the management of these complex patients, as well as to provide direction to policy makers and health systems regarding the management of patients with AF. Beginning with the 1994 CCS consensus conference on AF, the CCS AF guidelines program has provided comprehensive guideline updates in 2004 and 2010, with focused updates on the basis of emerging evidence in 2012, 2014, 2016, and 2018.¹⁻⁸ The 2020 iteration of the CCS AF guidelines is a comprehensive renewal that integrates, updates, and replaces the past decade of guidelines, recommendations, and practical tips. It is intended to be used by practicing clinicians across all disciplines who care for patients with AF.

The 2020 comprehensive AF guidelines address the following topics:

- 1. Classification and Definitions
- 2. Epidemiology
- 3. Pathophysiology

- 4. Clinical Evaluation
- 5. Screening and Opportunistic AF Detection
- 6. Detection and Management of Modifiable Risk Factors
- 7. Integrated Approach to AF management
- 8. Stroke Prevention
- 9. Arrhythmia Management
- 10. Sex Differences
- 11. AF and Special Populations

Preamble and Guideline Development Methodology

This document was developed in accordance with CCS best practices and the Grading of Recommendations, Assessment, Development, and Evaluation (GRADE) approach.⁹ The primary panelists developed the scope of the document, identified topics for review, performed the literature review, evaluated the quality of the evidence, and drafted the recommendations. A systematic search was performed to identify relevant studies within each topic, including systematic reviews and metaanalyses. Draft recommendations were presented, reviewed, and refined by the primary panel. Final review was performed by the primary and secondary panels, with each recommendation achieving more than 90% agreement.

Strength of recommendations and quality of evidence were evaluated according to the GRADE approach.



Figure 1. General overview of the management of atrial fibrillation (AF). CHADS-65, Canadian Cardiovascular Society algorithm; ED, emergency department; OAC, oral anticoagulation; LV, left ventricular; QOL, quality of life.

Evidence derived from randomized clinical trials (RCTs) was initially estimated as high-quality evidence, with observational evidence initially deemed low-quality evidence. These estimates were further refined through detailed appraisal into 4 categorical grades ("high," "moderate," "low," and "very low") after consideration of the risk of bias, confounding, consistency of results, directness of evidence, precision, publication bias, magnitude of effect, and dose-response gradient. After this evaluation, the strength of recommendation was determined by considering the balance between desirable and undesirable effects (ie, risk-benefit), confidence in the magnitude of effect (quality of evidence), patient values and preferences, as well as resource considerations. The strength of recommendation was graded as "strong" (the desirable effects outweigh the undesirable effects, and therefore, most individuals will be best served by the recommended course of action) or "weak" (the desirable effects probably outweigh the undesirable effects, indicating that there is a need to consider the individual patient's clinical details, circumstances, values, and preferences).

Peer review of the guideline document was provided by external content experts, patient partners, and the CCS Guidelines Committee (outlined in the *Acknowledgements* section). The final draft was presented and approved by the CCS Executive Committee. For the constitution and roles of the primary and secondary panels, systematic review strategy, methods for formulating the recommendations, and evidence see the Supplementary Material and www.ccs.ca.

1. Classification and Definitions

1.1. Classification on the basis of clinical pattern of AF

AF pattern is defined on the basis of clinical assessment of episode persistence. These patterns have been used to characterize the severity of disease, define patient populations in clinical trials, and are used to form the basis of therapeutic recommendations regarding pharmacological and invasive arrhythmia management.¹⁰

Four main clinical patterns of AF have been described. Paroxysmal AF is defined as a continuous AF episode lasting longer than 30 seconds but terminating within 7 days of onset. Persistent AF is defined as a continuous AF episode lasting longer than 7 days but less than 1 year. "Longstanding" persistent AF is defined as continuous AF \geq 1 year in duration, in patients in whom rhythm control management is being pursued. Permanent AF is defined as continuous AF for which a therapeutic decision has been made not to pursue sinus rhythm restoration. The mode of termination (spontaneous vs pharmacological/electrical cardioversion) does not influence the classification.

In many patients, AF progresses from short episodes of self-terminating paroxysmal AF to more frequent exacerbations of longer-lasting persistent AF. In the event that both paroxysmal and persistent episodes are present, the classification should be defined on the basis of the predominant AF pattern.

Although valuable, there are limitations to classifying AF patterns by clinical assessment. First, the distinction between paroxysmal and persistent AF is often inaccurate, because clinical assessment often underestimates the temporal persistence of AF/AF burden compared with long-term electrocardiogram (ECG) monitoring.¹¹⁻¹³ Second, although the AF patterns have been associated with adverse outcomes (eg, heart failure [HF], stroke, and death), there remains uncertainty regarding their independent role of these clinical patterns to predict response to therapy (eg, antiarrhythmic drugs or catheter ablation).¹³⁻¹⁷

1.2. Classification on the basis of pathophysiological pattern of AF

The likelihood of developing AF varies across physiological and pathological states. Despite similar clinical patterns the mechanisms underpinning AF vary substantially between patients. Within this context AF may be considered "primary" if the AF represents an established pathophysiological process or "secondary" if caused by a self-limited or acutely reversible precipitant (Figs. 2 and 3A).¹⁸ "Primary AF" should not be considered analogous to the antiquated term, "lone AF," which previously defined AF without known cause.¹⁹ Common causes of secondary AF include surgery, sepsis, acute myocardial infarction (MI), thyrotoxicosis, or acute pulmonary disease. Secondary AF can be further dichotomized on the basis of the underlying cardiac substrate and risk for AF recurrence into "reversible" or "provoked" AF.²⁰ Specifically, "reversible AF" defines AF that occurs solely secondary to an acute illness, with little to no abnormal underlying substrate and therefore limited future risk of AF. In contrast, "provoked AF" represents AF that is unmasked by the acute illness, occurring in patients with significant abnormal underlying substrate, and therefore ongoing risk for AF recurrence (Fig. 3, B and C). Examples of the former category include patients with hyperthyroidism or alcohol intoxication ("holiday heart") in the absence of previous heart disease or risk factors, and the latter would include patients developing AF after mitral valve surgery or in the context of a chronic obstructive pulmonary disease (COPD) exacerbation. These concepts are explored in greater depth in section 8.3.6.

1.3. Valvular and nonvalvular AF

The term, "nonvalvular AF" (NVAF) dates back to the late 1970s and was used interchangeably with nonrheumatic AF. This early distinction was on the basis of the observation of the high risk of stroke/systemic embolism associated with severe mitral stenosis. More recently this distinction was used to define candidacy for participation in the landmark phase III trials in which non-vitamin K direct-acting oral anticoagulants (DOACs) were compared with vitamin K antagonists (VKAs) for stroke prevention in patients with NVAF.²¹⁻²⁵ Although previous iterations of the CCS AF guidelines have considered rheumatic mitral stenosis, mitral valve repair, mechanical heart valves, and bioprosthetic heart valves to constitute valvular heart disease, the definition of "valvular AF" has continued to evolve on the basis of emerging evidence.^{6,26-28,884} The current definition of "valvular AF" is limited to AF in the presence of any mechanical heart valve, or in the presence

of moderate to severe mitral stenosis (rheumatic or nonrheumatic).

2. Epidemiology

2.1. Incidence and prevalence

AF is the most common sustained arrhythmia encountered in clinical practice.²⁹ Current evidence suggests that the prevalence of AF is 1%-2% in the general population, and increases significantly with age (< 1.0% up to 50 years of age, to 4% at 65 years, and 12% of those 80 years of age or older).^{29,30} Although the incidence has been relatively stable over time (approximately 28 per 1000 person-years), the overall prevalence of AF is increasing because of changing population demographics (eg, from 41 cases per 1000 in 1993 to 85 cases per 1000 in 2007).^{29,31}

However, the true prevalence of AF is likely to be substantially higher than 1-2%, as these historical estimates were derived from populations with AF diagnosed using ECG, and did not routinely account for patients with paroxysmal AF (which is estimated to be approximately two-thirds of the AF population) or patients with silent AF.³²⁻³⁵ When factoring in patients with paroxysmal and silent AF, the prevalence of AF increases from 500,000 to nearly 1 million Canadians.^{36,37}

2.2. Morbidity and mortality

Although rarely acutely life-threatening, AF is associated with significant impairments in functional capacity and health-related quality of life (HRQOL), as well as with an increased morbidity and mortality. These impairments have been noted across multiple HRQOL domains, with a magnitude comparable or worse than that observed in patients with HF or who are on long-term hemodialysis.³⁸⁻⁴² Even in the absence of perceived symptoms, AF patients objectively experience reduced global life satisfaction.³⁸

AF is independently associated with a 1.5- to 4-fold increased risk of mortality, which is predominantly due to increased risk of thromboembolic events and ventricular dysfunction.^{31,40,43-46} Nonanticoagulated patients with AF have a 3- to 5-fold increased risk of stroke, which are generally more severe (greater resource utilization, long-term disability, and mortality) and more recurrent than strokes unrelated to AF.^{43,47-50} To date, the only therapeutic intervention that has been consistently and definitively shown to improve survival in the AF population is the use of oral anticoagulation (OAC; see section 8).^{51,52} Strategies targeting modifiable cardiovascular risk factors and relevant comorbid conditions offer potential opportunities to further improve survival (see sections 3 and 6).

2.3. Health care resource utilization

The economic burden of AF care is substantial. A significant proportion of AF health care expenses are attributed to the direct costs associated with hospitalization and the provision of acute care.^{29,53-59} In Canada, AF resulted in 8815 same-day procedures, 76,964 ED visits, and 64,214 acute care admissions (25,892 with AF as the principal diagnosis and



Figure 2. A conceptual model of the life cycle of atrial fibrillation (AF). Early in its course AF is a disease of focal triggers that can be genetically determined. As patients grow older the contribution of abnormal substrate predominates, including nonmodifiable substrate related to age, sex, and genetically determined factors, modifiable substrate related to reversible risk factors (eg, hypertension), as well as substrate induced by the AF episodes themselves. In this model, control of the arrhythmia and therapies directed at cardiovascular risk reduction are complementary for control of the arrhythmia.

38,222 with AF as a comorbid diagnosis) in the 2007-2008 fiscal year. The annual direct cost of AF care adjusted to 2020 Canadian dollars (CAD\$) was \$956 million: \$66 million for ED visits with AF as the principal diagnosis and \$20 million with comorbid AF; \$204 million for hospitalization with AF as the principal diagnosis and \$634 million with comorbid AF; and \$32 million for AF-related day procedures.⁵⁸ On a per-patient basis, the excess annual direct cost of AF has been estimated to be \$16,944-\$19,529 (adjusted 2020 US dollars).^{55,56} In addition to these direct costs, the annual indirect costs (eg, days of work missed because of illness) have been estimated to be \$3082 higher for AF patients compared with those without AF.⁵⁷

It is important to recognize that the cost of care is not uniform across the spectrum of AF. Specifically, the annual inpatient and outpatient direct costs are more than twice as high for patients with "primary AF" compared with patients with "secondary AF" (see section 1.2).⁵⁷

3. Pathophysiology and Risk Factors

AF is a complex and multifaceted condition ranging from an isolated electrophysiological disorder or, more commonly, a manifestation or consequence of other cardiac and noncardiac pathologies (Table 1, Fig. 2).¹⁹ AF generally results from a combination of focal ectopic activity and reentry.^{29,60} Ectopic atrial foci arise from perturbations that cause cells to spontaneously depolarize, either secondary to enhanced automaticity or, more frequently, to triggered activity from afterdepolarizations. Discrete abnormalities in Ca²⁺ handling have been identified as centrally involved in afterdepolarization generation in paroxysmal and persistent AF, as well as postoperative AF (POAF).^{61,62} There is emerging evidence that inflammatory signalling plays a key role in promoting afterdepolarization generation, as well as other components of AF pathophysiology.⁶³ These repetitive rapid discharges predominantly originate from the pulmonary veins (PVs), which are a vulnerable region for triggered activity and micro reentry due to the shorter action potential duration, lower resting membrane potentials, and nonuniform myofibril arrangement.⁶⁴ When triggered, AF can be maintained by sustained rapid firing of focal impulses that disorganize into fibrillatory waves at their periphery or, in most cases, AF perpetuating reentry. Reentry requires specific conditions for initiation and maintenance. Although reentry is not sustained in the normal atrium, the presence of a vulnerable substrate can perpetuate AF through electrical heterogeneity (eg, regional differences in resting membrane potentials, refractory periods, action potential duration, and conduction velocities). In addition, conduction abnormalities can promote reentrant activity and stabilize reentrant circuits by creating functional barriers that allow recovery of tissue excitability. Structural abnormalities such as atrial fibrosis promote reentry through localized conduction slowing and structural conduction barriers, with atrial chamber dilatation promoting reentry through maintenance of the balance between rotor formation and rotor annihilation.

Recently, there has been a renewed focus on the contribution of modifiable cardiovascular risk factors to the causation of AF, because an improved understanding of this relationship is key to providing effective personalized primary and secondary prevention measures.^{29,65,66} Although the precise mechanistic links between risk factors and AF occurrence remain somewhat uncertain, information available from the literature provides many potential insights.²⁹ Hypertension, the most significant population-attributable modifiable risk factor for AF, causes activation of the sympathetic and renin-angiotensin-aldosterone systems as well as structural and electrophysiological atrial remodelling that enhances AF susceptibility. Diabetes mellitus promotes AF via structural and autonomic remodelling. Tobacco use promotes AF through a combination of the direct effects of nicotine on the atrium (eg,



Figure 3. Secondary atrial fibrillation (AF). (**A**) Components of AF susceptibility include the baseline modifiable substrate, nonmodifiable substrate, and substrate induced by AF triggers (as outlined in Fig. 2), as well as new substrate/transient triggers induced by the reversible event (eg, cardiac surgery). A reversible event might only transiently induce substrate or lead to new permanent substrate. In this model, if the sum of the substrate allows the trigger to reach threshold then an AF event might occur. Whether or not the AF recurs is a function of the baseline substrate and any new permanent substrate added by the trigger event. (**B**) "Reversible AF" defines AF that occurs solely secondary to an acute illness, with little to no abnormal underlying substrate and, therefore, limited future risk of AF. (**C**) In contrast, "provoked AF" represents AF that is unmasked by the acute illness, occurring in patients with significant abnormal underlying substrate and, therefore, ongoing risk for AF recurrence. ANS, autonomic nervous system; HTN, hypertension; OSA, obstructive sleep apnea.

altered atrial conduction and refractoriness) along with structural remodelling, inflammation, and oxidative stress. Alcohol, when consumed in excess, promotes AF through the induction of arrhythmia triggers (increased sympathetic activity/impairment of vagal tone) as well as atrial fibrosis (from the direct toxic effects of alcohol metabolites). Obesity promotes AF through weight-related structural (changes in atrial dimensions and interstitial fibrosis) and electrophysiological remodelling (conduction slowing and shortening of the effective refractory period), autonomic dysfunction, and inflammation. Obstructive sleep apnea (OSA) promotes AF acutely through strongly negative intrathoracic pressures leading to increased venous return (AF-promoting left atrial [LA] volume loading) and hypoxia-induced pulmonary vasoconstriction. Chronic OSA induces electrical and structural remodelling of the atria, autonomic dysregulation, oxidative stress, and inflammation. Regular exercise protects against AF by combating risk factors like obesity and metabolic syndrome but sustained intense exercise might promote AF occurrence (see section 11.3).

The relationship between key risk factors and AF are outlined in Table 2.

4. Clinical Evaluation

The purpose of the initial evaluation of a patient with AF is to establish the magnitude and severity of symptoms attributable to AF, identify the underlying etiology and precipitants of AF, establish prognosis, and develop a therapeutic strategy for symptom relief and morbidity mitigation (Fig. 4).

4.1. AF history

A comprehensive AF history should include the date of first symptomatic attack as well as the date of first ECG documentation. For patients in AF at the time of assessment, the timing of onset for the current AF episode should be determined.

The duration and frequency of episodes should be used to establish the predominant pattern of AF (paroxysmal vs persistent; see section 1). Of note, in some cases the symptom evaluation might be insufficient for the determination of AF pattern and additional monitoring might be required.^{12,13}

The presence and nature of AF-related symptoms, their severity, and their effect on QOL should be determined (see section 4.3). Symptoms might be absent or manifest as palpitations, dyspnea, dizziness, weakness, fatigue, or chest pain.⁶⁷ In addition, it is important to elicit any history of regular palpitations because any supraventricular tachycardia (SVT) can lead to the development of AF, and ablation of the SVT might eliminate or substantially reduce the likelihood of recurrent AF (see section 11.7).^{68,69} Symptoms at the termination of AF episodes, such as presyncope or syncope, should be determined because significant sinus pauses might limit the use of rate- or rhythm-controlling medications and might require the use of permanent pacing or prompt early ablation.

Table 1. Risk markers and comorbid conditions associated with AF

Established risk factors

- Advancing age
- Male sex
- Hypertension
- HF with reduced ejection fraction
- Valvular heart disease
- Overt thyroid diseaseObstructive sleep apnea
- Obstructive
 Obesity
- Obesity
- Excessive alcohol intake
- Congenital heart disease (eg, early repair of atrial septal defect) Emerging risk factors
 - Prehypertension and increased pulse pressure
 - Chronic obstructive pulmonary disease
 - HF with preserved ejection fraction
 - Subclinical hyperthyroidism
 - Coronary artery disease
- Morphometric (increased height, increased birth weight) Potential risk factors
 - Familial/genetic factors
 - Tobacco use
 - Left atrial dilatation
 - LV hypertrophy
 - Inflammation
 - Diabetes
 - Pericardial fat
 - Subclinical atherosclerosis
 - Chronic kidney disease
 - Excessive endurance exercise
 - Electrocardiographic (atrial conduction delay, PR interval prolongation)

AF, atrial fibrillation; HF, heart failure; LV, left ventricular.

Precipitating factors ("triggers for AF episodes"), reversible causes, and coexisting cardiovascular risk conditions should be determined. These include modifiable cardiovascular risk factors and comorbid conditions, which if treated, might reduce or eliminate AF recurrence and improve the overall outcome of the patient, independent of AF (see section 6).^{32,70}

Past evaluations and treatments should be explored, including a record of all previous pharmacologic and non-pharmacologic AF interventions (eg, cardioversion and catheter ablation).

AF-related health care utilization should be documented, including a record of emergency department (ED) visits, hospital admissions, and cardioversions.

Risk factors for stroke (see section 8.1) and bleeding (see section 8.5.1) should be elicited.

The precise frequency, duration, and intensity of sports participation (current and previous) needs to be assessed carefully for all AF patients (see section 11.3).

In addition, the evaluation should include: a comprehensive review of all prescription, over the counter, and nonprescription medications; a social history with a focus on alcohol, tobacco, and recreational drug intake; and a family history of cardiac dysrhythmia or relevant risk conditions.

4.2. Investigations

In addition to a comprehensive physical examination several routine investigations are warranted for all patients who present with AF (Fig. 4):

- AF must be electrocardiographically documented, because the perception of "irregularly irregular" palpitations might be the result of a variety of arrhythmias, including atrial tachycardia, atrial flutter (AFL), premature atrial and/or ventricular contractions, or nonarrhythmic causes.
- An ECG is useful in AF and sinus rhythm to identify LA enlargement, left ventricular (LV) hypertrophy (LVH), preexcitation, conduction disease, or evidence of MI.
- A transthoracic echocardiogram should also be performed in all patients to identify LVH or systolic dysfunction, significant valvular or congenital heart disease (CHD), LA enlargement, and rarely, complications such as intracardiac thrombus. LA dimension provides important information about the likelihood of AF recurrence or progression to persistent AF, which can help guide therapeutic decision-making.
- Transesophageal echocardiography (TEE) might be indicated for more detailed evaluation of valvular or CHD, or for the exclusion of LA appendage (LAA) thrombus.
- Routine blood biochemistry should be obtained at the time of the initial AF evaluation. A complete blood count and coagulation studies will inform decisions about the use of antithrombotic medications. Serum electrolytes and creatinine should be measured, and renal function determined to guide drug dosing (eg, creatinine clearance [CrCl], see section 8.3.1). Liver

Risk factor	Role in AF	Mechanism
Hypertension	Hypertension is a strong and independent predictor of incident AF ³⁰ AF incidence is linearly related to BP, with increased risk noted even in the prehypertensive range (adjusted HR, 1.3 for systolic BP 130-139 mm Hg vs < 120 mm Hg) ^{119,834} Increased pulse pressure has been associated with an increased AF risk (adjusted HR, 1.25 per 20 mm Hg increase) ¹¹⁹	 Hypertension might induce AF through: Neurohormonal activation (sympathetic nervous system and the renin-angiotensin-aldosterone system) Structural remodelling (atrial fibrosis) Electrical remodelling (conduction
Diabetes	Diabetes has been associated with approximately a 1.5 times increased risk of AF ^{835,836} AF risk is independently associated with a longer duration of treated diabetes (3% increased risk for each additional year of diabetes) and with worse glycemic control (13% increased risk with each 1% increase in HbA1c; and 33% increased risk with each 1 mmol/L increase in fasting glucose) ⁸³⁶⁻⁸⁴⁰ Coexistence of AF and diabetes portends a worsened prognosis increasing	disturbances) Diabetes might induce AF through: • Structural remodelling (atrial fibrosis) • Electrical remodelling (conduction slowing) • Autonomic remodelling
Tobacco	 all-cause mortality, cardiovascular death, and HF⁸⁴¹ Tobacco use has been associated with increased risk^{30,842-844} Active smokers have a higher risk than former smokers Risk is linked to cumulative exposure (greatest risk in the highest tertile, > 675 cigarette-years) Continued tobacco use is associated with worse outcomes after catheter ablation⁸⁴⁵ Pooled analyses have not established a relationship hetween smokeless 	 Tobacco use might induce AF through: Electrical remodelling (altered atrial conduction and refractoriness [direct effects of nicotine]) Structural remodelling (atrial fibrosis) Inflammation and oxidative stress
Alcohol	 tobacco products and AF, and there are limited data regarding electronic cigarettes or secondhand smoke⁸⁴⁶ Acute paroxysms of AF have been reported after binge consumption (> 5 standard drinks on a single occasion; "holiday heart" syndrome)⁸⁴⁷⁻⁸⁴⁹ Heavy habitual consumption has been associated with risk of incident AF in a dose-dependent relationship (8% increase in incident AF with each additional drink per day)^{847,849,850} Patients with established AF who continue to consume alcohol have higher rates of progression (eg, paroxysmal to persistent AF), and experience 	 Alcohol use might induce AF through: Neurohormonal activation (increased sympathetic activity, impairment of vagal tone) Electrical remodelling (increase in inter- and intra-atrial conduction time, shortening of atrial effective refractory period) Structural remodelling (atrial fibrosis)
Physical inactivity	Habitual moderate-intensity exercise is inversely associated with risk of incident AF, with a graded inverse relationship shown between increasing fitness and AF (7% lower risk of incident AF with every additional metabolic equivalent achieved on exercise testing) ^{711,714,716}	 Exercise might protect against AF through: Potentiation of parasympathetic tone Improved cardiovascular risk factor profile (weight loss; improved BP, glucose, and lipids) Improved and the list function
Intense exercise	 Intense exercise practices have been associated with an increased incidence of AF, and a greater rate of AF recurrence after cardioversion or catheter ablation^{714,721} In this athletic population, AF paroxysms are more than 3 times as likely to occur in vagal contexts (postprandial, sleep, at rest) compared with healthy controls⁸⁵² 	 Vigorous activity might induce AF through: Acute catecholamine fluxes Parasympathetic enhancement (baroreflex enhancement and acetylcholine sensitization) Long-term structural remodelling (atrial dilatation with associated atrial fibrosis) Electrical remodelling (heterogenous
Obesity	Compared with people with a normal BMI (< 25), overweight (25-30) and obese (> 30) individuals are at increased risk for AF development, with a linear relationship between BMI and AF incidence (AF incidence increasing 3%-7% for each unit increase in BMI) ^{120,853} Weight gain has been associated with AF risk independent of BMI (34% increased AF with a 16%-35% weight gain, and 61% increased AF with > 35% weight gain) ¹²¹ Coexistence of AF and obesity confers worsened prognosis, increasing all-	 Obesity might induce AF through: Structural remodelling (atrial dilatation and fibrosis) Electrical remodelling (conduction slowing) Autonomic dysfunction Inflammation and epicardial fat infiltration Left ventricular diastolic dysfunction
Sleep apnea	 OSA is highly prevalent among those with AF, with a rate that is double that of the general population⁸⁵⁵ Patients with moderate to severe OSA have a three- to sixfold increased risk of developing AF⁸⁵⁶ CPAP reduces the risk of AF, supporting a causal role of OSA in AF pathogenesis^{125,855,857,858} Coexistence of AF and OSA confers worsened prognosis, with higher recurrence after cardioversion and ablation^{125,857,858} 	 Acute OSA might induce AF through: Left atrial volume loading Hypoxia-induced pulmonary vasoconstriction Long-term OSA might induce AF through: Structural remodelling (atrial dilatation and fibrosis) Electrical remodelling (conduction anisotropy) Autonomic dysregulation
Dyslipidemia Age	The association between dyslipidemia and incident AF is unclear ^{859,860} Age is one the most powerful predictors of incident AF The lifetime risk of developing AF for individuals 40-55 years of age has been estimated to be 22%-26% ^{36,861}	 Oxidative stress and inflammation Unclear Aging might induce AF through: Structural remodelling Electrical remodelling

Table 2. The relationship between AF and modifiable and nonmodifiable risk factors

Andrade et al. 2020 CCS/CHRS Guidelines on Management of AF

Table 2. Continued.

Risk factor	Role in AF	Mechanism
Sex	Sex is one of most powerful predictors of incident AF After adjustment, male sex is associated with a 1.5-fold risk of developing AF ^{31-34,36,861}	Male sex might predispose to AF through: • Electrical and structural differences (greater atrial dimensions)
Valve disease	 Valvular heart disease has been associated with a 1.8- to 3.4-fold increased risk for AF³⁰ AF risk is greatest for stenotic left-sided valvular lesions, with the highest risk among those with severe stenosis⁸⁶² AF risk is increased with complexity of rheumatic heart disease (15% isolated MR, 30% isolated MS, 50% mixed MR and MS, to 70% with mixed mitral and tricuspid valve disease)⁸⁶³ 	Valvular disease might induce AF through:Structural remodelling
Cardiomyopathy	 AF increases with severity of HF symptomatology (< 5% in NYHA class I, 10%-25% in NYHA class II-III, and > 50% in NYHA class IV)⁸⁶⁴ Although AF prevalence increases with worsening systolic dysfunction, AF is more common in patients with HFpEF compared with those with HFrEF⁸⁶⁴⁻⁸⁶⁶ AF prevalence varies depending on the cause of cardiopathy³⁰⁰: Prevalence > 20% with nonischemic, ischemic, and restrictive cardiomyopathies Prevalence 15%-20% with amyloid, hypertrophic, Keshan, and LV noncompaction Prevalence < 15% with Takotsubo, Chagas, and arrhythmogenic BV weighter several several	 HF might induce AF through: Structural remodelling Electrical remodelling (abnormal Ca²⁺-handling and increased triggered activity) Neurohormonal activation (increased sympathetic activity, impairment of vagal tone)
Thyroid dysfunction	 AF increases with decreasing levels of thyroid stimulating hormone^{867,868} RR, 1.1 with high-normal euthyroidism RR, 1.2-4 with subclinical hyperthyroidism RR, 3-6 with overt hyperthyroidism 	 Hyperthyroidism might induce AF through: Neurohormonal activation (increased sympathetic tone) Structural remodelling Electrical remodelling Promotion of PV automaticity/triggered activity (thyroid hormone)
Genetic factors	AF is more common in those with a family history of AF in a first-degree relative ⁸⁶⁹ Monogenic and polygenic inheritance has been described, with multiple susceptibility signals identified at the chromosome 4q25 locus ⁸⁷⁰⁻⁸⁷⁵	Genetic causes of AF typically result from cardiac ion channel alterations (promoting reentry or ectopic activity), alterations in cellular coupling, or increasing susceptibility to AF

AF, atrial fibrillation; BMI, body mass index; BP, blood pressure; CPAP, continuous positive airway pressure; HbA1c, hemoglobin A1c; HF, heart failure; HFpEF, heart failure with reduced ejection fraction; HR, hazard ratio; LV, left ventricular; MR, mitral regurgitation; MS, mitral stenosis; NYHA, New York Heart Association; OSA, obstructive sleep apnea; PV, pulmonary vein; RR, relative risk; RV, right ventricular.

function should be assessed before the prescription of potentially hepatotoxic medications, such as amiodarone. Lipid profile, hemoglobin A1c, and fasting blood sugar are recommended in most patients as part of a comprehensive cardiovascular risk assessment. Thyroid function should be assessed because hyperthyroidism remains an important treatable cause of AF.⁷¹ In select cases N-terminal pro-B-type natriuretic peptide (NT pro-BNP) and inflammatory biomarkers might aid in patient management.

- Ambulatory ECG monitoring might aid in the documentation of AF, the identification of other arrhythmias, the assessment of ventricular rate control, and to correlate patient symptoms with heart rhythm or heart rate. In selected patients long-term monitoring using external loop recorders, wearable patch monitors, and cardiac implantable electronic devices (CIEDs) might be useful.^{12,13,72,73} In addition, consumer-facing devices (eg, handheld or wearable ECG devices) may be used to document heart rate or cardiac rhythm, potentially providing symptom-rhythm correlation and a measure of AF burden.
- Sleep study or overnight oximetry should be performed in most patients because typical symptoms are less prevalent and screening questionnaires are less accurate in the AF population.
- Exercise testing can be used to supplement ambulatory monitoring in certain patients with exerciserelated symptoms and might be helpful to exclude significant ischemia before class Ic antiarrhythmic drug prescription.
- Invasive electrophysiological studies may be considered in patients who are candidates for catheter ablation of AF or with suspected SVT, which could be triggering AF (see sections 9.4 and 11.7).

4.3. Evaluation of the effect of AF on well-being, symptoms, and QOL

Traditional rhythm-based outcome parameters, such as freedom from AF recurrence, are insufficient to evaluate the clinical effect of AF.^{74,75} Although rarely lifethreatening, AF causes a greater degree of impairment of QOL than is generally appreciated. AF can cause moderate and sometimes severe distress, and substantially alter

Comple	te AF History
Est • The date of first symptomatic attack a • The duration and frequency of episodo • The presence and nature of symptoms • Symptom severity (including impact of	tablish: nd the date of first objective confirmation es (e.g. Dominant pattern of AF) s related to AF n quality of life)
Id 1. Risk factors / Comorbid conditions • See Table 1 and Section 6 2. Triggers for AF episodes • Stimulants • Alcohol • Sleep deprivation • Emotional Stress • Physical Exertion • Sleep/Nocturnal • Digestive	entify: 3. Reversible causes / AF secondary to: Cardiac or non-cardiac surgery Acute cardiac pathology Acute pulmonary pathology Acute infection Thyrotoxicosis Alcohol Pharmacologic agents (e.g. Ibrutinib) Supraventricular tachycardia Ventricular pacing
 Ramily history to identify potentially h Prior pharmacologic and non-pharmacefficacy, tolerance and adverse effects AF-related healthcare utilisation (e.g. l 	eview: eritable causes of AF cologic AF interventions, with a focus on ED visits, hospitalisations, and cardioversions)
Exa	mination
 Measure blood pressure and heart rat Determine patient height, weight, wai Comprehensive cardiopulmonary exar causes of AF (e.g. comorbid risk condii 	e st circumference, and body mass index (BMI) nination with a focus on determination of the tions, or secondary causes of AF)
Routine	Investigations
1. 12-lead electrocardiogram Document presence of AF Document PR, QRS, and QT intervals (i Identify potential causes of AF (e.g. sti infarction, ventricular hypertrophy, atr Identify factors that increase risk of pc disturbances, sinus node dysfunction, Identify high risk conditions (e.g. mani 2. Echocardiogram Evaluate ventricular size, wall thickness Evaluate left atrial size and left atrial v Exclude significant valvular or congeni 3. Laboratory Investigations Complete blood count Coagulation profile Serum electrolytes including calcium a Renal function Liver function	e.g. baseline prior to therapy initiation) ructural heart disease such as myocardial rial enlargement, congenital heart disease) tential therapeutics (e.g. conduction or repolarization abnormalities) fest pre-excitation) s, and function olume tal heart disease (e.g. atrial septal defects) and magnesium
Fasting lipid profile	

Figure 4. Evaluation of the atrial fibrillation (AF) patient. ED, emergency department.

everyday functioning.⁷⁶ Impaired QOL is primarily the result of AF-associated symptoms but can be influenced by AF therapies, illness perceptions, and patient factors such as anxiety or depression.^{74,75} A consistent and standardized assessment of the effect of AF on HRQOL is recommended to evaluate the clinical effect of AF and quantitatively assess the changes in well-being resulting from therapeutic interventions. Specifically, multidimensional HRQOL instruments can be used to determine if an intervention had a beneficial effect across all domains concurrently or if a benefit in one domain (eg, physical health) was offset by a negative effect in another (eg, mental health). As such, the assessment of patientreported outcomes with validated multidimensional instruments offers a relevant and complementary means to evaluate the consequences of AF and the effect of therapeutic interventions on patients' functional status and health.

To date, a large number of instruments have been used to evaluate HRQOL (Table 3). In broad terms, these instruments can be dichotomized into generic and diseasespecific questionnaires. Generic instruments, such as the EuroQol-5D (EQ-5D) and Short Form-36 Health Survey (SF-36), are used to assess valuations of health and functioning across a predefined set of health-related domains. Generic instruments have the advantages of extensive

Table 3. Instruments for the assessment of QOL and symptoms in patients with AF

Instrument	Туре	Domains	Administration	Comments
SF-36 ⁸⁷⁶	Generic QOL	Vitality Physical functioning Bodily pain General health perceptions Physical role functioning Emotional role functioning Social role functioning	Patient	Extensive validation Widespread clinical use Good for comparing between diseases Can be used to evaluate cost- effectiveness Insensitive AF
EQ-5D ⁸⁷⁷	Generic QOL	Mental health Mobility Self care Usual activities Pain/discomfort Apyrigu/degreesion	Patient	
AFEQT ⁸⁷⁸	Specific QOL	Symptoms Daily activities Treatment concerns	Patient	Good reliability Good validity Fair responsiveness (sensitive to change)
AFQOL ⁸⁷⁹	Specific QOL	Physical Psychological Secuel activity	Patient	Poor reliability Good validity Poor responsiveness
AFSS ⁸⁸⁰	Symptom	Symptom frequency, duration, and severity	Patient	r oor responsiveness
AFS/B ⁸⁸¹ EHRA ⁸⁸²	Symptom Classification	Symptom severity and burden Impact of AF on ability to complete daily activities	Patient and caregiver Caregiver	Simple classification
CCS SAF ⁷⁷	Classification	Effect of AF on ability to complete daily activities	Caregiver	Detailed classification Effect of symptoms on QOL

Reliability refers to the extent to which the instrument is free of measurement error (eg, internal consistency, test-retest reliability, and measurement error); validity refers to the extent to which the instrument measures the construct it purports to measure (eg, content, construct, and criterion validity); responsiveness refers to extent to which the measure can detect change over time.⁸⁸³

AF, atrial fibrillation; AFEQT, Atrial Fibrillation Effect on QualiTy-of-Life questionnaire; AFQOL, Atrial Fibrillation Quality of Life; AFS/B, Atrial Fibrillation Symptom Severity and Burden; AFSS, Atrial Fibrillation Severity Scale; CCS SAF, Canadian Cardiovascular Society Symptoms of Atrial Fibrillation; EHRA, European Heart Rhythm Association; EQ-5D, EuroQol-5D; QOL, quality of life; SAF, Symptoms of Atrial Fibrillation; SF-36, Short Form-36 Health Survey.

validation across a wide range of populations and health conditions but lack precision for assessing the effect of AF. Disease-specific instruments include symptomspecific scales (eg, the University of Toronto Atrial Fibrillation Severity Scale [AFSS]), and AF-specific QOL symptom scales (eg, the Atrial Fibrillation Effect on Quality-of-Life [AFEQT] questionnaire). These instruments do not provide for the ability to compare between disease states (eg, the HRQOL of AF patients relative to HF patients) but are more sensitive to changes in AF patients' health status (spontaneous or as a result of intervention).

The Severity of Atrial Fibrillation (SAF; Table 4) is a semiquantitative scale ranging from 0 (no effect of AF or its treatment on overall QOL and patient functioning) to 4 (resulting in a severe impairment of functioning and overall QOL).⁷⁷ A multicentre Canadian study has shown that the results of this scale correlate well with previously validated symptom scores and generic measures of QOL and that it can be easily applied by a variety of caregivers at the bedside.^{67,78}

Equally important is consideration of advanced age, frailty, cognitive impairment, and mood disorders. Identifying these factors might substantially affect anticipated QOL benefits of various AF therapies and can also help to identify potential challenges in adherence to therapy and self-care (see section 7.2).^{79,80}

RECOMMENDATION

1. We recommend that the initial evaluation of a patient with newly diagnosed AF include: a complete history and physical examination, a 12-lead ECG, a transthoracic echocardiogram, and basic laboratory investigations as outlined in Figure 4 (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on a comprehensive evaluation of patients with AF and a lower value on initial costs to the health care system.

2. We recommend that reversible and secondary causes of AF should be identified and treated (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a high value on evidence supporting a reduction in AF recurrence with treatment of the underlying condition or reversible precipitating event, such as infection, surgery, or thyroid disease.

Practical tip. Elimination of the precipitating event does not completely eliminate the possibility of AF recurrence, meaning patients with secondary AF should be regularly screened for arrhythmia recurrence (see section 11.5.3).

- 3. We suggest that all individuals with AF should be assessed for their sports and exercise history, with special attention to frequency, duration, intensity, and type of sport (Weak Recommendation; Low-Quality Evidence).
- 4. We recommend that patient-reported AF-related symptoms and QOL be assessed with validated instruments as part of the longitudinal management of patients with AF (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on the recognition that symptom control has a powerful influence on disability and health care resource utilization. The assessment of patient-reported outcomes with validated multidimensional instruments offers a relevant and complementary means to evaluate the effect of therapeutic interventions.

Practical tip. The caregiver-administered CCS SAF scale can be used to assess functional status and the symptomatic effect of AF, and the self-administered disease-specific AFEQT questionnaire can be used to assess the effect of AF on QOL. Consideration can also be made to assess QOL using generic instruments such as the EQ-5D questionnaire.

5. We recommend that patients with AF should be assessed for multimorbidity, frailty, cognitive impairment, dementia, and depression (Strong Recommendation; Low-Quality Evidence).

Practical tip. In this population a structured, integrated, multidisciplinary, patient-focused approach to care can be particularly beneficial because these factors might have an effect on treatment decisions and might improve adherence to therapy and self-care.

5. Screening and Opportunistic AF Detection

5.1. Opportunistic AF detection in the general population

AF screening initiatives have emerged with the availability of safe and effective stroke prevention therapy, well defined stroke risk schemes, and new technologies that have simplified AF monitoring. Because a large number of patients with AF might be asymptomatic, screening might provide an opportunity for AF detection with early initiation of stroke prevention therapy to reduce the risk of AF-related complications.

The effects of screening (eg, opportunistic case finding or systematic screening) has been examined in numerous studies in various populations (eg, general population or high-risk subgroups) and settings (eg, community, outpatient clinics, or inpatient).^{81,82} Taken together, the rate of new AF detection was 0.9% (95% confidence interval [CI], 0.7-1.1) across 23 prospective cross-sectional studies, yielding a number needed to screen (NNS) of 111 individuals to detect 1 patient with AF.⁸¹ However,

Table 4. The CCS SAF scale for the assessment of quality of life⁷⁷

CCS SAF score	Effect on quality of life
Class 0	Asymptomatic with respect to AF
Class 1	Symptoms attributable to AF have minimal effect on patient general quality of life:
	• minimal and/or infrequent symptoms, or
	• single episode of AF without syncope or heart failure
Class 2	Symptoms attributable to AF have a minor effect on patient general quality of life:
	 mild awareness of symptoms in patients with persister permanent AF, or
	 rare episodes (eg, less than a few per year) in patien with paroxysmal or intermittent AF
Class 3	Symptoms attributable to AF have a moderate effect on patient's general quality of life:
	 moderate awareness of symptoms on most days in p tients with persistent/permanent AF, or
	 more common episodes (eg, more than every few month or more severe symptoms, or both, in patients with paroxysmal or intermittent AF
Class 4	Symptoms attributable to AF have a severe effect on patien general quality of life:
	 very unpleasant symptoms in patients with persister paroxysmal AF, and/or
	 frequent and highly symptomatic episodes in patien with paroxysmal or intermittent AF, and/or
	• syncope thought to be due to AF, and/or
	 congestive heart failure secondary to AF

CCS-SAF, Canadian Cardiovascular Society Severity of Atrial Fibrillation Scale; AF, atrial fibrillation.

the prevalence of AF varies as a function of age and sex. When age and sex adjustment was performed the AF detection rate after a single time point screen was noted to be 1.44% (95% CI, 1.13-1.82) for those 65 years of age or older, but 0.41% (95% CI, 0.31-0.53) for those younger than 65 years in a multicountry, patient level meta-analysis of 141,220 individuals. This yields a NNS of 69 to identify 1 new AF case for patients 65 years of age or older.⁸² The choice of screening methodology, device, geographical region, or setting did not influence the AF detection rate.

There is some debate regarding the utility of systematic screening compared with opportunistic case-finding. A comparison of the 2 approaches was addressed in 2 RCTs performed in a United Kingdom primary care setting.^{83,84} In the first study 3001 patients older than 65 years of age were randomized to single time point nurse-led systematic screening vs prompted opportunistic case finding.83 Despite substantially more patients who underwent assessment in the systematic screening arm (73% vs 29%) there was no significant difference in the identification of new AF cases (systematic 0.8% vs opportunistic 0.5%; odds ratio [OR], 1.72; 95% CI, 0.68-4.37).⁸³ The Screening for AF in the Elderly (SAFE) study was larger in scale (50 primary care practices, 14,802 patients), including 2 interventional groups (single time point systematic screening and opportunistic case finding) as well as a usual care control group. The SAFE study showed that both interventional groups identified significantly more new AF cases compared with usual care (OR, 1.6; 95% CI, 1.1-2.3; P = 0.009);

 Table 5. AF Screening

Technique	Device	Method	Advantage/disadvantage	Accuracy
Pulse-based	Pulse palpation	Pulse irregularity detected using manual palpation	Time efficient High sensitivity Cost effective Easy to apply across health care settings	Sensitivity 84%-97% Specificity 69%-75%
	Blood pressure monitor (eg, M6 Comfort, Blood Pressure Analyzer, or WatchBP)	Algorithm on the basis of pulse irregularity	Easy to perform at home Time efficient High sensitivity Easy to apply across health care settings Easy to perform at home	Sensitivity 92-100% Specificity 86-97%
	Plethysmography (eg, finger probe, Apple Watch, smartphone photoplethysmograph app)	Algorithm on the basis of pulse irregularity	Capable of intermittent (Smartphone Camera) and extended monitoring (watch-based) Easy to perform at home Potentially costly	Sensitivity 92-96% Specificity 92-98% (lower in continuous watch-based application)
Rhythm- based	Single-lead ECG (eg, Kardia, MyDiagnostick, Zenicor-ECG)	Automated algorithm on the basis of RR irregularity	Capable of intermittent (symptom- based) monitoring ECG can be reviewed by health care professional	Sensitivity 94%-99% Specificity 92%-97%

The blood pressure monitor, M6 Comfort is manufactured by Omron Healthcare (Kyoto, Japan), and the Blood Pressure Analyzer and Watch BP are manufactured by Microlife USA, Inc (Clearwater, FL). The Apple Watch is from Apple Inc (Cupertino, CA). The Kardia is manufactured by AliveCor, Inc (Mountain View, CA), the MyDiagnostick by Applied Biomedical Systems BV (Maastricht, The Netherlands), and the Zenicor-ECG by Zenicor Medical Systems AB (Stockholm, Sweden).

ECG, electrocardiogram.

however, only opportunistic case finding was deemed to be cost-effective. $^{84}_{\ }$

There is concern that single time point screening offers too finite a window of observation, which is particularly problematic in the context of intermittent AF. This limitation of single time point screening was shown in the STROKESTOP study, in which twice-daily intermittent ambulatory ECG recordings were used over a 2- week period. The authors reported newly diagnosed AF in 3.0% of 75- and 76-year-old subjects, with repeated rhythm assessments leading to a fourfold increase in AF detection over single time point screening (initial single-lead ECG recording [SL-ECG]).⁸⁵

Although effective in AF case-finding, the utility of screening is dependent on population engagement. In the studies outlined above,⁸¹⁻⁸⁵ only 50%-75% of eligible patients participated in systematic screening. Alternative screening environments might leverage existing health initiatives (eg, at time of influenza vaccination)⁸⁶ or other health care professionals (eg, pharmacy).^{87,88}

An economic evaluation of an AF screening program requires consideration of several key factors: (1) participation rate; (2) rate of undiagnosed AF in a targeted population; (3) difference in AF detection between screening and usual care; (4) stroke risk in a targeted population; (5) stroke risk reduction and increase in bleeding risk from OAC; and (6) acceptable threshold for willingness to pay.⁸⁹ Intermittent opportunistic case finding was estimated to cost between 10€ and 108€ per patient (depending on the device, calculation method, and intensity of screening), which was lower than that of systematic screening.^{81,87,90} A 2-part decision model to evaluate short- and long-term costs and quality-adjusted life years (QALYs) as part of the Program for the Identification of "Actionable" AF (PIAAF) in the pharmacy setting showed an incremental cost per QALY gained of CAD\$7480 compared with no screening.9 When

different screening strategies were compared in a Canadian family practice study, screening with pulse check had the lowest expected costs (CAD\$202) and screening with (SL-ECG) had the highest expected costs (CAD\$222). The no-screening arm resulted in the lowest number of QALYs, whereas pulse check and SL-ECG resulted in the highest expected QALYs.⁹²

In a pooled analysis, the sensitivity of pulse palpation, blood pressure (BP) monitors, non-12-lead ECG, and smart phone applications were similar, but specificity was lower with pulse palpation. Because of the simplicity and ease of use, pulse palpation remains the cornerstone for AF detection. Downstream confirmatory testing with a 12-lead ECG should be pursued after pulse palpation, when screening is performed with non-rhythm-based devices (Table 5), or when the diagnosis of AF remains uncertain after rhythm acquisition.⁹²⁻⁹⁴

Additional studies with robust methodology are needed to identify optimal screening strategies, screening tools, population, and settings with health outcomes for different health care systems. An approach to AF screening is shown in Figure 5.

RECOMMENDATION

6. We recommend that opportunistic screening for AF should be conducted in people 65 years of age and older at the time of medical encounters (Strong Recommendation; Low-Quality Evidence).

Practical tip. Screening can be efficiently and costeffectively performed using opportunistic pulse checks during routine medical encounters; consideration can also be made to use rhythm-based devices (eg, SL-ECG rhythm device). 7. We recommend downstream confirmatory testing when AF is suspected but not documented, or when the documentation method does not include electrocardiographic rhythm acquisition (Strong Recommendation; Low-Quality Evidence).

Values and preferences. Confirmatory testing for AF is highly dependent on the type of AF. The effectiveness of the various AF screening methods depends on duration of monitoring (eg, single 12-lead ECG vs continuous monitoring). The use of new technologies for screening require validation before implementation.

5.2. Opportunistic AF detection in patients with a CIED

Advances in CIED (pacemaker or defibrillator) technology, which allow for long-term cardiac monitoring, have enabled detection of atrial high-rate episodes (AHREs) in patients with an atrial lead.^{50,95,96} Because of the possibility of false positive results, a careful review of the electrograms should be performed.⁹⁷ Observational and registry data have shown at least a doubling of thromboembolic risk when AHREs are present compared with their absence.^{50,98-101} However, the adjusted stroke rates observed with AHREs appear to be lower than reported among patients with similar risk profiles and clinically apparent AF.^{50,99} Although evidence suggests a link between AHREs and elevated stroke risk, many uncertainties exist regarding relevant stroke risk factors, optimal atrial rate cut-off, minimum significant AF burden, and importantly, if OAC reduces AHRE-related stroke (see section 11.1).^{50,95,96}

RECOMMENDATION

8. We recommend that AHREs be assessed at the time of CIED (loop recorders, pacemakers, or implanted cardioverter-defibrillators) interrogation (Strong Recommendation; Low-Quality Evidence).

5.3. AF detection after embolic stroke of undetermined source

AF accounts for a substantial proportion of acute ischemic strokes or transient ischemic attacks (TIAs) of known etiology, and might be responsible for a significant amount of strokes of undetermined etiology (embolic stroke of undetermined source [ESUS] or cryptogenic stroke).¹⁰²⁻¹⁰⁴ In those with ESUS and no known history of AF, antiplatelet therapy remains the treatment of choice.¹⁰⁵ This was demonstrated in recent RCT, where empiric initiation of rivaroxaban 15mg daily was not superior to aspirin in preventing recurrent



Figure 5. Approach to opportunistic atrial fibrillation (AF) screening. In general, AF screening should be performed in enriched populations, in whom the identification of AF is likely to change management. When screening is pursued pulse-based screening (eg, using manual palpitation) is a reasonable first step, however, electrocardiographic confirmation of AF is required. In contrast, AF might be diagnosed when screening is performed using rhythm-based devices (eg, single-lead electrocardiogram [ECG]).

stroke, and was associated with an almost threefold higher risk of bleeding.^{105,106} However, it is known that antiplatelet therapy is inadequate for the treatment of AF-associated strokes (see section 8.2.1).^{51,107,108} As such, ESUS patients require extensive AF screening to identify patients who would benefit from OAC for the secondary prevention of AF-associated thromboembolism.^{51,107,108}

Detection of AF is related to burden and improves with increasing intensity of monitoring.¹⁰⁹ In a systematic review of inpatient screening, the proportion of stroke patients newly diagnosed with AF on an admission 12-lead ECG was 7.7% (95% CI, 5.0-10.8), with an additional 7.0% (95% CI, 3.9-10.8) diagnosed after inpatient cardiac telemetry, and an additional 4.5% (95% CI, 2.7-6.7) diagnosed after in-hospital Holter monitoring.¹¹⁰

Prolonged postdischarge cardiac monitoring strategies (> 24 hours) have been evaluated in 3 RCTs for the purpose of AF detection after stroke.¹¹¹ A small United Kingdom study showed that a 7-day event monitoring uncovered new AF in 18% of patients compared with 2% in the control arm (P < 0.05%).¹¹² The 30-Day Cardiac Event Monitor Belt for Recording Atrial Fibrillation After a Cerebral Ischemic Event (EMBRACE) trial (N = 572) showed a 30-day monitor detected AF in 16% of patients compared with 3.2% who underwent a 24-hour Holter monitoring within 90 days (P < 0.001; NNS = 8), which led to nearly a doubling in the rate of OAC use.⁷² The Cryptogenic Stroke and Underlying Atrial Fibrillation (CRYSTAL AF) trial (N = 441) showed a sixfold higher rate of AF detection at 6 months using an implantable cardiac monitor compared with usual care.⁷³ At 3 years, the rate of AF detection was 30%, with most patients prescribed an OAC.¹¹³

Prolonged rhythm monitoring was associated with a 51% reduction in the risk of recurrent stroke/TIA,¹¹⁴ with the rates of recurrent cerebrovascular events or mortality being similar in patients who had known AF before their stroke and patients in whom AF was only diagnosed after their stroke.¹¹⁵ Substudy analyses from the EMBRACE and CRYSTAL AF trials showed that prolonged rhythm monitoring was cost-effective for the prevention of recurrent stroke in patients with ESUS compared with usual care.^{116,117}

Although there is no minimal AF burden threshold stipulated for which long-term OAC should be initiated for secondary prevention,¹¹⁸ in the EMBRACE study a steep increase in OAC use was observed for patients found to have evidence of at least 30 seconds of AF on monitoring.⁷² The authors acknowledged the lack of data to inform this matter but argue that the AF duration threshold at which to begin OAC might reasonably be lower for secondary stroke prevention compared with primary prevention. Further clinical trials are needed to determine the optimal duration and method of rhythm monitoring, the ideal population for extended rhythm monitoring, the efficacy of a prolonged monitoring strategy for the end point of recurrent stroke.

RECOMMENDATION

9. We recommend at least 24 hours of ambulatory ECG monitoring to identify AF in patients with nonlacunar ESUS (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places relatively high value on the fact that stroke might be the first manifestation of previously undiagnosed AF.

10. We suggest additional monitoring for AF detection (eg, prolonged external loop recorder or implantable cardiac monitoring, where available) be performed for selected older patients with nonlacunar ESUS in whom AF is suspected but unproven (Weak Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places high value on aggressively investigating selected patients with embolic stroke of unknown source. The main rationale is to improve the identification of patients who would have an evidence-based change in management aimed at preventing recurrent strokes (eg, switching from antiplatelet to OAC) if a clear diagnosis of AF is found. There are currently insufficient data to indicate what the minimum AF duration should be for OAC to be instituted and expert opinion varies widely, therefore treatment decisions should be individualized.

6. Detection and Management of Modifiable Risk Factors

Modifiable cardiovascular risk factors are well recognized contributors to the development and progression of AF.^{29,65,66} These established, emerging, and potential risk factors for AF have been summarized in section 3, and Tables 1 and 2. The risk of developing AF increases with the severity and number of modifiable cardiovascular risk factors (such as hypertension, diabetes mellitus, and obesity). In many cases this risk increase is linear within and between risk factors and might be apparent even within the established "normal range." For example, a systolic BP in the prehypertensive range (130-139 mm Hg) has been associated with a 28% higher adjusted risk of developing AF compared with a systolic BP < 120 mm Hg.¹¹⁹ Similarly, although the corrected AF incidence increases approximately 5% for each unit increase in body mass index (BMI),¹²⁰ weight gain has been associated with incident AF independent of BMI (34% increased AF with a 16%-35% weight gain, and 61% increased AF with a > 35% weight gain).¹²

In light of the breadth of data supporting an association between these modifiable cardiovascular risk factors and AF incidence, it has been suggested that the implementation of lifestyle modification and risk factor intervention could significantly decrease the incidence of AF. Unfortunately, the evidence to support targeted risk factor modification to prevent incident AF is limited, in part because of the observation that the absolute risk increase of incident AF from any individual risk factor is low. As such, two complementary approaches have been proposed. The first is to target preventive intervention in individuals who are at highest risk for AF occurrence. Identification of these individuals could be accomplished through the use of risk prediction models or, potentially, with artificial intelligence-enabled screening algorithms.¹²²⁻¹²⁴ When identified, these high-risk populations could be targeted for comprehensive risk factor modification. However, it is important to recognize that a focus on only those identified as at highest risk of developing AF would miss the opportunity to prevent most incident cases of AF because these occur in the large segment of the population typically considered to be at "lower risk." As such, decreasing the population effect of AF will require broad implementation of lifestyle modification strategies (eg, a focus on physical activity, and avoidance of alcohol and tobacco consumption) in addition to targeted intervention of traditional cardiovascular risk factors (including hypertension, HF, diabetes, OSA, and obesity) in those at highest risk of AF development.

In patients with established AF, the relationship between risk factor intervention and AF outcomes is well recognized. Several studies have shown that continuous positive airway pressure usage in patients with OSA is associated with a lower risk of AF recurrence compared with nonusage, with rates of recurrent AF comparable with those in patients without OSA.¹²⁵⁻¹²⁷ Many studies have shown that targeted weight loss interventions significantly reduce AF burden (number and cumulative duration of AF episodes) and AF symptom severity scores.¹²⁸⁻¹³⁰ The Long-Term Effect of Goal-Directed Weight Management on Atrial Fibrillation Cohort: A Long-Term Follow-Up Study (LEGACY) cohort study showed that patients with a weight loss of > 10% was associated with a sixfold increase in the likelihood of being arrhythmia-free over a 5-year follow-up compared with those with lesser degrees of weight loss.¹ Abed et al. showed that weight reduction with intensive risk factor management resulted in a significant improvement in AF-related QOL, AF symptom scores, and AF burden.¹²⁸ The Impact of Cardiorespiratory Fitness on Arrhythmia Recurrence in Obese Individuals with Atrial Fibrillation (CARDIO-FIT) study showed that a > 2Metabolic Equivalents (METs) improvement in cardiorespiratory fitness was associated with a significantly reduced AF burden compared with a gain of < 2 METs over longterm follow-up.¹³¹ AF burden reduction was proportional to the increase in cardiorespiratory fitness, with an adjusted reduction in AF recurrence of 10% for each MET gained (hazard ratio [HR], 0.90; 95% CI, 0.83-1.00).¹³¹ Rienstra et al. showed improved maintenance of sinus rhythm at 1 year with a strategy of cardiac rehabilitation, HF medication optimization, and aggressive BP control (75% maintenance of sinus rhythm on a 7-day Holter vs 63% in the control group; OR, 1.77, P = 0.042).¹³² The Aggressive Risk Factor Reduction Study for Atrial Fibrillation and Implications for the Outcome of Ablation (ARREST-AF) single-centre cohort study showed that patients who chose to undergo aggressive risk factor modification had better QOL and symptom control, a significant reduction in AF burden, and greater arrhythmia-free survival after catheter ablation compared with those who did not (OR, 4.8; 95%



¹defined as containing 14 g of alcohol; 44 mL (1.5 fluid oz.) of 80-proof liquor, 148 mL (5 fluid oz.) of wine or 355 mL (12 fluid oz.) of beer

Figure 6. The components of modifiable risk factor management for atrial fibrillation patients. In addition to the key modifiable risk factors and treatment targets outlined in the illustration, consideration should be given to manage coexisting diabetes and dyslipidemia consistent with contemporary guideline recommendations. ACE-I, angiotensin-converting enzyme inhibitor; AHI, apnea-hypopnea index; ARB, angiotensin II receptor blocker; BMI, body mass index; CPAP, continuous positive airway pressure; HbA1c, hemoglobin A1c; OSA, obstructive sleep apnea.

CI, 2.04-11.4; P < 0.001).⁶⁵ Taken together, these data suggest that outcomes might be improved by using a comprehensive management strategy that includes suppression of triggers (targeted by risk factor modification, antiarrhythmic drugs, and/or catheter ablation) and amelioration of arrhythmogenic substrate (risk factor modification). Treatment targets for modifiable risk factor intervention are presented in Figure 6.

RECOMMENDATION

11. In patients with established AF or at high risk of developing AF, we recommend a systematic approach to the identification of traditional modifiable cardiovascular risk factors and/or conditions associated with AF, with strict guideline-adherent management to reduce major cardiovascular events (Strong Recommendation; High-Quality Evidence) and to prevent recurrence of the arrhythmia and/or decrease its symptom burden (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on a comprehensive, holistic, and systematic approach to the management of AF. Because of the contribution of modifiable risk factors to the development and progression of AF, a systematic approach to the identification of modifiable cardiovascular risk conditions offers a potential therapeutic target to improve outcomes in this population. This recommendation recognizes the association between these modifiable cardiovascular risk factors (including but not limited to hypertension, HF, diabetes mellitus, obesity, inactivity, sleep apnea, and alcohol misuse), and major adverse cardiovascular outcomes (eg, stroke, MI, cardiovascular death) and AF outcomes (AF burden/exacerbations, AF-related ED visits/ hospitalizations).

Practical tip. Screening for common cardiovascular risk factors and/or conditions (hypertension, obesity, inactivity, sleep apnea, diabetes, and alcohol misuse) should be performed in addition to screening for AF-specific risk conditions (HF, valvular heart disease, thyroid dysfunction).

7. Integrated Approach to AF Management

As with many other chronic cardiovascular conditions, the complex and multifaceted nature of AF necessitates a systematic approach to the management of the AF patient. Much of the initial management of AF can be provided by primary care providers with the support of specialist cardiology input to guide management decisions in selected AF patients who develop problems or complications during therapy. Dedicated multidisciplinary clinics specifically focused on integrated AF care have been developed to facilitate patient and provider education, provide advanced subspecialist treatment options, and deliver evidence-based care centred on chronic disease management principles.

The systematic approach to patient care in integrated multidisciplinary AF clinic networks typically includes holistic protocol-driven management beyond the heart rhythm, with a particular focus on known care gaps in the domains of stroke prevention, transitions between rate and rhythm control, and coordination of heart rhythm pro-cedures.^{49,133,134} Although individually tailored to the needs of their communities,¹³⁵ multidisciplinary AF clinics are broadly on the basis of the principles of: (1) timely access to specialist care, to reduce adverse outcomes (eg, stroke or hospitalization) imposed by treatment delays; (2) knowledge translation, because improved understanding of AF facilitates active participation by the patient in their care pathway; (3) guideline adherence, in particular in the domains of stroke prevention and comorbidity management; and (4) integration with community care providers to enhance treatment cohesiveness and support care transitions (eg, from the ED to the specialty clinic and back to community care). A conceptual framework for the integrated approach to AF management is presented in Figure 7.

Nonrandomized studies of specialized AF clinics have suggested that the greater coordination of care between specialist, nursing, and allied health interventions are associated with reduced wait times, enhanced transitions of care, improved adherence to guideline-based care, significant improvement in QOL, and more effective use of tertiary care resources.¹³⁶⁻¹³⁹ These observational findings were supported by the results of a randomized single-centre study, which showed that a nurse-led multidisciplinary integrated care approach reduced cardiovascular death (HR, 0.28 vs usual care) and cardiovascular hospitalization (HR, 0.66 vs usual care), with resultant cost-effectiveness compared with standard care.^{140,141} Although these data were not replicated in a multicentre study of nurse-led vs usual care (HR, 0.85; 95% CI, 0.70-1.05; P = 0.12 for the composite of cardiovascular death or cardiovascular hospitalization), a prespecified analysis showed significant benefit in experienced centres.¹⁴²

RECOMMENDATION

12. We suggest a structured, integrated, multidisciplinary, patient-focused approach to care should be implemented for patients with AF (Weak Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation recognizes that AF is a multifactorial disease that requires long-term treatment. An integrated patient-focused teambased approach to care has been shown to improve guideline adherence, reduce adverse clinical outcomes such as hospitalization and mortality, and improve QOL.

7.1. Self-management, shared decision-making, and patient education

As with other chronic illness, self-management plays a crucial role in the longitudinal care of AF patients. Self-management can be conceptualized as the "day-to-day management of chronic conditions by individuals over the course of an illness" with an ultimate goal of improving health outcomes by enabling individuals to effectively manage their own illness.¹⁴³ Pragmatically, this involves shifting from the traditional provider-patient relationship to one in which the patient takes the responsibility for guiding their care in partnership with their health care providers (eg, shared

decision-making).¹⁴⁴ Key areas of focus include medical management (eg, adherence to a therapeutic regimen), behaviour modification (eg, weight loss and exercise), and development of strategies to provide emotional and psychosocial support. Fundamental to the concept of self-management is the patient's perceived understanding about the cause(s), consequences, clinical manifestations, and controllability of their AF. Misalignment of the therapeutic interventions with the patient values and preferences can lead to dissatisfaction with therapy, nonadherence (not taking OAC as directed), and nonpersistence (therapy discontinuation), resulting in an increased risk of stroke.¹⁴⁵



Figure 7. Integrated approach to atrial fibrillation (AF) management. Shown at the top of the illustration is the current paradigm of care for the AF patient, whereby siloed care is delivered across the spectrum of AF patients (low-, moderate-, and high-risk) leading to inconsistent access to specialty care and allied health care providers, inappropriate health care resource utilization (eg, patients seeking nonacute care in the emergency department), and inconsistent management. Shown at the bottom of the illustration is the ideal integrated AF care model, whereby the integrated AF clinic acts as a central hub for the management of AF patients. In this model the AF clinic can act as a peripheral resource for lower risk/well managed patients, who are able to remain with their community providers. For higher-risk patients in need of complex or acute intervention the AF clinic acts as a central access point for comprehensive care from the multidisciplinary heart team.

Patient-centred care requires collaboration between clinicians and knowledgeable patients, considering the best available evidence in addition to the patient's values and preferences.¹⁴⁵ Unfortunately, AF patients often have a poor understanding of the cause, consequences, and controllability of their AF, with particular deficiencies noted in the domain of stroke prevention.¹⁴⁵⁻¹⁴⁷ Tailored patient education facilitates the construction of an accurate illness representation, improves patients' illness-treatment coherence, corrects beliefs about therapeutic options and goals, improves treatment adherence, relieves disease-associated anxiety and stress, and promotes self-management.¹⁴⁶⁻¹⁴⁸ However, the ideal educational strategy is uncertain. Despite written information, patient decision aids, and in-person education sessions all being commonly used, there is no consensus regarding the most effective structure, setting, or educator.149

RECOMMENDATION

- 13. We recommend that individualized goals of care and specific approaches to management should be developed in collaboration with patients and should consider their values and preferences to enhance engagement and improve adherence to long-term therapy (Strong Recommendation; Low-Quality Evidence).
- 14. We recommend that patients and care providers be supported with educational resources (in person and print/electronic) to enhance disease awareness and facilitate self-management (Strong Recommendation; Low-Quality Evidence).

7.2. Treatment adherence

Adherence has been defined as the "active, voluntary, and collaborative involvement of the patient in a mutually acceptable course of behaviour to produce a therapeutic result,"¹⁵⁰ which, in the case of pharmacotherapy can be further conceptualized as medication adherence (eg, the extent to which patients take their medications as prescribed) and persistence (eg, the duration of continuous use after index prescription). Unfortunately, nonadherence to pharmacotherapy is common for patients across the spec-trum of cardiovascular diseases.^{151,152} In the AF population, adherence and persistence to stroke prevention therapies are suboptimal. Beyond the observation that approximately one-third of higher-risk AF patients fail to receive appropriate OAC,¹⁵³⁻¹⁵⁵ of those who receive OAC approximately one-third of OAC-treated patients demonstrate suboptimal adherence.^{156,157} Furthermore, persistence in those who have started OAC treatment has been shown to be inadequate with 10% of those who have

Table 6. Potential interventions to improve adherence and persistence

- Establishment of a "blame-free" health care environment (eg, recognizing that adherence is not exclusively the responsibility of the patient)
- Assess health literacy and provide tailored education
- Use motivational strategies and focus the interventions on behavioural strategies
- Address rational nonadherence
- Modify care delivery (eg, provide more convenient care such as telemedicine; use structured follow-up)
- Simplify dosing regimens (eg, reduce the number of daily doses)
- · Consider medication delivery (eg, blister packs)

started VKA treatment failing to fill a second prescription, one-third discontinuing therapy within a year, and less than one-third continuing OAC after 5 years.¹⁵⁸ Likewise, in those who have started DOAC treatment the rates of discontinuation within a year have been reported to range between 13.6% and 52.7%.¹⁵⁹⁻¹⁶¹ Unfortunately, lower rates of OAC adherence has been associated with higher rates of all-cause mortality and stroke.^{156,157,160-162}

In consideration of the link between adherence and outcomes, it is paramount to pursue strategies that improve adherence and persistence. Although the focus is often limited to factors related to the patients themselves (eg, behavioural), nonadherence and nonpersistence are often multifactorial in origin. As such, any intervention that targets adherence and persistence must take into consideration factors related to the health care system (eg, lack of access or continuity of care), the patient-provider interaction (eg, a poor provider-patient relationship, poor communication, lack of patient education), the medical condition itself (eg, asymptomatic AF and the need for continued ongoing therapy), medical comorbidities (eg, physical disability, mental health disorders, or cognitive impairment), the therapeutic regimen (eg, complexity, medication side effects), or other socioeconomic factors (eg, low level of literacy, high medication cost, or lack of social support).¹⁶³

Solutions proposed to improve adherence and persistence are presented in Table 6. When used in isolation, these unimodal interventions have limited effect on adherence and clinical outcomes. Multimodal interventions, although complex, have shown the greatest benefit in improving outcomes. In the HF population an intensive pharmacist-led multimodal intervention increased medication adherence (78.8% vs 67.9%) and reduced ED visits/hospitalizations (relative risk [RR], 0.82; 95% CI, 0.73-0.93), saving USD\$2960 in annual health care costs compared with usual care.¹⁶⁴ Likewise, a large cluster-randomized study showed that a multimodal intervention improved OAC prescription at 1 year compared with usual care (80% vs 67%).¹⁶⁵ Despite the absolute difference in OAC prescription being 9.1% (95% CI, 3.8-14.4) between groups, there was a > 50% reduction in stroke (HR, 0.48; 95% CI, 0.23-0.99; P = 0.04). However, it is important to consider that the most effective interventions are often complex and time-consuming, which has limited their implementation outside of multidisciplinary clinical environments.

RECOMMENDATION

15. We recommend that adherence and persistence to pharmacotherapy be assessed at each clinical encounter and supported using patient-centred strategies (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation puts high value on the evidence that indicates poor long-term persistence and adherence with OAC treatment, as well as the recognition that strategies to improve persistence and adherence can substantially reduce the risk of stroke/ systemic embolism.

7.3. Electronic or mobile health

The term "eHealth" or "mHealth" commonly refers to the use of electronic media or mobile health technologies to provide or enhance the delivery of health care. eHealth can include the use of electronic interfaces (eg, Web sites or digital resources) along with mobile health devices (eg, handheld ECG devices or wearable smart devices).^{166,167} There are ample opportunities to incorporate eHealth throughout the provision of AF care (see section 5.1). Innovations in eHealth can improve access to services through outreach and telemedicine.¹⁶⁸ Smartphone applications and smart watches can measure the pulse using photoplethysmography or electrical sensors, providing an estimate of heart rate during daily activity and during arrhythmias. The future of AF care delivery will likely leverage a combination of point-of-care and other mHealth technologies to deliver a patient-centred experience.

8. Stroke Prevention

8.1. Stroke risk assessment

Observations from the Framingham cohort and subsequent clinical trials revealed that NVAF is an independent risk factor for stroke (annual incidence of approximately 4.1%-4.5%) and combined stroke/systemic embolism (annual



Figure 8. The "CCS algorithm" (CHADS-65) to guide antithrombotic therapy decision-making for patients with nonvalvular atrial fibrillation (AF) or atrial flutter. An oral anticoagulant (OAC) should be prescribed to most patients 65 years of age or older and for younger patients with 1 or more of the other **C**ongestive Heart Failure, **H**ypertension, **A**ge, **D**iabetes, **S**troke/Transient Ischemic Attack (CHADS₂) risk factors: history of congestive heart failure, hypertension, diabetes, or stroke/transient ischemic attack (TIA). If none of the previous factors are present, but the patient has coronary or peripheral vascular disease, we recommend acetylsalicylic acid (ASA) 81 mg daily alone or in combination with other antithrombotic therapy. If none of the factors (including vascular disease) are present, no antithrombotic therapy is indicated. When an OAC is prescribed, a nonvitamin K antagonist direct OAC (DOAC) is recommended in preference to warfarin. Bleeding risks should be modified whenever possible. bid, twice daily.

incidence of 50%).^{70,169} The risk was further refined by the delineation of various baseline characteristics that might affect the risk of the stroke.¹⁶⁹⁻¹⁷³ The first widely adopted tool for stroke risk assessment was the Congestive Heart Failure, Hypertension, Age, Diabetes, Stroke/Transient Ischemic Attack (CHADS₂) score, which assigns a single point for HF, hypertension, age 75 years or older, and diabetes, and 2 points for previous stroke/systemic embolism.^{26,172} Unfortunately, CHADS₂ was unable to adequately differentiate very low risk individuals (ie, in whom OAC is associated with a greater risk than benefit) from those at low but still clinically important stroke risk. This led to the development of the expanded Congestive Heart Failure, Hypertension, Age (≥75 years), Diabetes, Stroke/Transient Ischemic Attack, Vascular Disease, Age (65-74 years), Sex (Female) (CHA2DS2-VASc) score. This score includes the additional risk factors of vascular disease (1 point), female sex (1 point), and age between 65 and 74 years (1 point), and also increases the risk weight to 2 points for age 75 years or older.26,171

The 2010 CCS guidelines recommended stroke prevention therapies⁴ on the basis of the CHADS₂ whereas the 2012 update² differentiated among patients at low CHADS₂ risk on the basis of risk factors derived from the CHA2DS2-VASc.¹⁷⁴ The algorithm for antithrombotic prescription was further revised in 2014,¹⁷⁵ on the basis of data from a Danish national cohort study, which showed that the annual risk of "stroke" (defined as a thromboembolic event precipitating hospitalization or death) was 2.1% for patients aged 65-74 years and 4.4% for those 75 years of age or older.¹⁷⁴ Because age could be reliably determined in all patients, it was agreed that age 65 years or older should be the starting point for a revised algorithm.¹⁷⁵ Likewise, OAC was justified for younger patients with any CHADS₂ risk factors (annual risk of stroke: 2.4% with HF, 1.6% with hypertension, 2.3% with diabetes, and 7.9% with previous stroke/systemic embolism). The CCS Guidelines Committee judged that antiplatelet therapy has no role in AF-related stroke prevention and, therefore, no antithrombotic therapy was recommended for patients younger than 65 years with none of the CHADS₂ risk factors. However, in the presence of vascular disease daily antiplatelet therapy is indicated to prevent ischemic vascular events independent of the presence of AF.¹⁷⁶ As such, antiplatelet therapy is only indicated in the presence of established vascular disease in NVAF patients aged younger than 65 years with no CHADS₂ risk factors (see section 8.3.2.1). The CCS algorithm (CHADS-65) is presented in (Fig. 8).

Among several organizations that publish guidelines for antithrombotic therapies for patients with AF, there are variations in the definitions of the component risk factors, and in the selection and interpretation of data on the stroke risk associated with the individual risk factors according to their preferred risk schemes.^{10,26,177} These differences in definitions are unlikely to affect the categorization of patient stroke risk, however, they might provide different point estimates of the annual risk of stroke.

It has also become evident that the risk of stroke varies among cohorts reported from different countries.¹⁷⁸ For similar baseline characteristics, the Taiwanese¹⁷⁹ and Danish cohorts¹⁷⁴ appear to have particularly high stroke risk, whereas cohorts from Sweden¹⁸⁰ and the United States¹⁸¹ appear to be lower risk. The variation might be attributable to the dissimilarities in the definitions of stroke, treatment of comorbidities, and the protocols for data collection.

RECOMMENDATION

- 16. We recommend that all patients with AF should undergo annual assessment of their risk of stroke/ systemic embolism, irrespective of their clinical pattern of AF (Strong Recommendation; High-Quality Evidence).
- 17. We recommend that the "CCS Algorithm" (CHADS-65) be used to guide the choice of appropriate antithrombotic therapy for the purpose of stroke/systemic embolism prevention in patients with NVAF (Strong Recommendation; High-Quality Evidence).

Practical tip. NVAF is defined as AF in the absence of mechanical heart valves or moderate to severe mitral stenosis.

8.1.1. Atrial flutter

Atrial flutter (AFL) carries a significant risk of stroke/systemic embolism but, as opposed to AF, the relationship is somewhat less precisely established due to limited number of small and retrospective cohort studies, as well as significant heterogeneity of data.¹⁸² Because many patients with AFL also experience episodes of AF it becomes difficult to know the exact risk of stroke/systemic embolism related to AFL alone.¹⁸³ Moreover, there are no randomized trials of the value of OAC specifically in the AFL population.

However, there is direct and indirect evidence from mechanistic, observational, and prospective studies that AFL confers a significant thromboembolic risk. TEE evidence of atrial thrombi has been documented in patients with sustained AFL not receiving long-term anticoagulation (LA thrombus in 1.6% with at least moderate spontaneous LA echo contrast in 13%; median duration of 33 days).¹⁸⁴ OAC has been shown to reduce the incidence of stroke/systemic embolism after cardioversion, with an incidence similar to that of AF.^{185,18} In a metaregression analysis, the annual risk of stroke/systemic embolism was approximately 3% in patients with AFL.¹⁸⁷ Although the rate of stroke is higher than that in control participants (HR, approximately 1.4), the risk of stroke appears to be lower than for patients with AF (HR, approximately 0.70).^{183,188} For AFL patients, the risk of stroke is higher with increasing CHA₂ $D\hat{S}_2$ -VASc scores¹⁸⁹⁻¹⁹² and also increased in patients who develop AF after their initial diagnosis of AFL.¹⁹² As such, it is recommended that patients with AFL be stratified and treated in the same manner as patients with AF.¹⁹

8.2. Oral anticoagulation

The benefit of OAC must be weighed against the risk of hemorrhage. The relative importance of a stroke prevented, and a major bleed caused is a subjective judgement. There is considerable scope for physician-patient discussion (ie, shared decision-making) to ensure that patient values are concordant with the decision to prescribe OAC, particularly when the annual risk of stroke is < 2% per year.¹⁹⁴

The CCS rationale for recommending OAC for most patients with age 65 years or older or CHADS₂ score ≥ 1 is on the basis of the effects of OAC on the absolute risk reduction of stroke compared with the increase in major hemorrhage. In patients aged 65 years or older and without other risk factors for stroke, use of VKAs decreased the annual risk of stroke from 2.1% to 0.7% while it increased the risk of major bleeding by approximately 0.5% per year to 1.0%.^{51,195} Although the risk of major bleeding increases with increasing CHADS₂ scores, the rate of rise is not as steep as that for stroke; therefore, the benefit to risk ratio for OAC increases as stroke risk factors accumulate. Furthermore, 70% of strokes result in death or major disability, whereas most patients survive major hemorrhage without long-term effects.¹⁹⁶ Thus, these results favour use of OAC in patients with age 65 years or older or CHADS₂ score \geq 1. OACs with efficacy and safety evidence in AF include VKAs and DOACs.

RECOMMENDATION

18. We recommend that OAC be prescribed for most patients with AF and age 65 years or older or CHADS₂ score ≥ 1 (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places relatively greater weight on the absolute reduction of stroke risk with OAC compared with antiplatelet agents in patients aged 65 years or older or with CHADS₂ score ≥ 1 and less weight on the potentially increased risk of major hemorrhage with OACs compared with antiplatelet agents.

19. We suggest that no antithrombotic therapy be prescribed for stroke prevention for most patients with NVAF who are aged younger than 65 years with no CHADS₂ risk factors (Weak Recommendation; Moderate-Quality Evidence).

Values and preferences. For patients with NVAF who are aged younger than 65 years with no $CHADS_2$ risk factors the current evidence does not support antiplatelet monotherapy for stroke prevention. For those patients, with concomitant coronary or peripheral arterial vascular disease, the antithrombotic treatment should be directed at the underlying arterial disease as outlined in the CCS/ Canadian Association of Interventional Cardiology (CAIC) guidelines (see section 8.3.2.1).

20. We recommend that the longitudinal follow-up of patients receiving OAC include regular assessment of bleeding risk, potential drug-drug interactions, as well as adherence and persistence to pharmacotherapy (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. Consider the use of a standardized follow-up and monitoring tools to facilitate systematic review of efficacy and bleeding in patients receiving OAC, particularly for patients receiving DOACs who do not undergo regular laboratory monitoring.

8.2.1. Warfarin and vitamin K antagonists

Six studies (2900 patients) have compared VKAs with placebo in the setting of NVAF; 5 primary prevention trials (2 doubleblinded) and 1 secondary prevention trial. The mean achieved international normalized ratio (INR) ranged from 2.0 to 2.6 among patients who were assigned to VKAs in the primary prevention trials and was 2.9 in the secondary prevention trial. A meta-analysis of these 6 studies showed a RR reduction of 64% (95% CI, 49%-74%) in all stroke (ischemic or hemorrhagic)⁵¹ with an absolute risk reduction of 2.7% per year for primary prevention trials and 8.4% per year for secondary prevention. The absolute risk increase in major extracranial hemorrhage was 0.3% per year with use of VKAs. However, there was an absolute risk reduction of 1.6% per year in overall mortality with use of VKAs.

In a meta-analysis of 8 RCTs (3647 patients), the RR reduction with use of VKAs over aspirin was 39% (95% CI, 19%-53%) for all strokes, equivalent to an absolute risk reduction of approximately 0.7% per year for primary prevention and 7% per year for secondary prevention.⁵¹ There was an excess of bleeding in the VKA-treated patients leading to an absolute risk increase of 0.2% per year in major extracranial and intracranial hemorrhage (ICH). However, the absolute risk reduction for all-cause mortality remained in favour of VKAs at 0.5% per year. In an update of this meta-analysis to include the largest study of elderly patients¹⁹⁷ the absolute risk reduction increased to 0.9%, with the risk of major extracranial hemorrhage being no different between use of VKA and aspirin.⁵¹ A further meta-analysis showed that adjusted-dose VKA was associated with half the rate of stroke/systemic embolism events with no significant difference in the rate of major bleeding relative to adjusted-dose antiplatelet therapy.¹⁹

The benefit of VKAs for stroke prevention in patients with NVAF is optimized at a target INR of 2-3, with the time spent in this therapeutic range (TTR) directly correlated with clinical outcomes. The relationship between TTR and clinical outcomes was shown in a post hoc analysis of the AF Clopidogrel Trial With Irbesartan for Prevention of Vascular Events (ACTIVE W) trial.¹⁹⁹ In ACTIVE W there was no difference in stroke reduction between VKA and clopidogrel with aspirin therapy (RR, 0.93; 95% CI, 0.70-1.24; P = 0.61) at centres with a TTR below the median TTR (65%). However, VKAs had a marked benefit for patients at centres with a TTR above the median, reducing vascular events by more than twofold (RR, 2.14; 95% CI, 1.61-2.85; P < 0.0001). A subsequent systematic review reinforced this concept, showing a significant correlation between TTR and adverse outcomes (major hemorrhage and thromboembolic rate) in an analysis of 33,976 AF patients included in 38 studies.²⁰ More recently, a large worldwide observational study of 9934 VKA-treated patients enrolled from 35 countries between 2010 and 2015 showed an adjusted 2.6-fold increased risk of stroke/ systemic embolism, a 1.5-fold increased risk of major bleeding, and a 2.4-fold increased risk of all-cause mortality with TTR <65% vs $\geq 65\%$.²⁰¹ Further, the study highlighted the difficulty in managing VKAs because the mean TTR for the total study group was 55%, with more than half of the North American cohort having a TTR < 65%.²⁰¹

VKAs such as warfarin remain the preferred OAC in the presence of any mechanical prosthetic heart valve and in patients with moderate to severe mitral stenosis. With respect to the former, in the **R**andomized, Phase II Study to **E**valuate the

Table 7.	Recommendations	for	dosage	of	oral	anticoagulants

CrCl	Warfarin	Apixaban	Dabigatran	Edoxaban	Rivaroxaban
CrCl >50 mL/min	Dose adjusted for INR 2.0-3.0	5 mg BID†	150 mg BID*	60 mg daily∞	20 mg daily
CrCl 30-49 mL/min	Dose adjusted for INR 2.0-3.0	5 mg BID†	Consider 110 mg BID	30 mg daily	15 mg daily
CrCl 15-29 mL/min	No RCT Data**	Very limited RCT Data§	No RCT Data¶	Very limited RCT Data¶	No RCT Data
CrCl <15 mL/min (or on dialysis)	No RCT Data‡	Very limited RCT Data¶	No RCT Data¶	No RCT Data¶	Very limited RCT Data¶

BID, twice daily; CrCl, creatinine clearance, INR, international normalized ratio; RCT, randomized clinical trial.

*Dabigatran 110 mg po BID is recommended if age ≥80 years, or ≥75 years with other bleeding risk factors including CrCl 30-50mL/min

⁺Apixaban 2.5 mg po BID is recommended if 2 of the 3 following criteria are present: 1) age ≥80 years, 2) body weight ≤60 kg, or 3) serum creatinine ≥133 µmol/L ∞Consider Edoxaban 30mg daily if weight ≤60 kg or concomitant potent P-Gp inhibitor therapy EXCEPT amiodarone or verapamil

**Dose adjusted warfarin has been used, but data regarding safety and efficacy is conflicting

Dose adjusted warfarin has been used, but observational data regarding safety and efficacy is conflicting and suggests harm.

§The ARISTOTLE trial included a small number of patients with a CrCl as low as 25 mL/min

¶Product monographs suggest the drug is contraindicated for this level of renal function.

Safety and Pharmacokinetics of Oral Dabigatran Etexilate in Patients After Heart Valve Replacement (RE-ALIGN) study patients with mechanical heart valves were randomized to VKA or dabigatran (150 mg, 220 mg, or 300 mg twice daily [BID], on the basis of renal function).²⁰² The trial was terminated early because of an excess of thromboembolic and bleeding events with dabigatran. Although RE-ALIGN is the only trial to evaluate DOACs in the setting of mechanical valve prostheses, VKA remains the treatment of choice in this population. With respect to patients with moderate to severe mitral stenosis, there are no randomized trials of OAC for prevention of thromboembolic events. Retrospective studies have suggested a 4- to 15-fold decrease in the incidence of embolic events with OAC therapy.²⁰³ Because patients with severe mitral valve disease were excluded from the pivotal randomized DOAC trials,²¹⁻²⁵ VKAs remain the standard of care in this patient population until further evidence emerges. Ongoing trials of DOACs in patients with rheumatic valvular disease (Investigation of Rheumatic AF Treatment Using Vitamin K Antagonists, Rivaroxaban or Aspirin Studies [INVICTUS] studies; NCT02832531 and NCT02832544) will provide guidance.

8.2.2. Non-vitamin K direct oral anticoagulants

Four DOACs are currently approved for use in Canada: apixaban, dabigatran, edoxaban, and rivaroxaban. These four agents were developed to overcome the major limitations associated with VKAs and have been evaluated in large RCTs involving more than 70,000 patients.²¹⁻²⁵ Individually, DOACs were shown to be at least as effective as VKAs in decreasing the risk of NVAF-associated stroke/systemic embolism, with similar or less major bleeding. A meta-analysis of these four RCTs showed that the use of the higher approved DOAC dose resulted in a statistically significant reduction in stroke/systemic embolism (RR, 0.81; 95% CI, 0.73-0.91; P < 0.0001), ICH (RR, 0.48; 95% CI, 0.39-0.59; P < 0.0001), and all-cause mortality (RR, 0.90; 95% CI, 0.85-0.95; P = 0.0003) with less major bleeding (RR, 0.85; 95% CI, 0.73-1.00; P =0.06), compared with VKAs.⁵² Use of the lower-dose DOAC regimens resulted in similar rates of stroke/systemic embolism (RR, 1.03; 95% CI, 0.84-1.27; P = 0.74), with less major (RR, 0.65; 95% CI, 0.43-1.00; P = 0.05) and intracranial bleeding (RR, 0.31; 95% CI, 0.24-0.41; *P* < 0.0001), and lower mortality (RR, 0.89; 95% CI, 0.83-0.96; P = 0.003), but significantly more ischemic strokes (RR, 1.28; 95% CI, 1.02-1.60; P = 0.045) compared with VKAs. On the basis of these observations the CCS AF Guidelines Committee continues to recommend DOACs over VKAs for patients with NVAF.

There have been no published RCTs directly comparing the four DOACs. Differences in the designs of the RCTs, including the population studied (and by extension their baseline stroke risk), preclude definitive comparisons between the agents. Although so-called "real world" comparisons are plentiful in the published literature, it is not possible to make reliable therapeutic inferences from observational associations because these types of studies are subject to significant biases and limitations.²⁰⁴

Nevertheless, there are relevant differences in the pharmacological characteristics of the DOACs, which might influence their selection in individual patients such as: (1) bioavailability, with dabigatran being the least bioavailable, thereby requiring a special formulation for optimal absorption, and rivaroxaban requiring coadministration with food to optimize absorption of the 15- and 20-mg doses; (2) renal clearance, with dabigatran being predominantly (80%) renally cleared; (3) drug-drug interaction potential, with all of the DOACs influenced by P-glycoprotein interactions but only rivaroxaban and apixaban also being influenced by cytochrome P-450 isoenzyme function, specifically 3A4; (4) the elimination half-life and dosing schedule; and, (5) the presence and availability of an antidote.

Regardless of the specific DOAC agent selected, it is important to ensure that the prescribed dose is consistent with Health Canada labelling and the product monograph (Table 7). Specifically, it has been observed that overtreatment (or the prescription of a standard dose in patients with an indication for dose reduction) occurs in approximately 4% of patients and is associated with an increased risk of major bleeding, hospitalization, and death, without a significant additional reduction in stroke.²⁰⁵⁻²⁰⁷ Likewise, undertreatment (or prescription of a reduced dose without an indication to do so) occurs in 12%-15% of patients and is associated with a higher risk of thromboembolic events, hospitalization, and death, without a significant reduction in major bleeding.²⁰⁵⁻²⁰⁷

RECOMMENDATION

21. We recommend most patients should receive a DOAC (apixaban, dabigatran, edoxaban, or rivaroxaban) in preference to warfarin when OAC therapy is indicated for patients with NVAF (Strong Recommendation; High-Quality Evidence).

Values and preferences. This recommendation places a relatively high value on the results of several large RCTs showing that the DOACs are either noninferior or superior to warfarin in preventing AF-related stroke; that they have no more or less major bleeding compared with warfarin; that they are associated with less ICH compared with warfarin; and on the greater ease of use of DOACs compared with dose-adjusted warfarin.

Practical tip. Baseline renal function and complete blood counts should be measured before initiation of anticoagulation and at a regular intervals thereafter (see section 8.3.1).

Practical tip. The dose of DOAC prescribed should follow the doses used in the RCTs and Health Canadaapproved prescribing information (see Table 7). Receipt of a higher than recommended dose is associated with increased bleeding events and overall mortality. Receipt of a lower than recommended dose is associated with increased rates of stroke/systemic embolism.

Practical tip. Consideration should be given to switching eligible patients from warfarin to a DOAC, particularly if they are unable to maintain a therapeutic INR.

22. We recommend that warfarin be used for patients with a mechanical prosthetic valve and those with AF and moderate to severe mitral stenosis (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places high value on the evidence from 1 RCT of the inferiority of dabigatran compared with warfarin for the prevention of thromboembolism in patients with a mechanical prosthetic valve.

Values and preferences. This recommendation places a relatively high value on the long experience and clinical reports of the use of warfarin in patients with rheumatic mitral stenosis.

8.3. Anticoagulation in special populations

8.3.1. Anticoagulation in patients with chronic kidney disease and end-stage renal disease

AF patients with chronic kidney disease (CKD) represent a particularly high-risk subgroup because stroke, mortality, and major bleeding all increase as renal function deteriorates.^{208,209°} RCTs demonstrate that OAC, including DOACs, are safe and effective for AF patients with mild to moderate CKD (stage 1-3 CKD or eGFR > 30 mL/min).²¹⁰ There are few randomized data for patients with severe renal impairment (stage 4-5 CKD or eGFR < 25 to 30 mL/min) as these patients were excluded from the large phase III RCTs. Nonrandomized and very limited randomized data support the use of OAC in patients with stage 4 CKD (eGFR 15-30 mL/min),^{211,329} however, the benefit of OAC for AF patients with severe CKD (eGFR < 15 mL/min) or end-stage renal disease (ESRD) who require dialysis remains unclear. There are no prospective randomized trials that have evaluated OACs vs placebo for dialysis-dependent AF patients, and results of the observational studies are conflicting.^{209,212-215} The ongoing randomized Study of the Benefit/Risk Ratio of Oral Anticoagulation in Hemodialysis Patients with Atrial Fibrillation (AVKDIAL) is studying VKA (target INR 2-3) compared with no OAC in hemodialysis patients with AF (NCT02886962). Until these results are available there is insufficient evidence to support or deny routine OAC use in the AF population with ESRD who require dialysis.

When OAC is prescribed for AF patients with stage 1-4 CKD, the CCS AF guidelines panel recommends that a DOAC is preferred to a VKA. For those with moderate CKD in the landmark phase III trials, DOACs significantly reduced the risk of stroke/systemic embolism (RR, 0.79; 95% CI, 0.66-0.94) and major bleeding (RR, 0.80; 95% CI, 0.70-0.91), compared with VKAs.²¹⁶ Moreover, in patients with mild-moderate CKD the use of DOACs was associated with a lessened rate of adverse renal outcomes, including renal function decline, doubling in serum creatinine level, or acute kidney injury.²¹⁷

Although apixaban and rivaroxaban are approved for patients with stage 4 CKD (CrCl of > 15 mL/min), the evidence supporting DOAC use in preference to VKAs is limited.²¹⁸ In those with more severe stage 5 CKD or ESRD requiring dialysis the evidence supporting DOACs is incomplete, with the available observational data being subject to confounding and selection bias and the randomized studies being underpowered.^{219-222,885} The randomized Renal Hemodialysis Patients Allocated Apixaban versus VKA in AF trial (RENAL-AF; NCT02942407) was prematurely terminated following the enrollment of 154 patients with hemodialysisdependent ESRD (targeted enrollment, 760 patients). At study conclusion, the rates of bleeding and stroke/systemic embolism were similar between those randomized to apixaban 5 mg BID or VKAs. Ongoing studies include the Compare Apixaban and Vitamin-K Antagonists in Patients With Atrial Fibrillation (AF) and End-Stage Kidney Disease (ESKD) (AXADIA; NCT02933697), in which phenprocoumon vs is being compared to apixaban 2.5 mg twice daily (targeted enrollment 222 patients); and the Strategies for the Management of Atrial Fibrillation in Patients Receiving Dialysis (SAFE HD; NCT03987711) trial, in which VKA and

apixaban 5 mg BID, and no anticoagulation are being compared (targeted enrollment 150 patients).

Patients with clinically significant CKD require a reduction in DOAC dose in accordance with the approved prescribing information to avoid adverse clinical outcomes (Table 7).^{7,205-207,223} Although undertreatment (or prescription of a reduced dose without an indication to do so) is associated with a higher risk of thromboembolic events, hospitalization, and death,²⁰⁵⁻²⁰⁷ appropriate dose reduction in those with moderate CKD can achieve clinical outcomes comparable with that in patients with preserved renal function (CrCl > 50 mL/ min) who are receiving the higher DOAC dose.^{224,225}

When medication dose adjustment is performed, Cockcroft-Gault CrCl should be used. The rationale has been extensively outlined in the 2014 AF guidelines companion.²⁶ In brief, the Cockcroft-Gault CrCl was the formula used to assess eligibility in the landmark DOAC trials, and it is recommended to guide medication dose adjustment in the product monographs and approved prescribing information.^{21-23,25} Although the **Mo**dified **D**iet in **R**enal **D**isease (MDRD) or the Chronic Kidney Disease **Epi**demiology Collaboration (CKD-EPI) eGFR equations are commonly reported by laboratories, use of these formulae fail to identify a significant proportion of patients with contraindications to DOACs or those who require dose adjustment.²²⁶

RECOMMENDATION

23. We recommend that patients with AF who are receiving OAC should have their renal function assessed at baseline and at least annually to detect latent kidney disease, determine OAC eligibility, and to support drug dosing (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. Because DOAC eligibility and dosing are dependent on renal function, we recommend at least annual assessment of renal function with calculation of CrCl. In stable patients with an eGFR of 30-60 mL/min renal function should be monitored every 6 months, and every 3 months for patients with an eGFR of 15-30 mL/min. In patients with fluctuating renal function or acute dehydrating illness renal function should be assessed more frequently. Follow-up monitoring should also include assessment of bleeding risk, drug interactions, and adherence and persistence.

24. We recommend that CrCl, as estimated using the Cockcroft-Gault method, be used to support dosing decisions of anticoagulant medications (Strong Recommendation; High-Quality Evidence).

Practical tip. Multiple formulae have been developed to provide an estimate of renal function. The most commonly applied formulae estimate CrCl or filtration of creatinine by the glomerulus (glomerular filtration rate or eGFR). Although the eGFR equations (MDRD formula or the CKD-EPI formula) provide more accurate estimates of renal function, drug manufacturers have used the CrCl (Cockcroft-Gault formula) when recommending medication dosage adjustments for patients with renal dysfunction.

- 25. We recommend that antithrombotic therapy in AF patients with CKD be provided according to their risk of stroke/systemic embolism and the severity of renal dysfunction with selection of agent according to Table 7.
 - A. Stage 3 CKD or better (eGFR > 30 mL/min): we recommend that such patients receive antithrombotic therapy as determined by the "CCS algorithm" (Strong Recommendation; High-Quality Evidence).
 - B. Stage 4 CKD (eGFR 15-30 mL/min): we suggest that such patients receive antithrombotic therapy as determined by the "CCS algorithm" (Weak Recommendation; Low-Quality Evidence).
 - C. Stage 5 CKD (eGFR < 15mL/min or dialysisdependent): we suggest that such patients not routinely receive antithrombotic therapy for stroke prevention in AF (Weak Recommendation; Low-Quality Evidence).

Values and preferences. These recommendations place a relatively higher value on prevention of ischemic stroke than on bleeding complications associated with antithrombotic therapy.

Practical tip. Because of the lack of prospective data showing benefit in patients with a CrCl < 15 mL/min, the decision to use antithrombotic therapy should be individualized on the basis of physician and patient preference and considering the relative risks of stroke and bleeding. Therapy with antithrombotic therapy might be appropriate for some patients with AF and CrCl < 15 mL/min (or dialysis-dependent) in whom the benefit of preventing stroke outweighs the increased risk of bleeding.

8.3.2. Coronary artery disease

Up to 20%-30% of AF patients also have concomitant coronary artery disease (CAD), with a significant proportion who require percutaneous coronary intervention (PCI).^{227,228} OAC is indicated for the prevention of AF-related stroke/ systemic embolism, which also provides benefit in preventing ischemic coronary events. Antiplatelet therapy is indicated for the prevention of coronary events after acute coronary syndromes (ACS) and/or PCI; however, it is inferior to OAC for the prevention of stroke/systemic embolism in an AF population at increased risk of AF-related stroke.²²⁹ As such, management requires a careful and balanced assessment of the individual risks of bleeding vs the anticipated effect on thrombotic outcomes.

To clarify potentially confusing terminology in this area, single-agent antiplatelet therapy (SAPT) refers to the use of a single antiplatelet drug (eg, acetylsalicylic acid [ASA]), dual antiplatelet therapy (DAPT) refers to the concomitant use of 2 antiplatelet agents (eg, ASA with P2Y12 inhibitor), dual pathway therapy refers to the concomitant use of a SAPT with an OAC agent (eg, VKA with P2Y12 inhibitor), and triple antithrombotic therapy (TT), the combination of DAPT with an OAC (eg, VKA with ASA and P2Y12 inhibitor).

The comprehensive recommendations regarding antithrombotic treatment in AF patients indicated for OAC with concomitant coronary/arterial vascular disease are summarized in Figure 9.

RECOMMENDATION

26. We recommend that AF patients with coronary or arterial vascular disease (peripheral vascular disease or aortic plaque) receive an antithrombotic therapy regimen on the basis of a balanced assessment of their risk of AF-related stroke, ischemic coronary event, and clinically relevant bleeding associated with the use of antithrombotic agents (Strong Recommendation; High-Quality Evidence).

Practical tip. For patients who require combinations of antiplatelet and OAC agents for concomitant AF and coronary/arterial vascular disease, measures should be used to reduce the risk of bleeding, including careful consideration of modifiable bleeding risk factors with vigorous efforts to mitigate them; consideration of proton pump inhibitor (PPI) use; avoidance of prasugrel and ticagrelor in conjunction with OACs; the use of the lower target INR range (eg, 2.0-2.5) when a VKA is used as part of combination therapy; specific measures during PCI to reduce bleeding outcomes (radial access or ultrasound-guided femoral access, use of small diameter sheaths if appropriate, early sheath removal if feasible, and minimized use of acute periprocedural antithrombotic therapies); delaying non-urgent procedures until dual or triple therapy is no longer required; use of walking aids for those with gait or balance disorders; avoidance of concomittant use of nonsteroidal anti-inflammatory drugs (NSAIDs) or other drugs that might increase bleeding risk; and, strict BP control.

27. We recommend a DOAC in preference to a VKA when an OAC is indicated for AF patients with coronary or arterial vascular disease (Strong Recommendation; High-Quality Evidence).

Values and preferences. This recommendation places a relatively high value on the results of several large RCTs that showed that the DOACs are either noninferior or superior to VKAs in preventing AF-related stroke, that they cause no more or less major bleeding compared with VKAs, that they are associated with less ICH compared with VKAs, that they are associated with greater ease of use compared with dose-adjusted VKAs, and that DOACs are not associated with an increase in ischemic coronary outcomes.

8.3.2.1. Stable vascular disease and AF in patients at low risk of stroke/systemic embolism

SAPT (eg, ASA 81 mg/d) is recommended for patients with AF who are at low risk of stroke/systemic embolism (age younger than 65 years and CHADS₂ score of 0) if coronary or peripheral arterial vascular disease is present (CAD, peripheral vascular disease, or aortic plaque). The SAPT recommendation is on the basis of the efficacy of ASA therapy for the prevention of coronary events among patients with stable CAD (1.5% per year

absolute risk reduction for secondary prevention).^{51,230} Although there is extensive evidence for the efficacy of OAC for prevention of ischemic coronary events in patients with stable CAD,²³¹ the CCS AF guidelines recommend SAPT in preference to OAC in those at low risk of stroke because of the favourable safety profile and ease of use associated with antiplatelet therapy.

In a non-AF population, the Cardiovascular Outcomes for People Using Anticoagulation Strategies (COMPASS) trial showed that dual pathway therapy with ASA and "vascular dose" rivaroxaban (2.5 mg BID) was associated with significant reduction in cardiovascular mortality and ischemic stroke, albeit with a significantly increased risk of major bleeding.²³² As such, the combination of ASA and rivaroxaban 2.5 mg BID may be considered a reasonable alternative to ASA alone for NVAF patients at low risk of stroke (age younger than 65 years and CHADS₂ score of 0) who also have coronary or arterial vascular disease.

RECOMMENDATION

28. We suggest no oral anticoagulation for stroke prevention for most patients with NVAF aged younger than 65 years with no CHADS₂ risk factors and stable coronary or arterial vascular disease (Weak Recommendation; Moderate-Quality Evidence).

Practical tip. The risk of stroke associated with AF is not sufficiently elevated to justify routine OAC therapy for those patients with stable coronary or arterial vascular disease aged younger than 65 years with AF and no CHADS₂ risk factors. Treatment should be directed at the underlying coronary/peripheral arterial disease as outlined in the CCS/CAIC guide-lines. Therapeutic options include ASA 81 mg daily alone; or ASA 81 mg daily in combination with either clopidogrel 75 mg daily, ticagrelor 60 mg BID, or rivaroxaban 2.5 mg BID.

8.3.2.2. Stable vascular disease and AF in patients at high risk of stroke/systemic embolism

OAC is indicated for stroke prevention in patients with AF who are aged 65 years or older or with a CHADS₂ score ≥ 1 . When such a patient also has stable CAD (defined by the absence of ACS or PCI in the preceding 12 months), OAC provides protection against ischemic coronary events in addition to stroke and systemic embolism.²³³⁻²³⁹

The CCS AF Guidelines Committee does not recommend that antiplatelet agents be routinely prescribed in combination with OAC for AF patients at high risk of stroke with stable CAD. The Warfarin-Aspirin Reinfarction Study (WARIS)-II showed that the additional use of antiplatelet therapy with OAC did not confer a beneficial effect (eg, combined end point of death, MI, and stroke; 16.7% with OAC alone vs 15.0% with combination OAC and ASA; P = 0.18), but did increase the risk of adverse bleeding outcomes (overall bleeding 2.82% per year with OAC alone vs 3.27% per year with combination OAC and ASA).^{236,237} More recently, the Atrial Fibrillation and Ischemic Events With Rivaroxaban in Patients With Stable Coronary Artery Disease (AFIRE) trial²³⁹ randomized 2236 patients with AF with stable CAD to receive rivaroxaban monotherapy or combination therapy with rivaroxaban and SAPT. The trial was prematurely

AF Patients with Coronary or Vascular Disease and an Indication for OAC (Age \geq 65 years or CHADS₂ \geq 1)



1. PCI is considered high-risk based on clinical and angiographic features such as: diabetes mellitus, current smoker, chronic renal dysfunction (eGFR < 60 mL/min), prior ACS, multi-vessel disease, multiple stents implanted, complex bifurcation lesion, total stent length > 60 mm, prior stent thrombosis, chronic total occlusion intervention, or bioabsorbable vascular scaffold.

2. The OAC component evaluated as part of dual pathway therapy regimens include: warfarin daily, apixaban 5 mg BID (reduced to 2.5 mg if they met two or more of the following dose-reduction criteria: age > 80 years of age, weight < 60 kg, or Cr > 133 µmol per liter), dabigatran 110 mg or 150 mg PO BID, edoxaban 60 mg PO daily (30 mg in patients with CrCl 15–50 mL/min), bodyweight < 60 kg, or concomitant use of specified potent P-glycoprotein inhibitors), rivaroxaban 15 mg PO daily (10 mg in patients with CrCl 30-50 mL/min). A DOAC is preferred over warfarin, however if warfarin is to be used the lower end of the recommended INR target range is preferred. All patients should receive a loading dose of ASA 160 mg at the time of PCI (if previously ASA naïve).

3. The OAC component evaluated as part of triple therapy regimens include: warfarin daily, rivaroxaban 2.5 mg PO BID, or apixaban 5 mg BID (reduced to 2.5 mg if they met two or more of the following dose-reduction criteria: age > 80 years of age, weight < 60 kg, or Cr > 133 µmol per liter). A DOAC is preferred over warfarin, however if warfarin is to be used the recommended INR target is 2.0-2.5. All patients should receive a loading dose of ASA 160 mg at the time of PCI (if previously ASA naïve). Thereafter, ASA may be discontinued as early as the day following PCI or it can be continued longer. The timing of when to discontinue ASA will depend on individual patient's ischemic and bleeding risk.

4. The dose of OAC beyond one year after PCI should be standard stroke prevention doses. A combination of an OAC and single antiplatelet therapy may be used only in highly-selected patients with high-risk features for ischemic coronary outcomes, and who are also at low risk of bleeding

Figure 9. Management of antithrombotic therapy in patients with atrial fibrillation (AF) and coronary artery disease (CAD)/peripheral artery disease (PAD), who have an indication for an oral anticoagulant (OAC) for stroke prevention. ACS, acute coronary syndrome; ASA, acetylsalicylic acid; BID, twice per day; CHADS₂, **C**ongestive Heart Failure, **H**ypertension, **A**ge, **D**iabetes, **S**troke/Transient Ischemic Attack; Cr, creatinine; CrCI, creatinine clearance; DOAC, non-vitamin K direct oral anticoagulant; eGFR, estimated glomerular filtration rate; INR, international normalized ratio; PCI, percutaneous coronary intervention; PO, orally.

terminated because of increased all-cause mortality in the combination therapy group (1.9% per patient-year with rivaroxaban vs 3.4% per patient-year with combination therapy; HR, 0.55; 95% CI, 0.38-0.81; P < 0.05). Rivaroxaban monotherapy was noninferior to combination therapy for the primary efficacy end point (stroke/systemic embolism, MI, unstable angina requiring revascularization, or death)

with event rates of 4.14% and 5.75% per patient-year, respectively (HR, 0.72; 95% CI, 0.55-0.95; P < 0.001 for noninferiority). Rivaroxaban monotherapy was superior to combination therapy for the primary safety end point of International Society on Thrombosis and Haemostasis major bleeding (1.62% vs 2.76% per patient-year; HR, 0.59; 95% CI, 0.39-0.89; P = 0.01 for superiority).

RECOMMENDATION

29. We recommend OAC alone for patients with AF aged 65 years or older or with a $CHADS_2$ score ≥ 1 and stable coronary or arterial vascular disease (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. The use of combination antithrombotic therapy (eg, OAC with a single antiplatelet agent) is not routinely justified for patients with AF and stable coronary or arterial vascular disease (defined as the absence of ACS or revascularization procedure in the preceding 12 months) because of the observed increased risk of bleeding and all-cause mortality observed with combination therapy, without a significant reduction in ischemic coronary and cerebrovascular thrombotic events.

Practical tip. A combination of an OAC and single antiplatelet therapy may be considered only in highly selected patients with high-risk features for ischemic coronary outcomes, and who are also at low risk of bleeding.

8.3.2.3. PCI or ACS in patients with AF at low risk of stroke/systemic embolism

OAC is not recommended for stroke prevention for patients with AF who are at low risk of stroke/systemic embolism (age younger than 65 years and CHADS₂ score of 0). DAPT is generally prescribed for patients after elective PCI for a period of 6 months and for 12 months after ACS (with or without PCI), as outlined in the 2018 CCS/CAIC antiplatelet guidelines for non-AF patients.²⁴⁰ Some patients with low thrombotic or high bleeding risk may appropriately receive shorter durations of DAPT to decrease the risk of major bleeding. Conversely, longer durations of DAPT might be appropriate for those at higher thrombotic but lower bleeding risk, because premature DAPT discontinuation might increase the risk of stent thrombosis and MI.

8.3.2.4. PCI or ACS in patients with AF at high risk of stroke/systemic embolism

Combination OAC and antiplatelet therapy is required for patients with AF who are aged 65 years or older or with CHADS₂ score \geq 1, who are undergoing PCI or treatment of ACS. In these patients the optimal therapeutic regimen should be individualized on the basis of a balanced assessment of their risk of AF-related stroke/systemic embolism ("CCS algorithm", Fig. 8), ischemic coronary events, and clinically relevant bleeding (Fig. 10). The risk of ischemic coronary events is modulated by the clinical presentation (eg, ACS being higher risk than elective PCI), clinical characteristics (higher risk with comorbid diabetes mellitus or CKD, current tobacco use, or previous stent thrombosis), as well as PCI-related factors (higher risk with multivessel disease, multiple stent implantation, total stent length > 60 mm, bifurcation lesion, chronic total occlusion intervention, and stent type).^{240,241} The risk of bleeding can be estimated from clinical risk scores such as Hypertension, Abnormal Renal/Liver Function, Stroke, Bleeding History or Predisposition, Labile INR, Elderly (>65 Years), Drugs/Alcohol Concomitantly (HAS-BLED), Predicting Bleeding Complications in Patients Undergoing Stent Implantation and Subsequent Dual Anti



Figure 10. Risk factors associated with an increased risk of bleeding and an increased risk of ischemic coronary outcomes. CKD, chronic kidney disease; eGFR, estimated glomerular filtration rate; INR, international normalized ratio; LAD, left anterior descending artery; NSAID, nonsteroidal anti-inflammatory drug; TTR, time in therapeutic range.

Platelet Therapy (PRECISE-DAPT), and **Ca**rdiovascular Disease Research Using Linked Bespoke Studies and Electronic Health Records (CALIBER), with the former validated in a VKA-treated population, and the latter 2 in a population with CAD treated with PCI and DAPT.²⁴⁰

RECOMMENDATION

30. For patients with AF aged 65 years or older or with a $CHADS_2$ score ≥ 1 undergoing PCI without ACS or high-risk features, we recommend dual pathway therapy (OAC with P2Y12) (Strong Recommendation; High-Quality Evidence) for at least 1 month and up to 12 months after PCI (Weak Recommendation; Low-Quality Evidence).

Practical tip. The OAC component evaluated as part of dual pathway therapy regimens include: warfarin daily, apixaban 5 mg BID (reduced to 2.5 mg if they met 2 or more of the following dose-reduction criteria: age older than 80 years, weight < 60 kg, or creatinine > 133 μ mol/L), dabigatran 110 mg or 150 mg orally (PO) BID, edoxaban 60 mg PO daily (30 mg in patients with CrCl 15-50 mL/min, body weight \leq 60 kg, or concomitant use of specified potent P-glycoprotein inhibitors), rivaroxaban 15 mg PO daily (10 mg in patients with CrCl 30-50 mL/min). A DOAC is preferred over warfarin, however, if warfarin is used the lower end of the recommended INR target range is preferred. Clopidogrel is the preferred P2Y12 inhibitor. All patients should receive a loading dose of ASA 160 mg at the time of PCI (if previously ASA-naive).

31. For patients with AF aged older than 65 years or with a CHADS₂ score ≥ 1 undergoing PCI for ACS or elective PCI with high-risk features, we recommend an initial regimen of triple therapy (OAC with P2Y12 and ASA 81 mg/d) (Strong Recommendation; Low-Quality Evidence). After ASA discontinuation, which may occur as early as the day after PCI, we recommend that dual pathway therapy (OAC with P2Y12) be continued for up to 12 months after PCI (Strong Recommendation; High-Quality Evidence).

Practical tip. For some patients younger than 65 years of age with $CHADS_2$ score of 1 at the lower end of the stroke risk spectrum (eg, isolated hypertension), DAPT (eg, aspirin and ticagrelor) may be considered in preference to triple therapy (an OAC with P2Y12 and ASA).

Practical tip. A PCI is considered high-risk for ischemic coronary outcomes on the basis of the clinical presentation (eg, ACS), patient characteristics (comorbid diabetes mellitus treated with oral hypoglycemic agents or insulin, CKD with eGFR < 60 mL/min, current tobacco use, previous ACS, or previous stent thrombosis), as well as PCI-related factors (multivessel PCI, multiple [> 3] stents implanted, total stent length > 60 mm, complex bifurcation lesion, chronic total occlusion intervention, and stent type (eg, bioabsorbable vascular scaffold).

Practical tip. The OAC component evaluated as part of a triple therapy regimens include: warfarin daily, rivaroxaban 2.5 mg PO BID, or apixaban 5 mg BID (reduced to 2.5 mg if they meet 2 or more of the following dose-reduction criteria: age older than 80 years, weight < 60 kg, or creatinine > 133 μ mol/L). A DOAC is preferred over warfarin, however, if warfarin is used, the lower end of the recommended INR target is preferred. Clopidogrel is the preferred P2Y12 inhibitor. All patients should receive a loading dose of ASA 160 mg at the time of PCI (if previously ASA-naive). Thereafter, ASA may be discontinued as early as the day after PCI or it can be continued up to 30 days. The timing of when to discontinue ASA will depend on individual patient's ischemic and bleeding risk.

The population of patients with ACS not needing revascularization (PCI or coronary artery bypass graft [CABG] surgery) represents a heterogenous group. This population includes patients with thrombotic plaque rupture (type I MI) as well as those with supply-demand mismatch (type II MI). For patients with type II MI, it is unclear if there is an advantage to routine use of combined OAC and antiplatelet therapy. For patients with true ACS (type I MI) who are not revascularized, management should take into consideration the relative risk and benefits of combination therapy. Recently, the AUGUSTUS trial affirmed the benefits of apixaban-based dual pathway therapy in medically managed ACS patients.²⁴²

RECOMMENDATION

32. For patients with AF aged 65 years or older or with a $CHADS_2$ score ≥ 1 , we suggest that dual pathway therapy (OAC with P2Y12) be given without concomitant ASA for up to 12 months after medically managed type I ACS (Weak Recommendation; Low-Quality Evidence).

Values and preferences. For patients with AF and type I MI who do not undergo revascularization, the CCS AF Guidelines Committee places relatively greater emphasis on the reduction in ischemic coronary and cerebrovascular thrombotic events, rather than the increase in bleeding observed with combination therapy.

8.3.2.5. Key trials of dual pathway therapy vs triple therapy in patients with AF and ACS/PCI

The key trials that have compared dual pathway therapy with VKA-based TT include: the What Is the Optimal Antiplatelet and Anticoagulation Therapy in Patients With Oral Anticoagulation and Coronary Stenting (WOEST) study, the Prevention Of Bleeding In Patients With AF Undergoing PCI (PIONEER AF-PCI) study, the Randomized Evaluation of Dual Antithrombotic Therapy With Dabigatran versus Triple Therapy With

Trial	DT	$T^{*}\Gamma$	Follow-up	P2Y12 inhibitor	Bleeding* (DT vs VKA TT)	Efficacy outcom	tes (DT vs VKA TT)
WOEST $(n = 573)^{243}$	Warfarin and P2Y12 inhihitor	Warfarin with ASA and P2V12 inhibitor	12 months	Clopidogrel 100%	14.0% vs 31.3%; HR_040:95%	All-cause mortality	2.5% vs 6.3%; HR, 0.39; 95% CL 0 16-0 93
					CI, 0.27-0.58	Myocardial infarction	3.2% vs 4.6%; HR, 0.69; 95%
						Stroke	Li, 0.22-1.00 1.1% vs 2.8%; HR, 0.37; 95%
						Stent thrombosis	CI, 0.10-1.40 1.4% vs 3.2%; HR, 0.44; 95%
PIONEER AF-PCI	Rivaroxaban 15 mg	Rivaroxaban 2.5 mg	12 months	Clopidogrel 93%-96%	16.8% vs 26.7%;	Cardiovascular mortality	Cl, 0.14-1.44 2.4% vs 1.9%; HR, 1.29; 95%
$(n = 2124)^{-12}$	and P2Y12 inhibitor	BID and ASA with P2Y12 inhibitor		l icagrelor 5%-5% Prasugrel 0.7%-1.7%	нк, 0.47-0.76 СІ, 0.47-0.76	Myocardial infarction	CL, 0.59-2.80 3.0% vs 3.5%; HR, 0.86; 95%
		Warfarin and ASA with				Stroke	Cl, 0.46-1.59 1.3% vs 1.2%; HR, 1.07; 95% Cl 0.30.30
						Stent thrombosis	O.8% vs 0.7%; HR, 1.20; 95%
RE-DUAL PCI $(n = 2725)^{245}$	Dabigatran 110 BID	Warfarin and ASA with P2V12 inhibitor	14 months	Clopidogrel 87%-92% Ticaarelor 8%-13%	17.5% vs 26.9%; HR 0.65.95%	All-cause mortality	CI, 0.32-4.49 4.9% vs 4.9%; HR, 1.00; 95% CI 0.71-1.41
				0/01-0/0 Intelement	CI, 0.56-0.75	Myocardial infarction	4.0% vs 3.0%; HR, 1.36; 95%
	Dabigatran 150 BID				110 mg BID: HR, 0.52; 95% CI,	Stroke	CI, 0.89-2.08 1.5% vs 1.3%; HR, 1.13; 95%
	+ P2Y12 inhibitor				0.42-0.64	- - -	CI, 0.58-2.18
					0.72; 95% CI, 0.72; 95% CI, 0.58-0.89	Stent thrombosis	L.3% vs 0.8%; HK, 1.2% vs 0.8%; HK, 1.2% CI, 0.69-3.46
AUGUSTUS $(n = 4614)^{247}$	Apixaban 5 mg BID and P2Y12 inhibitor	Apixaban 5 mg BID and ASA with	6 months	Clopidogrel 93% Ticagrelor 6%	7.3% vs 18.7%; HR. 0.39: 95%	All-cause mortality	3.4% vs 3.0%; HR, 1.15; 95% CI, 0.73-1.81
		P2Y12 inhibitor		Prasugrel 1%	CI, 0.31-0.50	Myocardial infarction	3.3% vs 3.0%; HR, 1.12; 95%
	Warfarin	Warfarin + ASA				Stroke	CI, 0./1-1./0 0.4% vs 2.0%; HR, 0.42; 95%
	+ P2Y12 inhibitor	+ P2Y12 inhibitor				-	CI, 0.15-1.18
						Stent thrombosis	1.8% vs 1.0%; HR, 1.75; 95%
ENTRUST-AF PCI	Edoxaban 60 mg and	Warfarin and ASA with	12 months	Clopidogrel 92%	17.0% vs 20.1%;	All-cause mortality	6.1% vs 4.9%; HR, 1.25; 95%
$(n = 1506)^{246}$	P2Y12 inhibitor	P2Y12 inhibitor		Ticagrelor 7% Prasuorel 1%	HR, 0.85; 95% CL 0.68-1.05	Mvocardial infarction	CI, 0.82-1.90 3.9% vs 3.1%: HR. 1.27: 95%
				QQ			CI, 0.74-2.17
						Stroke	1.3% vs 2.6%; HR, 0.84; 95%
						Stent thrombosis	Cl, 0.50-1.95 1.1% vs 0.8%; HR, 1.34; 95%
							CI, 0.47-3.84
AF. atrial fibrillation	: ASA, aspirin: AUGUSTU	JS, Antithrombotic Therap	v after Acute Coronary	Syndrome or PCI in Atrial I	Fibrillation: BID, twice	e daily: CI, confidence inter	val: DT. dual pathway therapy:

ENTRUST-AF PCI, Edoxaban Treatment Versus Viramin K Antagonist in Patients With Arrial Fibrillation Undergoing Percutaneous Coronary Intervention; HR, hazard ratio; ISTH, International Society on Thrombosis and Haemostasis; PIONEER AF-PCI, An Open-label, Randomized, Controlled, Multicenter Study Exploring Two Treatment Strategies of Rivaroxaban and One of Oral Vitamin K Antagonist in Patients With Arrial Fibrillation Who Undergo Percutaneous Coronary Intervention; RE-DUAL PCI, Dual Antithrombotic Therapy With Dabigatran After PCI In Arrial Fibrillation; TIMI, Thrombolysis in Myocardial Infarction; TT, triple antithrombotic therapy; VKA, vitamin K antagonist; WOEST, What Is the Optimal Antiplatelet and Anticoagulation Therapy in Patients With Oral Anticoagulation and Coronary Stenting. *ISTH major and clinically relevant nonmajor bleeding, except WOEST TIMI major and minor. Warfarin in Patients With Nonvalvular Atrial Fibrillation Undergoing Percutaneous Coronary Intervention (RE-DUAL PCI) trial, the Edoxaban Treatment Versus Vitamin K Antagonist in Patients With Atrial Fibrillation Undergoing Percutaneous Coronary Intervention (ENTRUST-AF PCI) study, and the AUGUSTUS trial.²⁴³⁻²⁴⁷ Relevant details of these trials are presented in Table 8. Collectively these studies showed that the use of DOACs with a P2Y12 inhibitor was associated with a significant reduction in major bleeding (HR, 0.62; 95% CI, 0.47-0.81), without a statistically significant excess in the occurrence of MI (HR, 1.18; 95% CI, 0.93-1.52), definite stent thrombosis (HR, 1.55; 95% CI, 0.99-2.41), and stroke (HR, 0.89; 95% CI, 0.58-1.36).²⁴⁶

When considering these studies, it is important to recognize that a large proportion of patients were undergoing elective PCI for stable CAD (38%-72%), possibly affecting the absolute risk of thromboembolic complications relative to patients with ACS. Second, measures to decrease bleeding risk were underutilized, suggesting that the rates of bleeding in the TT arms might have been greater than in contemporary practice. Third, most of the patients who participated in these trials received clopidogrel as their P2Y12 inhibitor, meaning no conclusions can be drawn about the effects of prasugrel or ticagrelor.

RECOMMENDATION

33. We recommend that clopidogrel 75 mg daily be the preferred P2Y12 inhibitor when dual pathway or triple therapy is used (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation recognizes that clopidogrel was the predominant P2Y12 inhibitor used in the landmark randomized controlled trials of combination therapy in AF patients after PCI/ACS, and clopidogrel is associated with the lowest risk of bleeding among the P2Y12 inhibitors in non-AF studies.

8.3.2.6. Duration of triple therapy

The benefit of TT (reduction of recurrent MI and stent thrombosis) must be balanced against the increased bleeding risk with this therapeutic regimen. In the AF population, it is likely that shorter durations of DAPT after ACS or PCI are reasonable when concomitant OAC will be used for the prevention of stroke/ systemic embolism. This concept was shown in the Intracoronary Stenting and Antithrombotic Regimen: Triple Therapy in Patients on Oral Anticoagulation After Drug Eluting Stent Implantation (ISAR-TRIPLE) study, which showed no significant difference in the primary end point of "net clinical benefit" (combination of death, MI, stent thrombosis, stroke, or Thrombolysis in Myocardial Infarction [TIMI] major bleeding) between those who received 6 weeks vs 6 months of TT with a significant reduction in any bleeding in the group that received 6 weeks of TT.²⁴⁸ A post hoc analysis of the AUGUSTUS trial, which compared the risk of bleeding and ischemic outcomes from randomization to 30 days and from 30 days to 6 months showed that aspirin use (ie, TT) was associated with more severe bleeding (absolute risk difference 0.97%) but fewer ischemic events (absolute risk difference -0.91%) in the first 30 days following randomization, when compared to placebo (ie, dual pathway therapy). From 30 days to 6 months, the risk of severe bleeding was higher with aspirin than placebo (absolute risk difference 1.25%) whereas the risk of ischemic events was similar (absolute risk difference -0.17%). As such, it appears that the benefit of triple therapy on ischemic coronary outcomes is limited to the first 30 days of therapy, with continuation of triple therapy beyond 30 days resulting in increased rates of bleeding without reduction in ischemic outcomes. Likewise, the PIONEER AF-PCI and RE-DUAL PCI trials showed no relationship between major adverse cardiovascular events and TT duration.^{244,245} On balance, these findings suggest that limiting the course of TT to 1 month, and thereafter continuing dual pathway therapy (OAC and a P2Y12) might achieve the optimal balance of ischemic events prevented and bleeding caused.

8.3.3. Liver disease

The optimal management of patients with AF and advanced liver disease is complex. Patients with advanced liver disease are at increased risk of bleeding because of perturbations in coagulation factor synthesis,^{249,250} which was affirmed in a large Swedish registry-based study of 182,678 AF patients.²⁴⁹ In this study the presence of liver disease was associated with an increased risk of major bleeding (adjusted HR, 1.8; 95% CI, 1.45-2.23) even in the absence of concomitant OAC use.²⁴⁹ In addition, there is a concern that advanced liver disease results in a prothrombotic state, which might increase the risk of stroke/systemic embolism.^{249,251,252} Finally, liver disease results in altered metabolism of VKAs and DOACs, which further complicates OAC choice.²⁵³⁻²⁵⁷

Although OACs should be considered in most AF patients with liver disease, the optimal treatment is more challenging because such patients were excluded from the landmark anticoagulation trials. Specifically, patients with significant active liver disease and those with persistent elevation of liver enzymes or bilirubin were excluded from the large RCTs of a DOAC vs VKA.^{21-23,25} Although randomized data are limited, observational studies have shown that VKA use in patients with AF and advanced liver disease resulted in reduction of ischemic stroke risk with a higher risk of bleeding complications.^{250,252,256-258} Moreover, a post hoc analysis of a prospective observational multicentre study of AF patients showed that the presence of advanced liver fibrosis increased the risk of major bleeding in those who received a VKA, but not in those who received a DOAC.²⁵⁹ More recently, a nationwide retrospective Taiwanese cohort study reported outcomes in 2428 cirrhotic patients with NVAF who were taking apixaban (n = 171), dabigatran (n = 535), rivaroxaban (n = 732), or a VKA (n = 990).²⁵⁸ In this study a similar risk of ischemic stroke/systemic embolism between DOAC- and VKA-treated patients was observed, however, the rates of major bleeding (2.9% vs 5.4% per year; P = 0.0003) and gastrointestinal (GI) bleeding (1.9% vs 3.6% per year; P =0.0030) were significantly lower with a DOAC. For patients with advanced cirrhosis the DOAC group had a lower risk of ICH (HR, 0.17; 95% CI, 0.03-0.96; *P* = 0.04).

The Child-Pugh score can be used to guide OAC choice in patients with AF and advanced liver disease.²⁶⁰ This score is commonly used to express the severity of chronic liver disease on the basis of clinical (ascites and encephalopathy) and laboratory (INR, bilirubin, albumin) parameters. When OAC is

being prescribed for AF patients with Child-Pugh grade C cirrhosis it is recommended that a VKA be the preferred agent, because such patients were excluded from the landmark DOAC vs VKA trials.^{21-23,25} For those with Child-Pugh grade B cirrhosis it is recommended that rivaroxaban be avoided because it undergoes substantial hepatic metabolism,^{253,254} whereas apixaban, dabigatran, and edoxaban may be used with caution in such patients.^{256-258,261} Large studies appear to confirm a very low risk of hepatotoxicity with the approved DOACs.²⁶² The use of the appropriate antithrombotic agent in AF patients with advanced liver disease requires individualized therapy and might benefit from the guidance of an expert multidisciplinary team including a hepatologist and hematologist.

RECOMMENDATION

34. We recommend that OAC not be routinely prescribed for patients with AF and advanced liver disease (Child-Pugh grade C or liver disease associated with significant coagulopathy) (Strong Recommendation; Low-Quality Evidence).

Practical tip. In select patients, OAC might be appropriate even in light of advanced liver disease. In these patients, OAC treatment decisions should be made in collaboration with specialized expertise (eg, hepatologists). VKAs can be considered with careful and frequent monitoring but only if the baseline INR is < 1.7. There is no evidence regarding the safety and efficacy of DOACs in patients with advanced liver disease.

8.3.4. Cancer

AF and malignancy are relatively common conditions, with the prevalence of both increasing with age. Patients with cancer have an elevated risk of AF, possibly because of the presence of comorbid conditions, a local/direct tumour effect, systemic inflammation, altered sympathovagal balance, or as a complication of surgical or targeted therapies (eg, tyrosine kinase inhibitors such as ibrutinib).^{263,264} The management of coexisting AF and cancer is challenging, because of the increased rates of thrombosis and thromboembolism observed in association with malignancy, as well as the increased risk of bleeding associated with coagulation defects, blood vessel erosion, tumour vascularity, radiation injury, and the antiplatelet effects of chemotherapies.^{265,266} These factors increase the complexity of OAC decision-making.

The evidence for efficacy of anticoagulation among AF patients with cancer comes from several sources. These include 1 population-based retrospective cohort study, 2 retrospective cohort studies, 1 case-control study, and 3 post hoc analyses of the landmark DOAC vs VKA RCTs.²⁶⁷⁻²⁷³ Among patients where were receiving a DOAC, the annual incidence of thromboembolic events varied from 0.0% to 4.9% with cancer and from 1.3% to 5.1% without; the annual incidence of a major bleed during DOAC treatment varied from 1.2% to 4.4% with cancer and 1.2% to 3.1% without cancer.²⁷⁴ A large registry using prescription-based

analysis for AF patients receiving VKAs or DOACs, with and without cancer reported equivalence for bleeding and thromboembolic risk, although the rates of both were lower in the DOAC population.²⁷⁰ Within the landmark randomized trials of DOACs vs VKAs, there did not appear to be any significant difference in relative efficacy and safety of DOACs compared with VKAs in patients with and without a history of cancer.^{267,271,273} Likewise, RCTs evaluating the cancerassociated venous thromboembolism population have shown that DOACs were noninferior to low molecular-weight heparin (LMWH) for efficacy as well as safety.²⁷⁵

The use of the appropriate antithrombotic agent in a patient with a concomitant malignancy requires individualized therapy under the guidance of an expert multidisciplinary team. When myelosuppressive chemotherapy or radiation therapy is undertaken or intensified, further steps to reduce bleeding risk, such as dose reductions or temporary cessation of OAC might be necessary.

RECOMMENDATION

- 35. We suggest that OAC treatment decisions be individualized for patients with AF and active malignancy, in consideration of the goals of care, the risk of stroke/ systemic embolism, the risk of bleeding, and the concomitant antineoplastic therapy(ies) (Weak Recommendation; Low-Quality Evidence).
- 36. When an OAC is indicated in the presence of active malignancy, we suggest a DOAC in preference to a VKA (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places relatively greater value on the difficulties in ensuring stable INRs and the extensive drug-drug interactions between VKAs and active cancer therapeutic agents.

Practical tip. Although there are no randomized data on the use of DOACs in patients with active cancer and NVAF, this recommendation places a relatively high value on the recognition that DOACs cause no more or less major bleeding compared with VKAs; that they are associated with less ICH compared with VKAs; and on the greater ease of use of DOACs compared with doseadjusted VKAs. The specific choice of OAC should be tailored according to potential drug-drug interactions.

8.3.5. Congenital heart disease

8.3.5.1. Stratification for anticoagulation

Although the link between atrial arrhythmias and thromboembolic complications is less well established in patients with CHD compared with the general AF population, the existing data appear consistent. Absence of sinus rhythm was found to be the factor associated with the highest prevalence of stroke in a cohort of > 23,000 adults with CHD.²⁷⁶ Moreover, a strong association was observed between atrial arrhythmias and stroke in a Quebec administrative database study of > 38,000 patients with CHD.²⁷⁷ As such, questions regarding long-term anticoagulation for atrial arrhythmias frequently arise in the care of adults with CHD.²⁷⁸

Two retrospective studies have assessed anticoagulation practices, thromboembolic event rates, and bleeding complications in adults with CHD and atrial arrhythmias. A singlecentre retrospective study derived from the Congenital Corvitia (CONCOR) registry in the Netherlands followed 229 adults with CHD and intra-atrial reentrant tachycardia (IART), AF, or atrial tachycardia for a median of 6 years.²⁷ Overall, 67% of patients received a VKA and 7% antiplatelet therapy. The thromboembolic event rate in patients without a mechanical valve was 1.4% per year, with a major bleeding rate of 4.4% per year during treatment with a VKA. In univariable analyses, a CHA_2DS_2 -VASc score ≥ 2 was associated with a higher thromboembolic event rate (HR, 3.7; P = 0.021). Similarly, a HAS-BLED score ≥ 2 was associated with a higher major bleeding rate in univariable analyses (HR, 2.9; P = 0.017).

The Anticoagulation Therapy in Congenital Heart Disease (TACTIC) study enrolled 482 patients (age 32.0 \pm 18.0 years) from 12 North American centres.28 Although the study was retrospective, it adhered to clinical trial data management standards including blinded adjudication of arrhythmias and outcomes and multiple layers of data quality control. Patients were classified as having simple (18.5%), moderate (34.4%), or severe (47.1%) forms of CHD on the basis of an established classification scheme.²⁸¹ OAC, predominantly VKAs, were administered to 54% of participants and antiplatelet agents to 38% of participants. Freedom from thromboembolic events was 89% at 10 years and 85% at 15 years. Rates did not differ significantly according to whether patients had AF vs IART. In multivariable analyses, complexity of CHD was the only factor independently associated with thromboembolism. Corresponding thromboembolic event rates in patients with simple, moderate, and severe forms of CHD were 0.00%, 0.93%, and 1.95% per year, respectively (P < 0.001). Notably, no thromboembolic event occurred in patients with simple forms of CHD despite 44% not receiving OAC. In patients with moderate and severe forms of CHD, the thromboembolic event rate exceeded the major bleeding rate associated with a VKA (0.77% per year). Although 93% of the population had a HAS-BLED score of 0 or 1, a higher score was associated with a greater risk of bleeding in multivariable analyses (HR, 3.15; P = 0.047). These results were consistent with the concept of incorporating complexity of CHD in stratification decisions for anticoagulation, as proposed by the Pediatric and Congenital Electrophysiology Society (PACES)/Heart Rhythm Society (HRS) expert consensus statement on the recognition and management of arrhythmias in adults with CHD.²⁸

RECOMMENDATION

37. We recommend that patients who have CHD and concomitant AF or IART receive an antithrombotic therapy regimen on the basis of a balanced assessment of their risk of AF-related stroke, the complexity of CHD, and the risk of clinically relevant bleeding associated with the use of antithrombotic agents (Strong Recommendation; High-Quality Evidence). Values and preferences. This recommendation places a high value on thromboembolic and bleeding risk scores that are well established in the non-CHD population as well as observational studies specific to the CHD population. It recognizes that complexity of CHD was identified as the most powerful predictor of thromboembolism, with a thromboembolic event rate that exceeds the major bleeding rate during oral anticoagulant treatment in those with moderate or severe CHD, regardless of their stroke risk score.

Practical tip. Moderate or severe CHD is defined in the PACES/HRS expert consensus statement for the management of arrhythmias in adults with CHD.²⁸²

38. We suggest OAC for most patients with AF or IART and age 65 years or older, $CHADS_2$ score ≥ 1 , or CHD of moderate or severe complexity (Weak Recommendation; Moderate-Quality Evidence).

8.3.5.2. Choice of anticoagulant

Evidence on the use of DOACs in adults with CHD and atrial arrhythmias has recently emerged in the form of observational studies, as featured in a recent review.²⁸³ In the first series of 75 adults with CHD receiving DOACs, 57 (76%) were anticoagulated for atrial arrhythmias.²⁸⁴ No thrombotic or major hemorrhagic event occurred over a mean follow-up of 12 months. Nevertheless, minor bleeds were observed in 47%. An international registry reported a 1-year experience on the use of DOACs in 530 adults with CHD, 482 of whom had atrial arrhythmias.²⁸⁵ The DOACs consisted of rivaroxaban in 43%, apixaban in 39%, dabigatran in 12%, and edoxaban in 7%. Overall, complexity of CHD was simple in 15%, moderate in 45%, and severe in 40%. Nearly half the population (46%) was considered to have a significant valve lesion. Bioprosthetic valves were present in 11%, with no patient having a mechanical valve. At 1-year of follow-up, the thromboembolic event rate was 1.1%, major bleeding rate 1.3%, and minor bleeding rate 6.3%. In light of this reassuring data and considering that valve disease is common in adults with CHD, it appears reasonable to select a DOAC instead of a VKA in CHD patients with IART or AF and forms of valve disease that are similar to those with reassuring efficacy and safety data from large DOAC trials.²⁸³

However, thromboembolic and bleeding rates in the large international registry were disproportionately high in the 14% of patients with Fontan palliation, who experienced a remarkable 50% rate of thrombotic and bleeding complications.²⁸⁵ In a separate series 10 minor bleeding events occurred in 21 patients with Fontan surgery who received DOACs (12 for atrial arrhythmias).²⁸⁶ One patient with a right-to-left shunt through a fenestration developed deep vein thrombosis during treatment with dabigatran. Another patient receiving apixaban had progression of thrombosis within the Fontan circuit. A separate report described worsening thrombus in a patient with an intracardiac lateral tunnel Fontan and IART during treatment with apixaban.²⁸⁷ Thus, rigourous comparative safety and efficacy data of DOACs vs VKAs in certain subgroups of patients with severe forms of, including Fontan palliation, are required before DOACs can be endorsed for routine use in this setting.²⁸²

RECOMMENDATION

39. When OAC is indicated for atrial arrhythmias in adults with simple or moderate forms of CHD, we suggest a DOAC in preference to a VKA in the absence of recent cardiac surgery (< 3 months), a mechanical valve, and atrioventricular (AV) valve stenosis with enlarged and diseased atria (Weak Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a relatively high value on the growing observational literature that suggests that DOACs appear to have a favourable safety and short-term efficacy (1 year) profile when prescribed for atrial arrhythmias in adults with heterogeneous forms of CHD. It places a high value on established contraindications for DOACs in the context of valvular heart disease and recent cardiac surgery. It also recognizes that there are certain subgroups of patients with severe forms of CHD, such as those with cyanotic heart disease and single-ventricle physiology, in whom there is currently insufficient safety and efficacy data to endorse routine DOAC use. For example, Fontan patients prescribed DOACs were > 6-fold more likely to experience a thromboembolic event than other CHD patients receiving DOACs. There is a need for further research on the safety and efficacy of DOACs in subgroups of patients with severe forms of CHD.

8.3.6. Secondary AF

Although secondary AF has been associated with an increased risk for stroke and mortality irrespective of the precipitant, it is unclear whether the benefits of OAC apply equally to secondary AF patients compared with those with primary AF.^{288,289} As such, it remains uncertain whether such patients should be treated with long-term OAC on the basis of the standard stroke risk stratification schema. Until randomized trials assessing OAC in the secondary AF population are available an individualized approach to OAC is warranted. One such strategy would evaluate the likelihood that the secondary AF is "reversible" (ie, AF that occurs solely secondary to an acute illness, with little to no abnormal underlying substrate and therefore limited risk of recurrence) or 'provoked" (ie, AF that is unmasked by the acute illness, occurring in patients with significant abnormal underlying substrate and therefore significant risk for recurrence). In both instances OAC might be warranted in selected patients in the short-term (ie, until resolution of the acute AF episode) however the long-term need for OAC should take into consideration a given patient's underlying risk of stroke (eg, CHADS-65), the specific secondary precipitant for AF, and the likelihood of recurrence. In some cases, such as sepsis, the acute administration of intravenous (I.V.) anticoagulation increases the risk of bleeding but does not appear to reduce the risk of ischemic events.²⁸⁹⁻²⁹¹ Conversely, OAC is required for most patients with thyrotoxicosis-related AF because hyperthyroidism shifts the balance of the coagulationfibrinolytic system toward a hypercoagulable state, acting as an independent risk factor for thromboembolic events.²⁹²⁻² ⁷⁴ Of note, frequent monitoring of INR is advisable when a VKA is used in patients with hyperthyroid-associated AF, because the potency of VKAs will change as the degree of hyperthyroidism changes.²⁹⁵

RECOMMENDATION

40. We suggest that patients with secondary AF, which has resolved, not be routinely anticoagulated in the absence of recurrence (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places high value on the recognition that secondary AF is often a self-limited process. In the absence of AF recurrence, the current evidence base is insufficient to recommend longterm OAC. However, some patients and providers might elect to pursue long-term OAC on the basis of assessment of the patients underlying risk of stroke and values and preferences.

41. We suggest that most patients with secondary AF due to thyrotoxicosis be anticoagulated until a euthyroid state is restored (Weak Recommendation; Low-Quality Evidence).

8.3.7. Hypertrophic cardiomyopathy

The risk of stroke/systemic embolism in patients with AF and hypertrophic cardiomyopathy (HCM) is substantial and is significantly greater than in the non-HCM population with AF.²⁹⁶⁻³⁰¹ Stroke/systemic embolism in the HCM population appears to be unrelated to the type of AF (paroxysmal vs persistent) or AF burden.²⁹⁷ Moreover, traditional risk prediction scores (eg, CHADS₂ or CHA₂DS₂-VASc) have not been validated in the HCM population, and do not reliably predict thromboembolic outcomes in the HCM population.^{296,297,302} All subjects with HCM who develop AF should started treatment with OAC because there is currently insufficient evidence to support the use of stroke risk prediction tools in the HCM population.^{296,297,301,302}

VKAs have been traditionally used as the preferential OAC in subjects with HCM and AF.³⁰³ Observational data subject to selection and ascertainment bias suggest that DOACs are at least as effective as VKAs in preventing thromboembolic events with similar or lower bleeding and safety events, with some studies suggesting improved survival and greater treatment satisfaction with DOACs.³⁰⁴⁻³⁰⁶ As such, it is suggested that either DOACs or VKAs can be used to reduce thromboembolic events in subjects with HCM and AF.

RECOMMENDATION

42. We recommend that OAC be prescribed for most patients with AF and HCM (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a high value on the knowledge that the annual rate of stroke/systemic embolism in patients with HCM is substantial, and that the mechanism of stroke in HCM patients differs from the general population, which precludes the use of stroke risk prediction algorithms.

Practical tip. Although there are no randomized data on comparisons of DOACs with VKAs, observational studies suggest these agents might be safe and effective in the HCM population.

8.3.8. Cardiac amyloidosis

Patients with cardiac amyloidosis are at a particularly high risk for thromboembolism due to progressive atrial myopathy, electromechanical dissociation, and reduced LAA emptying velocity. LA thrombus has been reported in up to one-third of patients with cardiac amyloidosis in sinus rhythm, as well as in patients with AF who are receiving therapeutic anticoagulation.³⁰⁷⁻³⁰⁹ Among patients with cardiac amyloidosis who undergo cardioversion, 22.4% had intracardiac thrombus identified on TEE, which was significantly more common than in matched controls. Of patients with cardiac amyloidosis with intracardiac thrombus, 4 of 13 had received therapeutic OAC for a period of \geq 3 weeks and 2 of 13 had an AF episode duration of < 48 hours. Patients with amyloid light-chain (AL) amyloidosis appear to be at particularly high risk of intracardiac thrombosis.³¹⁰

There is limited information to inform a specific OAC strategy. Data regarding the safety and efficacy of warfarin and DOACs is limited, however, their use appears to be reasonably safe in patients with cardiac amyloidosis the absence of conventional contraindications.³¹¹

8.3.9. Frail elderly patients

Improvements in life expectancy have resulted in an increasing number of AF patients being cared for into advanced age.^{36,312} Advanced age is a well established risk factor for ischemic stroke and hemorrhagic events in subjects with AF.³¹³⁻³¹⁶ Unfortunately, elderly patients with AF might present with multiple health conditions that might affect QOL (eg, anemia or cognitive impairment/dementia), and/or increase the risk of adverse drug events (eg, impaired renal or hepatic function). In addition, polypharmacy is common in the elderly, which increases the risk of adverse drug reactions due to drug-drug interactions. Further, these patients are often excluded from RCTs, leading to a relative paucity of data to guide treatment decisions.

Unfortunately, despite clear benefit,^{21-23,25,270,317} OAC remains underutilized in this population.^{317,318} A recent study of older patients with AF noted 35% of OAC-eligible patients did not receive anticoagulation.³¹⁷ The major factors responsible for underutilization of anticoagulation in subjects with AF include older age, female sex, abnormal liver enzyme levels, history of falls, excessive alcohol consumption, and concomitant prescription of antiplatelet agents.³¹⁸⁻³²⁷ Despite these concerns, there is evidence that the net clinical benefit favours prescription of OAC in this population.^{328,329}

Although there are limited RCTs specifically in the elderly population with AF, secondary analyses of the landmark DOAC vs VKA trials did not show a difference in efficacy among subjects aged 75 years or older and those aged younger than 75 years.³³⁰⁻³³³ Specifically, a meta-analysis of 4 RCTs (20,165 AF participants 75 years of age or older) showed a significantly lower rate of stroke/systemic embolism in DOAC-treated elderly patients (3.3% vs 4.7%; OR, 0.65; 95% CI, 0.48-0.87), with no significant difference in the rates of major or clinically relevant nonmajor (CRNM) bleeding (6.2% vs 6.6%; OR, 0.82; 95% CI, 0.58-1.16).³³⁴ A pooled analysis from 2 European registries showed that DOAC use was associated with net clinical benefit in AF patients 75 years of age or older; with significant reduction in the composite end point of major bleeding and ischemic cardiovascular events (6.6% per year with DOACs vs 9.1% per year with VKAs; OR, 0.71; 95% CI, 0.51-0.99; *P* = 0.042), which was predominantly due to a lower rate of major bleeding (OR, 0.58; 95% CI, 0.38-0.90; P = 0.013) and a trend to reduction in ischemic events (OR, 0.71; 95% CI, 0.51-1.00; P =0.05) with DOACs.³²

In those at risk of falls, consideration should be made to: provide walking aids and proper footwear; promote cardiovascular, resistance, and balance activities; and, avoid medications that result in hypotension.

RECOMMENDATION

43. We recommend that OAC be prescribed for most frail elderly patients with AF (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places relatively greater value on the observation that elderly AF patients are at higher risk of stroke and, therefore, are more likely to benefit from OAC than younger patients, and places less value on the perceived increased risk of adverse treatment-related events (eg, the risk of bleeding if the patient falls). In general, the net clinical benefit is in favour of anticoagulant therapy in older patients because of the high risk of ischemic stroke.

Practical tip. Treatment decisions regarding specific OAC agents should carefully consider the patient's comorbidity profile, the risk for drug-drug interactions, and the risk of drug-disease interactions.

8.3.10. Increased BMI, obesity, and morbid obesity

Obesity is an established risk factor for the development of AF.¹²⁰ Clinicians face challenges regarding dosing of OAC in obese patients, because increasing BMI is independently associated with a higher risk of bleeding but lower risk of stroke/systemic embolism.335,336 Recent publications regarding outcomes in obese patients enrolled in the landmark DOAC trials are reassuring. A post hoc analysis of the Apixaban for Reduction in Stroke and Other Thromboembolic Events in Atrial Fibrillation (ARISTOTLE) trial revealed that the treatment effect of apixaban was consistent with the overall trial results across all of the weight categories (*P* for interaction = 0.64).³³⁷ Specifically, in 982 patients > 120 kg, there was a significant reduction in major and CRNM bleeding with apixaban (HR, 0.58; 95% CI, 0.35-0.95) with comparable efficacy for stroke/systemic embolism (HR, 0.39; 95% CI, 0.12-1.22), relative to warfarin. The Effective Anticoagulation With Factor Xa Next Generation in Atrial Fibrillation-Thrombolysis in Myocardial Infarction 48

(ENGAGE AF-TIMI 48) trial included 2099 patients (10%) with a BMI of 35-39, 1149 with a BMI > 40 (5.5%), and 148 with a BMI > 50 (0.7%).³³⁵ There was no significant difference in trough edoxaban plasma concentrations and anti-Factor Xa activity across BMI categories, and there was no significant interaction between BMI and clinical outcomes (stroke/systemic embolism, all-cause mortality, major bleeding, major or CRNM bleeding, or the net clinical outcome). The Rivaroxaban Once Daily Oral Direct Factor Xa Inhibition Compared With Vitamin K Antagonist for Prevention of Stroke and Embolism Trial in Atrial Fibrillation (ROCKET-AF) trial enrolled 5206 patients with a BMI $> 30.^{336}$ There was no significant treatment interaction by BMI category (< 25, 25-35, or > 35) for either the primary end point of stroke/systemic embolism, or major or CRNM bleeding.²² Finally, the Randomized Evaluation of Longterm Anticoagulation Therapy (RE-LY) trial included 3099 patients > 100 kg. There was no significant treatment interaction by weight category on the rate of stroke/systemic embolism or major bleeding, with dabigatran 110 mg BID and 150 mg BID being at least as effective as warfarin in obese patients.^{21,338} Meta-analyses of these phase III trials suggest that DOACs are more effective than warfarin for overweight patients (BMI 25-30: OR, 0.87; 95% CI, 0.76-0.99 for stroke/systemic embolism; OR, 0.83; 95% CI, 0.71-0.96 for major bleeding), and similarly effective in obese patients (BMI > 30: OR, 0.87; 95% CI, 0.76-1.00 for stroke/systemic embolism; OR, 0.91; 95% CI, 0.81-1.03 for major bleeding).³³⁸ Although it is important to note that severely obese patients were not well represented, these meta-analyses provide reassurance that DOACs are at least as effective as warfarin in overweight and obese patients.

8.4. Anticoagulation in special circumstances

8.4.1. Cardioversion

8.4.1.1. Increased thromboembolism risk at the time of cardioversion

The incidence of stroke/systemic embolism is increased after cardioversion. A pooled analysis of 16 studies published between 1960 and 1969 showed a 30-day postcardioversion incidence of thromboembolic events of 1.76% (2665 cardioversions) in patients not receiving OAC with AF of > 48 hours in duration.³³⁹ Similarly, a more contemporary pooled analysis of 10 studies describing outcomes after 3564 cardioversions of AF of > 48 hours duration published between 1986 and 2003 showed a 30-day incidence of thromboembolic events of 2.39% in patients not receiving OAC.³³⁹ These rates contrast with the monthly 0.5% background thromboembolic risk of patients with AF in the absence of OAC when cardioversion has not been performed.⁵¹

The increased thromboembolic risk associated with cardioversion might be related to 3 phenomena. The first is generation of thrombi during the persistent AF episode, with subsequent embolization after restoration of organized atrial contraction.³⁴⁰ The second relates to a period of transient atrial mechanical dysfunction after the restoration of sinus rhythm.³⁴¹⁻³⁴³ This "atrial stunning" might be responsible for development of new intracardiac thrombi post cardioversion despite the restoration of sinus rhythm.^{344,345} Such atrial mechanical dysfunction has been reported with pharmacologic, electrical, and spontaneous cardioversion and is maximal in the period immediately after cardioversion.³⁴⁴⁻³⁴⁷ The duration and severity of atrial stunning varies depending on the duration of the atrial arrhythmia, atrial size, and presence of underlying structural heart disease.³⁴⁴⁻³⁴⁷ The third possibility is that cardioversion does not directly cause stroke/systemic embolism but rather patients who require cardioversion have transient as well as permanent characteristics associated with a higher stroke/systemic embolism risk.³⁴⁸

8.4.1.2. Anticoagulation for cardioversion of AF of > 48 hours

In patients with continuous AF for > 48 hours, observational studies reported a lower 30-day incidence of post cardioversion thromboembolic events in patients who receive OAC than in patients who do not receive OAC (0.45% vs 1.76% in studies from 1960-1969; 0.20% vs 2.39% in studies from 1986-2003).³³⁹ Although none of these studies were randomized, the results are compelling in that patients who received OAC were those at highest risk of stroke/systemic embolism (eg, those with rheumatic mitral stenosis and/or previous thromboembolic events). Accordingly, there is widespread agreement that OAC is required for 3 weeks before and 4 weeks after planned cardioversion in patients who present with either "valvular AF," or NVAF of > 48 hours.

8.4.1.3. TEE instead of 3 weeks of OAC before cardioversion

When cardioversion is desired without 3 weeks of therapeutic OAC another approach is to establish immediate anticoagulation, then perform a TEE to exclude LA thrombi before immediate cardioversion. This practice is supported by the Assessment of Cardioversion Using Transesophageal Echocardiography (ACUTE) trial, which randomized 1222 patients with nonacute AF to TEE-guided cardioversion or to the conventional 3 weeks of OAC before cardioversion.³⁴⁹ Although there was no significant difference in the rate of periprocedural thromboembolic events (0.8% in the TEE group vs 0.5% in the conventional group; P = 0.50), patients in the TEE group had fewer hemorrhagic events (2.9% vs 5.5%; P = 0.03), a shorter time to cardioversion (3.0 \pm 5.6 vs 30.6 \pm 10.6 days; P < 0.001), and more frequently had sinus rhythm restored (71.1% vs 65.2%; P = 0.03).

Although other imaging modalities such as cardiac computed tomography (CT), cardiac magnetic resonance (MR) imaging (MRI), and intracardiac ultrasound might have sensitivities and specificities for the detection of LA clot that are comparable with those of TEE,³⁵⁰ only the use of TEE for screening patients for candidacy for cardioversion has been subjected to an RCT. Accordingly, TEE remains the gold-standard procedure for this purpose; nevertheless, the other procedures listed may be used for this purpose when TEE is contraindicated or is not available.

8.4.1.4. Anticoagulation for cardioversion of AF of \leq 48 hours

Patients who present with AF of \leq 48 hours have long been considered, on a theoretical basis, to have a low risk of stroke/ systemic embolism after cardioversion. This practice has been supported by observational reports of outcomes after cardioversion in patients with AF \leq 48 hours, suggesting a low risk of
stroke/systemic embolism in the 30 days after cardioversion (0.27% after 4836 cardioversions in 4380 patients).³³⁹ Although this risk is comparable with that of planned cardioversion in patients with AF of > 48 hours who receive OAC, these reports were limited by selection bias; describing low-risk patients who presented in a stable state early in the 48-hour window, with a significant proportion receiving OAC.³³⁹ Moreover, despite the significant selection bias the rate of stroke/systemic embolism substantially exceeded the established threshold for OAC initiation (1.5% per year or 0.12% per 30 days).³⁵¹

More recent data specifically focused on patients who underwent cardioversion for AF of < 48 hours in the absence of OAC provide a more disquieting viewpoint.352-359 In the Cleveland Clinic Study³⁵⁹ consecutive patients having cardioversion of AF of \leq 48 hours duration were examined, and a significantly higher 30-day postcardioversion rate of stroke/systemic embolism was reported in patients with no or subtherapeutic OAC (0.88%; 6 events after 683 cardioversions) compared with those with therapeutic OAC (0.22%; 2 events after 898 cardioversions; OR, 4.8; P = 0.03). In the ANTI-Kogulation Registry³⁵² 366 consecutive patients with AF of \leq 48 hours were examined, and a significantly higher rate of LA thrombosis (4% vs 0%; P = 0.02) and a nonsignificantly higher rate of 30-day postcardioversion stroke/TIA (1.3% vs 0.5%; P = 0.58) was observed in those who did not receive OAC. Although the latter was not statistically significant, it is noteworthy that each of the patients with a stroke/TIA had a CHADS₂ score of 0.

The FinCV studies were retrospective, observational studies that determined outcomes after cardioversion of AF in catchment area patients of 8 hospitals in Finland between 2003 and 2016.353-358 The program enrolled consecutive patients in whom electrical or pharmacological cardioversion was attempted for AF of \leq 48 hours (FinCV: 3143 patients; 7660 cardioversions), consecutive patients in whom elective electrical cardioversion was attempted for AF of > 48 hours (FinCV2: 1271 patients; 1894 cardioversions), and patients with AF who received DOACs in whom electrical or pharmacological cardioversion was attempted (FinCV3: 1028 patients; 1298 cardioversions). Together these 6 reports, on the short-term outcomes after 10,852 pharmacologic or electrical cardioversions in 5441 patients, demonstrated the following: (1) in the absence of OAC, time to cardioversion is a very strong predictor of 30day risk of stroke/systemic embolism354; (2) in the absence of OAC the incidence of stroke/systemic embolism after cardioversion is significantly higher in patients with AF duration of 12-48 hours than in patients with < 12 hours of AF (1.1% [30/ 2767 patients] vs 0.33% [8/2440 patients], respectively)³⁵⁴; (3) for patients who receive OAC, AF duration is not associated with the incidence of stroke/systemic embolism (0.1% for < 24hours; 0% for 24-48 hours; 0% for 48 hours to 30 days; and 0.2% for > 30 days)³⁵⁸; (4) in the absence of periprocedural OAC, independent predictors of stroke/systemic embolism postcardioversion are older age (OR, 1.05 per year; 95% CI, 1.02-1.08; P < 0.001), female sex (OR, 2.1; 95% CI, 1.1-4.0; P = 0.03), HF (OR, 2.9; 95% CI, 1.1-7.2; P = 0.03), and diabetes mellitus (OR, 2.3; 95% CI, 1.1-4.9; P = 0.03)^{353,355}; and (5) therapeutic OAC significantly decreases the risk of postcardioversion stroke/systemic embolism. 353-357,360

Specifically, for patients with AF of \leq 48 hours the 30 day incidence of stroke/systemic embolism was significantly lower

in the presence of OAC (0.13% [3 events in 2298 encounters] vs 0.71% [38 events in 5362 encounters]; P = 0.001).³⁵⁵ When parsed according to CHA₂DS₂-VASc score, OAC use significantly reduced the 30 day rate of definite stroke/systemic embolism in those with a CHA₂DS₂-VASc score of ≥ 2 (0.2% [3/1; 708] vs 1.1% [28/2590]; P = 0.001), but the benefit of OAC was not statistically significant in patients with a CHA₂DS₂-VASc score of 0-1 (0.0% [0/590] vs 0.40% [10/2772]; P = 0.23). Of note, 10 of the 38 (26%) definite thromboembolic events that occurred after successful cardioversion in patients without OAC occurred in patients with a CHA₂DS₂-VASc score of 0-1.

In 2 other recent reports the thromboembolic risk associated with cardioversion of recent-onset AF in the absence of OAC were indirectly assessed: a report from the Danish National Patient Registry³⁶¹ and a report from the Swedish National Patient Registry.³⁶² Each used administrative databases to evaluate the 30-day rate of stroke/systemic embolism after cardioversion of AF as a function of the presence or absence of precardioversion OAC. Although these databases did not include the duration of AF, the authors proposed that it would be reasonable to assume that cardioversion would only have been performed without previous OAC in patients with recentonset AF of \leq 48 hours. In the Danish analysis,³⁶¹ the 30-day incidence of stroke/systemic embolism after cardioversion was 1.06% (54 events in 5084 patients) without precardioversion OAC and 0.29% (32 events in 11,190 patients) with precardioversion OAC. Stroke/systemic embolism did occur in patients with CHADS₂ or CHA₂DS₂-VASc scores of 0 or 1 but the rates were not stated. In the Swedish analysis,³⁶² the crude 30-day incidence of stroke/systemic embolism in the absence of precardioversion OAC was significantly higher than in the presence of precardioversion OAC (0.86% [104 events in 12,152 patients] vs 0.33% [35 events in 10,722 patients]). After adjustment for unequal distributions of the CHA2DS2-VASc factors, the OR for stroke/systemic embolism in the 30 days after cardioversion was 2.54 (95% CI, 1.70-3.79; P < 0.001) without precardioversion OAC relative to patients with precardioversion OAC. Propensity analysis of 9500 patients who did not receive OAC matched with 9500 patients who did receive OAC showed that patients who did not receive OAC were more likely to have stroke/systemic embolism (OR, 2.51; 95% CI, 1.69-3.75; P < 0.001) with similar rates of major bleeding (OR, 1.00; 95% CI, 0.48-2.10) relative to those who received OAC.³⁶² Taken together, the authors concluded that cardioversion in patients with AF of < 48 hours without OAC conferred a greater risk of stroke/systemic embolism than cardioversion for patients with AF of > 48 hours who received OAC.

In summary, the risk of stroke/systemic embolism after cardioversion is elevated, even in patients who present within 48 hours of AF onset. This risk does not differ on the basis of the method of cardioversion (30 day risk of cardioversion of 0.26% with electrical vs 0.39% with pharmacological cardioversion).³⁶³ Immediate cardioversion without 3 weeks of therapeutic OAC appears to be associated with a low risk of stroke/systemic embolism only in patients who presenting within 12 hours of AF onset and in patients who presenting 12-48 hours after AF onset who have a low risk of stroke (eg, patients aged younger than 65 years with a CHADS₂ score of 0-1). Other patients deemed to be at higher risk of stroke should receive at least 3 weeks of therapeutic OAC before cardioversion (or undergo a TEE to exclude LA thrombus). After cardioversion, all patients should receive at least 4 weeks of OAC in the absence of a strong contraindication, and such therapy should be initiated as soon as possible and preferably before the cardioversion.³⁶⁴ Thereafter, the need for OAC should be on the basis of the risk of stroke/systemic embolism as determined by the "CCS Algorithm" (CHADS-65).

The approach to anticoagulation at time of cardioversion is presented in Figure 11.

RECOMMENDATION

- 44. We recommend that, in addition to appropriate rate control, most hemodynamically stable patients with AF for whom elective electrical or pharmacological cardioversion is planned should receive therapeutic anticoagulation for at least 3 weeks before cardioversion (Strong Recommendation; Moderate-Quality Evidence).
- 45. We suggest that TEE may be used to exclude cardiac thrombus, as an alternative to at least 3 weeks of therapeutic anticoagulation before cardioversion (Weak Recommendation; Moderate-Quality Evidence).
- 46. We suggest that pharmacological or electrical cardioversion of symptomatic AF without at least 3 weeks of previous therapeutic anticoagulation (or TEE) be reserved for patients with the following characteristics (Weak Recommendation; Low-Quality Evidence):
 - A. patients with NVAF who present with a clear onset of AF within 12 hours in the absence of recent stroke or TIA; or,
 - B. patients with NVAF and a $CHADS_2$ score of 0 or 1 who present after 12 hours but within 48 hours of AF onset.
- 47. When a decision has been reached that a patient will be undergoing unplanned pharmacological or electrical cardioversion of AF, we suggest that therapeutic anticoagulation therapy be initiated immediately (preferably before cardioversion) with either: (1) a DOAC; or (2) heparin followed by adjusted-dose VKA (Weak Recommendation; Low-Quality Evidence).
- 48. We suggest that, in the absence of a strong contraindication, all patients who undergo cardioversion of AF receive at least 4 weeks of therapeutic anticoagulation (adjusted-dose VKA or a DOAC) after cardioversion (Weak Recommendation; Low-Quality Evidence). Thereafter, we recommend that the need for ongoing antithrombotic therapy should be on the basis of the risk of stroke as determined by the CCS Algorithm (CHADS-65) (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places relatively greater emphasis on the benefits of stroke prevention and less emphasis on risk of bleeding with a short course of anticoagulation therapy. **Practical tip.** When OAC is to be used for only a short period (< 2 months) the use of a DOAC is preferred to adjusted-dose VKA.

8.4.1.5. DOAC or VKA for pericardioversion OAC

Three randomized trials have compared a DOAC with adjusted-dose VKAs in the setting of planned cardioversion of AF.³⁶⁵⁻³⁶⁷ A meta-analyses of these 3 RCTs³⁶⁵⁻³⁶⁷ showed that DOAC compared with adjusted-dose VKA was associated with significant reduction in stroke/systemic embolism (0.2% vs 0.6%; RR, 0.33; 95% CI, 0.12-0.91; P = 0.03,³⁶⁸ but no significant differences in any bleeding (1.8% vs 2.5%; RR, 0.85; 95% CI, 0.158-1.23; P = 0.38),³⁶⁹ major bleeding (0.4% vs 0.7%; RR, 0.61; 95% CI, 0.28-1.34; P = 0.22),³⁶⁸ or mortality $(0.3\% \text{ vs } 0.4\%; \text{RR}, 0.70; 95\% \text{ CI}, 0.23-2.10; P = 0.52).^{369}$ In a meta-analysis that combined these RCT³⁶⁵⁻³⁶⁷ data with that from patients having cardioversion in the 4 pivotal trials that compared DOAC with VKA therapy,³⁷⁰⁻³⁷³ it was reported that no significant differences were observed in the singular outcomes of ischemic stroke, hemorrhagic stroke, mortality, or major bleeding, or in the composite outcome of stroke/systemic embolism with DOACs compared with adjusted-dose VKAs.³⁷ It should be noted that these trials, individually and in their aggregate, were not sufficiently powered to exclude a clinically meaningful difference in either safety or efficacy.

8.4.2. Catheter ablation of AF

AF ablation is an invasive procedure associated with a risk of stroke/systemic embolism as well as a risk of bleeding. There has been a tremendous amount of research in the past few years regarding periablation OAC management. The use of uninterrupted OAC with a VKA was associated with a lower risk of bleeding and a decreased risk of thromboembolic complications compared with interrupted VKAs with LMWH bridging.³⁷⁶ More recently, several randomized trials have shown superior safety and efficacy of uninterrupted DOAC (apixaban, dabigatran, edoxaban, and rivaroxaban) compared with uninterrupted VKA.³⁷⁷⁻³⁸¹ For these reasons, uninterrupted OAC has been considered the standard of care for AF ablation. Uninterrupted OAC typically means initiating OAC 3-4 weeks before the procedure, continuing OAC for after the procedure.

More recently, 2 trials have assessed the safety of "minimal" DOAC interruption.^{382,383} On the basis of these trials, it might be reasonable to hold a DOAC for 1-2 doses before ablation with early reinitiation 6-12 hours post ablation.

Intraprocedural anticoagulation is critical to avoid the risk of clinical and subclinical cerebral thromboembolism.^{384,385} Anticoagulation during ablation is typically given as I.V. boluses and/or infusion of heparin with frequent measurement of the activated clotting time. More recent guidelines have suggested more aggressive activated clotting time targets of 350-400 seconds to prevent cerebral microembolism.^{384,386}

Post ablation management of OAC can be divided into 2 phases: the initial 2 months after the procedure and the period thereafter. Catheter ablation of AF transiently damages the LA endothelium, creating an early prothrombotic state, which

Andrade et al. 2020 CCS/CHRS Guidelines on Management of AF



Figure 11. Oral anticoagulation pathway in the context of cardioversion for atrial fibrillation (AF) or flutter. CHADS₂, **C**ongestive Heart Failure, **H**ypertension, **A**ge, **D**iabetes, **S**troke/Transient Ischemic Attack; LA, left atrial; NVAF, nonvalvular atrial fibrillation; OAC, oral anticoagulant; TEE, transesophageal echocardiogram; TIA, transient ischemic attack.

might lead to thromboembolism. Such a prothrombotic state can occur even in patients considered to have a low risk of stroke/systemic embolism according to traditional risk schema. For this reason, OAC is recommended for at least 2 months after AF ablation. Thereafter, the need for ongoing OAC should be on the basis of the CCS Algorithm (CHADS-65), rather than the apparent success of the ablation procedure. Although limited retrospective data series have suggested a low risk of stroke after successful AF ablation in a wide variety of patient risk profiles, other studies, including the large randomized Catheter Ablation vs Anti-Arrhythmic Drug Therapy for Atrial Fibrillation (CABANA) trial, have not shown a significant reduction of stroke post ablation.³⁸⁷⁻³⁹⁴ Furthermore, discontinuation of OAC is limited by asymptomatic episodes of AF (meaning patients would be unaware of AF recurrence), late AF recurrence (meaning short-term freedom from recurrent AF might not predict long-term success), as well as a lack of clear temporal association between AF recurrence and stroke/systemic embolism.^{75,95,395} Because of the lack of large RCTs to provide a definitive answer on how to manage OAC late after ablation, at this point in time, AF ablation should not be considered as an alternative to OAC. If a patient has a high thromboembolic risk profile, then the patient should continue OAC even after successful AF ablation. If discontinuation of anticoagulation is being considered on the basis of patient values and preferences, regular and prolonged monitoring for AF screening is suggested. The Optimal Anticoagulation for Enhanced Risk Patients Post-Catheter Ablation for Atrial Fibrillation (OCEAN) trial will provide guidance in the future (NCT02168829).³⁹⁰

RECOMMENDATION

- 49. We recommend that catheter ablation procedures for AF be performed with uninterrupted OAC (Strong Recommendation; High-Quality Evidence).
- 50. We suggest that after successful catheter or surgical ablation of AF, the decision to continue OAC beyond 2 months after ablation should be determined on the basis of the patient's risk of stroke ("CCS algorithm") and not according to the apparent success of the procedure (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on the use of predictive stroke risk indexes in determining long-term stroke risk in all AF and AFL patients, as well as the potential for ongoing risk even after successful ablation.

Practical tip. Limited retrospective and large data series have suggested a low risk of stroke and, therefore, the ability to stop OAC after successful AF ablation. However, there are no randomized trials, or even large prospective studies, to confirm these findings.

8.4.2.1. Catheter ablation of AFL

Patients in sustained AFL need to be anticoagulated for at least 3 weeks before the ablation procedure and at least 1 month after ablation, on the basis of the same principles underlying the rationale for OAC after cardioversion. Similarly, a preprocedural TEE can be used to rule out atrial

Table 9. Perioperative bleeding risk classification

Minimal bleed risk (OAC interruption generally not required)

- Cataract surgery
- · Dermatologic procedures (eg, biopsy)
- Gastroscopy or colonoscopy without biopsies
- Coronary angiography (using radial arterial approach)
- Permanent pacemaker insertion or internal defibrillator placement (if bridging anticoagulation is not used)
- Selected procedures with small-bore needles (eg, thoracentesis, paracentesis, arthrocentesis)
- Dental extractions (1 or 2 teeth)
- Endodontic (root canal) procedure
- Subgingival scaling or other cleaning
- Low/moderate bleed risk (OAC interruption as per Fig. 13)
 - Abdominal surgery (eg, cholecystectomy, hernia repair, colon resection) Other general surgery (eg, breast)

 - Other intrathoracic surgery Other orthopaedic surgery

 - Other vascular surgery Non-cataract ophthalmologic surgery
 - Gastroscopy or colonoscopy with biopsies
 - Coronary angiography*
 - Selected procedures with large-bore needles (eg, bone marrow biopsy, lymph node biopsy)
 - Complex dental procedure (eg, multiple tooth extractions)
- High bleed risk (OAC interruption as per Figure 13)
 - Any surgery or procedure with neuraxial (spinal or epidural) anaesthesia
 - Neurosurgery (intracranial or spinal)
 - Cardiac surgery (eg, CABG, heart valve replacement)
 - Major vascular surgery (eg, aortic aneurysm repair, aortofemoral bypass)
 - Major orthopaedic surgery (eg, hip/knee joint replacement surgery)
 - Lung resection surgery
 - Urological surgery (eg, prostatectomy, bladder tumour resection)
 - Extensive cancer surgery (eg, pancreas, liver)
 - Intestinal anastomosis surgery
 - Reconstructive plastic surgery
 - Selected procedures involving vascular organs (eg, kidney biopsy, prostate biopsy)
 - Selected high bleed risk interventions (eg, colonic polypectomy, spinal injection, pericardiocentesis)

CABG, coronary artery bypass graft.

* The radial approach might be considered minimal bleed risk compared with the femoral approach. An up-to-date risk classification tool is available at Thrombosis Canada (thrombosiscanada.ca).

thrombus before AFL ablation if the arrhythmia has been continuous for more than 48 hours without appropriate anticoagulation."

Analogous to AF, it is reasonable to perform catheter ablation procedures for AFL with uninterrupted OAC.

Continued anticoagulation after successful AFL ablation is generally warranted in light of recognition that patients with isolated typical right AFL have an increased risk of later developing AF. The risk of thromboembolic events remains high despite successful AFL ablation.^{189,398,399} The main predictors for the occurrence of stroke after AFL ablation are older age, higher CHA₂DS₂-VASc score, recurrence of AFL, and subsequent development of AF.^{189,190,399} Multivariate predictors of AF after ablation of AFL include AF before ablation, reduced LV ejection fraction (LVEF), hypertension, increased LA size, inducible AF during the electrophysiology study, endurance sports, and structural heart disease.⁴⁰⁰ In patients without a preexisting history of AF up to 20%-40% will develop AF over the 18-36 months after AFL ablation.^{189,398,400} In patients with a history of preexisting AF up to 40%-75% will continue to experience AF after AFL ablation.400 Although there is no RCT on the benefits of continued anticoagulation after AFL ablation, observation data suggest that patients with AFL and a CHA2DS2-VASc score ≥ 2 have a higher risk of stroke.^{189,190}

8.4.3. Management of OAC for patients undergoing invasive procedures

It is estimated that one in six AF patients who are receiving OAC will require an elective surgery or invasive procedure annually.401 For these patients, the periprocedural risk of a thromboembolic event with OAC interruption must be weighed against the risk of periprocedural bleeding. An OAC interruption interval that is longer than necessary might increase the thromboembolic risk, whereas an insufficient period of interruption might increase the bleeding risk, with consequent delays in OAC resumption.⁴⁰² As such, perioperative anticoagulation management necessitates an assessment of the patient's thrombotic risk, procedural bleeding risk (eg, minimal, low/moderate, and high bleed risk), specific anticoagulant used, renal function, and procedural bleeding risk (Table 9 and Figs. 12 and 13). An electronic tool incorporating these parameters to provide an OAC dosing schedule is provided by Thrombosis Canada (thrombosiscanada.ca).

RECOMMENDATION

51. We recommend that the decision to interrupt antithrombotic therapy for an invasive procedure must balance the risks of a thromboembolic event with those of a periprocedural bleeding event (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. This recommendation is intended to promote a balanced approach to minimize the outcomes of periprocedural thromboembolic events and major bleeding.

Practical tip. The Thrombosis Canada (thrombosiscanada.ca) Perioperative Anticoagulant Management Algorithm is a helpful tool to aid decisions regarding periprocedural anticoagulation.

- 52. We suggest that interruption of OAC is not necessary for most procedures with a minimal risk of bleeding (Weak Recommendation; Moderate-Quality Evidence).
- 53. We recommend interruption of OAC for most procedures with a low/moderate or high risk of bleeding, or those for which the bleeding risk associated with the procedure is uncertain (Strong Recommendation; Low-Quality Evidence).

8.4.3.1. No bridging for interrupted VKA in low thromboembolic risk patients

When a decision to interrupt VKA therapy has been made for an invasive procedure with a low/moderate or high risk of bleeding, interruption of VKA without LMWH or unfractionated heparin (UFH) bridging is recommended for most patients at low thromboembolic risk (Fig. 13). The rationale for the avoidance of bridging has come from several sources. A meta-analysis of 33 observational studies and 1 RCT involving

7118 predominantly non-AF patients showed that VKA interruption with bridging therapy was associated with increased overall bleeding (13.1% vs 3.4%; P < 0.0001) and increased major bleeding (4.2% vs 0.9%; P = 0.004), but without reduction in thromboembolic events (0.9% vs 0.6%; P = 0.50) compared with VKA interruption without bridging therapy.⁴⁰³ As well, in the randomized double-blinded placebo-controlled trial, Bridging Anticoagulation in Patients Who Require Temporary Interruption of Warfarin Therapy for an Elective Invasive Procedure or Surgery (BRIDGE),⁴ 1884 AF patients (most with $CHADS_2$ score < 4) undergoing interruption of VKA for an elective surgery/invasive procedure were evaluated. Patients were randomized to placebo or LMWH from 3 days to 1 day before the procedure, and for 5-10 days after the procedure. No bridging was noninferior to bridging for arterial thromboembolism (0.4% vs 0.3%; P =0.01 for noninferiority), but was associated with significantly less major bleeding (1.3% vs 3.2%; P = 0.005 for superiority) and significantly less minor bleeding (12.0% vs 20.9%; P <0.001). There were no significant differences for any other outcomes (all-cause mortality, MI, deep vein thrombosis, or pulmonary embolism).

RECOMMENDATION

54. When a decision to interrupt VKA therapy for an invasive procedure has been made, we suggest that the interruption begin 5 days before the procedure, that a procedure with a low bleeding risk may proceed when the INR is \leq 1.5, and a procedure with an intermediate or high bleeding risk may proceed when the INR is \leq 1.2 (Weak Recommendation; Low-Quality Evidence).

8.4.3.2. Bridging for interrupted VKA in high thromboembolic risk patients

Bridging should be considered only for patients at high risk of stroke/systemic embolism when a decision is made to interrupt VKA therapy for an invasive procedure with a low/ moderate or high risk of bleeding. Such patients would include those with "valvular AF" (mechanical heart valves or moderatesevere mitral valve stenosis), NVAF with a CHADS₂ score of 5-6, and those with a recent stroke or TIA. Before the procedure, patients should receive 3 days of LMWH bridging, although the dose of LMWH bridging should be reduced by 50% on the day before the procedure (Fig. 13).

RECOMMENDATION

55. When a decision to interrupt VKA therapy for an invasive procedure has been made, we suggest that bridging therapy with LMWH or UFH should be started when the INR is below therapeutic level only in patients at high risk of thromboembolic events (mechanical heart valves, moderate-severe mitral valve stenosis, NVAF with a CHADS₂ score of 5-6, and those with a recent stroke or TIA) (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a relatively higher value on prevention of stroke/systemic embolism, and a relatively lower value on the inconvenience and risk of major bleeding associated with heparin bridging.

8.4.3.3. No bridging for interrupted DOACs

Bridging is not necessary for DOAC-treated patients who need DOAC interruption. DOACs have a rapid onset and offset of action and relatively short elimination half-lives (ie, 10-14 hours), rendering bridging unnecessary from a pharmacological perspective.⁴⁰⁵ In addition, the use of bridging with interrupted DOAC has been associated with more bleeding and limited efficacy. Increased bleeding with bridging was shown in the prospective observational Dresden Registry (76% rivaroxaban, 24% dabigatran) in which heparin bridging did not reduce major cardiovascular events (stroke, venous thromboembolism [VTE], or ACS; 0.8% with no bridging vs 1.6% with bridging; P = 0.265), but led to significantly higher rates of major bleeding complications (2.7% vs 0.5%; P = 0.010).⁴⁰⁶

Post hoc analyses of the phase III DOAC trials²¹⁻²⁵ support no bridging for DOAC interruption. Interruption of dabigatran, rivaroxaban, and apixaban without bridging in these studies had comparable rates of bleeding and thromboembolic events compared with their respective VKA groups.407-409 The Perioperative Anticoagulant Use for Surgery Evaluation for Patients on a Direct Oral Anticoagulant Who Need an Elective Surgery or Procedure (PAUSE) study was a large prospective cohort study of 3007 DOAC-treated AF patients scheduled to undergo an elective surgery or procedure.⁴¹⁰ This study demonstrated the safety with preserved efficacy of a simple standardized DOAC interruption and resumption protocol that did not involve perioperative heparin bridging or preoperative coagulation function testing. Specifically, interruption of DOAC was associated with low rates of major bleeding (1.35% in the apixaban cohort, 0.90% in the dabigatran cohort, and 1.85% in the rivaroxaban cohort) and low rates of stroke/systemic embolism (0.16% in the apixaban cohort, 0.60% in the dabigatran cohort, and 0.37% in the rivaroxaban cohort).

RECOMMENDATION

56. When a decision to interrupt DOAC for an invasive procedure has been made for a patient with AF, we suggest that duration of interruption be on the basis of the risk of bleeding associated with the procedure and the patient's renal function (Weak Recommendation; Low-Quality Evidence).

Practical tip. Duration of preprocedural interruption of a DOAC should be adjusted according to renal function.

57. We recommend no bridging (LMWH or UFH) for NVAF patients requiring DOAC interruption for elective surgery or invasive procedures (Strong Recommendation; Moderate-Quality Evidence).



¹Inform patients/families regarding small thrombotic risk of PCC (e.g. stroke, myocardial infarction, venous thromboembolism), but consequences of uncontrolled bleeding likely exceed this risk

²idarucizumab is unlikely to improve outcomes in patients taking dabigatran with a dilute thrombin time TT <30 ng/mL, normal thrombin time, or a drug level <50 ng/mL ³Procedure bleeding risk is outlined in **Table 9**

⁴Patients considered to be high risk of thromboembolism include those with valvular AF (mechanical heart valves or moderate-severe mitral valve stenosis), non-valvular AF with a CHADS₂ score of 5-6, and those with recent TIA/Stroke (within 3 months).

Figure 12. Management of oral anticoagulant (OAC) use for patients requiring surgical procedures. aPCC, activated prothrombin complex concentrates; DOAC, non-vitamin K direct oral anticoagulant; dTT, dilute thrombin time; INR, international normalized ratio; I.V., intravenous; LMWH, low molecular-weight heparin; PCC, prothrombin complex concentrate; PTT, partial thromboplastin time; UFH, unfractionated heparin; VKA, vitamin K antagonist.

Interruption of OAC for Non-Urgent Procedures

OAC		Day -5	Day -4	Day -3	Day-2	Day-1	Procedure	Day +1	Day +2	Day +3	Day +4
Warfarin Usually no need to interrupt VKA for procedures with low bleeding risk	VKA	No VKA	No VKA	No VKA	No VKA	INR ²	None	VKA ^{5,6}	VKA ^{5,6}	VKA	VKA
	Heparin Bridging ¹	No LMWH	No LMWH	LMWH	LMWH	INR ^{2,3}	None	LMWH ^{6,7}	LMWH ^{6,7}	lmwh	LMWH
DOAC ⁴	Low/Moderate bleeding risk	DOAC	DOAC	DOAC	DOAC	None	None	DOAC ^{6,7}	DOAC ^{6,7}	DOAC	DOAC
	High bleeding risk	DOAC	DOAC	DOAC	None	None	None	None	DOAC ^{6,7}	DOAC	DOAC
Dabigatran + CrCl <50 mL/min	Low/Moderate bleeding risk	DOAC	DOAC	DOAC	None	None	None	DOAC ^{6,7}	DOAC ^{6,7}	DOAC	DOAC
	High bleeding risk	DOAC	None	None	None	None	None	None	DOAC ^{6,7}	DOAC	DOAC

¹Patients in need of bridging during interrupted VKA therapy include those with valvular AF (mechanical heart valves or moderate-severe mitral valve stenosis), non-valvular AF with a CHADS₂ score of 5-6, and those with a recent stroke or transient ischemic attack.

 2 INR should be performed the day prior to the procedure. If >1.5 then consider administering vitamin K PO/IV.

³Give morning LMWH for bid dosed regimens (or ½ daily LMWH dose for once daily dosed regimens).

⁴This schedule applies to factor Xa inhibitors (apixaban, edoxaban, rivaroxaban) and dabigatran (but only when dabigatran is used in patients with a CrCl ≥50 mL/min).

⁵VKA therapy resumption following an invasive procedure may occur almost immediately given it will take several days for the INR to become therapeutic.

⁶Consider withholding anticoagulation therapy for the first 72 hours following cardiac surgery

⁷DOAC/LMWH resumption following an invasive procedure should only occur once hemostasis has been achieved.

Figure 13. Anticoagulation interruption schedule for patients undergoing elective or nonurgent surgery. The need for oral anticoagulation (OAC) interruption is outlined in Figure 12. There is usually no need to interrupt vitamin K antagonists (VKAs) for procedures with low bleeding risk. Certain low bleeding risk procedures may be performed with uninterrupted direct oral anticoagulant (DOAC) use, depending on the patients' risk of stroke/ systemic embolism and physician judgement. For patients undergoing OAC interruption for procedures with low/moderate or high risk of bleeding, the suggested OAC interruption schedule is outlined. Low molecular-weight heparin (LMWH) bridging should be performed for VKA patients at high risk of stroke/systemic embolism (CHADS₂ score 5-6, mechanical heart valves, or recent thromboembolism) with enoxaparin 1 mg/kg twice daily, enoxaparin 1.5 mg/kg once daily, dalteparin 100 IU/kg twice daily, dalteparin 200 IU/kg once daily, or tinzaparin 175 IU/kg once daily. AF, atrial fibrillation; bid, twice daily; CHADS₂, **C**ongestive Heart Failure, **H**ypertension, **A**ge, **D**iabetes, **S**troke/Transient Ischemic Attack; INR, international normalized ratio; CrCl, creatinine clearance; I.V., intravenous; PO, orally.

8.4.3.4. OAC resumption

After the invasive procedure, DOAC is reintroduced after hemostasis has been achieved (usually 24 hours after a low/ moderate bleed risk procedure and 48-72 hours after a high bleed risk procedure). VKA therapy may be reintroduced on the same evening after the procedure, recognizing that it will take several days for the INR to achieve therapeutic range. When bridging LMWH is used, we suggest using a therapeutic-dose regimen for 3 days before the procedure and to resume 24 hours after a low/ moderate bleed risk procedure and 48-72 hours after a high bleed risk procedure. In selected patients, for example, those undergoing a cardiac, intracranial, or spinal procedure, it is suggested to forego postprocedure LMWH bridging altogether, although low-dose heparin regimens, typically used for VTE prophylaxis, can be considered post procedure.

The recommended approach to OAC interruption is presented in Figures 12 and 13 and Table 9.

RECOMMENDATION

58. When OAC has been interrupted for an invasive procedure we suggest that such therapy be restarted when hemostasis is established (within 24 hours for a procedure with a low risk of bleeding and within 48-72 hours for a procedure with a high risk of bleeding) (Weak Recommendation; Low-Quality Evidence).

Values and preferences. Recommendations regarding the timing of postprocedural reintroduction of antithrombotic therapy are intended to promote a balanced approach to minimizing the combined outcome of postprocedural thromboembolic events and major bleeding.

8.5. Bleeding

8.5.1. Prevention

Bleeding is a known adverse effect of OAC. Despite the similar or lower risk of bleeding with DOACs vs VKAs, there remains a significant residual 2%-4% annual risk of major bleeding. Assessment of bleeding risk is important to understand the risk-benefit balance of OAC, to identify and optimize modifiable bleeding risk factors, to identify patients who might benefit from more frequent follow-up, and to provide a framework for ongoing monitoring. The use of a validated bleeding risk prediction algorithm appears to be better at predicting major bleeding than using individual modifiable bleeding risk factors, the identification of modifiable bleeding risk factors before OAC initiation and at each point of contact should be routinely done.

GI protection for patients at high risk of GI events and require concomitant ASA or NSAID therapy has been associated with a reduction in GI bleeding and ulcer complications. In one of the earliest studies, albeit a small study with limited number of events, patients with acute GI bleeding or obstruction due to ulcer who received *Helicobacter pylori* eradication therapy restarted ASA 100 mg daily treatment with lansoprazole 30 mg daily or placebo.⁴¹⁴ The trial was stopped early after the second of 2 planned interim analyses revealed a statistically significant reduction in recurrent bleeding in the patients who received lansoprazole (1.6% vs 15%; HR, 10.6; 95% CI, 1.3-86.1). Subsequently the benefit of using a PPI for patients with no baseline evidence of peptic ulcer disease who required daily ASA (in doses ranging from 75 to 325 mg daily) was shown in the Randomized, Double-Blind, Placebo-Controlled Study to Assess the Prevention of Low Dose Acetylsalicylic Acid (ASA) Associated Gastroduodenal Lesions and Upper Gastrointestinal Symptoms in Patients Taking Esomeprazole 20 mg Once Daily for Two Weeks (ASTERIX) study.⁴¹⁵ In 991 patients 60 years of age or older, the additional use of esomeprazole 20 mg daily reduced the risk of endoscopically diagnosed gastric and duodenal ulcers (1.6% with esomeprazole vs 5.4% with placebo; P = 0.0007). In addition, the use of PPIs lowered the risk of upper GI bleeding in patients who required DAPT with ASA and clopidogrel in the Clopidogrel and the Optimization of Gastrointestinal Events Trial (COGENT) trial, and decreased the occurrence of peptic ulcers in patients at high concomitant risk of cardiovascular and GI events who were taking ASA in the Randomized, Double-Blind, Parallel Group, Multicentre, Phase III Study to Assess the Effect of Esomeprazole 20 or 40 mg od Versus Placebo on the Occurrence of Peptic Ulcers During 26 Weeks in Subjects on Continuous Low Dose Acetylsalicylic Acid (OBERON) trial.416,41

There is no published RCT in which PPI use was investigated in patients with AF who were receiving standard stroke prevention dose OAC alone or in combination with or. In a retrospective cohort study of > 1.6 million Medicare beneficiaries who were receiving anticoagulation, PPI co-therapy was associated with a lower rate of upper GI bleeding hospitalizations: incidence rate ratio, 0.66 (95% CI, 0.62-0.69).418 However, more recently in the COMPASS - Proton Pump Inhibitor (COMPASS-PPI) study⁴¹⁹ the effect of PPI use in patients receiving ASA in combination with rivaroxaban 2.5 mg BID was investigated. In this study, patients with stable atherosclerotic vascular disease and no clinical need for a PPI who were receiving 1 of 3 regimens: rivaroxaban 2.5 mg BID with ASA 100 mg daily, rivaroxaban 5 mg BID, or ASA 100 mg daily without anticoagulation, were randomized to pantoprazole 40 mg daily or placebo. The main reason for exclusion from the PPI arm of this trial was physician judgement that their patients had a clinical need for a PPI; this accounted for more than 35% of eligible patients (n = 9797). Of the 17,598 patients randomized, the mean age was 68 years, 78% were male, and 2.6% had baseline peptic ulcer disease. These patients were followed for a mean of 3 years. Clinically significant GI events were no different in the 2 arms: 1.2% in the PPI arm vs 1.3% in the placebo arm (HR, 0.88; 95% CI, 0.67-1.15). The event rate in the nonrandomized cohort was 0.6%. In an analysis of the individual GI events included in the primary outcome, there was a reduction in overt gastroduodenal bleeding with pantoprazole (0.4% per year) over placebo (1.2% per year; HR, 0.52; 95%) CI, 0.28-0.94; P = 0.03). In addition, the rate of discontinuation of pantoprazole was similar to that of placebo overall (21.4% vs 22.4%) and for serious adverse events (0.9% vs 0.75%). Of note, a Cochrane review is currently under way to assess the benefits and harms of PPIs vs histamine-2 receptor antagonists or placebo for the prevention of upper GI bleeding in adults receiving single-agent and combination antithrombotic therapy for cardiovascular conditions.⁴²⁰

RECOMMENDATION

59. We recommend initial and ongoing evaluation of bleeding risk for all patients with AF whose stroke risk warrants antithrombotic therapy, with the use of strategies to mitigate the increased risk of bleeding associated with OAC (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places greater emphasis on routinely and systematically evaluating bleeding risk factors in patients who are receiving antiplatelet or OAC with emphasis on modifying bleeding risk when possible and less emphasis on using a bleeding risk stratification score to determine whether to initiate antithrombotic therapy or not.

Practical tip. Individual patient bleeding risk should be taken into account when determining the strategy for stroke risk management. The use of the HAS-BLED algorithm might be valuable to identify patients at high risk of major bleeding who might benefit from more frequent follow-up.

Practical tip. Emphasis should be placed on the identification of modifiable risk factors to decrease the risk of major bleeding at each medical encounter. These include BP control, avoidance of unnecessary antiplatelet and NSAID therapy, management of anemia, limiting alcohol intake, improving INR control, and ensuring patients are receiving DOAC doses that align with the Health Canada-approved dosing recommendations.

Practical tip. Patients receiving OAC should be educated regarding self-monitoring for bleeding, including when to seek medical help.

60. We suggest the additional use of a PPI to decrease the risk of GI adverse effects, for patients who require daily antithrombotic therapy that includes ASA (Weak Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places greater value on the published RCT evidence of decreased GI complications, including GI bleeding, with the additional use of a PPI with antithrombotic therapy, and lesser value on the short duration of PPI treatment used in the trials and the fact that the trials did not specifically enroll patients with AF.

Practical tip. Gastroprotection with a PPI has been shown to decrease the risk of GI ulceration and other adverse bleeding events in patients receiving daily ASA who are older than 60 years or have at least 1 other risk factor for bleeding. It stands to reason that this benefit would be extended to patients receiving the combination of daily ASA and anticoagulation, despite the lack of published evidence in this patient group.

8.5.2. Management of a bleeding event

DOACs are the preferred agents for stroke prevention in NVAF patients who merit anticoagulation. Although less life-threatening bleeding was shown with DOACs, annual rates of

major bleeding were observed to be 2%-4%.^{21-23,25} Bleeding management protocols for DOACs have included supportive therapy alone, antifibrinolytic therapy, hemostatic factors such as fresh-frozen plasma, 3- or 4-factor prothrombin complex concentrates, and recombinant activated factor VII; activated charcoal for overdose or unintentional ingestion; dialysis or continuous renal replacement therapy; and, specific reversal agents such as idarucizumab and andexanet alfa. An approach to bleeding is outlined in Figure 14.

RECOMMENDATION

61. In patients experiencing a bleeding event with OAC treatment we recommend investigation into the cause of the bleeding (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a relatively high value on the recognition that patients who experience a GI or genitourinary bleeding event after OAC initiation are at higher risk of having an underlying malignancy discovered as the cause of their bleeding.

62. We recommend that anticoagulant therapy should be recommenced in patients at high risk of stroke as soon as possible after the cause of bleeding has been identified and corrected (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a relatively high value on the recognition that OAC discontinuation after a bleeding event is associated with a significant increase in the risk of stroke and all-cause mortality.

Idarucizumab is a humanized monoclonal antibody fragment that binds to protein-bound and unbound dabigatran with high affinity, neutralizing dabigatran and its active metabolites.⁴²¹ In phase I trials with > 200 volunteers, idarucizumab was well tolerated.⁴²²⁻⁴²⁴ In the Reversal Effects of Idarucizumab on Active Dabigatran (REVERSE-AD) trial, adults with overt, life-threatening bleeding (n = 301) or those requiring urgent invasive procedures (n = 202) while receiving dabigatran received idarucizumab as two 2.5-g bolus infusions up to 15 minutes apart.⁴²⁵ In those with overt bleeding, 137 (45.5%) presented with GI bleeding and 98 (32.6%) with ICH. Urgent surgery was defined as a procedure requiring normal hemostasis that could not be delayed for at least 8 hours. The median maximum percent reversal of the anticoagulant effect of dabigatran was 100% within 4 hours of idarucizumab administration in those who had a prolonged dilute thrombin time or ecarin clotting time at baseline (92%). Reversal was rapid and occurred independently of age, sex, renal function, and baseline dabigatran concentration. was 2.5 hours after idarucizumab administration in 134 patients (45%) with uncontrolled bleeding, and all had confirmed bleeding cessation within 24 hours. Of the 197 patients who underwent surgery, periprocedural hemostasis was assessed as normal or mildly abnormal in 98.5%. At 30 days, thrombotic

events occurred in 4.8% and the mortality rate was 13.5% in those with uncontrolled bleeding and 12.6% in those requiring an urgent procedure. Postmarketing experience in large, practice-based cohorts of consecutive patients, has shown hemostatic effectiveness and clinical outcomes similar to those seen in REVERSE-AD.^{426,427}

RECOMMENDATION

63. We recommend administering idarucizumab for emergency reversal of dabigatran's anticoagulant effect in patients with uncontrollable or potentially life-threatening bleeding and/or in patients who require urgent surgery for which normal hemostasis is necessary (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a relatively greater value on the ability of idarucizumab to reverse coagulation parameters indicative of dabigatran's effect, its potential to decrease bleeding-related outcomes, and risks of urgent surgery, and its safety and tolerability profile, and less value on the absence of a control group in the REVERSE-AD trial and on the cost of the drug.

Practical tip. Reversing OAC exposes patients to the thrombotic risk of their underlying disease. OAC should be reintroduced as soon as medically appropriate.

Andexanet alfa is a recombinant modified human factor Xa decoy protein with high affinity for the active site of factor Xa inhibitors, sequestering them within the vascular space.⁴²⁸ In patients requiring anticoagulation that involves inhibition of Factor Xa, the presence of andexanet alfa will inhibit the anticoagulant effectiveness. The pharmacodynamic half-life is approximately 1 hour. The Andexanet Alfa, A Novel Antidote to the Anticoagulation Effects of Factor Xa Inhibitors (ANNEXA-4) was a single-group cohort study designed to assess the efficacy and safety of andexanet alfa in patients with acute major bleeding occurring while taking a factor Xa inhibitor. Patients who had received 1 of 4 factor Xa inhibitors (apixaban, edoxaban, enoxaparin, or rivaroxaban) within 18 hours were treated with a bolus and 2-hour infusion of andexanet alfa.⁴²⁹ The dose of andexanet alfa was dependent on the timing of the last dose of factor Xa inhibitor. Patients were excluded if they required surgery within 12 hours, if they received other bleeding treatments such as prothrombin complex concentrate, recombinant factor VIIa, whole blood, or plasma, and if they had a Glascow Coma Scale of < 7, an estimated ICH volume of > 60 mL, or an expected survival of < 1 month. Patients enrolled (n = 352) had a mean age of 77 years. Bleeding was predominantly ICH (64%) and GI (26%). Most of the patients were receiving apixaban (54%) or rivaroxaban (36%). Anti-factor Xa activity decreased by 92% for apixaban (149.7 ng/mL to 11.1 ng/mL) and rivaroxaban (211.8 ng/mL to 14.2 ng/mL). Of 249 patients who were eligible for the hemostatic efficacy analysis, 204 were adjudicated as having excellent or good hemostasis at 12 hours. Reduction in anti-factor Xa activity was not predictive of hemostatic efficacy overall but was modestly predictive in patients with ICH. At 30 days of follow-up of the overall

Andexanet alfa received accelerated Food and Drug Administration approval in the United States in 2018 for the management of life-threatening or uncontrolled bleeding in patients treated with rivaroxaban and apixaban.⁴³⁰ A condition of this accelerated approval was the requirement to complete an RCT including patients with acute ICH, the results of which might affect the modification or withdrawal of approval. The Food and Drug Administration also mandated a black box warning that and exanet alfa has been associated with thromboembolic events, ischemic events, cardiac arrest, and sudden death. There appears to be some increase in endogenous thrombin generation potential upon bolus administration of andexanet alfa although the mechanism is not yet known⁴³¹ and did not appear to be associated with clinical thrombotic events in healthy volunteers.⁴²⁸ The thrombosis rate in a single health system retrospective cohort study of 13 patients who received and exanet alfa was 30% (4 patients) and mortality was 15% (2 patients).⁴³² In another, the thrombosis rate was 0 and mortality was 40%. 433 Practical challenges reported with andexanet alfa include access to product, lack of availability of anti-factor Xa levels, and drug preparation requiring reconstitution of multiple vials.⁴³²

RECOMMENDATION

64. We recommend administering and exanet alfa (when available) for emergency reversal of the anticoagulant effect of apixaban, edoxaban, and rivaroxaban in patients who present with uncontrollable or potentially life-threatening bleeding who have received any of these agents within the preceding 18 hours (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places relatively greater value on the ability of andexanet alfa to reverse anti-factor Xa activity and its potential to facilitate hemostasis, and less value on the low risk of infusion reactions and the absence of a control group in the ANNEXA-4 trial.

Practical tip. Patients who are anticipated to require unfractionated or low molecular-weight heparin within 12-24 hours should not receive and exanet alfa.

Practical tip. The dose of andexanet alfa is on the basis of the specific FXa inhibitor, the dose the patient is taking, and the timing of the last dose before the bleed.

Practical tip. Reversing OAC exposes patients to the thrombotic risk of their underlying disease. There is evidence of thrombin generation after administration of andexanet alfa. OAC should be reintroduced as soon as medically appropriate.

8.6. Secondary stroke prevention

8.6.1. Stroke during OAC treatment

Use of OAC is associated with a substantial risk reduction in the occurrence of ischemic stroke (see section 8.2), however, some patients treated with OAC will still experience ischemic stroke or TIA despite use of OAC.^{51,52} Studies have shown that strokes that occur in patients receiving OAC tend to be less severe than strokes that occur in AF patients not taking OAC; in VKA-treated patients, stroke severity has been inversely correlated with INR level at the time of the event.^{434,435}

After confirmation of the stroke diagnosis, an on-treatment event should prompt an evaluation for potential causes (other than AF) and ensure optimization of OAC and risk factor management.

RECOMMENDATION

65. We recommend that patients with AF who experience an ischemic stroke while receiving OAC be managed acutely according to the secondary stroke prevention practice guidelines (eg, Canadian stroke best practice recommendations¹¹⁸), with emphasis on addressing OAC medication adherence, ensuring correct OAC dosing and avoidance of drug interactions, identifying and treating other potential causes for the stroke other than AF, and promoting general vascular risk factor modification and healthy lifestyle choices (Strong Recommendation; Moderate-Quality Evidence).

8.6.1.1. Potential causes of an ischemic stroke in a patient receiving OAC

- I. Inadequate intensity of anticoagulation. In patients treated with VKAs, a subtherapeutic INR is associated with an increased risk of stroke as well as greater stroke severity.^{434,435} In those treated with DOACs, undertreatment (eg, low plasma levels) has been associated with higher risk of thromboembolic events, and increased stroke severity (eg, large vessel occlusion).^{205-207,436} For DOAC-treated patients, undertreatment might take the form of the prescription of a reduced dose without an indication to do so, failure to anticipate drug-drug interactions (eg, OAC levels might be decreased by interaction with anticonvulsants such as phenytoin) or, in the case of rivaroxaban, a failure to account for GI absorption. Specifically, rivaroxaban should be taken with food for maximum GI absorption.
- II. Suboptimal medication adherence or persistence. Poor patient adherence is a common problem with many chronic medications, adherence and persistence to stroke prevention therapies being particularly problematic in the AF population. As outlined in section 7.2 suboptimal adherence and persistence with OAC has been associated with higher rates of all-cause mortality and stroke.^{156,157,160-162} Missed doses is a particular concern for DOACs because of their short half-life (average 12 hours). Strategies to improve persistence and adherence are outlined in section 7.2 and Table 6.
- III. Alternate stroke etiology. Patients with AF might have other etiologies for stroke, some of which require specific management. Examples include large artery atherosclerosis of the intra- or extracranial circulation, cerebral small-vessel disease, other cardiac sources of embolism

(eg, patent foramen ovale) or another determined cause (carotid or vertebral artery dissection, vasculitis, etc).

IV. Suboptimal risk factor management. Untreated or suboptimally controlled stroke risk factors (eg, hypertension, dyslipidemia, etc) might contribute to the occurrence of stroke, independent of the presence of AF. Of particular importance, reduction in BP reduces the risk of recurrence of ischemic stroke as well as incident hemorrhagic stroke.¹¹⁸

8.6.1.2. Practical clinical approach to patients with an ischemic stroke who are receiving OAC therapy

- I. Ensure an accurate diagnosis of the event. Brain imaging with CT or MRI is recommended for the diagnostic evaluation of acute stroke to exclude ICH, hemorrhagic transformation of ischemic stroke, and other structural pathology (MRI has greater diagnostic sensitivity than CT for identifying acute ischemia and small lesions). TIAs are frequently overdiagnosed, with up to 60% of patients suspected to have had a TIA having nonischemic causes for their symptoms after systemic evaluation.^{437,438} Although TIA remains a clinical diagnosis mainly on the basis of a detailed history of the event, brain imaging is useful in assessing for mimics (eg, migraine aura, seizure, peripheral vertigo, presyncope, neuropathy, multiple sclerosis) including small sub-arachnoid hemorrhages.⁴³⁹
- II. Investigate for the most likely etiology of the stroke event. A typical etiological stroke workup for a patient with known AF consists of brain imaging (CT or MRI); vascular imaging (carotid ultrasound at a minimum; ideally CT angiography or MR angiography from the aortic arch to vertex), echocardiography, and laboratory investigations.
- III. Determine if any other treatments are indicated for stroke risk reduction. For example, symptomatic carotid artery stenosis that is moderate (50%-69%) or severe (70%-99%) might benefit from urgent revascularization, whereas nonsurgical management is usually recommended for asymptomatic carotid stenosis.¹¹⁸
- IV. Assess medication adherence. Patients and/or family members should be questioned regarding missed doses or prolonged interruptions (eg, periprocedural or in association with a bleeding event). For VKA-treated patients, the INR record for the preceding months should be reviewed. Patients should be counselled about the importance of daily medication adherence, the dangers of missed doses, and avoidance of prolonged or unnecessary treatment interruptions. Encourage adherence- and persistence-enhancing strategies as outlined in section 7.2 and Table 6.
- V. Determine whether or not any changes are needed to the patient's current OAC (agent or dose). For VKA-treated patients with suboptimal INR control, either switch to a DOAC (if eligible) or aim for improved INR control (consider a higher target INR and more frequent INR monitoring). For DOAC-treated patients, it is reasonable to continue the same DOAC (after ensuring it is at the correct dose for the patient's age, renal function, and body weight) or switch to a different DOAC. For patients unable to comply with DOACs, then a switch to a VKA



nitiate OAC as soon as possible after the cause of bleeding has been identified and corrected. Re-evaluate concomitant medications which may contribute to bleeding (e.g. ASA, NSAIDs)

¹Examples of mild bleeding include extremity bruising, hemorrhoid bleeding, subconjunctival bleed, self-limited epistaxis.

²Examples of significant non-life-threatening bleeding include hemodynamically stable gastrointestinal bleed, epistaxis, hematuria, or menstrual bleeding, requiring medical attention and/or intervention.

³Examples of severe life-threatening bleeding include intracranial hemorrhage, or severe gastrointestinal bleed with actual or impending hemodynamic instability, retroperitoneal bleed, intramuscular bleed with compartment syndrome.

⁴Inform patients/families regarding small thrombotic risk of PCC (e.g. stroke, myocardial infarction, venous thromboembolism), but consequences of uncontrolled bleeding likely exceed this risk.

⁵Idarucizumab is unlikely to improve outcomes in patients taking dabigatran with a dilute thrombin time (dTT) <30 ng/mL, normal thrombin time, or a drug level <50 ng/mL. ⁶Pending approval and availability.

⁷Do not administer hemostatic agents (PCC, aPCC, rVIIa) if antidote has been given.

⁸A delayed restart is recommended when the bleed occurred in a critical site, the patient has a high risk of rebleeding or of death from a rebleed, the source of the bleed cannot be identified, or future surgical interventions are planned.

Figure 14. Management of bleeding for patients receiving oral anticoagulation (OAC). aPCC, activated prothrombin complex concentrates; ASA, acetylsalicylic acid; DOAC, non-vitamin K direct oral anticoagulant; dTT, dilute thrombin time; FFP, fresh frozen plasma; INR, international normalized ratio; I.V., intravenous; NSAID, nonsteroidal anti-inflammatory drug; PCC, prothrombin complex concentrate; PO, orally; PTT, partial thromboplastin time; RBC, red blood cell; rVIIa, recombined activated factor VII; VKA, vitamin K antagonist.

might be reasonable because of the longer half-life and ability to monitor the INR. The additional use of an antiplatelet agent is sometimes considered, however, the additional use of an antiplatelet agent with therapeutic OAC is associated with a significant increase in bleeding events without an additional benefit for stroke prevention. LAA occlusion (LAAO) might be an option for some patients but the optimal candidates for this procedure remains unclear. VI. Identify and address any untreated vascular risk factors. Optimize BP and lipids, recommend smoking cessation, and promote general secondary stroke prevention lifestyle recommendations (diet, exercise).

For further details on the etiological stroke workup and secondary stroke prevention treatment recommendations, see the Canadian stroke best practice recommendations¹¹⁸ available at: strokebestpractices.ca.

8.6.2. Timing of OAC initiation after acute ischemic stroke in patients with AF

In the 2-week period immediately after a TIA or ischemic stroke, patients with AF have an increased risk of ischemic stroke that ranges from 0.5%-1.3% per day.^{440,441} Initiation of OAC during this phase must balance the benefit of OAC with the risk of hemorrhagic transformation of the acute brain infarct. Hemorrhagic transformation can range from asymptomatic petechiae to symptomatic parenchymal hematoma with mass effect.^{442,443} Hemorrhagic transformation is more frequent in large-volume infarcts and in the setting of thrombolysis or mechanical revascularization of the index stroke.^{442,443}

The optimal timing for OAC initiation post stroke has not yet been clearly defined and practice patterns are highly variable.⁴⁴⁴ RCTs are under way to help address this uncertainty. For now, the heterogeneity of strokes and the lack of randomized data (particularly for DOACs in the early poststroke period) preclude any specific recommendations regarding the timing of OAC initiation. In practice, treatment decisions are typically individualized on the basis of a benefit-risk assessment guided by brain imaging appearance and clinical context. Bridging with low-dose aspirin is often prescribed until OAC treatment is initiated. Serial head CT scans within the first days or weeks post stroke can be helpful to monitor the evolution of an acute infarct, especially for moderate or large infarcts, and to assess for the presence and extent of any hemorrhage, before OAC treatment initiation.

Outcomes in relation to timing of OAC initiation have been studied in prospective cohort studies.445,446 In the Clinical Relevance of Microbleeds in Stroke-2 study 1490 participants with AF and stroke or TIA in whom anticoagulation was indicated were enrolled.⁴⁴⁶ Of these, 1335 (90%) had a known date of OAC initiation post stroke that was dichotomized as early (≤ 4 days) or late (\geq 5 days or never). The groups differed significantly, with lower stroke severity, better premorbid functioning, and a lower probability of thrombolysis for the index event in those in the early OAC group. At 90 days of follow-up the combined end point of intracerebral hemorrhage, ischemic stroke, and death was similar between groups (2% with early and 5% with late initiation; adjusted OR, 1.17; 95% CI, 0.48-2.84; P = 0.736). Recognizing that the event rate was low, the authors concluded that there did not appear to be a hazard associated with early anticoagulation in carefully selected patients. Results of a prospective cohort study consisting of 1029 individuals with ischemic stroke and AF suggested that initiation of anticoagulation between 4 and 14 days optimized the combined outcome of stroke, TIA, systemic embolus, symptomatic intracranial bleeding, and bleeding within 90 days of onset. 445 Most participants were treated with VKAs or mixed treatment protocols including heparin, with only 12% treated with DOACs.

Results of 2 small RCTs have suggested the early use of DOACs is safe for small strokes using an MRI measure of hemorrhagic transformation.^{447,448} Participants (N = 195) randomized to VKA or rivaroxaban with a median National Institutes of Health Stroke Scale (NIHSS) score of 2 (ie, very mild strokes) did not differ in the rate of hemorrhagic transformation detected using MRI, an imaging modality with high sensitivity for the detection of hemorrhage.⁴⁴⁷ The Dabigatran Following Acute Transient Ischemic Attack and minor stroke trial (DATAS II) randomized 301 individuals

with a median NIHSS score of 1 (ie, very mild strokes) to ASA or dabigatran within 72 hours of symptom onset. No symptomatic hemorrhagic transformation was observed and the rate of minor petechial hemorrhage within the acute infarct did not differ between the treatment arms. Infarct volume predicted the probability of hemorrhagic transformation.⁴⁴⁸ These trials do not establish optimal time points for OAC initiation but suggest that an identifiable group of patients with very mild stroke severity and small volume infarcts can be safely anticoagulated early.

RECOMMENDATION

66. We recommend that the timing of initiation of anticoagulant therapy after an ischemic stroke should be individualized and take into account the competing risks of recurrent stroke against the risk of hemorrhagic transformation of infarction (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. The timing of OAC initiation or resumption might be as short as within 24 hours in patients with a TIA, 3 days in patients with a mild stroke (NIHSS score < 8), 6 days in patients with a moderate stroke (NIHSS score 8-15), and 12-14 days in patients with a severe or large stroke (NIHSS score \geq 16). Factors favouring delayed OAC initiation include: high NIHSS score (> 8), moderate-large brain infarction on imaging, hemorrhagic transformation, neurological instability, advanced patient age, and uncontrolled hypertension. In these patients OAC treatment decisions should be made in collaboration with specialized expertise (eg, neurology).

In the absence of definitive data, the following is a pragmatic approach to the timing of OAC initiation or resumption after an ischemic stroke event.

- I. It is reasonable to initiate a DOAC 1 day post event for patients with a TIA, 3 days for patients with a small infarct/mild stroke severity (NIHSS score < 8), 6 days in patients with a moderate-sized infarct/moderate severity stroke (NIHSS score 8-15), and 12-14 days in patients with a large infarct/severe stroke (NIHSS score >15). For a brief-duration TIA with no residual symptoms or deficits and no acute infarct or hemorrhage on head CT scan, it is reasonable to consider DOAC initiation on the day of the event provided timely follow-up is available (Figure 15).
- II. It is reasonable to delay OAC initiation for more than 2 weeks post stroke if, in the judgement of the clinician the risk of bleeding is believed to be high (eg, for some patients with large infarcts and those with hemorrhagic transformation).
- III. If OAC initiation is recommended after hospital discharge, coordination with the treating physicians and close followup is suggested because postdischarge recommendations are implemented in only two-thirds of patients.⁴⁴⁹
- IV. Hypertension is a significant risk factor for intracerebral hemorrhage, and it is reasonable to delay OAC in the setting of uncontrolled hypertension.



1895

Figure 15. Oral anticoagulation (OAC) initiation or reinitiation after an acute ischemic stroke. The timing of OAC initiation should be evaluated on the basis of the severity of stroke, the risk of short-term recurrence, and the risk of secondary hemorrhagic transformation. AF, atrial fibrillation; NIHSS, National Institutes of Health Stroke Scale; TIA, transient ischemic attack.

- V. In most instances, brain imaging should be repeated before initiating OAC to detect asymptomatic hemorrhagic transformation that might influence the timing for OAC initiation. Consideration should be given to delaying OAC in patients with hemorrhage other than minor petechial changes, because OAC in the presence of existing hemorrhage might increase bleeding. Delayed initiation in patients with hemorrhagic transformation was not associated with increased risk of ischemic stroke in an observational study of > 2000 patients with stroke and AF.⁴⁵⁰
- VI. Neurological instability might suggest recurrent ischemia or hemorrhage and in either case, might require a repeat CT examination or reassessment of the timing of OAC initiation.

8.6.3. OAC initiation after hemorrhagic stroke

For patients with AF who have had an acute primary intracerebral hemorrhage, subdural, or subarachnoid hemorrhage, OAC is typically avoided in the acute-subacute period, and decisions regarding OAC initiation should be made in consultation with a stroke/neurology specialist on the basis of careful risk stratification guided by brain MRI appearance, the presumed etiology for the hemorrhage, and the estimated risk of recurrence. Patients with a lobar (superficial) ICH attributed to cerebral amyloid angiopathy have a higher risk of recurrent ICH than patients with a deep hypertensive ICH. If OAC is initiated post-ICH, a DOAC is preferred over VKAs because of the lower rate of incidence of ICH with these agents.⁵² Observational studies support the cautious use of OAC for selected patients post ICH, usually initiated weeks to months post ICH, but clinical equipoise exists.^{451,452} RCTs are currently under way to evaluate the safety and efficacy of DOAC vs aspirin in AF patients who have had an ICH.

8.7. Left atrial appendage occlusion

Imaging and postmortem studies suggest that most AFassociated ischemic strokes are cardioembolic, and that the majority arise from the LAA.^{453,454} In sinus rhythm the LAA has pulsatile flow, however in AF appendage emptying is reduced leading to stasis and clot formation. Physical elimination of the LAA (removal or occlusion) from the circulation is postulated to prevent thrombus formation and subsequent embolization, without suffering from the limitations of pharmacological therapy. Specifically, although OAC is effective at preventing AF-associated stroke/systemic embolism, its use is limited by short- and long-term nonadherence, nonpersistence, and side effects such as the risk of major bleeding. Conversely, when performed, the effects of the physical elimination of the LAA are considered permanent and are not reliant on patient compliance.

8.7.1. Percutaneous LAAO

Two RCTs and a number of single and multicentre registries that have evaluated percutaneous device closure have been performed. The WATCHMAN LAA Closure Technology for Embolic **Protect**ion in Patients With **A**trial **F**ibrillation (PROTECT AF) study enrolled 707 patients with AF and a CHADS₂ score of \geq 1, and the **P**rospective **R**andomized **Eva**luation of the WATCHMAN LAA Closure Device in Patients With Atrial Fibrillation vs Long-Term Warfarin

Therapy (PREVAIL) study enrolled 407 patients with a CHADS_2 score of $\geq 2.^{455,456}$ Both studies randomized patients to LAAO or VKA. The 5-year outcome data from these 2 studies were combined in a meta-analysis.457 Although LAAO was deemed noninferior to VKAs for the combined end points, much of the benefit of LAAO is due to a reduction in intracranial bleeding (0.2% per year with LAAO vs 0.9% per year with VKA; HR, 0.20; 95% CI, 0.07-0.56; P = 0.002).⁴⁵⁷ However, the effect of LAAO on ischemic stroke remains to be determined, and is suggested to be inferior (1.6% per year with LAAO vs 0.95% per year with VKA; HR, 1.71; 95% CI, 0.94-3.11; P = 0.08).⁴⁵⁷ Higher risk of ischemic stroke might relate to device-related thrombosis, which occurs in 2%-4% of cases and is a known independent predictor of stroke (HR, 4.4; 95% CI, 1.05-18.43).⁴⁵⁸ In the recently published randomized multicentre Left Atrial Appendage Closure Versus Novel Anticoagulation Agents in Atrial Fibrillation (PRAGUE-17) study the safety and efficacy of LAAO was compared with DOAC in high-risk AF patients with a history of significant bleeding or thromboembolic event while receiving OAC (CHA₂DS₂-VASc score 4.7 \pm 1.5; HAS-BLED score 3.1 \pm 0.9). This study showed no significant difference between LAAO and DOAC, with similar rates of all-cause stroke/TIA (2.20% with LAAO vs 2.68% with DOACs), ischemic stroke/ TIA (2.20% with LAAO vs 2.38% with DOACs), and major and CRNM bleeding (5.5% per year with LAAO vs 7.42% per year with DOACs). There was no difference in the end points of cardiovascular death, noncardiovascular death, or all-cause mortality. Major LAAO implant-related complications occurred in 4.5%. LAAO implantation was unsuccessful in 10.0% of the patients, however, the results of the study did not differ when analyzed according to treatment received (P =0.31) or per protocol (P = 0.40).

RECOMMENDATION

67. We suggest that percutaneous LAAO be considered for stroke prevention in patients with NVAF who are at moderate to high risk of stroke and have absolute contraindications to OAC (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on the evidence from multiple RCTs which showed significant benefit of VKAs and DOACs in the prevention of stroke/systemic embolism, relatively low risk of bleeding, and significant survival advantage, as well as the relative lack of data supporting LAAO for OACeligible patients.

Practical tip. Patients should receive individualized counselling regarding the risks and benefits of stroke prevention therapy, with a focus on the evidence supporting long-term OAC as the preferred strategy.

Practical tip. Contraindications to long-term OAC include but are not limited to: recurrent nontraumatic intracranial bleeding with high risk of recurrence, recurrent irreversible pulmonary bleed, recurrent irreversible urogenital bleed, recurrent irreversible GI bleed, recurrent irreversible retroperitoneal bleed, esophageal varices, intraocular bleeds, hereditary hemorrhagic telangiectasia.

8.7.2. Surgical LAAO

The device-based trials provide proof of concept that LAAO might provide benefit, however they do not address the surgical AF population, do not inform on the combination of LAAO and OAC, and do not evaluate a surgical intervention that can be offered with little increase in risk.

Results of existing observational studies suggest that standalone surgical LAAO can be safely and effectively performed, however, there are limited high-quality data to suggest that a stand-alone surgical LAAO approach is reasonable in patients eligible for a percutaneous approach.⁴⁵⁹

In patients with AF who are already undergoing a thoracotomy, it is possible that concomitant surgical LAAO might offer significant benefit with minimal incremental risk. However, it is important to recognize that patients who undergo surgical LAAO often have higher CHA2DS2-VASc scores and more often have coexisting valvular heart disease (ie, are more likely to develop AF-associated thrombosis outside of the LAA). 454 To date the observational data of concomitant surgical LAAO are somewhat conflicting, with some observational series suggesting that surgical LAAO is associated with significantly greater freedom from stroke/ systemic embolism, and others suggesting that LAAO did not reduce the incidence of stroke/systemic embolism in patients with AF (P = 0.69), with the apparent stroke reduction being related to continued VKA use and not the LAAO procedure itself.^{460,461} However, these observational studies do not have sufficient power or freedom from bias to provide the level of evidence needed to clearly answer this important question. A large RCT of surgical exclusion of the LAA vs VKA therapy has completed recruitment of > 4800patients and is in the follow-up phase.⁴⁶² The primary end point is stroke/systemic embolism over 4 years. Total mortality and safety end points will also be compared. Until this trial is completed, the only basis for recommendations regarding surgical LAA removal is consensus. As such, we have made a weak recommendation on the basis of the existing low-quality evidence.

RECOMMENDATION

68. We suggest surgical LAAO be considered for stroke prevention in patients with NVAF who are at moderate to high risk of stroke and have contraindications to OAC, and who are not suitable for percutaneous LAAO (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation is qualified by the patient's stroke risk in relation to their perioperative risk.

Practical tip. Patients should receive individualized counselling regarding the risks and benefits of stroke prevention therapy, with a focus on the evidence supporting long-term OAC treatment as the preferred prophylactic strategy.

69. We suggest concomitant surgical LAAO be considered in patients with AF who are undergoing an open chest cardiac surgical procedure and who are ineligible for long-term OAC (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation is qualified by the patient's stroke risk in relation to their incremental surgical risk with concomitant LAAO, and that the contraindication for OAC is absolute.

Practical tip. In patients with AF who are eligible for long-term OAC the evidence to date is insufficient to support a recommendation for or against concomitant surgical LAAO. In these patients it is recommended to continue OAC after surgical LAAO.

8.8. Cognitive function/dementia

An association between AF and cognitive impairment has been frequently reported over the past 30 years. 463,464 Although the association has not been consistently observed, 465,466 recent meta-analyses have shown a significant correlation between AF and cognitive impairment, irrespective of a history of previous stroke.⁴⁶⁷⁻⁴⁶⁹ It has been estimated that the risk of dementia in individuals with AF and no overt cerebrovascular events is increased 30%-60% compared with patients without AF, even after adjustment for age, sex, multiple comorbidities, and cardiovascular medications.⁴⁷⁰⁻⁴⁷² An American study of community-dwelling individuals showed that AF was associated with a 50% increase in risk of developing Alzheimer disease compared with individuals without AF.473 The findings are not just unique to North America and Europe, with a Taiwanese study involving 332,665 patients,⁴⁷⁰ and a South Korean study of 262,611 patients 472 demonstrating that patients with AF had a 42% and a 63% higher risk of dementia, respectively, even after adjustment for age, sex, baseline differences, and medication. More recent data suggest that AF increases the risk of vascular and degenerative (eg, Alzheimer disease) dementia, 474,475 and also that the relationship between AF and cognitive decline appears to be strongest when AF develops in middle age, 476,477 when the AF is persistent and of long duration, 477,478 and when the ventricular rate in AF is fast. 479,480

Although AF is related to cognitive impairment and dementia, the etiological mechanisms remain a matter for conjecture. Shared risk factors have been proposed as part of the explanation. The CHADS₂ and CHA₂DS₂-VASc scores have been shown to predict cognitive impairment as well as stroke in patients with AF. Nevertheless, the association between AF and cognitive decline persists even after controlling for age and these comorbidities.^{467,470,481} Proposed mechanisms include silent cerebral infarcts due to microemboli, cerebral microbleeds, disruptions of the blood-brain barrier, cerebral vascular disease, cerebral hypoperfusion due to variability in the cardiac cycle and reduced cardiac output, oxidative stress, inflammation, and endothelial dysfunction.^{471,482-485} Of these, subclinical embolic cerebrovascular ischemia is thought to be the most likely mechanism.⁴⁸⁵ Although white matter disease had been raised as another potential driver, recent work appears to refute this.^{482,486,487}

Treatments that have been proposed to reduce the risk of cognitive impairment include risk factor management,⁴⁸⁸ simaintenance with ablation,⁴⁸⁹ rhythm and nus anticoagulation. 490-492 A modest effect has been observed with specific risk factor treatments, such as statins or angiotensin conversion enzyme inhibitor and angiotensin receptor blocker therapy in some studies,^{481,488} but not in others.⁴⁹³ There also remains controversy concerning the ability of anticoagulants to decrease the incidence of cognitive impairment and dementia in patients with AF,⁴⁹⁴ with reviews of the available literature failing to find convincing evidence that they do so.^{495,496} Although positive data continue to accumulate, including from large population-based studies, 497,498 there have been no clinical trials to eliminate potential confounders. VKA use appears to be most effective when TTR is kept consistently very high (eg, > 75%).^{499,500} Data relating to the DOACs and dementia are sparse, with some studies showing relative benefit compared with VKAs, 491,501,502 whereas a nationwide Danish cohort study did not.503 Several clinical trials are being undertaken to allow greater clarity. In 2 of these studies, VKAs are being compared with dabigatran (Cognitive Impairment Related to Atrial Fibrillation Prevention Trial [GIRAF; NCT01994265] and Impact of Anticoagulation Therapy on the Cognitive Decline and Dementia in Patients With Non-Valvular Atrial Fibrillation [CAF; NCT03061006]) and in a third study, Blinded Randomized Trial of Anticoagulation to Prevent Ischemic Stroke and Neurocognitive Impairment in Atrial Fibrillation (BRAIN-AF; NCT02387229), rivaroxaban is being compared with standard of care.

There are currently only 2 observational studies on the association between catheter ablation and cognitive dysfunction. The largest was a retrospective case-control study from Intermountain Health in Utah, in which outcomes in 4212 patients who underwent AF ablation were compared with 16,848 age- and sex-matched patients with AF who did not undergo ablation. At 3 years, the incidence of dementia was low in both groups, but it was significantly less in the AF patients who had undergone ablation.⁴⁸⁹ A small prospective case-control study was recently published, which involved 308 AF patients treated with ablation and 50 medically managed control participants.⁵⁰⁴ Cognitive function was assessed at baseline, 3 months, and 1 year. Significant cognitive improvement was observed at 3 and 12 months in the ablation group compared with the control group, and no major adverse events were reported at either point in time. A larger prospective case-control study (Cognitive Impairment in Atrial Fibrillation [DIAL-F]; NCT01816308) is specifically addressing the effect of AF catheter ablation on dementia and is currently under way.

Finally, LAAO might be expected to have a potential role on decreasing the incidence of cognitive impairment on the basis of subclinical strokes. However, whether the available devices can prevent microemboli remains to be seen.

9. Arrhythmia Management

9.1. Acute management of AF

The acute management of AF is centred on the following domains:

- 1. Determination if AF is the primary concern ("primary AF") or secondary to another acute medical illness ("secondary AF"). AF in the setting of critical illness has been associated with an increased risk of death (see section 11.5).^{505,506} Unfortunately, there is a paucity of high-quality evidence on whether or how to treat AF patients in the setting of critical illness,^{507,508} and there is a wide variety of reported approaches to AF management in this setting.⁵⁰⁹ In the ED setting, results of a retrospective study suggested that rate and rhythm control efforts in patients with AF secondary to acute medical illness (predominantly sepsis and acute HF) might be associated with higher rates of adverse events.⁵¹⁰
- 2. Determination of hemodynamic stability, defined as AF causing hypotension, ACS, or pulmonary edema. Acute unstable AF should be treated with synchronized direct current cardioversion (DCCV), however, instability solely due to AF is rare, therefore, an underlying precipitant should also be aggressively sought and managed.⁵¹¹
- 3. Determination of an arrhythmia management strategy, defined as rate vs rhythm control. In patients with established AF multiple RCTs have shown no significant difference in cardiovascular outcomes between patients treated with a strategy with rate control vs rhythm control.⁵¹²⁻⁵¹⁴ In patients with newly diagnosed AF (ie, within a year) an initial strategy of rhythm control has been associated with reduced cardiovascular death and reduced rates of stroke.⁵¹⁵
- 4. Determination of the need for hospitalization. Most patients with AF can be safely discharged home after acute management. However, hospitalization might be required for highly symptomatic patients with AF in association with acute medical illness or complex medical conditions, in highly symptomatic patients in whom adequate rate control cannot be achieved, or in those who require monitoring or ancillary investigations not readily available in the outpatient setting.
- 5. Determination of the need for OAC. There is evidence that OAC prescription in the ED results in improved long-term use.^{364,516,517} As such, it is of paramount importance that OAC be initiated as soon as time allows in patients who undergo cardioversion for new-onset AF (see section 8.4.1), as well as in patients at risk of stroke (see the "CCS algorithm" in Fig. 8) whether or not attempts at rhythm control are made.
- 6. Early follow-up. Patients discharged from the ED with AF benefit from early follow-up. Ideally this should occur within a week of discharge, because early follow-up has been associated with lower rates of readmission and death.⁵¹⁸ In addition, cardiology assessment within 3 months of hospital discharge for new-onset AF has been associated with lower rates of death, stroke, and major bleeding.⁵¹⁹

A general overview of rate and rhythm management of AF is provided in Figure 16, and the approach to the management of AF in the acute care setting is provided in Figure 17.

RECOMMENDATION

70. We recommend that the management of patients who present with recent-onset AF due to a reversible or secondary cause should be directed at the primary illness (Strong Recommendation; Low-Quality Evidence).

Values and preferences. AF in the acute care environment can be secondary to a primary cardiac pathology or can occur secondary to a specific precipitating event, such as infection, surgery, or thyroid disease.

71. We recommend immediate electrical cardioversion for patients whose recent-onset AF is the direct cause of instability with hypotension, ACS, or pulmonary edema (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on immediately addressing instability by attempting cardioversion and a lower value on reducing the risk of cardioversion-associated stroke with a period of anticoagulation before cardioversion.

Practical tip. Therapeutic anticoagulation should be initiated as soon as possible, ideally prior to cardioversion if time allows.

72. We suggest that a rhythm control strategy be considered for most stable patients with recent-onset AF (Weak Recommendation; Moderate-Quality Evidence).

Values and preferences. In patients with established AF multiple RCTs have shown no significant difference in cardiovascular outcomes between patients treated with a strategy with rate control vs rhythm control, recognizing that most of these trials did not specifically address recent-onset AF. In patients with newly diagnosed AF (ie, within a year) an initial strategy of rhythm control has been associated with reduced cardiovascular death and reduced rates of stroke.

73. In patients with AF and manifest pre-excitation we recommend against acute rate control (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. See section 11.8 for the management of patients with AF and pre-excitation.

74. We recommend that patients who present with AF in the acute care setting have their need for long-term antithrombotic therapy be determined using the CCS Algorithm (CHADS-65) (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. See section 8.4.1 for the recommendations regarding OAC in the context of cardioversion.

9.1.1. Acute rate control

In the setting of recent onset AF, the rate control agent and the formulation chosen will be influenced by clinical circumstance (eg, the presence of HF or hypotension) and patient comorbidities (eg, known LV dysfunction, reactive airways disease, hypotension, history of MI, or angina; Supplemental Table S10). Options include oral or I.V. β -blockers, oral or I.V. nondihydropyridine calcium channel blockers (ND-CCBs), I.V. digoxin, and I.V. amiodarone (recognizing that the latter is also a rhythm control agent).

I.V. rate control agents might be initially considered, however if the patient is hemodynamically stable oral agents might be preferred. If an I.V. agent is used as the initial therapy, it is important to coadminister an oral rate control agent as soon as possible to maintain rate control/avoid rebound tachycardia as the I.V. formulation wears off.⁵²⁰

In patients without contraindications, β -blockers and calcium channel blockers (CCBs) are considered first-line agents for rate control. Only two small RCTs totalling 92 patients have compared I.V. β -blockers with CCBs in patients with recent-onset AF (one of which also included AFL), both showed that I.V. diltiazem was more effective at controlling the heart rate (< 100 beats per minute [bpm]) at 20-30 minutes compared with I.V. metoprolol (RR, 1.8; 95% CI, 1.2-2.6).⁵²¹⁻⁵²³ A retrospective study of 110 AF patients showed similar results at 60 minutes, although the success rate with diltiazem in that study (57%) was lower than those reported in the RCTs (90%-95%).^{521,522,524}

In patients with ACS who require acute rate control, β -blockers are the agent of choice.

Digoxin or amiodarone might be considered for acute rate control in the setting of decompensated HF, known significant LV systolic dysfunction (defined as LVEF \leq 40%), or mild hypotension. However, it is important to recall that I.V. formulations of amiodarone can lower BP.525 Moreover, in this population the selective use of I.V. CCBs (and β blockers) has been safely and successfully used in several randomized and nonrandomized studies, often with an improvement in BP when the recent-onset AF is ratecontrolled. Jandali performed a retrospective cohort study on the use of I.V. diltiazem in 162 patients with LV systolic dysfunction (LVEF \leq 50%; with 52 having an LVEF \leq 30%), and compared them with 473 patients with preserved LVEF (\geq 50%).⁵²⁶ There was no difference in the rates of hypotension, intensive care unit (ICU) transfer, or mortality between the patients with LVEF > 50% vs those with LVEF < 50%, or those with LVEF \geq 30% vs those with LVEF <30%. Hirschy et al. performed a retrospective cohort study on the use of I.V. metoprolol (14 patients) and I.V. diltiazem (34 patients) in patients with known LV systolic dysfunction (mean LVEF 23% [15-35] vs 25% [15-30], respectively). Successful rate control within 30 minutes occurred in 62% of the metoprolol group and 50% of the diltiazem group (P =0.49), with no difference in complications or worsening HF.⁵²⁷ Goldenberg et al. performed a small double-blind RCT of 37 AF patients with reduced ejection fraction (EF; LVEF 36% \pm 14%) and symptomatic HF (New York Heart Association [NYHA] class III [62%] and IV [38%]), in whom the cautious use of I.V. diltiazem resulted in therapeutic response (97%) with self-limited hypotension in 11%, and no patients experiencing worsening of HF.528 Although these studies have shown that the use of I.V. CCBs for rate control in highly selected patients can be safe, caution must be used because of the risk of precipitating cardiac decompensation.

RECOMMENDATION

75. We recommend that either β -blockers or ND-CCBs (diltiazem or verapamil) be first-line agents for AF rate control in patients without significant LV dysfunction (eg, patients with an LVEF > 40%) (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. Use caution when administering I.V. formulations of β -blockers or ND-CCBs (verapamil and diltiazem) because of the risk of precipitating hypotension.

Practical tip. The selection of a β -blocker or ND-CCB for rate control of AF should be on the basis of patient comorbidities, contraindications, and side effect profile.

Practical tip. Oral formulations should be introduced as soon as possible because of the need for ongoing control of ventricular rate.

- 76. We recommend evidence-based β -blockers (bisoprolol, carvedilol, metoprolol) be first-line agents for rate control of hemodynamically stable AF in the acute care setting in patients with significant LV dysfunction (LVEF \leq 40%) (Strong Recommendation; Moderate-Quality Evidence).
- 77. We suggest I.V. amiodarone or I.V. digoxin be considered for acute rate control in patients with significant LV dysfunction (LVEF \leq 40%), decompensated HF, or hypotension, when immediate electrical cardioversion is not indicated (Weak Recommendation; Moderate-Quality Evidence).

Practical tip. Use caution when administering I.V. amiodarone for rate control because of the possibility of hypotension and/or conversion to sinus rhythm, and the subsequent risk of stroke in patients who are not adequately anticoagulated.

9.1.1.1. Acute rate control targets

There have been no RCTs that specifically examined rate control targets in the acute care setting, 529 nor in patients with paroxysmal AF. 530

RECOMMENDATION

78. We recommend titrating rate-controlling agents to achieve a heart rate target of ≤ 100 bpm at rest for patients who present with a primary diagnosis of AF in the acute care setting (Strong Recommendation; Low-Quality Evidence).

Practical tip. There is no evidence to support a specific heart rate target in acutely ill patients with AF secondary to a reversible or secondary cause. Treatment targets should be individualized in this patient population after consideration of the risk/benefits of pharmacological rate control.



²See Figure 19 for long-term rhythm control

Figure 16. Approach to rate and rhythm management of atrial fibrillation (AF). AAD, antiarrhythmic drug; QOL, quality of life.

9.1.2. Acute rhythm control

For stable patients with recent-onset AF who are eligible for cardioversion, the choice to pursue sinus rhythm restoration should be made on the basis of patient symptoms and goals of care, recognizing that early rhythm control has been associated with a lower risk of stroke and cardiovascular death.⁵¹⁵ Because cardioversion increases the risk of systemic embolism, it is important to start appropriate anticoagulation as soon as time allows for all patients (see section 8.4.1).³³⁹ For patients with recent-onset AF who are eligible for cardioversion, rhythm control is preferred and can be established via either pharmacological or electrical cardioversion. In general, electrical cardioversion is more effective than pharmacological cardioversion, especially for more prolonged AF episode durations.^{363,531-534} Pharmacological cardioversion has the advantage of being immediately feasible in a nonfasting patient, as well as avoiding the delays and risks associated with procedural sedation. However, most pharmacological agents have cautions or contraindications that limiting their use in patients with significant cardiac comorbidities, and their use requires a monitored bed, access to a crash cart, and a dedicated nurse to monitor for potential complications.

RECOMMENDATION

79. We recommend that synchronized direct current or pharmacologic cardioversion may be used for sinus rhythm restoration in hemodynamically stable patients with recent-onset AF (Strong Recommendation; Moderate-Quality Evidence). **Practical tip.** In treatment environments in which procedural sedation is readily available, DCCV might be the preferred initial means to restore sinus rhythm, because it is more than 90% effective in the acute care environment and reduces ED length of stay.

Practical tip. A strategy of pharmacological conversion followed by DCCV (if necessary) might be preferred in environments in which procedural sedation is not readily available, because pharmacological conversion might avert the need for DCCV in approximately half of the treated patients.

9.1.2.1. Pharmacologic cardioversion

Antiarrhythmic medication selection is typically dictated by the patient's comorbidities as well as physician preference. Characteristics, indications, contraindications, and monitoring details of antiarrhythmic medications used for acute

RECOMMENDATION

80. We recommend that the choice of antiarrhythmic drug used for acute pharmacological cardioversion be defined according to patient characteristics (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. The choice of medication in patients without any contraindications will depend on physician experience, duration of AF, and considerations unique to the practice setting (See Supplemental Table S11).



¹Hemodynamically unstable acute AF is defined as AF causing hypotension, acute coronary syndrome, or pulmonary edema. ²Initiate OAC as outlined in section 8.4.1/Figure 11.

³Initiate OAC as outlined in the CCS Algorithm.

⁴Rhythm-control is preferred in patients with newly diagnosed AF (i.e. within a year)

⁵Second line therapy – use if suboptimal control or contraindications.

⁶May be cautiously utilised in the absence of decompensated heart failure or hypotension.

⁷Use caution when administering IV amiodarone given the possibility of hypotension and/or conversion to sinus rhythm, with risk of stroke in underanticoagulated patients.

⁸TEE-guided cardioversion may be considered an alternate to 3 weeks of pre-CV OAC as outlined in section 8.4.1.3.

⁹See Supplementary Table 11 for indications and contraindications.

Figure 17. Approach to the management of atrial fibrillation (AF) in the acute care setting. CCS, Canadian Cardiovascular Society; CHADS₂, **C**ongestive Heart Failure, **H**ypertension, **A**ge, **D**iabetes, **S**troke/Transient Ischemic Attack; CV, cardioversion; DCCV, direct current electrical cardioversion; HR, heart rate; ND-CCB, nondihydropyridine calcium channel blocker; HF, heart failure; I.V., intravenous; LVEF, left ventricular ejection fraction; NVAF, nonvalvular atrial fibrillation; OAC, oral anticoagulation; TEE, transesophageal echocardiography; TIA, transient ischemic attack.

pharmacological cardioversion are presented in Supplemental Table S11.

Practical tip. All patients require monitoring after cardioversion, the length of which is dependent on the method of conversion. A general guideline is to observe patients for a duration of time that is equal to half of the medication's therapeutic half-life.

Practical tip. If the patient's history is unknown, electrical cardioversion should be used in preference to pharmacological cardioversion.

9.1.2.1.1. Procainamide

Procainamide, a class Ia agent, is the most common I.V. medication used for cardioversion of recent-onset AF in the Canadian ED setting.⁵³⁵ Although procainamide can be administered as a 1-g bolus over 30 minutes followed by an infusion of 2 mg/min or a dose of 15-18 mg/kg administered over 60 minutes, most clinical evidence (including safety outcomes) was derived on the basis of a single infusion of 1 g over 60 minutes.^{531,536-538}

Procainamide is more effective for the conversion of recent-onset AF (50%-60% conversion) than for AFL (30% conversion). $^{531,536-538}$ The most common side effect is hypotension (approximately 5%), although premature ventricular contractions, runs of ventricular tachycardia and QRS widening might occur. 531,536,537 This medication, like all drugs with class I (Na⁺ channel-blocking) action, should be avoided in patients with Brugada syndrome. 539

9.1.2.1.2. Ibutilide

Ibutilide is an I.V. class III agent that has been shown to effectively terminate AFL (50%-75%) and AF (30%-50%), with cardioversion typically occurring within 30-60 minutes.⁵⁴⁰⁻⁵⁴⁶ However, widespread clinical uptake has been limited by a significant risk of torsades de pointes (TdP) and ventricular tachycardia (most frequently nonsustained), each of which occurs in approximately 2%-3% of patients.⁵⁴⁰⁻⁵⁴⁸ Consequently, ibutilide should not be used in patients with a prolonged QTc (> 440 ms), a history of HF (typically defined as clinically symptomatic or NYHA classification > II), or reduced EF, signs of an ACS, and/or low serum potassium or magnesium levels.⁵⁴⁰⁻⁵⁴⁷ Periprocedural

I.V. magnesium (typically given pre- and post treatment) appears to improve ibutilide cardioversion rates, with higher doses (eg, ≥ 4 g total) more effective than lower doses (1-3 g total).^{549,550} Patients must be observed with continuous ECG monitoring for a minimum of 4 hours after ibutilide administration.

9.1.2.1.3. Vernakalant

Vernakalant is an atrial-selective antiarrhythmic approved for conversion of AF. In patients treated within 48 hours of AF onset, randomized trials report a conversion rate at 90 minutes ranging from 52% to 69%, which is not significantly better than other active agents (combined comparator of ibutilide and amiodarone).^{544,545,551-553} However, the median time to cardioversion of 10-12 minutes is shorter than the next fastest pharmacological agent (ibutilide, median time to conversion 26 minutes). The major side effects are hypotension and bradycardia after cardioversion.^{551,554,555} Transient but fairly common side effects include dysgeusia, paresthesia, and nausea.^{544,551,554,555} Vernakalant is not effective for the conversion of AFL; and should be avoided in patients with hypotension, severe HF (NYHA classification III/IV), bradycardia, recent ACS, or severe aortic stenosis.^{552,555,556}

9.1.2.1.4. Amiodarone

With the exception of patients with structural heart disease, amiodarone is not recommended for acute rhythm control because of a delay in conversion (approximately 8 hours).^{525,532,557} The most common adverse drug reactions with I.V. administration are phlebitis, hypotension, and bradycardia.^{525,532} Although there is potential for prolongation of the QT interval, the incidence of TdP is rare.^{532,557}

9.1.2.1.5. Flecainide and propafenone

I.V. flecainide and propafenone are superior to placebo but are not currently available in Canada for acute care cardioversion.⁵ The oral formulations, however, have similar, if slightly delayed, efficacy as their I.V. counterparts. 558,559 Three hours after administration of a single dose of oral flecainide, between 57% and 68% of patients will convert.⁵³² Success rates with oral propafenone are similar.^{532,559} Although the time to cardioversion (approximately 2-6 hours) is longer than with I.V. formulations, the major clinical benefit is that patients are able to treat their AF episodes at home ("pill-in-the-pocket"), which reduces the need to visit the ED for recurrences. A key caveat to this approach is that the first treatment attempt must be administered in a monitored environment, to verify efficacy and exclude treatment-related adverse reactions. $^{557,560-563}$ A β -blocker or ND-CCB should be given ≥ 30 minutes before administration of a class Ic antiarrhythmic to prevent the risk of 1:1 AV conduction during AFL. One study suggests that rare adverse events can occur even after successful use in a monitored environment⁵⁶³; therefore, clear instructions must be given to these patients about when to seek emergency care (Supplemental Table S12). It is important to note that flecainide and propafenone should not be used in patients with structural heart disease, including a history of ischemic heart disease.

9.1.2.2. Electrical cardioversion

In patients with hemodynamically stable recent-onset AF in whom sinus rhythm restoration is desired, there is a wealth

of observational data supporting the safety and efficacy (approximately 90%) of direct-current cardioversion (DCCV).^{363,531,533,535,564-567} Since the late 1990s patient sedation and cardioversion have typically been performed by emergency physicians in the Canadian ED setting.^{531,535,565} Adverse events attributable to DCCV such as bradyarrhythmia, acute HF, and skin burns are rare.^{533,566,567} The time to patient discharge using DCCV is shorter than when using pharmacological cardioversion.^{534,568} Importantly, AF patients who receive DCCV in the ED rate their care as more effective compared with those who receive only rate control, however, their QOL scores at 30 days were not different than those treated with only rate control.⁵⁶⁹

Biphasic shocks are preferred over monophasic because less energy is required.⁵⁷⁰ Pad placement (anterolateral vs anteroposterior) does not seem to influence cardioversion efficacy.^{538,571} In obese patients, using paddles and applying force might improve success rates with DCCV over adhesive pads.^{572,573}

Pretreatment with antiarrhythmic drugs (eg, ibutilide and amiodarone) has been shown to improve the effectiveness of DCCV.^{363,574}

RECOMMENDATION

81. We recommend at least a 150-J biphasic waveform as the initial energy setting for DCCV (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on the avoidance of repeated shocks after ineffective attempts at low-energy cardioversion.

Practical tip. Electrical cardioversion should ideally be performed with one trained operator managing the sedation and airway and a second trained operator managing the synchronized DCCV. Atropine and pacing capability must be immediately available in case of prolonged sinus pause after cardioversion.

- 82. We suggest antiarrhythmic drug therapy be considered to enhance the efficacy of electrical cardioversion and the maintenance of sinus rhythm, particularly in patients with persistent and long-standing persistent AF (Weak Recommendation; Low-Quality Evidence).
- 83. We suggest that the use of antiarrhythmic drug therapy after sinus rhythm restoration be on the basis of the estimated probability of AF recurrence (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on minimizing the risk of infrequent but serious side effects associated with long-term antiarrhythmic drugs. A high value is also placed on the appropriate use of specialty care to make patient-specific decisions to minimize these risks.

9.2. Long-term rate control

9.2.1. Agents

Pharmacotherapy for long-term AF rate control revolves around agents with negative dromotropic properties such as β blockers and ND-CCBs (verapamil and diltiazem). The choice



Figure 18. Approach to long-term rate control. For patients with difficult to control symptoms and heart rate despite combination therapy, consideration should be made to refer to electrophysiology for evaluation (and/or pacemaker implantation and atrioventricular junction [AVJ] ablation). AF, atrial fibrillation; bpm, beats per minute; LVEF, left ventricular ejection fraction; ND-CCB, nondihydropyridine calcium channel blocker.

of a specific rate-controlling regimen should be on the basis of patient's characteristics and the drug's efficacy/side effect profile (Supplemental Table S10; Fig. 18).⁵⁷⁵

In patients without significant LV dysfunction (LVEF > 40%), β -blockers and ND-CCBs are first-line options. There are no randomized long-term data to support choosing a β blocker over an ND-CCB. Several retrospective studies of AF patients have shown conflicting results when rates of hospital admission after using β -blockers vs CCBs were compared: one showed no difference whereas another showed that use of CCBs was associated with a higher rate of hospitalization compared with use of $\beta\text{-blockers.}^{520,576}$ In the longer-term, $\beta\text{-}$ blockers might be more effective at slowing ventricular rates at rest and during exercise, however, their use is associated with a higher risk of adverse effects, notably fatigue and exercise intolerance.⁵⁷⁷⁻⁵⁸⁰ Moreover, there is emerging evidence suggesting that CCBs might have favourable dose-response characteristics for AF rate control vs β -blockers, such that they might be preferred in patients with a preserved LVEF and without another indication for a β -blocker.⁵ Specific patient characteristics might favour the use of one pharmacological class (eg, ND-CCBs with hypertension or reactive airway disease, vs β-blockers with CAD). Caution should be used when β -blockers are used with ND-CCBs.

In patients with significant LV systolic dysfunction (LVEF \leq 40%), maximally tolerated doses of evidence-based β -blockers (extended-release metoprolol succinate, bisoprolol, carvedilol) remain first-line therapy for rate control, although the benefits of

adrenergic blockade, in addition to that provided by the control of the ventricular response rate, are uncertain.⁵⁸³⁻⁵⁸⁸

RECOMMENDATION

84. We recommend β -blockers or ND-CCBs (diltiazem or verapamil) be first-line agents for rate control of AF in patients without significant LV dysfunction (LVEF > 40%) (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a high value on the extensive clinical experience and record of safety and efficacy of β -blockers and ND-CCBs (verapamil and diltiazem) for AF rate control.

Practical tip. The choice of specific rate-controlling agents should be guided by the patient's characteristics and the drug efficacy/side effect profile.

Practical tip. ND-CCBs (verapamil and diltiazem) have favourable pharmacological properties for rate control and might be the preferred choice in patients without a compelling indication for β -blocker usage.

85. We recommend evidence-based β -blockers (bisoprolol, carvedilol, metoprolol) be first-line agents for rate control of AF in patients with significant LV dysfunction (LVEF $\leq 40\%$) (Strong Recommendation; Moderate-Quality Evidence). 86. We recommend against rate control as a treatment strategy in patients with AF and who manifest preexcitation (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. See section 11.8 for the management of patients with AF and pre-excitation.

87. We suggest combination therapy (eg, a β-blocker with a ND-CCB) in patients who do not achieve satisfactory symptom or heart rate control with monotherapy (Weak Recommendation; Low-Quality Evidence).

Practical tip. Combination therapy should be used with caution in patients at risk of significant bradycardia/ AV block (eg, patients with resting sinus bradycardia or with significant conduction disease). These patients might require pacemaker implantation to facilitate pharmacological rate control.

Monotherapy with digoxin is generally ineffective in younger patients because of its inability to control ventricular rate during exertion or stress. Moreover, digoxin has a narrow therapeutic window, with observational evidence suggesting potential harmful effects when used for ventricular rate control.⁵⁸⁹⁻⁵⁹² However, a recent meta-analysis of 28 trials of digoxin for AF rate control showed no increase in all-cause mortality vs control intervention (RR, 0.82; 95% CI, 0.24-11.5).⁵⁹³ As such, digoxin remains a reasonable choice for selected older or sedentary patients with HF and for those with inadequate rate control while receiving maximally tolerated doses of a β -blocker/ND-CCB. Although there is no direct evidence to support digoxin concentration monitoring for AF rate control, it might be reasonable to monitor patients at risk of digoxin-related adverse events (eg, female sex with low body weight and impaired renal function), at the clinician's discretion, aiming for trough levels between 0.5 and 0.9 ng/mL.⁵⁹⁴ Furthermore, it is important to note that coadministration of ND-CCBs and amiodarone will decrease digoxin clearance, resulting in a propensity toward toxicity.

RECOMMENDATION

88. We suggest that digoxin be considered as a monotherapy in older or sedentary individuals with permanent AF; or those with side effects or contraindications to first-line agents; or in addition to first-line agents in those who fail to achieve satisfactory symptom or heart rate control (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a lesser value on observational cohort studies in which adverse outcomes from digoxin have been reported.

Practical tip. Digoxin is relatively ineffective for heart rate control in younger patients during activity but might be useful in older or sedentary individuals with HF, particularly in combination therapy.

Practical tip. Therapeutic drug monitoring might be useful in adjusting digoxin dose, particularly in patients at risk of digoxin-related adverse events (eg, female sex with low

body weight and impaired renal function). In patients with HF with reduced EF (HFrEF) trough levels between 0.5 and 0.9 ng/mL were associated with a significant decrease in all-cause mortality and hospitalizations compared with levels \geq 1.0 ng/mL, however, the optimal trough level for AF patients is unknown.

Amiodarone is a class III antiarrhythmic with complex pharmacological properties and potential serious adverse effects. However, selected patients such as the critically ill or those with side effects from, or contraindication to, first-line agents might benefit from amiodarone for rate control after careful consideration of alternative agents, alternative approaches (eg, transition to rhythm control), and risk/benefits of continued amiodarone therapy.^{595,596}

RECOMMENDATION

89. We recommend that amiodarone be used for AF rate control only in highly-selected patients such as the critically ill or those with significant side effects from or contraindication to first-line agents after careful consideration of alternative agents and risk/benefits of amiodarone therapy (Strong Recommendation; Low-Quality Evidence).

Dronedarone should not be used for AF rate control because it was associated with excess HF, stroke, and cardiovascular death in the **P**ermanent **A**trial Fibri**lla**tion Outcome **S**tudy Using Dronedarone on Top of Standard Therapy (PALLAS) trial.⁵⁹⁷

RECOMMENDATION

90. We recommend dronedarone not be used for AF rate control or in patients with HF (Strong Recommendation; High-Quality Evidence).

Values and preferences. This recommendation places a high value on randomized controlled trial data that have shown that the use of dronedarone resulted in excess of cardiovascular death, and unplanned cardiovascular hospitalization.

9.2.2. Targets

The goals of ventricular rate control are the reduction of AFrelated symptoms and the prevention of adverse cardiovascular events, rather than the achievement of a specific heart rate target. It is known that overly aggressive rate control is associated with adverse outcomes (eg, risk of symptomatic bradycardia with subsequent pacemaker implantation) and increased frequency of medical encounters. Conversely, overly lenient rate control might lead to HF (eg, tachycardia-mediated cardiomyopathy). In addition, specific populations might need stricter HR targets (eg, patients with cardiac resynchronization therapy [CRT], HF, tachycardia-mediated cardiomyopathy, mitral stenosis, stable angina), whereas others might do well with a more lenient target. As such, the intensity of AF rate control beyond the target HR of \leq 100 bpm should be individualized on the basis of clinical characteristics and coexisting cardiovascular diagnoses.

Most of the evidence to guide clinical decision-making for ventricular rate control targets has been acquired in patients with preserved LVEF.^{512-514,598} Previous guidelines have recommended heart rate targets of < 80 bpm at rest and < 110 bpm with exercise, because these targets were used in the Atrial Fibrillation Follow-up Investigation of Rhythm Management (AFFIRM) study.⁵¹² However, retrospective analyses of the Rate Control versus Electrical Cardioversion for Persistent Atrial Fibrillation (RACE) and AFFIRM studies suggested that cardiovascular morbidity, mortality, and QOL did not differ between those achieving or not achieving the prespecified heart rate target (but still maintaining a resting heart rate < 100 bpm).⁵⁹⁹ The prospective randomized Ratecontrol Efficacy in Permanent Atrial Fibrillation: a Comparison between Lenient Versus Strict Rate-control-II (RACE-II) trial⁵⁹⁸ showed that lenient rate control (resting heart rate target < 110 bpm) was noninferior to strict rate control (resting heart rate target < 80 bpm, < 110 bpm with exercise), with fewer medications, lower medication doses, fewer adverse events, and reduced health care utilization. However, it is important to recognize that the mean heart rates achieved were 76 \pm 14 bpm in the strict group and 85 \pm 14 bpm in the lenient group, with very few of those patients randomized to lenient rate control having resting heart rate of > 100bpm.⁶⁰⁰ As such, there remain some questions as to the ideal heart rate target. A recent analysis of the Outcomes Registry for Better Informed Treatment of Atrial Fibrillation (ORBIT-AF) registry and a retrospective analysis of the AFFIRM and Atrial Fibrillation and Congestive Heart Failure (AF-CHF) studies provides further insight. In both studies a U-shaped relationship between the resting heart rate and adverse outcomes was observed, with increased adverse event rates at the 2 extremes of resting heart rates. Although the combined AFFIRM/AF-CHF analysis suggested lower rates of mortality for those with resting heart rates in AF between 76 and 89 bpm, and lower rates of hospitalization for those with resting heart rates in AF between 76 and 114 bpm, the optimal heart rate in the ORBIT-AF registry was noted to be approximately 65 bpm.^{601,602} On the basis of the best available evidence, a heart rate-target of < 100 bpm at rest appears to be associated with an acceptable risk/benefit profile in patients without significant LV systolic dysfunction (LVEF > 40%). Because of the paucity of data to guide heart rate targets in patients with significant LV systolic dysfunction it is reasonable to target a heart rate of < 100 bpm, although clinically driven targets should be individualized to symptoms and hemodynamics as per in other populations.

RECOMMENDATION

91. We recommend titrating rate-controlling agents to achieve a resting heart rate of < 100 bpm during AF (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a high value on the small number of RCTs that have shown outcomes comparable between strict vs lenient strategies for AF rate control.

Practical tip. The goal of AF rate control should be the control of AF-related symptoms and cardiovascular complications. Specific populations (eg, those with HF, tachycardia-mediated cardiomyopathy, mitral stenosis, stable angina, patients with CRT) might need stricter HR targets whereas others might do well with a more lenient target.

Practical tip. Paroxysmal AF and AFL can be more challenging to rate-control than persistent/permanent AF. Rhythm control should be considered for these patients.

92. We recommend maximizing evidence-based β blocker dose in patients with reduced LVEF (LVEF \leq 40%), in addition to achieving a resting heart rate of \leq 100 bpm (Strong Recommendation; Moderate-Quality Evidence).

In patients with AF and a CRT device, the goals of care should be to maximize biventricular pacing (ie, as close to 100% as possible) rather than target a specific heart rate. If use of combination pharmacotherapy does not achieve a high percentage of biventricular pacing, then AV junction (AVJ) ablation should be considered.

RECOMMENDATION

93. We recommend that pharmacological atrioventricular blockade in patients with AF and CRT should target maximal biventricular pacing (as close to 100% as possible) and not a specific heart rate target (Strong Recommendation; High-Quality Evidence).

In patients with paroxysmal AF a rhythm control approach should be preferentially considered because ventricular rate control can be challenging.

Finally, there is insufficient evidence to support routine monitoring of exercise heart rates. In the subset of very active individuals or patients with exercise-related symptoms it might be reasonable to monitor heart rate during exercise, and additionally target a heart rate of < 110 bpm on moderate exertion (6-minute walk test) and a maximal heart rate of < 110% of age-predicted maximum heart rate at peak exertion.

RECOMMENDATION

94. We suggest monitoring of the heart rate during exercise only in patients with exercise-related symptoms or in highly active individuals (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on the lack of evidence to support heart rate monitoring during exercise.

9.2.3. Atrioventicular junction ablation and pacing

Pharmacological agents achieve satisfactory heart rate control in most patients; however, a subset of patients fail to achieve adequate control of ventricular rate despite maximally tolerated/combination doses of rate-controlling agents. Implantation of a permanent pacemaker followed by AVJ ablation can be a useful treatment strategy to achieve definitive rate control in this population with permanent AF. An AVJ ablation strategy ensures reliable control of the ventricular rate, regularization of the RR intervals, and discontinuation of rate/rhythm control drugs in most patients.⁶⁰³ In patients with HF and CRT devices, AVJ ablation optimizes the delivery of biventricular pacing (see section 9.2.3.1).

Compared with medical therapy, AVJ ablation and pacemaker implantation results in significant improvements in symptoms and QOL despite no significant changes in exercise capacity or functional status (eg, treadmill exercise or VO₂ max).^{604,605} In general the greatest improvement in QOL is observed in patients who are able to discontinue rate-limiting pharmacotherapies.⁶⁰⁶ It is important to note that these studies were performed in patients with permanent AF refractory to pharmacological rate control. As such, there are no data to support AVJ ablation and pacemaker implantation as a first-line treatment (ie, prior to attempts at pharmacological rate control).

Patients with paroxysmal/persistent AF should be considered for a rhythm control strategy including catheter ablation (section 9.4) prior to pursuing AVJ ablation given the fact that AVJ ablation is irreversible, rendering patients pacemakerdependent, further complicating the management of device infection and lead failure, and increasing the risk of pacinginduced cardiomyopathy.⁶⁰³

In patients with permanent AF who undergo AVJ ablation, a single-chamber right ventricular pacemaker is preferred for those with normal ventricular function. In those with normal ventricular function CRT implantation before AVJ ablation has not been shown to result in substantial benefit over the single-chamber right ventricular pacemaker. In a meta-analysis of 4 trials comparing de novo CRT vs right ventricular pacemaker in patients with AF treated with AVJ ablation, CRT therapy resulted in only a small improvement in QOL (Minnesota Living with Heart Failure Questionnaire, 2.72 fewer points; 95% CI, 1.45-3.99) and LVEF (+2.6%; 95% CI 1.69%-3.44%), at the expense of a trend toward increased procedure-related complications (RR, 1.96; 95% CI, 0.71-5.45; P = 0.2).⁶⁰⁷ Although there are emerging data on leadless pacemaker implantation and conduction system pacing (His bundle/left bundle branch area pacing) in the context of AVJ ablation, further studies are required before these strategies can be routinely recommended in preference to conventional singlechamber right ventricular pacemakers.⁶⁰⁸⁻⁶¹⁰

Canadian Journal of Cardiology Volume 36 2020

For patients with LV systolic dysfunction being considered for AVJ ablation for permanent AF, a CRT device is preferable to a single-chamber right ventricular pacemaker. In susceptible patients chronic right ventricular pacing might lead to progressive LV dysfunction and HF. Biventricular pacing has been shown to significantly lower incidence of death or HF hospitalization (HR, 0.74; 95% CI, 0.60-0.90) compared with right ventricular pacing in patients with LVEF $\leq 50\%$.⁶¹¹

RECOMMENDATION

95. We recommend permanent pacemaker implantation with AVJ ablation in patients ineligible for rhythm control who have an uncontrolled heart rate during AF despite maximally tolerated combination pharmacological rate control (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a high value on the efficacy and safety of permanent pacemaker implantation with AVJ ablation for patients with refractory permanent AF.

Practical tip. Patients with paroxysmal or persistent AF should be considered for a rhythm control strategy (eg, catheter ablation of AF) before proceeding with permanent pacemaker implantation with AVJ ablation.

96. We recommend against permanent pacemaker implantation with AVJ ablation in patients with an uncontrolled heart rate during AF without previous attempts at pharmacological rate control (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a high value on the lack of evidence to support a strategy of early pacemaker implantation and AVJ ablation (eg, before an adequate trial of pharmacological rate control), because of the longer-term consequences of iatrogenic pacemaker dependence.

9.2.3.1. HF, AF, and biventricular devices

HF and AF often coexist, with up to 25% of patients with a CRT device having coexisting AF.⁶¹² In this population, the outcomes of CRT are suboptimal with a larger proportion of "non-responders" and a higher mortality rate vs patients in sinus rhythm.⁶⁰³ Therapeutic failure is largely due to the irregularity of the AF interfering with the ability of the device to deliver optimal CRT.^{613,614} Management options to optimize CRT include increasing the lower rate limit, uptitrating atrioventricular nodal-blocking drugs, and/or AVJ ablation. A meta-analysis of 6 studies that included 768 patients with AF, symptomatic HF, LVEF \leq 35%, and a CRT device reported that AVJ ablation was associated with lower all-cause mortality (RR, 0.42; 95% CI, 0.26-0.68; P < 0.001), lower cardiovascular mortality (RR, 0.44; 95% CI, 0.24-0.81; P = 0.008), greater improvement in NYHA class (-0.34; 95% CI, -0.56 to -0.13; P = 0.002), and a reduction in appropriate/ inappropriate defibrillator shocks, compared with medical therapy for ventricular response rate control.^{615,616} These

potential benefits of AVJ ablation have to be weighed against the implications of rendering a patient pacemakerdependent. The Cardiac Resynchronisation Therapy and AV Nodal Ablation Trial in Atrial Fibrillation Patients (CAAN-AF; NCT01522898) and the Resynchronization/ Defibrillation for Ambulatory Heart Failure Trial in Patients With **Perm**anent **AF** (RAFT-Perm AF; NCT01994252) trials should provide additional insight into the role of CRT in patients with AF and HF.

RECOMMENDATION

97. We suggest AVJ ablation in HF patients with AF who are CRT nonresponders with biventricular pacing < 95% despite maximally tolerated doses of ratecontrolling drugs (Weak Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a high value on the documented importance of a high percentage of biventricular pacing for effective resynchronization therapy.

9.3. Long-term pharmacologic rhythm control

9.3.1. Antiarrhythmic drugs

A strategy of sinus rhythm maintenance using long-term antiarrhythmic drug therapy is preferred for those with recently diagnosed AF (ie, within a year), and might be considered for other symptomatic patients with established AF (Fig. 16). Because long-term antiarrhythmic therapy might not completely suppress AF, the focus of rhythm control should be on symptom relief, improving functional capacity and QOL, and reducing health care utilization while balancing potential adverse drug effects. Moreover, a recent study showed that an initial rhythm control strategy for patients with recently diagnosed AF was associated with decreased cardiovascular mortality and a reduced incidence of thromboembolic events compared with rate control alone.⁵¹⁵

Efficacy and safety data of common antiarrhythmic drugs used for rhythm control have been summarized in several systematic reviews.⁶¹⁷⁻⁶¹⁹ In a meta-analysis of 59 RCTs, the pooled recurrence rates of AF was 64%-84% at 1 year in control participants. Antiarrhythmic therapy reduced recurrence rates to 20%-50%. The most effective drug was amiodarone (OR, 0.22; 95% CI, 0.16-0.29 for recurrence vs placebo).⁶¹⁷ Proarrhythmic events (ventricular or bradyarrhythmia) were significantly more frequent with sotalol (OR, 6.44; 95% CI, 1.03-40.24; P = 0.047) and propafenone (OR, 4.06; 95% CI, 1.13-14.52; P = 0.035), but were not significantly more frequent for flecainide (OR, 6.77; 95% CI, 0.85-54.02; P = 0.067) or amiodarone (OR, 5.45; 95% CI, 0.69-42.93; P = 0.095). Antiarrhythmic drugs have not been associated with a beneficial effect on mortality, and long-term use of sotalol and amiodarone have been associated with increased mortality (OR, 4.32; 95% CI, 1.59-11.70; P = 0.013 and OR, 2.73; 95% CI, 1.00-7.41; P = 0.049, respectively).⁶¹⁷ A single study showed stroke risk reduction with dronedarone use compared with placebo (OR, 0.69;

95% CI, 0.57-0.84) however, this finding has not been confirmed by other studies. 620,621

The antiarrhythmic drug doses, contraindications, or precautions are summarized in Supplemental Table S11. The initial choice of antiarrhythmic therapy is primarily driven by safety and tolerability, because these agents have a relatively similar efficacy (Fig. 19). If the initial drug does not achieve the desired results, an alternative antiarrhythmic may be used or catheter ablation may be considered. When the decision is made to abandon pharmacologic rhythm control and favour a rate control strategy, the antiarrhythmic drug should be discontinued.

RECOMMENDATION

98. We recommend a rhythm control strategy for patients with established AF who remain symptomatic with rate control therapy, or in whom rate control therapy is unlikely to control symptoms (Strong Recommendation; Moderate-Quality Evidence). We suggest consideration be given to rhythm control rather than rate control for patients with newly diagnosed AF (Weak Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a high value on the significant reductions in cardiovascular mortality and stroke observed in patients with newly diagnosed AF (within a year) treated with rhythm control, and lesser value on the increased adverse events observed in such patients. For those with established AF this recommendation places a high value on a decision-making process shared with patients, which considers the likelihood of improved symptoms, QOL, and health care utilization while minimizing adverse drug effects compared with other treatment strategies (rate control or catheter ablation).

Practical tip. In select cases ablation might be preferred as first-line therapy (eg, rather than oral antiar-rhythmic therapy) for patients with recurrent AF in whom long-term rhythm control is desired.

Practical tip. Long-term oral antiarrhythmic therapy should not be continued in patients when AF becomes permanent.

- 99. We recommend that the goal of rhythm control therapy should be an improvement in cardiovascular outcomes, patient symptoms, and health care utilization, and not necessarily the elimination of all AF episodes (Strong Recommendation; Moderate-Quality Evidence).
- 100. We recommend that the choice of antiarrhythmic drug used for long-term pharmacologic rhythm control be defined according to patient characteristics (Strong Recommendation; Moderate-Quality Evidence).
- 101. We recommend that pharmacologic rhythm control should be avoided in patients with AF and advanced sinus or AV nodal disease unless the patient has a pacemaker or implantable defibrillator (Strong Recommendation; Low-Quality Evidence).
- 102. We recommend an AV nodal-blocking agent (β-blockers and ND-CCBs) be used in combination with class I antiarrhythmic drugs (eg, flecainide or propafenone) (Strong Recommendation; Low-Quality Evidence).

103. We recommend the use of amiodarone for pharmacologic rhythm control only when the potential for drug toxicities is considered and when other choices are contraindicated or have failed (Strong Recommendation; Low-Quality Evidence).

9.3.1.1. Flecainide and propafenone

Class Ic antiarrhythmic agents, such as flecainide and propafenone, are use-dependent sodium channel-blocking drugs.⁶²² The concomitant use of AV nodal blocking agents is recommended with class Ic antiarrhythmic drug therapy because of the potential risk of AF organization into AFL, with the potential of 1:1 AV conduction and rapid ventricular rate. Class Ic agents should be avoided in patients with: (1) preexisting advanced AV block (second- or third-degree AV block) or significant conduction system disorders (left bundle branch block, or right bundle branch block when associated with left hemiblock); (2) LV systolic dysfunction (LVEF \leq 40%); (3) significant LV hypertrophy; (4) severe hepatic or severe renal impairment (CrCl < 35 mL/min); and (5) ischemic heart disease (active ischemia or history of MI). Because the use of flecainide has been associated with increased mortality when administered to suppress ventricular ectopy in the context of recent MI,⁶²³ a formal ischemia assessment (eg, stress test) should be considered before initiation of class Ic antiarrhythmic drugs in patients older than 50 years of age or with significant atherosclerotic risk factors. In addition, it is reasonable to consider annual assessment of symptoms of CAD for patients receiving long-term class Ic antiarrhythmic use, with formal stress testing being performed if significant symptoms are present. An ECG should be performed at baseline and after initiation to monitor for PR and QRS interval prolongation. An increase in QRS duration >25% compared with baseline increases proarrhythmia risk. 624

9.3.1.2. Sotalol

Sotalol is a drug with reverse use dependence I_{kr} inhibition and is also a β -blocker. At lower doses the β -blocker effects predominate whereas the class III effects emerge with higher doses. The major risk of sotalol is QT prolongation and TdP. Sotalol should be avoided in patients with: (1) preexisting QTc prolongation (congenital or acquired long QT syndromes); (2) high-degree AV conduction disorders; (3) severe renal impairment (dose adjustment required for CrCl 40-60 mL/min; avoid with CrCl < 40 mL/min); (4) significant LV systolic dysfunction (LVEF \leq 40%); or (5) significant risk factors for TdP (eg, women aged older than 65 years who are receiving diuretics or those with renal insufficiency). An ECG should be performed at baseline and 48-72 hours after outpatient initiation to monitor for QT interval prolongation.

9.3.1.3. Amiodarone

Amiodarone is a multichannel blocker and a nonselective β blocker. A loading regimen of 10-12 g is recommended and then maintenance of \leq 200 mg daily. Amiodarone has a long half-life and long-term use increases the risk for numerous extracardiac toxicities that affect the skin, thyroid, pulmonary, liver, and neurological systems. It should not be used as a firstline therapy when another drug might be an option. Amiodarone should be avoided in patients with: (1) high-degree AV conduction disorders; (2) active hepatitis or significant chronic liver disease; (3) pulmonary interstitial abnormalities; (4) preexisting QTc prolongation (congenital or acquired long QT syndromes); (5) hypersensitivity to the drug components, including iodine; or (6) concomitant use of strong cytochrome P450 3A4 (CYP3A4) inhibitors (eg, ketoconazole, cyclosporin, clarithromycin, ritonavir). Surveillance investigations are recommended with the use of amiodarone (eg, liver and thyroid tests every 6 months, annual chest radiograph).⁶²⁵ Amiodarone inhibits CYP3A, CYP2C9, and P-glycoprotein drugs, which requires close monitoring and dose adjustment of other affected medications (eg, digoxin, VKA, HMG-CoA reductase inhibitors). Patients should be counselled to be diligent in use of sun protection because of photosensitivity and report any symptoms indicative of pulmonary toxicity (eg, persistent nonproductive cough), optic neuropathy (changes in visual acuity and decreases in peripheral vision), or hepatic injury.

9.3.1.4. Dronedarone

Dronedarone resembles amiodarone but removal of iodine and the addition of a methane-sulfonyl group results in shortening the half-life (approximately 24 hours) and less tissue accumulation.⁶²⁶ Dronedarone is the only antiarrhythmic drug shown to reduce hospitalization and cardiovascular mortality in patients with paroxysmal or persistent AF with at least 1 additional risk factor for death.⁶²⁰ Dronedarone should be avoided in patients with: (1) permanent AF; (2) HF with recent decompensation requiring hospitalization or LV systolic dysfunction (LVEF \leq 40%), owing to the observed increase in mortality observed with dronedarone use in this population⁵⁹⁷; (3) highdegree AV conduction disorders; (4) patients with previous lung or liver injury related to previous use of amiodarone; (5) preexisting QTc prolongation; or (6) severe hepatic impairment. Because of the risk of hepatotoxicity it is recommended that liver function tests be performed every 3 months for the first year of use, then every 6 months thereafter.

9.3.2. Pill-in-the-pocket antiarrhythmic drug therapy

In selected patients with symptomatic, infrequent, longerlasting episodes of AF, the use of intermittent class Ic antiarrhythmic therapy ("pill-in-the-pocket") shortly after symptom onset to restore sinus rhythm might be an alternative approach to daily antiarrhythmic use.^{557,560-563} Indications, contraindications, and monitoring details are presented in Supplemental Table S12.

RECOMMENDATION

104. We recommend intermittent antiarrhythmic drug therapy ("pill-in-the-pocket") as an alternative to daily antiarrhythmic therapy in patients with infrequent, symptomatic episodes of AF (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation is on the basis of the results of observational cohort studies that have shown efficacy and safety of intermittent antiarrhythmic drug therapy in selected patients. It places a high value on patient preferences and capabilities.



¹Consider AF symptom burden, possibility of adverse drug reactions and patient preference ²Consider alternative AADs or ablation rather than long-term amiodarone (significant risk of extra-cardiac side-effects) ³Sotalol should be used with caution in patients with high-risk features for torsade de pointes (≥ 65 years, women, reduced renal function, concomitant potassium-wasting diuretics). Sotalol is not recommended for patients with left ventricular hypertrophy. ⁴Dronedarone should be used with caution in combination with digoxin ⁵Class IC agent should be combined with AV-nodal blocking agent. Use caution for patients with left ventricular hypertrophy.

Figure 19. Approach to long-term rhythm control. AAD, antiarrhythmic drug; AF, atrial fibrillation; AV, atrioventricular; CAD, coronary artery disease; LVEF, left ventricular ejection fraction; LVH, left ventricular hypertrophy.

9.3.3. Trial of cardioversion

Many patients with persistent AF can present with the insidious onset of vague symptoms, such as fatigue or decreased exercise tolerance. In this population it can be difficult to determine the contribution of AF to their clinical presentation. A trial of sinus rhythm using cardioversion can often be useful in this population with established AF to determine if these nonspecific symptoms are secondary to AF. If so, then rhythm control might be the preferred treatment strategy (Fig. 16).

9.4. Catheter ablation of AF

Catheter ablation of AF has emerged as an important therapeutic modality for this common arrhythmia. The goal of catheter ablation is to eliminate the triggers and substrate responsible for the initiation and maintenance of AF. Ablation is performed as a percutaneous procedure in which catheters are inserted through venous access into the heart using fluoroscopic and electroanatomical mapping systems to identify regions of interest. These regions can then be ablated using thermal energy, namely radiofrequency or cryothermy.⁶²⁷ Irreversible electroporation (pulsed electrical fields) is a recent modality that might achieve the same effect, without the risk of thermal energy-related complications.⁶²⁸

Most of the triggers for AF originate within the PVs of the left atrium.⁶²⁹ Therefore, the "cornerstone" of all AF ablation procedures is ablating around the PV to electrically isolate them (Fig. 20). PV isolation (PVI) for paroxysmal AF is associated with success rates ranging from 60% to 80% after 1 procedure. For persistent AF, single-procedure success rates are lower (50%-70%), and it is unclear if ablation beyond PVI is required.⁶³⁰ Although the success rates seem suboptimal,



Figure 20. Ablation lesion set for pulmonary vein isolation. Three-dimensional electroanatomic voltage map of the left atrium (LA) including the pulmonary veins. The **purple colour** indicates normal ("healthy") myocardium. The **red dots** represent point-by-point ablation lesions created by radiofrequency energy in the left atrium surrounding the pulmonary veins, thus electrically disconnecting them from the rest of the atrium (**red colouring** indicates electrical silence). Veins are labelled as follows: LSPV, left superior pulmonary vein; LIPV, left inferior pulmonary vein; RSPV, right superior pulmonary vein; RIPV, right inferior pulmonary vein.

these are often determined on the basis of the total elimination of AF. However, in many cases substantial clinical benefit can be achieved without total elimination of AF.⁶³¹ When considered quantitatively, AF ablation substantially reduces the burden of AF (ie, time spent in arrhythmia) by > 98%.⁶²⁷

The CABANA trial (2204 patients) is the largest trial of catheter ablation vs medical therapy.³⁹⁴ In this trial patients with previous antiarrhythmic drug failure (47%) as well as antiarrhythmic-naive patients (53%) were enrolled. Although in the intention to treat analysis the trial did not show a difference between ablation and medical therapy for the primary end point of death, stroke, bleeding, and cardiac arrest, there was a significant reduction in the secondary end point of mortality and cardiovascular hospitalization in the ablation group (HR, 0.83; 95% CI, 0.74-0.93; P = 0.001). Moreover, ablation was associated with a significant reduction in AF recurrence (HR, 0.52; 95% CI, 0.45-0.60; P < 0.001) with the per-protocol analysis showing that ablation was associated with a significant reduction in the 73; 95% CI, 0.54-0.99; P = 0.046).

9.4.1. Procedural considerations for catheter ablation of AF

AF ablation is a complex procedure that requires a high degree of operator expertise. Recent balloon-based technologies have helped to make the procedures shorter and more consistent, but there remains evidence that such procedures should only be performed by individuals with appropriate experience and technological support. Recent evidence has shown that the rate of complications is directly related to the number of procedures performed at an institution and/or by the operator.^{632,633}

RECOMMENDATION

105. We suggest that catheter ablation of AF should be performed by electrophysiologists with a high degree of expertise and high annual procedural volumes (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation recognizes that the risks of catheter ablation are directly related to operator experience and procedural volume at a given centre. Although it is difficult to specify exact numerical values, the threshold seems to be 25-50 procedures per operator per year.

9.4.2. Catheter ablation of AF after a trial of antiarrhythmic drugs

The primary indication for AF ablation is to achieve rhythm control in patients in whom antiarrhythmic drug therapy has already failed. Two systematic reviews have been performed, one before (2272 patients in 17 RCTs)⁶³⁴ and one including CABANA (4464 patients in 18 RCTs).⁶³⁵ Overall, the quality of studies was high and risk of bias low. Most patients included in these trials were patients in whom at least 1 antiarrhythmic drug had already failed. Despite the large difference in sample size, outcomes were remarkably consistent with reasonably large reductions in cardiovascular hospitalizations (RR, 0.63; 95%)

CI, 0.46-0.87; P = 0.01, and RR, 0.56; 95% CI, 0.39-0.81; P = 0.0001, respectively) and AF recurrences (RR, 0.44; 95% CI, 0.31-0.61; P = 0.001, and RR, 0.42; 95% CI, 0.33-0.53; P < 0.00001, respectively) among patients who underwent catheter ablation, compared with antiarrhythmic therapy.

All-cause mortality was also reduced in both meta-analyses but was driven by studies that enrolled patients with HF and reduced EF; sensitivity analysis showed no statistically significant reduction in all-cause mortality when these studies were removed from the analysis (RR, 0.48; 95% CI, 0.23-1.01; P = 0.05, and RR, 0.67; 95% CI, 0.23-1.99; P = 0.47, respectively). In patients without HF, mortality rates were too low to assess the potential for mortality benefit.

Stroke rates were very low across all study populations and no effect of catheter ablation was shown on the rate of stroke (0.68% vs 1.23% for medical therapy; RR, 0.56; 95% CI, 0.26-1.22; P = 0.14). Likewise, major bleeding was not significantly more frequent after catheter ablation (3.4% vs 2.5% for medical therapy; RR, 1.55; 95% CI, 0.83-2.91; P = 0.17).

One RCT specifically addressed QOL as the primary outcome in patients in whom previous antiarrhythmic drug therapy had failed.⁶³⁶ Similar to contemporary trials,⁶³¹ it showed important improvements in AF-specific and general QOL driven predominantly by improvements in AF symptoms and AF-related hospitalizations.

RECOMMENDATION

106. We recommend catheter ablation of AF in patients who remain symptomatic after an adequate trial of antiarrhythmic therapy and in whom a rhythm control strategy remains desired (Strong Recommendation; High-Quality Evidence).

Values and preferences. This recommendation recognizes the positive effect of catheter ablation on AF burden, symptoms, QOL, and cardiovascular hospitalizations, as well as the declining risks of the procedure.

Practical tip. Catheter ablation might be the preferred means to maintain sinus rhythm in select patients with symptomatic AF and mild-moderate structural heart disease, particularly systolic HF, who are refractory or intolerant to ≥ 1 antiarrhythmic medication.

9.4.3. Catheter ablation of AF as first-line treatment

Early intervention for AF can prevent progression to persistent AF and avoid some of the long-term risks of the arrhythmia including stroke and HF. Furthermore, studies have shown that success rates for catheter ablation are higher when used earlier in the disease.⁶³⁷ Therefore, several clinical trials have addressed whether AF ablation should be used as first-line therapy before the use of antiarrhythmic drugs. Three previous randomized studies suggested a benefit of first-line ablation over antiarrhythmic drugs but these were small.⁶³⁸⁻⁶⁴⁰ More recently the large randomized Early Agressive Invasive Intervention for Atrial Fibrillation (EARLY-AF) trial demonstrated that first-line cryoballoon catheter ablation resulted in a significant reduction in arrhythmia recurrence and AF burden, and a significant improvement in quality of life relative to first line antiarrhythmic drugs.⁶⁴¹

RECOMMENDATION

107. We suggest catheter ablation to maintain sinus rhythm as first-line therapy for relief of symptoms in select patients with symptomatic AF (Weak Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation recognizes that patients might have relative or absolute contraindications to pharmacologic rhythm control.

9.4.4. Candidates for AF ablation

Similar to long-term antiarrhythmic drug therapy, the decision to pursue a strategy of sinus rhythm maintenance should be aimed primarily at reduction of patient symptoms to improve QOL and reduce health care utilization. Patients who have truly asymptomatic AF are not generally considered candidates for ablation, however ablation might be pursued in those in whom AF is thought to adversely affect LV function even in the absence of overt symptoms.

A specific subgroup that deserves mention are patients with HF, in whom AF is a causative or major contributing factor. A number of studies and systematic reviews have shown clinically important improvements in HRQOL, exercise tolerance, and LV function associated with catheter ablation over pharmacological rate control or rhythm control strategies.^{642,643} In addition, 2 RCTs, including the Catheter Ablation vs Standard Conventional Therapy in Patients With Left Ventricular Dysfunction and Atrial Fibrillation (CASTLE-AF) trial, have recorded reduced hospitalizations in patients with HFrEF who underwent catheter ablation compared with medical therapy.^{644,645} A recent systematic review showed that catheter ablation in this population was associated with a statistically significant reduction in all-cause mortality (5.3% vs 7.9% for medical therapy; RR, 0.69; 95% CI, 0.54-0.88; P = 0.003; 9 studies, 3576 patients).⁶³⁵ Although a low level of bias was seen, the mortality benefit was driven by a single study (CASTLE-AF) that exclusively enrolled patients with HFrEF.⁶⁴⁵ A recent observational study showed that catheter ablation was associated with a long-term reduction in all-cause mortality and HF rehospitalizations.⁶⁴⁶ There are no data showing a mortality benefit in patients with HF and preserved EF. The large multicentre Canadian randomized ablation-based Randomized Ablation-Based Atrial Fibrillation Rhythm Control vs Rate Control Trial in Patients With Heart Failure and High Burden Atrial Fibrillation (RAFT-AF) study has closed enrollment and is due to report later this year (NCT01420393). These data will provide considerable insight into the effect of catheter ablation on HF with preserved EF and HFrEF.

First-line therapy might be considered in younger patients who wish to avoid the risks of long-term antiarrhythmic agent use. Patients might also have cardiac or noncardiac absolute or relative contraindications to antiarrhythmic drugs. Some patients with tachy-brady syndrome are unable to tolerate drug therapy because of bradycardia complications in the absence of a pacemaker. If the AF can be successfully ablated, then antiarrhythmic therapy and the need for permanent pacing might be avoided.⁶⁴⁷

9.4.5. Catheter ablation of AFL

Although AF and AFL might coexist in the same patient, AFL is an arrhythmia that is distinct from AF. In contrast to the apparent disorganization of AF, AFL usually involves a single macroreentrant atrial circuit rotating around a large central functional or anatomic obstacle (valves, veins, scar). AFL can be classified as "typical" (cavo-tricuspid isthmusdependent) or "atypical" (eg, non-cavo-tricuspid isthmusdependent right or left AFL). Atypical AFL arising from scar related to previous heart surgery (eg, atriotomy or prosthesis) or catheter ablation and is also known as "incisional" flutter.

Catheter ablation of typical right AFL is preferred to pharmacological therapy because of the relatively high success rate and relatively low rate of periprocedural complications. Ablation is typically performed with the patient under conscious sedation, using the femoral approach, guided by fluoroscopy or 3-D electroanatomic mapping. The goal is to create a complete ablation line along the cavo-tricuspid isthmus between the tricuspid annulus and the inferior vena cava to achieve bidirectional electrical conduction block. This intervention has an excellent acute and long-term success rate > 90%, which is more effective than pharmacological rhythm control. 400,648,649 In addition, ablation improves QOL, reduces symptoms, and might also avoid the development of a tachycardia-mediated cardiomyopathy.⁶⁴⁸ The acute complication rate is 2.6%, most commonly related to vascular access problems.⁴⁰⁰ Major complications, like complete heart block or pericardial effusion, are uncommon.⁶⁴⁹

In patients with coexisting AF and AFL it is perfectly cromulent to consider a hybrid treatment approach, in which AFL ablation is performed to prevent recurrent AFL while antiarrhythmic drug therapy is used to control AF.⁶⁵⁰ This approach has been effective for patients who derive clinical benefit from class Ic antiarrhythmic treatment of AF but manifest recurrent AFL.

RECOMMENDATION

108. We recommend catheter ablation of typical right AFL as a reasonable alternative to pharmacologic rhythm or rate control therapy (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation recognizes the high success rate and low complication rate of typical right AFL ablation, as well as the observation that typical right AFL is challenging to control medically with antiarrhythmic drugs and adequate rate control is difficult to achieve.

9.5. Arrhythmia surgery

Surgical treatment for AF can be accomplished as a "standalone" procedure or performed coincident with a planned surgical procedure (valve replacement/repair and/or CABG). A number of factors need to be considered when contemplating surgical AF ablation therapy, including the indication for the procedure (eg, the potential benefits of sinus rhythm), the likely efficacy of the procedure (considering local surgeon/ institutional experience), and the safety of the procedure. Moreover, in the context of adjuvant surgical ablation (eg, surgical ablation at the time of another surgery) the type of concomitant cardiac surgery (eg, mitral valve, which necessitates left atriotomy, vs CABG surgery, which does not) needs to be considered.

Recent studies have shown that LA ablation in addition to valve surgery and/or CABG was beneficial in terms of sinus rhythm maintenance. 651,652 In the first study, 224 patients who underwent CABG and valve surgery were randomized to LA cryoablation or control; 60.2% of cryoablation patients showed sinus rhythm according to Holter monitoring at 1 year, vs 35.5% (P = 0.002) of patients without ablation.⁶⁵¹ In the second study 63.2% of 260 mitral valve surgery patients randomized to ablation (either PVI alone or biatrial ablation [secondary randomization]) were in sinus rhythm at 1 year vs 29.4% of those randomized to no ablation (P < 0.001).⁶⁵² A recent systematic review combined 9 small RCTs and showed a large effect on the maintenance of sinus rhythm at 12 months in 481 patients (OR, 10.41; 95% CI, 5.30-20.44) as well as beyond 12 months (OR, 11.61; 95% CI, 4.53-29.79; 4 studies, 154 patients).65 Although no heterogeneity was observed among these trials, no individual study randomized more than 95 patients.

Despite the high reported rates of sinus rhythm maintenance after surgical ablation, the effect on other outcomes is controversial. Although there is evidence for an improvement in HRQOL outcomes with a surgical procedure that includes concomitant surgical AF ablation, in several RCTs the magnitude of improvement was observed to be similar to patients who did not undergo ablation at the time of cardiac surgery.⁶⁵⁴⁻⁶⁵⁶ Moreover, surgical ablation appears to have no effect on the incidence of late stroke/TIA (6 RCTs; OR, 1.01; 95% CI, 0.41-2.49; P = 0.98) or long-term survival (15 RCTs; OR, 0.91; 95% CI, 0.59-1.41; P = 0.67).⁶⁵⁷

A significant concern with surgical ablation relates to the possibility of increased postoperative morbidity secondary to the additional surgical procedure. In a large meta-analysis, concomitant surgical AF ablation was not associated with an increase in the incidence of perioperative morbidity (which included deep sternal wound infection, pneumonia, reoperation for bleeding, and renal failure) or perioperative ICU length of stay.⁶⁵⁷ Likewise, there was no difference in the rate of readmission within 30 days, short-term survival (< 30 days), or perioperative stroke/TIA (< 30 days).^{653,657} However, as highlighted in the two aforementioned RCTs, surgical ablation was associated with a significant increase in the rates of pacemaker implantation (6% vs 1% in Budera et al.⁶⁵¹ and 21.5% vs 8.1% in Gillinov et al.⁶⁵²).

RECOMMENDATION

109. We suggest that a surgical AF ablation procedure be considered in association with a planned cardiac surgical procedure (eg, mitral valve, aortic valve, or coronary artery bypass surgery) in patients with symptomatic nonpermanent AF when the likelihood of success is deemed to be high, the additional risk is low, and sinus rhythm is expected to achieve substantial symptomatic benefit (Weak Recommendation; Low-Quality Evidence). Values and preferences. This recommendation recognizes that there is no evidence that surgical AF ablation influences hard outcomes (eg, stroke, mortality, thromboembolic complications). Patient considerations and individualized risk-benefit analyses should determine for whom the surgical procedure is performed.

Practical tip. The symptomatic benefit of sinus rhythm needs to be balanced with the attendant risks of ablation surgery, including the increased need for permanent pacing (particularly for biatrial and/or Maze procedures).

There has been recent interest in hybrid AF ablation, achieved through collaboration between surgeons and electrophysiologists. This procedure encompasses surgical ablation (usually minimally invasive) coupled with percutaneous endocardial ablation of any residual gaps 6-8 weeks later. Similar to all invasive procedures, centre expertise and operator experience are important determinants of success. Comparative metaanalyses suggest a modestly improved freedom from antiarrhythmic drug use after hybrid AF ablation in comparison to traditional Cox-Maze procedures, but former approach is associated with a higher rate of complications.^{657,658} Some have advocated for the hybrid approach to involve surgical bilateral PVI with LAA closure coupled with endocardial ablation protocols, although this approach has not been tested in an RCT. Often a bilateral thoracoscopic intervention is needed in this setting. Other hybrid approaches might include unilateral thoracoscopic PVI encircling box lesions and posterior LA wall epicardial ablation lesion sets without concomitant LAAO.⁶⁵⁷ Although stand-alone surgical ablation presently comprises a small percentage of AF ablations, the ability to offer these newer approaches in a minimally invasive fashion without sternotomy or cardiopulmonary bypass is potentially attractive. Nevertheless, these approaches will need to be assessed in outcome studies.

RECOMMENDATION

110. We suggest that stand-alone surgical or hybrid ablation of AF may be considered for patients with symptomatic nonpermanent AF that is refractory to attempts at percutaneous catheter ablation and whose symptoms warrant the additional risk of a surgical procedure (Weak Recommendation; Low-Quality Evidence).

10. Sex Differences in Patients With AF

Recognition of sex differences offers an opportunity to improve outcomes in women with AF. 659

10.1. Epidemiology and pathophysiology

Age and sex are the two most powerful predictors of incident AF. Although the prevalence of AF doubles with each decade of age (increasing from 1%-4% at 60 years to 6%-15% at 80 years), male sex is associated with a 1.5-fold risk of AF, even after adjusting for age and predisposing conditions. Although the age-adjusted prevalence of AF is consistently observed to be higher in men (eg, a sex-based prevalence of

9.2% for women vs 15.0% for men in a community-based, randomized, controlled AF screening study performed in Sweden)^{31,85} the absolute number of female patients with AF exceeds the number of male patients with AF because of the longer life span of female patients.

Although the exact mechanism responsible for the reported sex-related differences in AF remains inadequately understood, several theories have been suggested. First, anthropomorphic differences between the sexes result in a larger LA dimension and volume in male patients.^{660,661} Second, female patients with AF have been shown to have a relatively greater burden of atrial fibrosis using delayed-enhancement MRI.⁶⁶² Third, male patients with AF have greater expression of repolarizing ion channel subunits, which could favour reentry.^{660,663} Fourth, the contribution of sex hormones has been explored in several studies, with testosterone deficiency having been linked to increased atrial arrhythmogenicity.⁶⁶⁴; progesterone associated with shortened action potentials.⁶⁶⁵; and estrogen has been postulated to play a central role in arrhythmogenesis due to prolongation in conduction time, action potential duration, and the atrial effective refractory period.⁶⁶⁶

10.2. Presentation

Female patients with AF are more likely to have underlying hypertension and valvular disease, whereas male patients with AF are more likely to have CAD and abnormal LV function. Female patients with AF report more atypical symptoms, with a relatively greater symptom burden and lower QOL compared with male patients.^{667,668} As a result, women are more likely to seek care for AF symptoms and are more likely to experience depression related to AF.⁶⁶⁹

10.3. Outcomes

Important sex-specific differences in cardiovascular outcomes have been described. AF in female patients is associated with a greater all-cause mortality relative to male patients (RR, 1.12; 95% CI, 1.07-1.17).⁶⁷⁰ Compared with male patients with AF, strokes experienced by female patients tend to be larger, and are associated with poorer functional outcomes and greater need for institutionalization.⁶⁷¹

10.4. Stroke prevention

Sex-specific differences in antithrombotic therapy have been observed: female patients with AF are more likely to be prescribed antiplatelet agents; when OACs are prescribed, they are more likely to receive a DOAC and, are more likely to be inappropriately prescribed the lower approved dose.⁶⁷²⁻⁶⁷⁵ In terms of efficacy, female patients with AF have a greater residual risk of stroke despite VKA therapy, which might reflect sex-specific differences in VKA metabolism or underlying risk factor control.^{676,677} Although no sex-specific difference in DOAC efficacy has been observed, there is a significant reduction in major and clinically-relevant nonmajor bleeding in female patients with AF treated with a DOAC.^{21-23,25,52,677}

10.5. Rate and rhythm management

Female patients with AF are more likely to receive rate control, compared with male patients with AF. 678,679 In those

who receive rhythm control, female patients with AF are preferentially managed with pharmacologic antiarrhythmic therapy, and are less likely to undergo ablation (OR, 0.5-0.8 compared with men).^{680,681} Moreover, female patients who do undergo ablation tend to be older, have more comorbidities, have more advanced AF (eg, longer duration of AF, more likely to be persistent), and show that treatment with a larger number of antiarrhythmic agents have failed.^{395,663,682,683} Despite their more complex clinical profile before ablation, female patients who undergo ablation have comparable acute and longer-term success rates compared with male patients.^{668,682,683}

11. AF and Special Populations

11.1. Device-detected AF

By convention, and on the basis of somewhat arbitrary definitions, the diagnosis of AF requires ECG documentation of an irregular rhythm with no discernible, distinct P waves, lasting at least 30 seconds. Contrary to this widely accepted threshold for AF diagnosis, the minimal duration of incessant AF that a patient should manifest before warranting OAC for stroke prevention remains a matter of debate, even in the presence of other stroke risk factors. The uncertainty relates to the few studies in which this was examined and the fact that the duration of subclinical AF associated with stroke differed widely across the studies. Specifically, AF episode durations associated with increased risk of stroke/systemic embolism ranged from 5 minutes in the Mode Selection Trial (MOST; HR, 2.79),⁹⁹ 6 minutes in the Asymptomatic Atrial Fibrillation and Stroke Evaluation in Pacemaker Patients and the Atrial Fibrillation Reduction Atrial Pacing Trial (ASSERT; HR, 2.5), > 1 hour in Stroke Prevention Strategies on the Basis of Atrial Fibrillation Information From Implanted Devices (SOS AF; HR, 2.1), 684 > 5.5-hour daily burden in the The Relationship Between Daily Atrial Tachyarrhythmia Burden From Implantable Device Diagnostics and Stroke Risk (TRENDS) study (HR, 2.4),¹⁰⁰ and > 24 hours in a prospective observational study (HR, 3.1).¹⁰¹

It is not entirely clear why there is a discrepancy between arrhythmia durations and stroke risk, but there are several hypotheses. First, the studies used CIEDs, which specifically detect AHRE, meaning not all of these arrhythmia episodes were AF or AFL.^{685,686} Second, the AF duration cut points across studies were not empirically derived. The 5-minute cutoff in the MOST study was chosen to avoid false positive results from oversensing⁹⁹; whereas the 5.5-hour threshold used in the TRENDS study corresponded to the median AHRE duration that was measured.¹⁰⁰ Third, studies did not specifically adjudicate strokes as being cardioembolic and, in some cases, these were combined with TIA and systemic emboli.⁶⁸⁵ Fourth, the incidence rates for stroke were generally low and, notwithstanding significant HR, the absolute risk imparted by AF was often small or unreported. Fifth, differences in study populations, and specifically their stroke risk profiles, could have accounted for some of the discrepancies. Patients with higher AF burden are typically older and/or have other conditions that independently increase stroke risk.^{685,686} Finally, it is unlikely that any influence of AF duration on stroke risk would manifest consistently across individual patients.

More recent analyses have questioned the association between shorter AF episodes and stroke risk. A reanalysis of ASSERT showed that the risk of ischemic stroke/systemic embolism in patients with an AF episodes lasting between 6 minutes and 24 hours was comparable with those without subclinical AF.⁹⁶ Conversely, episodes of AF lasting > 24 hours were associated with a greater than threefold increased risk of stroke/systemic embolism (HR, 3.2).

Clear insight into the correlation between the risk of stroke and the duration of incidentally discovered AF is overdue. The issue has become increasingly important because of the rapid proliferation of wearable technologies that detect arrhythmias, some of which have been marketed specifically to detect AF.⁶⁸⁷ Putting aside the relative inaccuracy of many consumer devices, ^{688,689} the surge in individuals requiring more prolonged and diagnostically precise rhythm monitoring will uncover far greater numbers of people with silent AF than is presently the case. Discovery of subclinical AF will generate patient anxiety over having a condition with potential serious consequences and pressure on physicians to implement treatment. The clinical quandary remains the point at which the upfront hazard of antithrombotic therapy, in terms of major and fatal bleeding, is outweighed by the preventable stroke risk conferred by a specific amount and duration of paroxysmal AF. Addressing these gaps are the focus of 2 ongoing clinical trials, Apixaban for the Reduction of Thrombo-Embolism Sub-Clinical Atrial Fibrillation Due to (ARTESiA; NCT01938248) and Non-Vitamin K Antagonist Oral Anticoagulants in Patients With Atrial High Rate Episodes (NOAH-AFNET 6; NCT02618577). In the absence of conclusive data, the CCS AF Guidelines Committee recognizes that it is reasonable to prescribe OAC for patients with AF who are aged 65 years or older or with a CHADS₂ score ≥ 1 who have episodes of subclinical AF lasting ≥ 24 continuous hours.

Last, rather than focus simply on the correlation between AF episode duration and stroke risk, a more useful exercise might be to incorporate AF burden into established stroke risk calculators to improve their predictive ability and better identify patients who might benefit from antithrombotic therapy. A 2009 retrospective study showed better stroke risk stratification as a result of combining AF episode duration (< 5 minutes, 5 minutes to 24 hours, and > 24 hours) with CHADS₂ score (scores of 0, 1, 2, and \geq 3).⁶⁹⁰ A similar study from 2019, which involved 28,032 patients improved stroke risk stratification by combining AF duration (no AF, 6 minutes to 23.5 hours, and > 23.5 hours) with CHA₂DS₂-VASc score (0-1, 2, 3-4, \geq 5).⁶⁹¹

RECOMMENDATION

111. We suggest that it is reasonable to prescribe OAC for patients with AF who are aged 65 years or older or with a CHADS₂ score ≥ 1 who have episodes of subclinical AF lasting > 24 continuous hours (Weak Recommendation; Low-Quality Evidence).

Practical tip. In patients with subclinical or devicedetected (implanted pacemaker, defibrillator, or cardiac monitor) AF, there appears to be an association between the duration of device-detected AF and the risk of stroke/ systemic embolism. Even in the absence of clinical AF, observational data suggest that continuous therapy with OAC should be considered in patients with episodes of device-detected AF lasting longer than 24 hours. For shorter episodes the risk-benefit ratio of OAC remains unclear because the stroke risk with device-detected AF appears to be lower than for clinical AF. While OAC may be considered in some patients with shorter-lasting episodes, ongoing RCTs will determine the value of routine OAC in these patients.

11.2. Hypertrophic cardiomyopathy

HCM is transmitted as an autosomal dominant trait and has an annual incidence of 0.3-0.5 per 100,000.^{692,693} The prevalence of AF in subjects with HCM varies between 18% and 40%, which is four- to sixfold more frequent than that in the non-HCM population.^{300,694-696} The development of AF in subjects with HCM might precipitate a sudden clinical deterioration and has been associated with a significant increase in morbidity and mortality.^{298,302,305,695,696}

ND-CCBs or β -blockers may be used for rate control in subjects with HCM. These agents are negative inotropes with bradycardic effects, which might improve symptoms related to obstruction, diastolic dysfunction, and myocardial demand. Digoxin should be avoided in subjects with HCM.

Amiodarone and dofetilide are the preferential pharmacological rhythm control agents, with sotalol reserved for subjects with mild hypertrophy.^{697,698} Flecainide and propafenone should be avoided because of risk of proarrhythmia. Select subjects may be considered for catheter ablation, typically after a trial of antiarrhythmic drug therapy. The AF-free survival is lower in subjects with HCM compared with the general population with AF. The safety outcomes were comparable with those in the general population that underwent catheter ablation for AF. Subjects with longstanding persistent AF, older subjects, and subjects with evidence of advanced atrial remodelling are less likely to benefit from catheter ablation.⁶⁹⁹⁻⁷⁰⁶ In subjects with HCM and AF who undergo surgical myectomy a concomitant Cox-Maze procedure might be considered.^{707,708}

Anticoagulation management of patients with HCM are discussed in section 8.3.7.

11.3. AF and athletic participation

The relationship between AF and athletic participation is complex. In individuals who exercise < 5-7 hours per week, it is entirely unclear whether sport or exercise bears any consistent contribution to the observation of AF, which is particularly true for recreational athletes who do not train for competition.

For individuals with a history of participation in endurance sports, the data are inconsistent. Retrospective, poorly controlled cohort studies of former or current endurance athletes have suggested RR ratios for AF occurrence of between 3 and 5 compared with sedentary individuals; however, higher-quality data (eg, prospective cohort studies) have observed the lowest RR.⁷⁰⁹ The largest prospective cohort study followed 17,000 physicians (median age 50 years) for 15-19 years.⁷¹⁰ The absolute risk of AF in the highest exercise group (defined as exercising 5-7 days per week) was 0.7%-0.8% per year, equivalent to an absolute increase in risk over sedentary persons of 0.1%-0.2% per year. In contrast, several large meta-analyses have shown an inverse relationship between physical fitness and AF incidence, with the lowest rates of AF occurrence observed in the most fit group compared with the least fit group.⁷¹¹⁻⁷¹³

Similarly, the relationship between AF and exercise duration and intensity is complex. Mozaffarian et al. reported that the individuals older than 65 years with the greatest time spent in leisure activity had the lowest risk of AF, whereas individuals with the highest intensity of exercise had no such reduction in the risk of AF.⁷¹⁴ Similarly, Ricci et al. reported that men with the highest intensity of exercise (> 20 MET hours per week) did not have a reduction in risk of AF compared with sedentary individuals, whereas those with < 20 MET hours per week did show a significant reduction.⁷¹⁵ Conversely, the top quintile of exercisers in the Women's Health Study (> 23 MET hours per week) experienced a decrease in the incidence of newly documented AF (1.87 vs 2.43/1000 person-years of follow-up in sedentary individuals).⁷¹⁶ A similar sex-based difference was observed in a large study of 208,654 long-distance skiers in Sweden, in which female skiers had a lower incidence of AF than female nonskiers (HR, 0.55; 95% CI, 0.48-0.64).⁷¹⁷ Conversely, male skiers had an AF incidence similar to that of nonskiers (HR, 0.98; 95% CI, 0.93-1.03). Moreover, there was an increase in incident AF in men who completed \geq 3 races (HR, 1.24; 95% CI, 1.03-1.51). Taken together, these data indicate that levels of physical exercise that are less than intense competitive training are not risk factors and might actually be protective against AF. Conversely, competitive endurance sport is a risk factor for AF in men, however the absolute risk is very low.

To our knowledge, there have been no randomized or controlled trials of AF management in athletes.⁷¹⁸

11.3.1. Careful attention to standard risk factors for AF

In addition to the factors outlined in section 6, particular attention should be placed on the assessment of hypertension, including overt and masked hypertension (not present in the clinic but present during daily life, assessed using ambulatory BP monitoring); unexpected OSA (in the absence of obesity); occult valvular disease such as mitral regurgitation; and alcohol consumption.

11.3.2. Rate and rhythm management

In most athletes, AF presents as paroxysmal as opposed to persistent AF. Paroxysms often occur at rest or at nighttime and are often not present during exercise or training. In such patients, treatment should be directed at improving QOL rather than suppressing AF episodes or reducing AF burden because there is no convincing evidence that AF suppression will alter morbidity or stroke in the athletic population.

Specific treatment might not be required in patients with infrequent AF episodes that occur at rest, particularly if the symptoms are mild, the episodes are self-limiting, and the spontaneous ventricular rate is < 110 bpm. Highly symptomatic patients with impaired QOL might warrant a "pill-in-the-pocket" approach if the episodes are infrequent or, alternatively, antiarrhythmic drug therapy if the episodes are more significantly symptomatic or prolonged (as outlined in section 9.3.2). If "pill-in-the-pocket" therapy is used, then it is recommended that vigorous exercise be avoided within 6 hours of administration to avoid 1:1 conduction of AFL.

For individuals who have very frequent AF episodes, continuous antiarrhythmic drug therapy may be considered. Class Ic antiarrhythmics such as flecainide or propafenone (if the individual does not have LV dysfunction or CAD, which is unusual in athletes), combined with an AV nodal blocker are typically preferred. Some authors favour CCBs as the AV nodal blocking agents of choice in light of empirical and anecdotal evidence that suggests that β -blockers are very poorly tolerated by athletes. Specifically, most studies of CCBs show no change or improved exercise tolerance in AF, whereas β -blockers lead to no change or decreased exercise tolerance despite effective rate control.⁵⁷⁷

Catheter ablation should be considered for individuals whose QOL is substantially impaired, particularly if the "pillin-the-pocket" strategy or continuous antiarrhythmic drug therapies are ineffective, poorly tolerated, or strongly not desired by the patient. Anecdotal evidence and small case series suggest that PVI is at least as effective in athletes as it is in sedentary individuals, particularly because risk factors for decreased efficacy such as longstanding persistent AF, obesity, OSA, and LV dysfunction are rarely present in athletes.⁷¹⁹

Although a reduction in exercise intensity or duration ("detraining") has been considered in the management of AF in athletes, most athletes are reluctant to contemplate this strategy. Moreover, there are no controlled or prospective studies to suggest that detraining can reduce the frequency, severity, or duration of AF episodes in athletes.⁷²⁰

Athletes might also frequently present with AFL as the initial presenting arrhythmia rather than AF,⁷²¹ and AFL ablation can be safely performed as the primary treatment strategy in these individuals. One study suggests that the risk of AF after AFL ablation might be higher in athletes than in nonathletes, but AF still occurs in < 25% of patients at 1 year.⁷²¹

RECOMMENDATION

- 112. We suggest a period of decreased exercise intensity ("detraining") be considered as a possible management strategy in individuals engaged in highintensity long-duration endurance activity, taking into account values and preferences (Weak Recommendation; Low-Quality Evidence).
- 113. We suggest early catheter ablation (PVI) be considered for athletes, considering their values and preferences (Weak Recommendation; Low-Quality Evidence).

11.3.3. Stroke risk in athletes vs nonathletes with AF

There are few data to assess the relative and absolute risk of stroke in athletes with AF vs that in nonathletes. Confounding the assessment of a potential risk of stroke are the observations that athletes tend to be younger than nonathletes when they develop AF; they tend to have fewer stroke risk factors; and their AF episodes tend to be more often infrequent and paroxysmal. These observations suggest a lower risk of stroke in athletes when controlling for other stroke risk factors. The risk of stroke in cross-country skiers with AF (some of whom received anticoagulant therapy) was significantly lower than in nonskiers without AF (HR, 0.64; 95% CI, 0.60-0.69).⁷¹⁷ These caveats notwithstanding, it seems prudent to manage athletes similarly to nonathletes, with the decision regarding OAC being on the basis of the "CCS algorithm."

Some athletes might be concerned about the possibility of injury leading to a higher risk of bleeding, but there are no data to support a different treatment approach in such individuals except possibly those engaged in competitive contact sports. Using the shared decision-making model, athletes should be counselled regarding stroke risk and advisability of OAC when warranted according to guidelines.

11.4. Congenital heart disease

Atrial tachyarrhythmias are highly prevalent in patients with CHD,⁷²² are the leading cause of morbidity/hospitalizations, and have been linked to HF, sudden death, and stroke.^{277,280,723-725} Contemporary prevalence estimates for atrial arrhythmias in adults with CHD range from 10% to 15%,²⁷⁷ with > 50% of those with severe CHD projected to develop an atrial arrhythmia by 65 years of age.²⁷⁷

Whereas organized atrial macroreentrant arrhythmias are the most common supraventricular arrhythmia in CHD patients, the prevalence of AF is rising, particularly with atrial or AV septal defects, Ebstein anomaly, tetralogy of Fallot, univentricular hearts, and left-sided obstructive lesions.^{726,727}

The type and prevalence of arrhythmias depend, in part, on the subtype of CHD, nature of the surgical intervention, residual defects, and age.^{728,729} Potential contributors include surgical incisions, natural conduction barriers (eg, valve orifices, venous structures, septal defects, crista terminalis), and sequelae of chronic hemodynamic or hypoxic stress (eg, fibrosis, hypertrophy). Defects associated with the highest prevalence of atrial arrhythmias include single ventricles with Fontan palliation, transposition of the great arteries with atrial redirection surgery (eg, Mustard or Senning baffle), tetralogy of Fallot, Ebstein anomaly, and atrial septal defects.⁷³⁰ Factors associated with IART with some consistency include older age, more complex CHD, later repair, coexisting sick sinus syndrome, and right atrial dilation.^{731,732} In contrast, AF is predicted by factors such as older age, residual left-sided obstructive lesions, lower systemic ventricular EF, and LA dilation.^{731,733} It is also noteworthy that acquired comorbidities associated with AF in the general population are applicable to adults with CHD, including hypertension, obesity, OSA, and male sex.734,735

In a multicentre North American study of adults with CHD and atrial arrhythmias, AF accounted for 29% of presenting arrhythmias, IART for 62%, and focal atrial tachycardia for 9.5%.⁷³⁵ A strong association between AF and age was observed, with AF surpassing IART as the most common presenting arrhythmia in patients 50 years of age or older. Moreover, although the predominant pattern of AF was paroxysmal (> 60%), permanent AF (20%) more frequently occurred in older patients. The coexistence of AF with IART was frequently observed, with the most common scenarios being IART progressing toward AF, and paroxysmal forms transforming to permanent.^{735,736}

11.4.1. Rate and rhythm management

11.4.1.1. Surgical considerations

The timing and type of surgery can have a major effect on arrhythmia outcomes.⁷²⁹ Knowledge of arrhythmic complications during follow-up has led to surgical modifications to improve outcomes in subsequent generations. Although preventive arrhythmia surgery could be considered in specific circumstances, there is currently no evidence to support a prophylactic left-sided Maze procedure in adults with CHD undergoing cardiac surgery for other indications.²⁸² However, in patients with preexisting atrial arrhythmias, AF can be addressed surgically with a modified Cox-Maze III procedure, albeit with scarce supportive data.³⁷ A modified right atrial Maze in conjunction with an LA Cox-Maze III procedure has been performed in patients with univentricular hearts and AF undergoing Fontan conversion or revision. In general, these decisions are best guided by an interdisciplinary team that includes an electrophysiologist with expertise in the care of adults with CHD. Prophylactic arrhythmia surgery should not be performed if it substantially increases surgical morbidity or mortality.

11.4.1.2. Pharmacological therapy

A paucity of data exist to inform optimal pharmacological strategies for AF in adults with CHD. Conversion rates with ibutilide and sotalol for the acute termination of AF have ranged from 50% to 80%, with hypotension, bradycardia, and TdP as reported complications.^{738,739} Successful pharmacological cardioversion has also been observed with dofe-tilide in 41% of patients with CHD.⁷⁴⁰ There exists no efficacy or safety data on acute cardioversion of AF in patients with CHD treated with class Ia, Ic, or other class III agents.

Regarding long-term management, in the absence of CHDspecific data, rhythm control is generally preferred to rate control as an initial treatment strategy in those with moderate or complex CHD.²⁸² Rhythm control is justified on the basis that AF can be poorly tolerated in the context of conditions such as a single ventricle, systemic right ventricle, cyanosis, concomitant pulmonary hypertension, and/or significant residual hemodynamic lesions.²⁸² The global clinical scenario, including coexisting bradyarrhythmia and ventricular dysfunction should be taken into consideration in the selection of pharmacological agents.²⁸² Current expert recommendations discourage the use of class I antiarrhythmic agents in patients with CAD or systolic dysfunction of a subaortic or subpulmonary ventricle.^{282,741} Observational studies have suggested that class III antiarrhythmic agents are most effective in reducing recurrences of atrial arrhythmias in patients with CHD.742 Amiodarone is considered a drug of choice in the context of HF. However, particular to the CHD population, amiodarone-induced thyrotoxicosis is four- to sevenfold higher in those with cyanotic heart disease and Fontan palliation.^{743,744} Dofetilide appears to be a reasonable alternative to amiodarone in the setting of ventricular dysfunction or refractory arrhythmias.745 In a multicentre study, dofetilide led to better control of atrial arrhythmias in 49% of patients with CHD at 3 years of follow-up.⁷

11.4.1.3. Catheter ablation

Small studies have recently emerged on catheter ablation for AF in adults with CHD centred on PVI. A single-centre series of 36 patients, most of whom had atrial (61%) or ventricular (17%) septal defects, included 72% with paroxysmal and 28% with persistent AF.746 Freedom from recurrent AF without antiarrhythmic drugs was 42% at 300 days and 27% at 4 years after a single procedure. The largest series of AF ablation in patients with CHD included 57 patients, with a mean age 51 years, of whom 61% had simple, 18% moderate, and 21% severe forms of CHD.747 The pattern of AF was paroxysmal in 37% and persistent in 63%. Freedom from recurrent atrial arrhythmias with or without antiarrhythmic drugs was 63% at 1 year and 22% at 5 years. Recovery of PV conduction was observed in 65% of patients who underwent a second intervention. One case series described the feasibility and safety of cryoballoon ablation in 10 patients with AF and CHD.⁷⁴⁸ PVI was acutely successful in all, with no major complication. One year after a single ablation procedure, 60% remained free from AF. The totality of evidence thus far suggests that catheter ablation is feasible, safe, and modestly efficacious. A more thorough understanding of underlying mechanisms and extra-PV triggers carries the potential to further improve the selection of optimal candidates and procedural outcomes.⁷²

11.5. Secondary AF or AF due to reversible precipitants

As outlined in section 1.2, the likelihood of developing AF varies across physiological and pathological states. Conceptually, AF may be considered "primary" if the AF represents an established pathophysiological process or "secondary" if the AF is caused by a self-limited or at least partially reversible precipitant.¹⁸ Within secondary AF, these episodes can be conceptualized as arising from a combination of several complementary elements. The first is the patient's underlying propensity to AF due to modifiable and nonmodifiable substrates, the second is the new substrate imparted by the acute/secondary precipitant, and the third is the reversible AF trigger provoked by the secondary cause (Fig. 3).749 This conceptual model suggests that secondary AF will be more likely in patients with a propensity to AF (eg, more baseline substrate) and, even when the secondary cause for the disease is completely reversible these patients might still be at higher risk for future events.

11.5.1. Common causes of secondary AF

Common causes of secondary AF include surgery (cardiac and noncardiac), acute cardiac pathology (eg, MI and myopericarditis), acute pulmonary pathology (eg, COPD, pulmonary emboli, pneumonia), thyrotoxicosis, infection and sepsis, acute alcohol consumption and substance use, electrocution, and other metabolic disorders.¹⁸

RECOMMENDATION

114. We recommend that secondary causes for AF be identified and treated (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a high value on evidence that supports a reduction in recurrence with treatment of the underlying reversible trigger (see Tables 1 and 2, and Fig. 6).

11.5.2. Rate and rhythm management of secondary AF

The decision to provide rate or rhythm control for secondary AF depends on the clinical context. In all cases, management of the underlying precipitant is paramount. For AF associated with acute medical illness, the risk-benefit balance must take into consideration the underlying medical condition (see section 9.1). In the case of AF associated with hyperthyroidism, the hyperadrenergic state encourages a rapid ventricular response and often necessitates β -blockade. In this case, propranolol might be the preferred agent because can effectively manage the ventricular rate, symptoms of hyperthyroidism (tremor, anxiety, palpitations), and might act to inhibit the monodeiodinase type I enzyme responsible for conversion of T4 to the more potent T3.⁷⁵⁰

11.5.3. Prognosis and follow-up of secondary AF

An increased risk of adverse outcomes in association with new-onset AF secondary to MI, sepsis, or surgery has been reported.⁷⁵¹⁻⁷⁵³ Increased mortality has been observed with recent-onset AF in the presence of ACS, sepsis, and COPD exacerbation.⁷⁵⁴ ACS-associated new-onset AF acts as an independent prognostic indicator for adverse cardiovascular events such as early reinfarction, stroke, and HF. Recent-onset AF in patients with sepsis is associated with increased risks of HF, ischemic stroke, and prolonged length of ICU stay.^{288,752,755,756}

AF in patients with severe sepsis has been associated with increased mortality and in-hospital stroke.^{752,755} New-onset AF in the HF population has been associated with increased mortality.⁷⁵⁷ New-onset AF in the setting of an ACS has been associated with increased mortality, stroke, and reinfarction, resulting in a significantly higher in-hospital mortality compared with those with preexisting AF.^{751,758,759}

Irrespective of the cause of secondary AF, long-term recurrence of AF is frequently observed.^{18,288,756,760} At present, there are limited data to predict which patients will develop recurrent AF beyond an assessment of the traditional risk factors for AF outlined in section 3. As such, it is recommended that patients with secondary AF undergo careful reassessment and follow-up.

RECOMMENDATION

115. We recommend that patients who have experienced secondary AF be followed indefinitely for the possible emergence of recurrent clinical AF, with opportunistic screening for AF conducted at the time of medical encounters (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. Elimination of the trigger is not a guarantee that AF will not recur. Because patients with secondary AF might develop AF later in life it is important to follow them regularly to screen for recurrence (see section 11.5.3 regarding screening).

Practical tip. Patients should be counselled regarding AFassociated symptomatology, when to report for medical evaluation, and about risk factor modification (see section 6).

11.6. AF after cardiac and noncardiac surgery

11.6.1. Incidence of postoperative atrial tachyarrhythmias

AF commonly occurs in the perioperative setting because of the relationship between AF and atrial stretch, atrial ischemia, epicardial inflammation, hypoxia, acidosis, electrolyte disturbances, and high sympathetic tone.⁷⁶¹⁻⁷⁶⁷ The incidence of POAF after cardiac surgery is approximately 30% for isolated CABG, approximately 40% for valve replacement or repair, and approximately 50% for combined CABG/valve procedures.⁷⁶⁸ The peak incidence of POAF is between postoperative days 2 and 4.⁷⁶⁷ Although POAF might be transient and cause little morbidity in many cases, POAF has been independently associated with increased in-hospital duration and health care costs.⁷⁶⁷

11.6.2. Risk factors for postoperative atrial tachyarrhythmias

Independent factors for POAF after cardiac surgery include a preexisting history of AF, older age, male sex, a history of hypertension, the procedure performed, the number of bypass grafts, the duration of surgery, the duration of aortic cross-clamp time, a requirement for an intraoperative balloon pump, a requirement for ventilation > 24 hours, and withdrawal of β -blocker therapy.⁷⁶¹⁻⁷⁶⁷ Of these, age has the highest predictive value.³²

11.6.3. Prevention of postoperative atrial tachyarrhythmias

11.6.3.1. β-Blockers

Current evidence suggests that β -blocker therapy reduces the incidence of POAF in patients who undergo cardiac and noncardiac surgery.^{769,770} For patients who underwent cardiac surgery, use of β -blockers significantly reduced the occurrence of AF (RR, 0.50; 95% CI, 0.42-0.59; 40 studies, 5650 participants) and ventricular arrhythmias (RR, 0.40; 95% CI, 0.25-0.63; 12 studies, 2296 participants) without significant bradycardia, hypotension, or MI.⁷⁷⁰ In patients who underwent noncardiac surgery use of β -blockers significantly reduced the occurrence of AF (RR, 0.41; 95% CI, 0.21-0.79; 9 studies, 9080 participants) and MI (RR, 0.72; 95% CI, 0.60-0.87; 12 studies, 10,520 participants) at the expense of bradycardia (RR, 2.49; 95% CI, 1.74-3.56; 49 studies, 12,239 participants) and hypotension (RR, 1.40; 95% CI, 1.29-1.51; 49 studies, 12,304 participants).⁷⁶⁹ All-cause mortality at 30 days after cardiac or noncardiac surgery was not influenced by β -blocker use.^{769,770}

Relative to cardiac surgery, some trials required preoperative withdrawal of preexisting β -blocker therapy in patients randomized not to receive study β -blocker therapy (β -blocker withdrawal-mandated trials); other trials continued preexisting β -blocker therapy in patients randomized not to receive study β -blocker therapy (β -blocker withdrawal-not mandated trials). In the β -blocker withdrawal-mandated trials, study β -blocker therapy substantially reduced POAF (25 RCTs; 2600 patients; 10.5% vs 28.7%; OR, 0.30; 95% CI, 0.22-0.40; P <0.001).⁷⁷¹ In the β -blocker withdrawal-not mandated trials, study β -blocker therapy reduced POAF to a lesser extent (3 trials; 1163 patients; 31.5% vs 40.2%; OR, 0.69; 95% CI, 0.54-0.87; P = 0.002). This observation might relate to preoperative β -blocker withdrawal increasing POAF in the control groups of the β -blocker withdrawal-mandated trials.

Subgroup analyses did not identify any differences in outcomes according to the β -blocker used, or according to

whether β -blocker therapy was initiated before, during, or immediately after surgery.⁷⁷²

RECOMMENDATION

- 116. We recommend that patients who have been receiving a β -blocker before cardiac or noncardiac surgery have that therapy continued postoperatively in the absence of contraindications (Strong Recommendation; High-Quality Evidence).
- 117. We suggest that patients who have not been receiving a β -blocker before cardiac surgery have β -blocker therapy initiated immediately after the surgical procedure in the absence of a contraindication (Weak Recommendation; Low-Quality Evidence).

Values and preferences. These recommendations place a high value on reducing POAF and a lower value on adverse hemodynamic effects of β -blockade during or after cardiac surgery.

11.6.3.2. Amiodarone

Perioperative amiodarone therapy has been shown to reduce the occurrence of POAF after cardiac surgery compared with placebo or usual care (33 RCTs; 5402 patients; 19.4% vs 33.3%; OR, 0.43; 95% CI, 0.34-0.54; P < 0.001).⁷⁷³ However, a meta-analysis⁷⁷⁴ of head-to-head trials showed no significant difference between amiodarone and standard βblockers for POAF prophylaxis after cardiac surgery, which might reflect the withdrawal of preexisting β-blocker therapy in the amiodarone group, which biases the comparison in favour of β-blocker prophylaxis. Results of 1 small comparative trial suggest that the combination of amiodarone and β-blocker prophylaxis is more effective than β-blockers alone.⁷⁷⁵

In a recent network meta-analysis⁷⁷⁶ the effect of timing and route of amiodarone administration on its efficacy and adverse effects were examined. Regimens that included only oral amiodarone reduced POAF similarly as did regimens that included I.V. administration. Regimens that included a least 1 day of preoperative amiodarone reduced POAF to an equivalent degree as did regimens that started amiodarone the day of or after surgery. This meta-analysis⁷⁷⁶ also showed that bradycardia, hypotension, or QT prolongation were less common with oral-only regimens than with I.V. regimens.

In addition to reducing POAF, amiodarone has been shown to reduce postoperative hospital length of stay by approximately 1 day.^{773,777}

RECOMMENDATION

118. We recommend that patients who have a contraindication to β-blocker therapy before or after cardiac surgery be considered for prophylactic therapy with amiodarone to prevent POAF (Strong Recommendation; High-Quality Evidence).

Values and preferences. This recommendation places a high value on minimizing the patient population exposed to the potential adverse effects of amiodarone and a lower value on data that suggest that amiodarone is more effective than β -blockers for this purpose.
11.6.4. Other apparently effective therapies less commonly used

11.6.4.1. Sotalol

A Cochrane intervention review⁷⁷³ showed that perioperative sotalol therapy reduced POAF after cardiac surgery compared with placebo or usual care (11 trials; 1609 patients; 18.1% vs 40.0%; OR, 0.34; 95% CI, 0.26-0.43; P < 0.001). Another meta-analysis⁷⁷⁸ showed a significantly higher efficacy of sotalol compared with standard β -blocker drugs for prevention of POAF (5 RCTs; 1043 patients; 14.3% vs 22.7%; RR, 0.64; 95% CI, 0.50-0.84; P < 0.001). It also showed similar efficacies when sotalol and amiodarone were compared.⁷⁷⁸ Thus, sotalol prophylaxis appears to be effective for prevention of POAF after cardiac surgery but is not considered a first-line therapy because of its risk of QTinterval prolongation and TdP.

11.6.4.2. I.V. magnesium

A meta-analysis⁷⁷⁹ indicated that postoperative I.V. magnesium supplementation care reduced POAF after cardiac surgery compared with placebo or standard care (21 RCTs; 3248 patients; RR, 0.69; 95% CI, 0.56-0.86; P = 0.006). Although most magnesium trials reported no adverse effects of magnesium therapy, one trial³⁶ showed more postoperative hypotension with a combination of I.V. magnesium and propranolol therapy than with propranolol therapy alone.⁷⁸⁰

11.6.4.3. Overdrive atrial pacing

Trials have shown a significant reduction in POAF after cardiac surgery with overdrive atrial pacing (14 RCTs; 1885 patients; 17.7% vs 35.3%; OR, 0.60; 95% CI, 0.47-0.77; P < 0.001), and in particular with biatrial pacing.⁷⁷¹ However, the **A**trial **Fibrillation Suppression Trial II** (AFIST-II) showed that amiodarone is better at prevention of POAF than overdrive septal atrial pacing.⁷⁸¹

11.6.4.4. Colchicine

Colchicine has been shown to reduce POAF (5 RCTs; 1412 patients; 18.1% vs 26.8%; RR, 0.69; 95% CI, 0.57-0.84; P = 0.0002) and hospital stay after cardiac surgery (3 RCTs; 692 patients; mean difference, -1.2 days; 95% CI, -1.9 to -0.4 days; P = 0.002) compared with placebo or standard care.⁷⁸² However, these potential benefits were counterbalanced by an increased incidence of drug-related adverse effects (4 RCTs; 1196 patients; 21.0% vs 8.2%; RR, 2.52; 95% CI, 1.62-3.93; P < 0.0001), which was dominated by adverse GI side effects.

11.6.4.5. Posterior pericardial drainage

Drainage of the posterior pericardial space at the time of cardiac surgery with a posterior pericardiotomy and/or a posterior pericardial drainage tube reduces early and late pericardial effusion and cardiac tamponade, POAF (17 RCTs; 3245 patients; 12.6% vs 24.8%; OR, 0.42; 95% CI, 0.29-0.59; P < 0.001), hospital length of stay (13 RCTs; 2068 patients; mean difference, -0.82 days; 95% CI, -1.12 to -0.51 days; P = 0.005), and death or cardiac arrest (10 RCTs; 2141 patients;

1.1% vs 2.3%; OR, 0.49; 95% CI, 0.25-0.94; P < 0.001).⁷⁸³ A large RCT, **P**osterior Left Pericardiotomy for Prevention of Atrial Fibrillation After Cardiac Surgery (PALACS; NCT02875405), is under way.⁷⁸⁴

11.6.4.6. HMG-CoA reductase inhibitors

A meta-analysis⁷⁸⁵ of RCTs of an HMG-CoA reductase inhibitor vs placebo or standard care showed that perioperative statin therapy reduces POAF after cardiac surgery (16 RCTs; 3985 patients; 20.3% vs 23.7%; OR, 0.50; 95% CI, 0.34-0.73; P = 0.0004) and shortens hospital stay (12 RCTs; 1185 patients; mean difference, -0.43 days; 95% CI, -0.70to -0.15; P = 0.002). However, no differences were observed with respect to stroke or short-term mortality.⁷⁸⁵

11.6.4.7. Antioxidants

A meta-analysis⁷⁸⁶ of RCTs of vitamin C treatment vs placebo or standard care for prevention of POAF after cardiac surgery showed that vitamin C reduces POAF (10 RCTs; 1956 patients; 26.2% vs 36.4%; RR, 0.68; 95% CI, 0.54-0.87; P = 0.002) and reduces hospital length of stay (9 RCTs; 1158 patients; mean difference, -0.95 days; 95% CI, -1.64 to -0.26 days; P = 0.007).

A meta-analysis⁷⁸⁷ of RCTs of N-acetylcysteine vs placebo or standard care for prevention of POAF showed that Nacetylcysteine reduces POAF (10 RCTs; 1026 patients; 20.5% vs 28.8%; OR, 0.56; 95% CI, 0.40-0.77; P = 0.0005) without a reduction in hospital length of stay but with a possible reduction in postoperative mortality.

11.6.4.8. Other potential therapies

Class Ia (disopyramide, procainamide, quinidine) and class Ic (propafenone, flecainide) antiarrhythmic agents might prevent POAF^{788,789} but are generally avoided in cardiac surgical patients because of concerns of proarrhythmia. Small RCTs have supported the use of dofetilide (1 RCT; 133 patients; 17.9% vs 36.4%; OR, 0.32; 95% CI, 0.16-0.83; P = 0.017) or ranolazine (1 RCT; 102 patients; 8.8% vs 30.8%; RR, 0.29; 95% CI, 0.09-0.89; P < 0.001).⁷⁹⁰

Perioperative steroid use, renin-angiotensin system inhibitors, digoxin prophylaxis, calcium antagonist prophylaxis, n-3 polyunsaturated fatty acids, glucose-insulinpotassium therapy, vasopressin, and NSAIDs have been shown to lack promise.^{771,791-796} Intraoperative injection of botulinum toxin into epicardial neural ganglia has been shown to be promising in small RCTs with a larger study in progress.⁷⁹⁷

RECOMMENDATION

119. We suggest that patients who have a contraindication to β -blocker therapy and to amiodarone be considered for prophylactic therapy to prevent POAF with I.V. magnesium, biatrial pacing, colchicine, and/or posterior pericardiotomy (Weak Recommendation; Low-Quality Evidence). Values and preferences. This recommendation places a high value on preventing POAF using novel therapies that are supported by lower-quality data; with a higher value on the lower probability of adverse effects from magnesium vs colchicine. The use of atrial pacing needs to be individualized according to the patient and institution, because the potential for adverse effects might outweigh benefit on the basis of local expertise.

11.6.5. Treatment of postoperative atrial tachyarrhythmias

The management of POAF centres on the identification of potential contributors (pulmonary thromboembolism, volume overload, congestive HF, sepsis), prevention of thromboembolic events, slowing of the ventricular response rate, and consideration of sinus rhythm restoration.

The natural history of POAF is dominated by self-terminating but frequently recurrent AF episodes, and resolution in 6-12 weeks regardless of the therapy used.⁷⁹⁸⁻⁸⁰⁰ Furthermore, the hyperadrenergic state lessens the effective-ness of therapies that do not include β -blockade.

There is an association between POAF, thromboembolic cerebrovascular events, 766,767,801,802 and cognitive impairment. ⁸⁰³ Nevertheless, early anticoagulation might predispose the post-operative cardiac surgery patient to delayed pericardial bleeding and tamponade. ⁸⁰⁴ In recognition of this risk and in the absence of controlled trials, the initiation of anticoagulation is generally not recommended in the first 72 hours after cardiac surgery. When initiated, anticoagulation is usually continued for > 6 weeks.

Therapy for ventricular rate control is usually required for patients who experience POAF. Nevertheless, cardiac surgery might also predispose patients to bradyarrhythmia after conversion of AF. Accordingly, the availability of back-up ventricular pacing is important. Because the postsurgical state includes adrenergic discharge, β -blocker therapy is often effective. Alternatives to β -blocker therapy include ND-CCBs or amiodarone because digoxin is usually ineffective.

The advantages and disadvantages of sinus rhythm restoration are similar to those in other settings. In the POAF population specifically, a recent RCT failed to show a significant difference between a rate control vs a rhythm control approach in the outcomes of total number of days in hospital, length of the index hospitalization, readmission rate, time to sustained sinus rhythm, or adverse events including the need for pacemaker.⁸⁰⁵ Accordingly, either the rate control or the rhythm control approach to the treatment of POAF seems appropriate. If rhythm control is pursued, antiarrhythmic drug therapy is preferred to isolated DCCV because early recurrence of POAF is common. Regardless of the approach chosen, therapy provided for POAF can usually be withdrawn 6-12 weeks later.

RECOMMENDATION

120. We recommend that AF after cardiac or noncardiac surgery may be appropriately treated with a rate control strategy or a rhythm control strategy (Strong Recommendation; Moderate-Quality Evidence). Values and preferences. This recommendation places a high value on the randomized controlled trials in which rate control as an alternative to rhythm control for AF has been investigated, including 2 trials with a specific focus on the cardiac postoperative period. Choice of strategy should therefore be individualized on the basis of the degree of symptoms experienced by the patient.

Practical tip. The management of POAF should follow the principles outlined in sections 9.1-9.3.

121. We suggest that consideration be given to withhold anticoagulation therapy for the first 72 hours after cardiac surgery, on the basis of an individualized assessment of the risks of a thromboembolic event and the risk of postoperative bleeding (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation recognizes that risk of postoperative bleeding decreases with time. The benefit-to-risk ratio favours a longer period without anticoagulation in the postoperative setting than that suggested in other settings.

122. We recommend that, when anticoagulation therapy, rate control therapy, and/or rhythm control therapy has been prescribed for POAF, formal reconsideration of the ongoing need for such therapy should be undertaken 6-12 weeks later (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation reflects the high probability that POAF will be a self-limiting process that does not require long-term therapy. At present, the possibility that long-term anticoagulation therapy might prevent the suggested increased risk of stroke in patients with transient POAF with a favourable riskbenefit ratio is insufficient to recommend long-term anticoagulation therapy in these patients.

11.6.6. Prognosis of postoperative atrial tachyarrhythmias

Because patients with POAF are older and have more comorbidities than patients without POAF, the challenge has been to determine if POAF is an independent predictor of long-term adverse events. In one of the first studies⁸⁰⁶ to address these issues, the outcomes of consecutive survivors of CABG surgery from a single centre in Sweden who did not have a preoperative AF history or a pacemaker were retrospectively analyzed. Of these 571 patients, 165 (28.9%) experienced POAF. After a median follow-up of more than 6 years, patients with POAF had a higher probability of recurrent AF (25.4% vs 3.6%; adjusted OR, 8.31; 95% CI, 4.20-16.43; P < 0.001), a higher probability of late all-cause mortality (29.7% vs 14.8%; adjusted HR, 1.57; 95% CI, 1.05-2.34; P = 0.027), and a higher probability of death from cerebral ischemia (4.2% vs 0.2%; P < 0.001) compared with patients without POAF. A very recent analysis⁸⁰⁷ took this observational comparison further with a larger sample size, prospectively collected data, longer follow-up, and a matched cohort evaluation. This report considered 7145 consecutive

survivors of CABG surgery who did not have a preoperative AF history or a pacemaker from a single centre in Sweden. Of these 7145 patients, 2183 (30.6%) experienced POAF. Patients in the POAF group and in the non-POAF group were matched (on age, sex, and county) to 5 control patients who did not undergo cardiac surgery. The cumulative incidence of late AF (at 10 years of follow-up) was higher in patients with POAF compared with those without POAF (15.3% vs 4.9%; adjusted HR, 3.20; 95% CI, 2.73-3.76; P < 0.001) or matched, nonsurgical controls (15.3% vs 6.4%; P < 0.001). There was no difference in the incidence of late AF in patients without POAF compared with their matched, nonsurgical control participants (4.9% vs 5.7%; P = 0.043). Patients with POAF had higher probabilities of late ischemic stroke, HF, cardiac mortality, cerebrovascular mortality, and all-cause mortality, compared with patients without POAF. Most other analyses of this sort have shown similar findings.753,808,809 Accordingly, POAF after cardiac surgery is associated with long-term risk of future AF (15%-25% over 10 years, which is higher than in patients without POAF and higher than in the general population). As such, this patient group should be followed more closely for the ultimate development of AF, particularly because of the observation that approximately half of the AF episodes in this setting are asymptomatic.⁸¹⁰

RECOMMENDATION

123. We recommend that patients who have experienced AF after cardiac or noncardiac surgery be followed indefinitely for the possible emergence of recurrent clinical AF (Strong Recommendation; Moderate-Quality Evidence).

Practical tip. Elimination of the trigger (eg, resolution of postsurgical change) is not a guarantee that AF will not recur. Because patients with secondary AF can often go on to develop AF later in life it is important to follow them regularly to screen for recurrence (see section 11.5.3).

Practical tip. Patients should be counselled regarding AFassociated symptomatology, when to report for medical evaluation, and about risk factor modification (see section 6).

11.7. AF and supraventricular arrhythmia

AF might occur in conjunction with SVT. Previous cohort studies have observed AF in association with SVT in almost 20% of patients, with a gradient dependent on the underlying arrhythmia (58% with AFL, 27% with atrial tachycardia, 14% with AV reentrant tachycardia [AVRT], and 10% with AV nodal reentrant tachycardia [AVNRT]).⁸¹¹ Conversely, studies of AF patients who underwent electrophysiology studies before catheter ablation of AF have shown inducible SVT, in 4.3%-10.0%, including AFL (3.7%), typical AVNRT (1.7%), AVRT (1.2%), and atrial tachycardia (1%).⁸¹²⁻⁸¹⁴ In those with inducible SVT, there was a transition from SVT into AF during electrophysiology study in almost one-half of the patients.⁸¹⁴

AF patients with inducible SVT tend to be younger, more likely to have paroxysmal AF, less likely to have structural

heart disease, and are more likely to be free of AF after a single ablation procedure (88% were AF-free after SVT ablation alone or SVT and AF ablation).^{812,814} Enriched populations of younger AF patients, those free of structural heart disease, and those with a history of regular palpitations or narrow QRS tachycardia have been shown to have much higher rates of inducible SVT, with one series documenting AVNRT in 18% and AVRT in 9% of AF patients younger than 65 years,⁸¹⁵ and another series inducing SVT in up to 57% of patients younger than 35 years of age without medical comorbidities.⁸¹⁶

11.7.1. Role of the electrophysiology study

Because of the variable rates of inducible SVT in patients with AF, an electrophysiology study should be primarily considered in AF patients who are younger, without medical comorbidities, have a history of regular palpitations or narrow complex tachycardia, and/or are planned to undergo catheter ablation of AF.

It is reasonable to perform catheter ablation of the SVT alone. In younger patients, successful ablation of SVT can treat the underlying cause and might be associated with lower rates of AF recurrence.^{812,814} Older patients generally have higher rates of AF occurring after SVT ablation (up to 20%-30% after several years) and may benefit from combined SVT and AF ablation.^{69,817,818}

RECOMMENDATION

124. We suggest an electrophysiology study to exclude reentrant tachycardia as a cause of AF be performed in younger patients without medical comorbidities or those with a history suggestive of concomitant SVT (eg, regular palpitations). If present, we suggest ablation of the tachycardia (Weak Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation recognizes that SVT can initiate AF when the substrate for AF is present and can be ablated with a high success rate and minimal risk.

11.8. Wolff-Parkinson-White syndrome

11.8.1. Acute management of preexcited AF

AF can occur in up to 50% of young patients with Wolf-Parkinson-White (WPW) syndrome.⁸¹⁹ Rapid conduction of AF from the atrium to the ventricle via the accessory pathway is rare but can result in ventricular fibrillation and sudden cardiac death.^{820,821} In patients who present with preexcited AF, we recommend urgent synchronized cardioversion in the presence of hemodynamic instability^{822,823} or when pharma-cological cardioversion of hemodynamically stable preexcited AF with I.V. ibutilide or procainamide is ineffective.^{824,825} The use of I.V. β -blockers, diltiazem, verapamil, digoxin, and amiodarone are not recommended for acute management of preexcited AF because of the potential to precipitate a

life-threatening arrhythmia.⁸²⁶⁻⁸²⁹ Conduction over the accessory pathway is enhanced with use of drugs that block the AV node (β -blockers, diltiazem, verapamil, or amiodarone) and they might cause hypotension and a catecholamine release, resulting in an increase in the ventricular rate. Digoxin shortens the refractory period of the accessory pathway, further enhancing antegrade conduction of AF.

RECOMMENDATION

- 125. We recommend electrical cardioversion to restore sinus rhythm in hemodynamically unstable patients with evidence of ventricular preexcitation during AF (Strong Recommendation; Low-Quality Evidence).
- 126. We recommend electrical or pharmacologic cardioversion using I.V. procainamide or ibutilide to restore sinus rhythm in hemodynamically stable patients with evidence of ventricular preexcitation during AF (Strong Recommendation; Low-Quality Evidence).

Practical tip. Electrical cardioversion might be preferred on the basis of the relatively limited evidence base, as well as the small size of the studies of pharma-cologic cardioversion for preexcited AF.

127. We recommend that AV nodal-blocking agents, digitalis, and amiodarone should be avoided in patients with evidence of ventricular preexcitation (Strong Recommendation; Low-Quality Evidence).

11.8.2. Chronic management of preexcited AF

The treatment of choice for patients with preexcited AF is catheter ablation. Ablation of an accessory pathway has a very high acute success rate (> 90%) with a low rate of complications (overall 3%; death 0.1%-0.3%; stroke 0.1%-0.2%; tamponade 0.4%-0.6%; AV block requiring pacemaker 0.3%-1.0%).^{649,830} Certain high-risk features that increase the risk of ventricular fibrillation include: an accessory pathway with a short refractory period, arrhythmia-associated syncope, the presence of multiple pathways, and orthodromic AV reciprocating tachycardia that degenerates into preexcited AF. When present, these factors should prompt urgent catheter ablation.^{821,831-833}

RECOMMENDATION

128. In patients with evidence of ventricular preexcitation and AF, we recommend catheter ablation of the accessory pathway (Strong Recommendation; Low-Quality Evidence).

Values and preferences. This recommendation places a high value on the high efficacy of catheter ablation of the accessory pathway and the prevention of sudden cardiac death in patients at high risk, and a low value on the small risk of complications. **Practical tip.** Urgent or in-patient ablation should be strongly considered for preexcited AF with high-risk features. High-risk features include AF associated with syncope, rapid ventricular rates (eg, short accessory pathway refractory period), the presence of multiple accessory pathways, or regular narrow complex tachycardia that degenerated into preexcited AF.

11.9. Screening for AF in patients with a history of atrial dysrhythmia

Patients with a history of atrial dysrhythmia (AFL, atrial tachycardia, AVRT, and AVNRT) should be followed for the development of AF.

RECOMMENDATION

129. We recommend that patients who have undergone ablation of an atrial dysrhythmia (atrial flutter, atrial tachycardia, AVRT, AVNRT) should be followed for the occurrence of AF with opportunistic screening conducted at the time of medical encounters (Strong Recommendation; Moderate-Quality Evidence).

Values and preferences. This recommendation places a relatively high value on the evidence supporting the observation that AF might occur after ablation of AFL, SVT, or WPW, particularly in those with increasing age and a history of AF. This guideline recognizes that the risk of AF recurrence is relatively higher with a history of AFL, intermediate for WPW, and lowest for a history of other SVTs.

Practical tip. In those with SVT but no previous AF history, it remains unclear if SVT alone is enough to increase the risk of AF in the absence of significant risk factors.

Practical tip. Patients should be counselled regarding AFassociated symptomatology, when to report for medical evaluation, and about risk factor modification (see section 6).

Acknowledgements

The authors thank Ms Christianna Brooks (CCS) for her assistance and outstanding contribution throughout the guideline writing process. We are indebted to Marc Bains (patient partner, HeartLife Foundation), Andrew Campbell (Emergency Medicine), James Douketis (Thrombosis Canada), Jasmine Grewal (Adult Congenital Cardiology), Saurabh Gupta (Cardiac Surgery), Thalia Field (Stroke Neurology), Jenny MacGillivray (Pharmacist), Arianne Marelli (Adult Congenital Cardiology), Michael McDonald (Canadian Heart Failure Society), Candice Silversides (Adult Congenital Cardiology), and David Wood (Canadian Association of Interventional Cardiology) and the CCS Guidelines Committee for their external review of this document.

References

- 1. Canadian Cardiovascular Society Consensus Conference on Atrial Fibrillation. Can J Cardiol 1996;12(suppl A):1A-61A.
- Andrade JG, Nattel S, Macle L. The Canadian Cardiovascular Society Atrial Fibrillation Guidelines Program: a look back over the last 10 years and a look forward. Can J Cardiol 2020;36:1839-42.

- **3.** 2004 Canadian Cardiovascular Society Consensus Conference: Atrial Fibrillation. Can J Cardiol 2005;21(suppl B):9B-73B.
- Cairns JA, Connolly S, McMurtry S, et al. Canadian Cardiovascular Society atrial fibrillation guidelines 2010: prevention of stroke and systemic thromboembolism in atrial fibrillation and flutter. Can J Cardiol 2011;27:74-90.
- Skanes AC, Healey JS, Cairns JA, et al. Focused 2012 update of the Canadian Cardiovascular Society atrial fibrillation guidelines: recommendations for stroke prevention and rate/rhythm control. Can J Cardiol 2012;28:125-36.
- Verma A, Cairns JA, Mitchell LB, et al. 2014 focused update of the Canadian Cardiovascular Society Guidelines for the management of atrial fibrillation. Can J Cardiol 2014;30:1114-30.
- Macle L, Cairns J, Leblanc K, et al. 2016 Focused update of the Canadian Cardiovascular Society guidelines for the management of atrial fibrillation. Can J Cardiol 2016;32:1170-85.
- Andrade JG, Verma A, Mitchell LB, et al. 2018 Focused update of the Canadian Cardiovascular Society guidelines for the management of atrial fibrillation. Can J Cardiol 2018;34:1371-92.
- Guyatt GH, Oxman AD, Vist GE, et al. GRADE: an emerging consensus on rating quality of evidence and strength of recommendations. BMJ 2008;336:924-6.
- Andrade JG, Macle L, Nattel S, Verma A, Cairns J. Contemporary atrial fibrillation management: a comparison of the current AHA/ACC/HRS, CCS, and ESC guidelines. Can J Cardiol 2017;33:965-76.
- Charitos EI, Ziegler PD, Stierle U, et al. Atrial fibrillation burden estimates derived from intermittent rhythm monitoring are unreliable estimates of the true atrial fibrillation burden. Pacing Clin Electrophysiol 2014;37:1210-8.
- Charitos EI, Purerfellner H, Glotzer TV, Ziegler PD. Clinical classifications of atrial fibrillation poorly reflect its temporal persistence: insights from 1,195 patients continuously monitored with implantable devices. J Am Coll Cardiol 2014;63:2840-8.
- Andrade JG, Yao RRJ, Deyell MW, et al. Clinical assessment of AF pattern is poorly correlated with AF burden and post ablation outcomes: a CIRCA-DOSE sub-study. J Electrocardiol 2020;60:159-64.
- Banerjee A, Taillandier S, Olesen JB, et al. Pattern of atrial fibrillation and risk of outcomes: the Loire Valley Atrial Fibrillation Project. Int J Cardiol 2013;167:2682-7.
- Vanassche T, Lauw MN, Eikelboom JW, et al. Risk of ischaemic stroke according to pattern of atrial fibrillation: analysis of 6563 aspirin-treated patients in ACTIVE-A and AVERROES. Eur Heart J 2015;36. 281-7a.
- Atar D, Berge E, Le Heuzey JY, et al. The association between patterns of atrial fibrillation, anticoagulation, and cardiovascular events. Europace 2020;22:195-204.
- Andrade JG, Deyell MW, Verma A, et al. Association of atrial fibrillation episode duration with arrhythmia recurrence following ablation: a secondary analysis of a randomized clinical trial. JAMA Netw Open 2020;3:e208748.
- Quon MJ, Pilote L. Is secondary atrial fibrillation different? Or Is atrial fibrillation just atrial fibrillation? Canadian Journal of General Internal Medicine 2018;13:27-30.
- Wyse DG, Van Gelder IC, Ellinor PT, et al. Lone atrial fibrillation: does it exist? J Am Coll Cardiol 2014;63:1715-23.
- Cheung CC, Andrade JG. Reversible or provoked atrial fibrillation?: The devil in the details. JACC Clin Electrophysiol 2018;4:563-4.

- Connolly SJ, Ezekowitz MD, Yusuf S, et al. Dabigatran versus warfarin in patients with atrial fibrillation. N Engl J Med 2009;361:1139-51.
- 22. Patel MR, Mahaffey KW, Garg J, et al. Rivaroxaban versus warfarin in nonvalvular atrial fibrillation. N Engl J Med 2011;365:883-91.
- Granger CB, Alexander JH, McMurray JJ, et al. Apixaban versus warfarin in patients with atrial fibrillation. N Engl J Med 2011;365: 981-92.
- 24. Connolly SJ, Eikelboom J, Joyner C, et al. Apixaban in patients with atrial fibrillation. N Engl J Med 2011;364:806-17.
- Giugliano RP, Ruff CT, Braunwald E, et al. Edoxaban versus warfarin in patients with atrial fibrillation. N Engl J Med 2013;369:2093-104.
- Macle L, Cairns JA, Andrade JG, et al. The 2014 atrial fibrillation guidelines companion: a practical approach to the use of the Canadian Cardiovascular Society guidelines. Can J Cardiol 2015;31:1207-18.
- Noseworthy PA, Yao X, Shah ND, Gersh BJ. Comparative effectiveness and safety of non-vitamin K antagonist oral anticoagulants versus warfarin in patients with atrial fibrillation and valvular heart disease. Int J Cardiol 2016;209:181-3.
- Siontis KC, Yao X, Gersh BJ, Noseworthy PA. Direct oral anticoagulants in patients with atrial fibrillation and valvular heart disease other than significant mitral stenosis and mechanical valves: a meta-analysis. Circulation 2017;135:714-6.
- Andrade J, Khairy P, Dobrev D, Nattel S. The clinical profile and pathophysiology of atrial fibrillation: relationships among clinical features, epidemiology, and mechanisms. Circ Res 2014;114:1453-68.
- Kannel WB, Wolf PA, Benjamin EJ, Levy D. Prevalence, incidence, prognosis, and predisposing conditions for atrial fibrillation: populationbased estimates. Am J Cardiol 1998;82:2N-9N.
- Piccini JP, Hammill BG, Sinner MF, et al. Incidence and prevalence of atrial fibrillation and associated mortality among Medicare beneficiaries, 1993-2007. Circ Cardiovasc Qual Outcomes 2012;5:85-93.
- Kannel WB, Abbott RD, Savage DD, McNamara PM. Epidemiologic features of chronic atrial fibrillation: the Framingham study. N Engl J Med 1982;306:1018-22.
- Furberg CD, Psaty BM, Manolio TA, et al. Prevalence of atrial fibrillation in elderly subjects (the Cardiovascular Health Study). Am J Cardiol 1994;74:236-41.
- 34. Rietbrock S, Heeley E, Plumb J, van Staa T. Chronic atrial fibrillation: incidence, prevalence, and prediction of stroke using the Congestive heart failure, Hypertension, Age >75, Diabetes mellitus, and prior Stroke or transient ischemic attack (CHADS2) risk stratification scheme. Am Heart J 2008;156:57-64.
- 35. Chiang CE, Naditch-Brule L, Murin J, et al. Distribution and risk profile of paroxysmal, persistent, and permanent atrial fibrillation in routine clinical practice: insight from the real-life global survey evaluating patients with atrial fibrillation international registry. Circ Arrhythm Electrophysiol 2012;5:632-9.
- **36.** Go AS, Hylek EM, Phillips KA, et al. Prevalence of diagnosed atrial fibrillation in adults: national implications for rhythm management and stroke prevention: the AnTicoagulation and Risk Factors in Atrial Fibrillation (ATRIA) Study. JAMA 2001;285:2370-5.
- 37. Miyasaka Y, Barnes ME, Gersh BJ, et al. Secular trends in incidence of atrial fibrillation in Olmsted County, Minnesota, 1980 to 2000, and implications on the projections for future prevalence. Circulation 2006;114:119-25.

- Dorian P, Jung W, Newman D, et al. The impairment of health-related quality of life in patients with intermittent atrial fibrillation: implications for the assessment of investigational therapy. J Am Coll Cardiol 2000;36:1303-9.
- **39**. Savelieva I, Paquette M, Dorian P, Luderitz B, Camm AJ. Quality of life in patients with silent atrial fibrillation. Heart 2001;85:216-7.
- 40. Singh SN, Tang XC, Singh BN, et al. Quality of life and exercise performance in patients in sinus rhythm versus persistent atrial fibrillation: a Veterans Affairs Cooperative Studies Program substudy. J Am Coll Cardiol 2006;48:721-30.
- Thrall G, Lane D, Carroll D, Lip GY. Quality of life in patients with atrial fibrillation: a systematic review. Am J Med 2006;119. 448.e1-19.
- 42. Kang Y. Relation of atrial arrhythmia-related symptoms to health-related quality of life in patients with newly diagnosed atrial fibrillation: a community hospital-based cohort. Heart Lung 2006;35:170-7.
- Wolf PA, Mitchell JB, Baker CS, Kannel WB, D'Agostino RB. Impact of atrial fibrillation on mortality, stroke, and medical costs. Arch Intern Med 1998;158:229-34.
- 44. Andersson T, Magnuson A, Bryngelsson IL, et al. All-cause mortality in 272,186 patients hospitalized with incident atrial fibrillation 1995-2008: a Swedish nationwide long-term case-control study. Eur Heart J 2013;34:1061-7.
- 45. Benjamin EJ, Virani SS, Callaway CW, et al. Heart disease and stroke statistics-2018 update: a report from the American Heart Association. Circulation 2018;137:e67-492.
- 46. Lee E, Choi EK, Han KD, et al. Mortality and causes of death in patients with atrial fibrillation: a nationwide population-based study. PLoS One 2018;13:e0209687.
- Petty GW, Brown RD Jr, Whisnant JP, et al. Ischemic stroke subtypes: a population-based study of functional outcome, survival, and recurrence. Stroke 2000;31:1062-8.
- Bruggenjurgen B, Rossnagel K, Roll S, et al. The impact of atrial fibrillation on the cost of stroke: the Berlin Acute Stroke study. Value Health 2007;10:137-43.
- Gladstone DJ, Bui E, Fang J, et al. Potentially preventable strokes in high-risk patients with atrial fibrillation who are not adequately anticoagulated. Stroke 2009;40:235-40.
- 50. Healey JS, Connolly SJ, Gold MR, et al. Subclinical atrial fibrillation and the risk of stroke. N Engl J Med 2012;366:120-9.
- Hart RG, Pearce LA, Aguilar MI. Meta-analysis: antithrombotic therapy to prevent stroke in patients who have nonvalvular atrial fibrillation. Ann Intern Med 2007;146:857-67.
- Ruff CT, Giugliano RP, Braunwald E, et al. Comparison of the efficacy and safety of new oral anticoagulants with warfarin in patients with atrial fibrillation: a meta-analysis of randomised trials. Lancet 2014;383: 955-62.
- Stewart S, Murphy NF, Walker A, McGuire A, McMurray JJ. Cost of an emerging epidemic: an economic analysis of atrial fibrillation in the UK. Heart 2004;90:286-92.
- Humphries KH, Jackevicius C, Gong Y, et al. Population rates of hospitalization for atrial fibrillation/flutter in Canada. Can J Cardiol 2004;20:869-76.
- Wu EQ, Birnbaum HG, Mareva M, et al. Economic burden and comorbidities of atrial fibrillation in a privately insured population. Curr Med Res Opin 2005;21:1693-9.

- 56. Lee WC, Lamas GA, Balu S, et al. Direct treatment cost of atrial fibrillation in the elderly American population: a Medicare perspective. J Med Econ 2008;11:281-98.
- Kim MH, Lin J, Hussein M, Kreilick C, Battleman D. Cost of atrial fibrillation in United States managed care organizations. Adv Ther 2009;26:847-57.
- 58. O'Reilly DJ, Hopkins RB, Healey JS, et al. The burden of atrial fibrillation on the hospital sector in Canada. Can J Cardiol 2013;29: 229-35.
- Meyre P, Blum S, Berger S, et al. Risk of hospital admissions in patients with atrial fibrillation: a systematic review and meta-analysis. Can J Cardiol 2019;35:1332-43.
- Nattel S. New ideas about atrial fibrillation 50 years on. Nature 2002;415:219-26.
- Voigt N, Heijman J, Wang Q, et al. Cellular and molecular mechanisms of atrial arrhythmogenesis in patients with paroxysmal atrial fibrillation. Circulation 2014;129:145-56.
- 62. Voigt N, Li N, Wang Q, et al. Enhanced sarcoplasmic reticulum Ca2+ leak and increased Na+-Ca2+ exchanger function underlie delayed afterdepolarizations in patients with chronic atrial fibrillation. Circulation 2012;125:2059-70.
- 63. Yao C, Veleva T, Scott L Jr, et al. Enhanced cardiomyocyte NLRP3 inflammasome signaling promotes atrial fibrillation. Circulation 2018;138:2227-42.
- Nishida K, Datino T, Macle L, Nattel S. Atrial fibrillation ablation: translating basic mechanistic insights to the patient. J Am Coll Cardiol 2014;64:823-31.
- 65. Pathak RK, Middeldorp ME, Lau DH, et al. Aggressive risk factor reduction study for atrial fibrillation and implications for the outcome of ablation: the ARREST-AF cohort study. J Am Coll Cardiol 2014;64: 2222-31.
- Lau DH, Nattel S, Kalman JM, Sanders P. Modifiable risk factors and atrial fibrillation. Circulation 2017;136:583-96.
- Dorian P, Guerra PG, Kerr CR, et al. Validation of a new simple scale to measure symptoms in atrial fibrillation: the Canadian Cardiovascular Society Severity in Atrial Fibrillation scale. Circ Arrhythm Electrophysiol 2009;2:218-24.
- Haissaguerre M, Fischer B, Labbe T, et al. Frequency of recurrent atrial fibrillation after catheter ablation of overt accessory pathways. Am J Cardiol 1992;69:493-7.
- 69. Brembilla-Perrot B, Olivier A, Villemin T, et al. Prediction of atrial fibrillation in patients with supraventricular tachyarrhythmias treated with catheter ablation or not. Classical scores are not useful. Int J Cardiol 2016;220:102-6.
- Wolf PA, Dawber TR, Thomas HE Jr, Kannel WB. Epidemiologic assessment of chronic atrial fibrillation and risk of stroke: the Framingham study. Neurology 1978;28:973-7.
- Kim DD, Young S, Cutfield R. A survey of thyroid function test abnormalities in patients presenting with atrial fibrillation and flutter to a New Zealand district hospital. N Z Med J 2008;121:82-6.
- 72. Gladstone DJ, Spring M, Dorian P, et al. Atrial fibrillation in patients with cryptogenic stroke. N Engl J Med 2014;370:2467-77.
- 73. Sanna T, Diener HC, Passman RS, et al. Cryptogenic stroke and underlying atrial fibrillation. N Engl J Med 2014;370:2478-86.

- 74. Hindricks G, Piorkowski C, Tanner H, et al. Perception of atrial fibrillation before and after radiofrequency catheter ablation: relevance of asymptomatic arrhythmia recurrence. Circulation 2005;112:307-13.
- **75.** Verma A, Champagne J, Sapp J, et al. Discerning the incidence of symptomatic and asymptomatic episodes of atrial fibrillation before and after catheter ablation (DISCERN AF): a prospective, multicenter study. JAMA Intern Med 2013;173:149-56.
- 76. Walters TE, Wick K, Tan G, et al. Psychological distress and suicidal ideation in patients with atrial fibrillation: prevalence and response to management strategy. J Am Heart Assoc 2018;7:e005502.
- 77. Dorian P, Cvitkovic SS, Kerr CR, et al. A novel, simple scale for assessing the symptom severity of atrial fibrillation at the bedside: the CCS-SAF scale. Can J Cardiol 2006;22:383-6.
- Reynolds MR, Ellis E, Zimetbaum P. Quality of life in atrial fibrillation: measurement tools and impact of interventions. J Cardiovasc Electrophysiol 2008;19:762-8.
- 79. Miyazaki M, Nakashima A, Nakamura Y, et al. Association between medication adherence and illness perceptions in atrial fibrillation patients treated with direct oral anticoagulants: an observational crosssectional pilot study. PLoS One 2018;13:e0204814.
- Miura K, Ikemura N, Kimura T, et al. Treatment strategies and subsequent changes in the patient-reported quality-of-life among elderly patients with atrial fibrillation. Am Heart J 2020;222:83-92.
- 81. Mairesse GH, Moran P, Van Gelder IC, et al. Screening for atrial fibrillation: a European Heart Rhythm Association (EHRA) consensus document endorsed by the Heart Rhythm Society (HRS), Asia Pacific Heart Rhythm Society (APHRS), and Sociedad Latinoamericana de Estimulacion Cardiaca y Electrofisiologia (SOLAECE). Europace 2017;19:1589-623.
- 82. Lowres N, Olivier J, Chao TF, et al. Estimated stroke risk, yield, and number needed to screen for atrial fibrillation detected through single time screening: a multicountry patient-level meta-analysis of 141,220 screened individuals. PLoS Med 2019;16:e1002903.
- Morgan S, Mant D. Randomised trial of two approaches to screening for atrial fibrillation in UK general practice. Br J Gen Pract 2002;52(373-4):377-80.
- 84. Hobbs FD, Fitzmaurice DA, Mant J, et al. A randomised controlled trial and cost-effectiveness study of systematic screening (targeted and total population screening) versus routine practice for the detection of atrial fibrillation in people aged 65 and over. The SAFE study. Health Technol Assess 2005;9. iii-iv, ix-x, 1-74.
- Svennberg E, Engdahl J, Al-Khalili F, et al. Mass screening for untreated atrial fibrillation: the STROKESTOP study. Circulation 2015;131: 2176-84.
- 86. Kaasenbrood F, Hollander M, Rutten FH, et al. Yield of screening for atrial fibrillation in primary care with a hand-held, single-lead electrocardiogram device during influenza vaccination. Europace 2016;18: 1514-20.
- Lowres N, Neubeck L, Salkeld G, et al. Feasibility and cost-effectiveness of stroke prevention through community screening for atrial fibrillation using iPhone ECG in pharmacies. The SEARCH-AF study. Thromb Haemost 2014;111:1167-76.
- Sandhu RK, Dolovich L, Deif B, et al. High prevalence of modifiable stroke risk factors identified in a pharmacy-based screening programme. Open Heart 2016;3:e000515.
- Freedman B, Camm J, Calkins H, et al. Screening for atrial fibrillation: a report of the AF-SCREEN International Collaboration. Circulation 2017;135:1851-67.

- Moran PS, Teljeur C, Ryan M, Smith SM. Systematic screening for the detection of atrial fibrillation. Cochrane Database Syst Rev 2016;6: CD009586.
- Tarride JE, Dolovich L, Blackhouse G, et al. Screening for atrial fibrillation in Canadian pharmacies: an economic evaluation. CMAJ Open 2017;5:E653-61.
- **92.** Tarride JE, Quinn FR, Blackhouse G, et al. Is screening for atrial fibrillation in Canadian family practices cost-effective in patients 65 years and older? Can J Cardiol 2018;34:1522-5.
- Andrade JG, Godin R, Nault I. Large-scale implementation of a pragmatic atrial fibrillation screening program in Canadian community practice. Pacing Clin Electrophysiol 2020;43:768-9.
- 94. Perez MV, Mahaffey KW, Hedlin H, et al. Large-scale assessment of a Smartwatch to identify atrial fibrillation. N Engl J Med 2019;381: 1909-17.
- **95.** Daoud EG, Glotzer TV, Wyse DG, et al. Temporal relationship of atrial tachyarrhythmias, cerebrovascular events, and systemic emboli on the basis of stored device data: a subgroup analysis of TRENDS. Heart Rhythm 2011;8:1416-23.
- 96. Van Gelder IC, Healey JS, Crijns H, et al. Duration of device-detected subclinical atrial fibrillation and occurrence of stroke in ASSERT. Eur Heart J 2017;38:1339-44.
- 97. Swiryn S, Orlov MV, Benditt DG, et al. Clinical implications of brief device-detected atrial tachyarrhythmias in a cardiac rhythm management device population: results from the Registry of Atrial Tachycardia and Atrial Fibrillation Episodes. Circulation 2016;134:1130-40.
- 98. Gorenek BC, Bax J, Boriani G, et al. Device-detected subclinical atrial tachyarrhythmias: definition, implications and management-an European Heart Rhythm Association (EHRA) consensus document, endorsed by Heart Rhythm Society (HRS), Asia Pacific Heart Rhythm Society (APHRS) and Sociedad Latinoamericana de Estimulacion Cardiaca y Electrofisiologia (SOLEACE). Europace 2017;19:1556-78.
- 99. Glotzer TV, Hellkamp AS, Zimmerman J, et al. Atrial high rate episodes detected by pacemaker diagnostics predict death and stroke: report of the Atrial Diagnostics Ancillary Study of the MOde Selection Trial (MOST). Circulation 2003;107:1614-9.
- 100. Glotzer TV, Daoud EG, Wyse DG, et al. The relationship between daily atrial tachyarrhythmia burden from implantable device diagnostics and stroke risk: the TRENDS study. Circ Arrhythm Electrophysiol 2009;2:474-80.
- 101. Capucci A, Santini M, Padeletti L, et al. Monitored atrial fibrillation duration predicts arterial embolic events in patients suffering from bradycardia and atrial fibrillation implanted with antitachycardia pacemakers. J Am Coll Cardiol 2005;46:1913-20.
- 102. Marini C, De Santis F, Sacco S, et al. Contribution of atrial fibrillation to incidence and outcome of ischemic stroke: results from a populationbased study. Stroke 2005;36:1115-9.
- 103. Cotter PE, Martin PJ, Ring L, et al. Incidence of atrial fibrillation detected by implantable loop recorders in unexplained stroke. Neurology 2013;80:1546-50.
- 104. Hart RG, Diener HC, Coutts SB, et al. Embolic strokes of undetermined source: the case for a new clinical construct. Lancet Neurol 2014;13:429-38.
- 105. Johnston SC, Easton JD, Farrant M, et al; Clinical Research Collaboration, Neurological Emergencies Treatment Trials Network, and the POINT Investigators. Clopidogrel and aspirin in acute ischemic stroke and high-risk TIA. N Engl J Med 2018;379:215-25.

- 106. Hart RG, Sharma M, Mundl H, et al. Rivaroxaban for stroke prevention after embolic stroke of undetermined source. N Engl J Med 2018;378:2191-201.
- 107. Investigators AWGotA, Connolly S, Pogue J, et al. Clopidogrel plus aspirin versus oral anticoagulation for atrial fibrillation in the Atrial fibrillation Clopidogrel Trial with Irbesartan for prevention of Vascular Events (ACTIVE W): a randomised controlled trial. Lancet 2006;367: 1903-12.
- 108. Ntaios G, Papavasileiou V, Diener HC, Makaritsis K, Michel P. Nonvitamin-K-antagonist oral anticoagulants in patients with atrial fibrillation and previous stroke or transient ischemic attack: a systematic review and meta-analysis of randomized controlled trials. Stroke 2012;43:3298-304.
- 109. Andrade JG, Field T, Khairy P. Detection of occult atrial fibrillation in patients with embolic stroke of uncertain source: a work in progress. Front Physiol 2015;6:100.
- 110. Sposato LA, Cipriano LE, Saposnik G, et al. Diagnosis of atrial fibrillation after stroke and transient ischaemic attack: a systematic review and meta-analysis. Lancet Neurol 2015;14:377-87.
- 111. Kishore A, Vail A, Majid A, et al. Detection of atrial fibrillation after ischemic stroke or transient ischemic attack: a systematic review and meta-analysis. Stroke 2014;45:520-6.
- 112. Higgins P, MacFarlane PW, Dawson J, et al. Noninvasive cardiac event monitoring to detect atrial fibrillation after ischemic stroke: a randomized, controlled trial. Stroke 2013;44:2525-31.
- 113. Brachmann J, Morillo CA, Sanna T, et al. Uncovering atrial fibrillation beyond short-term monitoring in cryptogenic stroke patients: three-year results from the Cryptogenic Stroke and Underlying Atrial Fibrillation trial. Circ Arrhythm Electrophysiol 2016;9:e003333.
- 114. Tsivgoulis G, Katsanos AH, Kohrmann M, et al. Duration of implantable cardiac monitoring and detection of atrial fibrillation in ischemic stroke patients: a systematic review and meta-analysis. J Stroke 2019;21:302-11.
- 115. Yang XM, Rao ZZ, Gu HQ, et al. Atrial fibrillation known before or detected after stroke share similar risk of ischemic stroke recurrence and death. Stroke 2019;50:1124-9.
- 116. Yong JH, Thavorn K, Hoch JS, et al. Potential cost-effectiveness of ambulatory cardiac rhythm monitoring after cryptogenic stroke. Stroke 2016;47:2380-5.
- 117. Diamantopoulos A, Sawyer LM, Lip GY, et al. Cost-effectiveness of an insertable cardiac monitor to detect atrial fibrillation in patients with cryptogenic stroke. Int J Stroke 2016;11:302-12.
- 118. Wein T, Lindsay MP, Cote R, et al. Canadian stroke best practice recommendations: secondary prevention of stroke, sixth edition practice guidelines, update 2017. Int J Stroke 2018;13:420-43.
- 119. Mitchell GF, Vasan RS, Keyes MJ, et al. Pulse pressure and risk of newonset atrial fibrillation. JAMA 2007;297:709-15.
- 120. Dublin S, French B, Glazer NL, et al. Risk of new-onset atrial fibrillation in relation to body mass index. Arch Intern Med 2006;166: 2322-8.
- 121. Rosengren A, Hauptman PJ, Lappas G, et al. Big men and atrial fibrillation: effects of body size and weight gain on risk of atrial fibrillation in men. Eur Heart J 2009;30:1113-20.
- 122. Schnabel RB, Sullivan LM, Levy D, et al. Development of a risk score for atrial fibrillation (Framingham Heart Study): a community-based cohort study. Lancet 2009;373:739-45.

- 123. Alonso A, Krijthe BP, Aspelund T, et al. Simple risk model predicts incidence of atrial fibrillation in a racially and geographically diverse population: the CHARGE-AF consortium. J Am Heart Assoc 2013;2: e000102.
- 124. Attia ZI, Noseworthy PA, Lopez-Jimenez F, et al. An artificial intelligence-enabled ECG algorithm for the identification of patients with atrial fibrillation during sinus rhythm: a retrospective analysis of outcome prediction. Lancet 2019;394:861-7.
- 125. Fein AS, Shvilkin A, Shah D, et al. Treatment of obstructive sleep apnea reduces the risk of atrial fibrillation recurrence after catheter ablation. J Am Coll Cardiol 2013;62:300-5.
- 126. Li L, Wang ZW, Li J, et al. Efficacy of catheter ablation of atrial fibrillation in patients with obstructive sleep apnoea with and without continuous positive airway pressure treatment: a meta-analysis of observational studies. Europace 2014;16:1309-14.
- 127. Shukla A, Aizer A, Holmes D, et al. Effect of obstructive sleep apnea treatment on atrial fibrillation recurrence: a meta-analysis. JACC: Clin Electrophysiol 2015;1:41-51.
- 128. Abed HS, Wittert GA, Leong DP, et al. Effect of weight reduction and cardiometabolic risk factor management on symptom burden and severity in patients with atrial fibrillation: a randomized clinical trial. JAMA 2013;310:2050-60.
- 129. Pathak RK, Middeldorp ME, Meredith M, et al. Long-term effect of goal-directed weight management in an atrial fibrillation cohort: a longterm follow-up study (LEGACY). J Am Coll Cardiol 2015;65:2159-69.
- 130. Mahajan R, Lau DH, Brooks AG, et al. Electrophysiological, electroanatomical, and structural remodeling of the atria as consequences of sustained obesity. J Am Coll Cardiol 2015;66:1-11.
- 131. Pathak RK, Elliott A, Middeldorp ME, et al. Impact of CARDIOrespiratory FITness on arrhythmia recurrence in obese individuals with atrial fibrillation: the CARDIO-FIT study. J Am Coll Cardiol 2015;66: 985-96.
- 132. Rienstra M, Hobbelt AH, Alings M, et al. Targeted therapy of underlying conditions improves sinus rhythm maintenance in patients with persistent atrial fibrillation: results of the RACE 3 trial. Eur Heart J 2018;39:2987-96.
- 133. Ogilvie IM, Newton N, Welner SA, Cowell W, Lip GY. Underuse of oral anticoagulants in atrial fibrillation: a systematic review. Am J Med 2010;123:638-45.
- 134. Boodhoo L, Bordoli G, Mitchell AR, et al. The safety and effectiveness of a nurse led cardioversion service under sedation. Heart 2004;90: 1443-6.
- 135. Cruz J, Mariano Z, Dorian P. Atrial fibrillation clinics in Canada: a nationwide project report. Can J Cardiol 2018;34:1219-24.
- 136. Gillis AM, Burland L, Arnburg B, et al. Treating the right patient at the right time: an innovative approach to the management of atrial fibrillation. Can J Cardiol 2008;24:195-8.
- 137. Elmouchi DA, VanOosterhout S, Muthusamy P, et al. Impact of an emergency department-initiated clinical protocol for the evaluation and treatment of atrial fibrillation. Crit Pathw Cardiol 2014;13:43-8.
- 138. Angaran P, Mariano Z, Dragan V, et al. The Atrial Fibrillation Therapies after ER visit: outpatient care for patients with acute AF - the AFTER3 study. J Atr Fibrillation 2015;7:1187.
- 139. Carter L, Gardner M, Magee K, et al. An integrated management approach to atrial fibrillation. J Am Heart Assoc 2016;5:e00295.

- 140. Hendriks JM, de WR, Crijns HJ, et al. Nurse-led care vs usual care for patients with atrial fibrillation: results of a randomized trial of integrated chronic care vs routine clinical care in ambulatory patients with atrial fibrillation. Eur Heart J 2012;33:2692-9.
- 141. Hendriks J, Tomini F, van Asselt T, Crijns H, Vrijhoef H. Costeffectiveness of a specialized atrial fibrillation clinic vs usual care in patients with atrial fibrillation. Europace 2013;15:1128-35.
- 142. Wijtvliet E, Tieleman RG, van Gelder IC, et al. Nurse-led vs usual-care for atrial fibrillation. Eur Heart J 2020;41:634-41.
- 143. Grady PA, Gough LL. Self-management: a comprehensive approach to management of chronic conditions. Am J Public Health 2014;104: e25-31.
- 144. Holman H, Lorig K. Patients as partners in managing chronic disease. Partnership is a prerequisite for effective and efficient health care. BMJ 2000;320:526-7.
- 145. Andrade JG, Krahn AD, Skanes AC, et al. Values and preferences of physicians and patients with nonvalvular atrial fibrillation who receive oral anticoagulation therapy for stroke prevention. Can J Cardiol 2016;32:747-53.
- 146. Lane DA, Ponsford J, Shelley A, Sirpal A, Lip GY. Patient knowledge and perceptions of atrial fibrillation and anticoagulant therapy: effects of an educational intervention programme. The West Birmingham Atrial Fibrillation Project. Int J Cardiol 2006;110:354-8.
- 147. Hernandez Madrid A, Potpara TS, Dagres N, et al. Differences in attitude, education, and knowledge about oral anticoagulation therapy among patients with atrial fibrillation in Europe: result of a self-assessment patient survey conducted by the European Heart Rhythm Association. Europace 2016;18:463-7.
- 148. Clifford S, Barber N, Horne R. Understanding different beliefs held by adherers, unintentional nonadherers, and intentional nonadherers: application of the Necessity-Concerns Framework. J Psychosom Res 2008;64:41-6.
- 149. Wofford JL, Wells MD, Singh S. Best strategies for patient education about anticoagulation with warfarin: a systematic review. BMC Health Serv Res 2008;8:40.
- 150. Meichenbaum D, Turk DC. Facilitating Treatment Adherence: A Practitioner's Guidebook. New York: Plenum Press, 1987.
- 151. Ho PM, Magid DJ, Shetterly SM, et al. Medication nonadherence is associated with a broad range of adverse outcomes in patients with coronary artery disease. Am Heart J 2008;155:772-9.
- 152. Ruppar TM, Cooper PS, Mehr DR, Delgado JM, Dunbar-Jacob JM. Medication adherence interventions improve heart failure mortality and readmission rates: systematic review and meta-analysis of controlled trials. J Am Heart Assoc 2016;5:e002606.
- 153. Kakkar AK, Mueller I, Bassand JP, et al. Risk profiles and antithrombotic treatment of patients newly diagnosed with atrial fibrillation at risk of stroke: perspectives from the international, observational, prospective GARFIELD registry. PLoS One 2013;8:e63479.
- 154. Camm AJ, Accetta G, Ambrosio G, et al. Evolving antithrombotic treatment patterns for patients with newly diagnosed atrial fibrillation. Heart 2017;103:307-14.
- 155. Huisman MV, Rothman KJ, Paquette M, et al. The changing landscape for stroke prevention in AF: findings from the GLORIA-AF registry phase 2. J Am Coll Cardiol 2017;69:777-85.
- **156.** Shore S, Carey EP, Turakhia MP, et al. Adherence to dabigatran therapy and longitudinal patient outcomes: insights from the veterans health administration. Am Heart J 2014;167:810-7.

- 157. Alberts MJ, Peacock WF, Fields LE, et al. Association between onceand twice-daily direct oral anticoagulant adherence in nonvalvular atrial fibrillation patients and rates of ischemic stroke. Int J Cardiol 2016;215: 11-3.
- **158.** Gomes T, Mamdani MM, Holbrook AM, Paterson JM, Juurlink DN. Persistence with therapy among patients treated with warfarin for atrial fibrillation. Arch Intern Med 2012;172:1687-9.
- **159.** Beyer-Westendorf J, Ehlken B, Evers T. Real-world persistence and adherence to oral anticoagulation for stroke risk reduction in patients with atrial fibrillation. Europace 2016;18:1150-7.
- 160. Lip GYH, Pan X, Kamble S, et al. Discontinuation risk comparison among 'real-world' newly anticoagulated atrial fibrillation patients: apixaban, warfarin, dabigatran, or rivaroxaban. PLoS One 2018;13: e0195950.
- 161. Ruigomez A, Vora P, Balabanova Y, et al. Discontinuation of nonvitamin K antagonist oral anticoagulants in patients with non-valvular atrial fibrillation: a population-based cohort study using primary care data from The Health Improvement Network in the UK. BMJ Open 2019;9:e031342.
- 162. Yao X, Abraham NS, Alexander GC, et al. Effect of adherence to oral anticoagulants on risk of stroke and major bleeding among patients with atrial fibrillation. J Am Heart Assoc 2016;5:e003074.
- 163. Ho PM, Bryson CL, Rumsfeld JS. Medication adherence: its importance in cardiovascular outcomes. Circulation 2009;119:3028-35.
- 164. Murray MD, Young J, Hoke S, et al. Pharmacist intervention to improve medication adherence in heart failure: a randomized trial. Ann Intern Med 2007;146:714-25.
- 165. Vinereanu D, Lopes RD, Bahit MC, et al. A multifaceted intervention to improve treatment with oral anticoagulants in atrial fibrillation (IMPACT-AF): an international, cluster-randomised trial. Lancet 2017;390:1737-46.
- 166. Pandya E, Bajorek BV. Assessment of Web-based education resources informing patients about stroke prevention in atrial fibrillation. J Clin Pharm Ther 2016;41:667-76.
- 167. Giebel GD, Gissel C. Accuracy of mHealth devices for atrial fibrillation screening: systematic review. JMIR Mhealth Uhealth 2019;7:e13641.
- 168. Linz D, Pluymaekers N, Hendriks JM. TeleCheck-AF for COVID-19. Eur Heart J 2020;41:1954-5.
- 169. Risk factors for stroke and efficacy of antithrombotic therapy in atrial fibrillation. Analysis of pooled data from five randomized controlled trials. Arch Intern Med 1994;154:1449-57.
- 170. Predictors of thromboembolism in atrial fibrillation: I. Clinical features of patients at risk. The Stroke Prevention in Atrial Fibrillation Investigators. Ann Intern Med 1992;116:1-5.
- 171. Olesen JB, Torp-Pedersen C, Hansen ML, Lip GY. The value of the CHA2DS2-VASc score for refining stroke risk stratification in patients with atrial fibrillation with a CHADS2 score 0-1: a nationwide cohort study. Thromb Haemost 2012;107:1172-9.
- 172. Gage BF, Waterman AD, Shannon W, et al. Validation of clinical classification schemes for predicting stroke: results from the National Registry of Atrial Fibrillation. JAMA 2001;285:2864-70.
- 173. Lip GY, Nieuwlaat R, Pisters R, Lane DA, Crijns HJ. Refining clinical risk stratification for predicting stroke and thromboembolism in atrial fibrillation using a novel risk factor-based approach: the Euro Heart survey on atrial fibrillation. Chest 2010;137:263-72.

- 174. Olesen JB, Lip GY, Hansen ML, et al. Validation of risk stratification schemes for predicting stroke and thromboembolism in patients with atrial fibrillation: nationwide cohort study. BMJ 2011;342:d124.
- 175. Verma A, Cairns JA, Mitchell LB, et al. 2014 Focused update of the Canadian Cardiovascular Society guidelines for the management of atrial fibrillation. Can J Cardiol 2014;30:1114-30.
- 176. Bell AD, Roussin A, Cartier R, et al. The use of antiplatelet therapy in the outpatient setting: Canadian Cardiovascular Society guidelines. Can J Cardiol 2011;27(suppl A):S1-59.
- 177. Proietti M, Lane DA, Boriani G, Lip GYH. Stroke prevention, evaluation of bleeding risk, and anticoagulant treatment management in atrial fibrillation contemporary international guidelines. Can J Cardiol 2019;35:619-33.
- 178. Quinn GR, Severdija ON, Chang Y, Singer DE. Wide variation in reported rates of stroke across cohorts of patients with atrial fibrillation. Circulation 2017;135:208-19.
- 179. Chao TF, Wang KL, Liu CJ, et al. Age threshold for increased stroke risk among patients with atrial fibrillation: a nationwide cohort study from Taiwan. J Am Coll Cardiol 2015;66:1339-47.
- 180. Friberg L, Skeppholm M, Terent A. Benefit of anticoagulation unlikely in patients with atrial fibrillation and a CHA2DS2-VASc score of 1. J Am Coll Cardiol 2015;65:225-32.
- 181. Singer DE, Chang Y, Borowsky LH, et al. A new risk scheme to predict ischemic stroke and other thromboembolism in atrial fibrillation: the ATRIA study stroke risk score. J Am Heart Assoc 2013;2:e000250.
- 182. Vadmann H, Nielsen PB, Hjortshoj SP, et al. Atrial flutter and thromboembolic risk: a systematic review. Heart 2015;101:1446-55.
- 183. Biblo LA, Yuan Z, Quan KJ, Mackall JA, Rimm AA. Risk of stroke in patients with atrial flutter. Am J Cardiol 2001;87:346-349, A349.
- 184. Corrado G, Sgalambro A, Mantero A, et al. Thromboembolic risk in atrial flutter. The FLASIEC (FLutter Atriale Societa Italiana di Ecografia Cardiovascolare) multicentre study. Eur Heart J 2001;22:1042-51.
- 185. Elhendy A, Gentile F, Khandheria BK, et al. Thromboembolic complications after electrical cardioversion in patients with atrial flutter. Am J Med 2001;111:433-8.
- 186. Gallagher MM, Hennessy BJ, Edvardsson N, et al. Embolic complications of direct current cardioversion of atrial arrhythmias: association with low intensity of anticoagulation at the time of cardioversion. J Am Coll Cardiol 2002;40:926-33.
- 187. Ghali WA, Wasil BI, Brant R, Exner DV, Cornuz J. Atrial flutter and the risk of thromboembolism: a systematic review and meta-analysis. Am J Med 2005;118:101-7.
- 188. Al-Kawaz M, Omran SS, Parikh NS, et al. Comparative risks of ischemic stroke in atrial flutter versus atrial fibrillation. J Stroke Cerebrovasc Dis 2018;27:839-44.
- 189. Gula LJ, Redfearn DP, Jenkyn KB, et al. Elevated incidence of atrial fibrillation and stroke in patients with atrial flutter-a population-based study. Can J Cardiol 2018;34:774-83.
- 190. Jin MN, Song C, Kim TH, et al. CHA(2)DS(2)-VASc score in the prediction of ischemic stroke in patients after radiofrequency catheter ablation of typical atrial flutter. Yonsei Med J 2018;59:236-42.
- 191. Lin YS, Chen YL, Chen TH, et al. Comparison of clinical outcomes among patients with atrial fibrillation or atrial flutter stratified by CHA2DS2-VASc score. JAMA Netw Open 2018;1:e180941.

- **192.** Chen YL, Lin YS, Wang HT, et al. Clinical outcomes of solitary atrial flutter patients using anticoagulation therapy: a national cohort study. Europace 2019;21:313-21.
- 193. Parikh MG, Aziz Z, Krishnan K, Madias C, Trohman RG. Usefulness of transesophageal echocardiography to confirm clinical utility of CHA2DS2-VASc and CHADS2 scores in atrial flutter. Am J Cardiol 2012;109:550-5.
- 194. Eckman MH, Singer DE, Rosand J, Greenberg SM. Moving the tipping point: the decision to anticoagulate patients with atrial fibrillation. Circ Cardiovasc Qual Outcomes 2011;4:14-21.
- 195. You JJ, Singer DE, Howard PA, et al. Antithrombotic therapy for atrial fibrillation: antithrombotic therapy and prevention of thrombosis, 9th ed: American College of Chest Physicians evidence-based clinical practice guidelines. Chest 2012;141:e531S-75S.
- 196. Saposnik G, Gladstone D, Raptis R, et al. Atrial fibrillation in ischemic stroke: predicting response to thrombolysis and clinical outcomes. Stroke 2013;44:99-104.
- 197. Mant J, Hobbs FDR, Fletcher K, et al. Warfarin versus aspirin for stroke prevention in an elderly community population with atrial fibrillation (the Birmingham Atrial Fibrillation Treatment of the Aged Study, BAFTA): a randomised controlled trial. Lancet 2007;370: 493-503.
- 198. Andersen LV, Vestergaard P, Deichgraeber P, et al. Warfarin for the prevention of systemic embolism in patients with non-valvular atrial fibrillation: a meta-analysis. Heart 2008;94:1607-13.
- 199. Connolly SJ, Pogue J, Eikelboom J, et al; ACTIVE W Investigators. Benefit of oral anticoagulant over antiplatelet therapy in atrial fibrillation depends on the quality of international normalized ratio control achieved by centers and countries as measured by time in therapeutic range. Circulation 2008;118:2029-37.
- 200. Wan Y, Heneghan C, Perera R, et al. Anticoagulation control and prediction of adverse events in patients with atrial fibrillation: a systematic review. Circ Cardiovasc Qual Outcomes 2008;1:84-91.
- 201. Haas S, Camm AJ, Bassand JP, et al. Predictors of NOAC versus VKA use for stroke prevention in patients with newly diagnosed atrial fibrillation: results from GARFIELD-AF. Am Heart J 2019;213:35-46.
- 202. Eikelboom JW, Connolly SJ, Brueckmann M, et al. Dabigatran versus warfarin in patients with mechanical heart valves. N Engl J Med 2013;369:1206-14.
- 203. Adams GF, Merrett JD, Hutchinson WM, Pollock AM. Cerebral embolism and mitral stenosis: survival with and without anticoagulants. J Neurol Neurosurg Psychiatry 1974;37:378-83.
- 204. Rush CJ, Campbell RT, Jhund PS, Petrie MC, McMurray JJV. Association is not causation: treatment effects cannot be estimated from observational data in heart failure. Eur Heart J 2018;39:3417-38.
- 205. Bell AD, Gross P, Heffernan M, et al. Appropriate use of antithrombotic medication in Canadian patients with nonvalvular atrial fibrillation. Am J Cardiol 2016;117:1107-11.
- 206. Steinberg BA, Shrader P, Thomas L, et al. Off-label dosing of nonvitamin K antagonist oral anticoagulants and adverse outcomes: the ORBIT-AF II Registry. J Am Coll Cardiol 2016;68:2597-604.
- 207. Yao X, Shah ND, Sangaralingham LR, Gersh BJ, Noseworthy PA. Nonvitamin K antagonist oral anticoagulant dosing in patients with atrial fibrillation and renal dysfunction. J Am Coll Cardiol 2017;69:2779-90.
- 208. Olesen JB, Lip GY, Kamper AL, et al. Stroke and bleeding in atrial fibrillation with chronic kidney disease. N Engl J Med 2012;367: 625-35.

- 209. Bonde AN, Lip GY, Kamper AL, et al. Net clinical benefit of antithrombotic therapy in patients with atrial fibrillation and chronic kidney disease: a nationwide observational cohort study. J Am Coll Cardiol 2014;64:2471-82.
- 210. Nielsen PB, Lane DA, Rasmussen LH, Lip GY, Larsen TB. Renal function and non-vitamin K oral anticoagulants in comparison with warfarin on safety and efficacy outcomes in atrial fibrillation patients: a systemic review and meta-regression analysis. Clin Res Cardiol 2015;104:418-29.
- 211. Stanton BE, Barasch NS, Tellor KB. Comparison of the safety and effectiveness of apixaban versus warfarin in patients with severe renal impairment. Pharmacotherapy 2017;37:412-9.
- 212. Shah M, Avgil Tsadok M, Jackevicius CA, et al. Warfarin use and the risk for stroke and bleeding in patients with atrial fibrillation undergoing dialysis. Circulation 2014;129:1196-203.
- 213. Garg L, Chen C, Haines DE. Atrial fibrillation and chronic kidney disease requiring hemodialysis - does warfarin therapy improve the risks of this lethal combination? Int J Cardiol 2016;222:47-50.
- 214. Randhawa MS, Vishwanath R, Rai MP, et al. Association between use of warfarin for atrial fibrillation and outcomes among patients with endstage renal disease: a systematic review and meta-analysis. JAMA Netw Open 2020;3:e202175.
- 215. Pokorney SD, Black-Maier E, Hellkamp AS, et al. Oral anticoagulation and cardiovascular outcomes in patients with atrial fibrillation and endstage renal disease. J Am Coll Cardiol 2020;75:1299-308.
- 216. Del-Carpio Munoz F, Gharacholou SM, Munger TM, et al. Metaanalysis of renal function on the safety and efficacy of novel oral anticoagulants for atrial fibrillation. Am J Cardiol 2016;117:69-75.
- 217. Yao X, Tangri N, Gersh BJ, et al. Renal outcomes in anticoagulated patients with atrial fibrillation. J Am Coll Cardiol 2017;70:2621-32.
- 218. Stanifer JW, Pokorney SD, Chertow GM, et al. Apixaban versus warfarin in patients with atrial fibrillation and advanced chronic kidney disease. Circulation 2020;141:1384-92.
- 219. Chan KE, Edelman ER, Wenger JB, Thadhani RI, Maddux FW. Dabigatran and rivaroxaban use in atrial fibrillation patients on hemodialysis. Circulation 2015;131:972-9.
- 220. Chokesuwattanaskul R, Thongprayoon C, Tanawuttiwat T, et al. Safety and efficacy of apixaban versus warfarin in patients with end-stage renal disease: meta-analysis. Pacing Clin Electrophysiol 2018;41:627-34.
- 221. Reed D, Palkimas S, Hockman R, et al. Safety and effectiveness of apixaban compared to warfarin in dialysis patients. Res Pract Thromb Haemost 2018;2:291-8.
- 222. Siontis KC, Zhang X, Eckard A, et al. Outcomes associated with apixaban use in patients with end-stage kidney disease and atrial fibrillation in the United States. Circulation 2018;138:1519-29.
- 223. Gong IY, Kim RB. Importance of pharmacokinetic profile and variability as determinants of dose and response to dabigatran, rivaroxaban, and apixaban. Can J Cardiol 2013;29:S24-33.
- 224. Fox KA, Piccini JP, Wojdyla D, et al. Prevention of stroke and systemic embolism with rivaroxaban compared with warfarin in patients with non-valvular atrial fibrillation and moderate renal impairment. Eur Heart J 2011;32:2387-94.
- 225. Bohula EA, Giugliano RP, Ruff CT, et al. Impact of renal function on outcomes with edoxaban in the ENGAGE AF-TIMI 48 trial. Circulation 2016;134:24-36.

- 226. Andrade JG, Hawkins NM, Fordyce CB, et al. Variability in nonvitamin K antagonist oral anticoagulants dose adjustment in atrial fibrillation patients with renal dysfunction: the Influence of Renal Function Estimation Formulae. Can J Cardiol 2018;34:1010-8.
- 227. Kralev S, Schneider K, Lang S, Suselbeck T, Borggrefe M. Incidence and severity of coronary artery disease in patients with atrial fibrillation undergoing first-time coronary angiography. PLoS One 2011;6:e24964.
- 228. Lip GY, Laroche C, Dan GA, et al. A prospective survey in European Society of Cardiology member countries of atrial fibrillation management: baseline results of EURObservational Research Programme Atrial Fibrillation (EORP-AF) Pilot General Registry. Europace 2014;16: 308-19.
- 229. Connolly S, Pogue J, Hart R, et al. Clopidogrel plus aspirin versus oral anticoagulation for atrial fibrillation in the Atrial fibrillation Clopidogrel Trial with Irbesartan for prevention of Vascular Events (ACTIVE W): a randomised controlled trial. Lancet 2006;367:1903-12.
- 230. Antithrombotic Trialists' (ATT) Collaboration, Baigent C, Blackwell L, et al. Aspirin in the primary and secondary prevention of vascular disease: collaborative meta-analysis of individual participant data from randomised trials. Lancet 2009;373:1849-60.
- 231. Anand SS, Yusuf S. Oral anticoagulants in patients with coronary artery disease. J Am Coll Cardiol 2003;41. 62S-9S.
- Eikelboom JW, Connolly SJ, Bosch J, et al. Rivaroxaban with or without aspirin in stable cardiovascular disease. N Engl J Med 2017;377:1319-30.
- 233. Randomised double-blind trial of fixed low-dose warfarin with aspirin after myocardial infarction. Coumadin Aspirin Reinfarction Study (CARS) Investigators. Lancet 1997;350:389-96.
- 234. Fiore LD, Ezekowitz MD, Brophy MT, et al. Department of Veterans Affairs Cooperative Studies Program Clinical Trial comparing combined warfarin and aspirin with aspirin alone in survivors of acute myocardial infarction: primary results of the CHAMP study. Circulation 2002;105: 557-63.
- 235. Brouwer MA, van den Bergh PJ, Aengevaeren WR, et al. Aspirin plus coumarin versus aspirin alone in the prevention of reocclusion after fibrinolysis for acute myocardial infarction: results of the Antithrombotics in the Prevention of Reocclusion In Coronary Thrombolysis (APRICOT)-2 trial. Circulation 2002;106:659-65.
- 236. Hurlen M, Abdelnoor M, Smith P, Erikssen J, Arnesen H. Warfarin, aspirin, or both after myocardial infarction. N Engl J Med 2002;347: 969-74.
- 237. van Es RF, Jonker JJ, Verheugt FW, et al. Aspirin and coumadin after acute coronary syndromes (the ASPECT-2 study): a randomised controlled trial. Lancet 2002;360:109-13.
- 238. Herlitz J, Holm J, Peterson M, et al. Effect of fixed low-dose warfarin added to aspirin in the long term after acute myocardial infarction; the LoWASA study. Eur Heart J 2004;25:232-9.
- 239. Yasuda S, Kaikita K, Ogawa H, et al. Atrial fibrillation and ischemic events with rivaroxaban in patients with stable coronary artery disease (AFIRE): protocol for a multicenter, prospective, randomized, openlabel, parallel group study. Int J Cardiol 2018;265:108-12.
- 240. Mehta SR, Bainey KR, Cantor WJ, et al. 2018 Canadian Cardiovascular Society/Canadian Association of Interventional Cardiology focused update of the guidelines for the use of antiplatelet therapy. Can J Cardiol 2018;34:214-33.
- 241. Angiolillo DJ, Goodman SG, Bhatt DL, et al. Antithrombotic therapy in patients with atrial fibrillation undergoing percutaneous coronary

intervention: a North American perspective-2016 update. Circ Cardiovasc Interv 2016;9:e004395.

- 242. Windecker S, Lopes RD, Massaro T, et al. Antithrombotic therapy in patients with atrial fibrillation and acute coronary syndrome treated medically or with percutaneous coronary intervention or undergoing elective percutaneous coronary intervention: insights from the AUGUSTUS trial. Circulation 2019;140:1921-32.
- 243. Dewilde WJ, Oirbans T, Verheugt FW, et al. Use of clopidogrel with or without aspirin in patients taking oral anticoagulant therapy and undergoing percutaneous coronary intervention: an open-label, randomised, controlled trial. Lancet 2013;381:1107-15.
- 244. Gibson CM, Mehran R, Bode C, et al. Prevention of bleeding in patients with atrial fibrillation undergoing PCI. N Engl J Med 2016;375: 2423-34.
- 245. Cannon CP, Bhatt DL, Oldgren J, et al. Dual antithrombotic therapy with dabigatran after PCI in atrial fibrillation. N Engl J Med 2017;377: 1513-24.
- 246. Vranckx P, Valgimigli M, Eckardt L, et al. Edoxaban-based versus vitamin K antagonist-based antithrombotic regimen after successful coronary stenting in patients with atrial fibrillation (ENTRUST-AF PCI): a randomised, open-label, phase 3b trial. Lancet 2019;394: 1335-43.
- 247. Lopes RD, Heizer G, Aronson R, et al. Antithrombotic therapy after acute coronary syndrome or PCI in atrial fibrillation. N Engl J Med 2019;380:1509-24.
- 248. Fiedler KA, Maeng M, Mehilli J, et al. Duration of triple therapy in patients requiring oral anticoagulation after drug-eluting stent implantation: the ISAR-TRIPLE trial. J Am Coll Cardiol 2015;65:1619-29.
- 249. Friberg L, Rosenqvist M, Lip GY. Evaluation of risk stratification schemes for ischaemic stroke and bleeding in 182 678 patients with atrial fibrillation: the Swedish Atrial Fibrillation cohort study. Eur Heart J 2012;33:1500-10.
- 250. Choi J, Kim J, Shim JH, Kim M, Nam GB. Risks versus benefits of anticoagulation for atrial fibrillation in cirrhotic patients. J Cardiovasc Pharmacol 2017;70:255-62.
- 251. Khoury T, Ayman AR, Cohen J, et al. The complex role of anticoagulation in cirrhosis: an updated review of where we are and where we are going. Digestion 2016;93:149-59.
- 252. Kuo L, Chao TF, Liu CJ, et al. Liver cirrhosis in patients with atrial fibrillation: would oral anticoagulation have a net clinical benefit for stroke prevention? J Am Heart Assoc 2017;6:e005307.
- 253. Kubitza D, Roth A, Becka M, et al. Effect of hepatic impairment on the pharmacokinetics and pharmacodynamics of a single dose of rivaroxaban, an oral, direct Factor Xa inhibitor. Br J Clin Pharmacol 2013;76: 89-98.
- Mueck W, Stampfuss J, Kubitza D, Becka M. Clinical pharmacokinetic and pharmacodynamic profile of rivaroxaban. Clin Pharmacokinet 2014;53:1-16.
- 255. Lauschke VM, Ingelman-Sundberg M. The importance of patientspecific factors for hepatic drug response and toxicity. Int J Mol Sci 2016;17:1714.
- **256.** Intagliata NM, Henry ZH, Maitland H, et al. Direct oral anticoagulants in cirrhosis patients pose similar risks of bleeding when compared to traditional anticoagulation. Dig Dis Sci 2016;61:1721-7.
- 257. Hum J, Shatzel JJ, Jou JH, Deloughery TG. The efficacy and safety of direct oral anticoagulants vs traditional anticoagulants in cirrhosis. Eur J Haematol 2017;98:393-7.

- 258. Lee HF, Chan YH, Chang SH, et al. Effectiveness and safety of nonvitamin K antagonist oral anticoagulant and warfarin in cirrhotic patients with nonvalvular atrial fibrillation. J Am Heart Assoc 2019;8: e011112.
- 259. Pastori D, Lip GYH, Farcomeni A, et al. Incidence of bleeding in patients with atrial fibrillation and advanced liver fibrosis on treatment with vitamin K or non-vitamin K antagonist oral anticoagulants. Int J Cardiol 2018;264:58-63.
- 260. Pugh RN, Murray-Lyon IM, Dawson JL, Pietroni MC, Williams R. Transection of the oesophagus for bleeding oesophageal varices. Br J Surg 1973;60:646-9.
- 261. Qamar A, Antman EM, Ruff CT, et al. Edoxaban versus warfarin in patients with atrial fibrillation and history of liver disease. J Am Coll Cardiol 2019;74:179-89.
- 262. Alonso A, MacLehose RF, Chen LY, et al. Prospective study of oral anticoagulants and risk of liver injury in patients with atrial fibrillation. Heart 2017;103:834-9.
- 263. Hu YF, Liu CJ, Chang PM, et al. Incident thromboembolism and heart failure associated with new-onset atrial fibrillation in cancer patients. Int J Cardiol 2013;165:355-7.
- **264.** Farmakis D, Parissis J, Filippatos G. Insights into onco-cardiology: atrial fibrillation in cancer. J Am Coll Cardiol 2014;63:945-53.
- 265. Caine GJ, Stonelake PS, Lip GY, Kehoe ST. The hypercoagulable state of malignancy: pathogenesis and current debate. Neoplasia 2002;4: 465-73.
- 266. Lenneman CG, Sawyer DB. Cardio-oncology: an update on cardiotoxicity of cancer-related treatment. Circ Res 2016;118:1008-20.
- 267. Melloni C, Dunning A, Granger CB, et al. Efficacy and safety of apixaban versus warfarin in patients with atrial fibrillation and a history of cancer: insights from the ARISTOTLE trial. Am J Med 2017;130: 1440-1448.e1441.
- 268. Laube ES, Yu A, Gupta D, et al. Rivaroxaban for stroke prevention in patients with nonvalvular atrial fibrillation and active cancer. Am J Cardiol 2017;120:213-7.
- 269. Ianotto JC, Couturier MA, Galinat H, et al. Administration of direct oral anticoagulants in patients with myeloproliferative neoplasms. Int J Hematol 2017;106:517-21.
- 270. Ording AG, Horvath-Puho E, Adelborg K, et al. Thromboembolic and bleeding complications during oral anticoagulation therapy in cancer patients with atrial fibrillation: a Danish nationwide population-based cohort study. Cancer Med 2017;6:1165-72.
- 271. Flack KF, Desai J, Kolb JM, et al. Major gastrointestinal bleeding often is caused by occult malignancy in patients receiving warfarin or dabigatran to prevent stroke and systemic embolism from atrial fibrillation. Clin Gastroenterol Hepatol 2017;15:682-90.
- 272. Russo V, Rago A, Papa AA, et al. Use of non-vitamin K antagonist oral anticoagulants in atrial fibrillation patients with malignancy: clinical practice experience in a single institution and literature review. Semin Thromb Hemost 2018;44:370-6.
- 273. Chen ST, Hellkamp AS, Becker RC, et al. Efficacy and safety of rivaroxaban vs warfarin in patients with non-valvular atrial fibrillation and a history of cancer: observations from ROCKET AF. Eur Heart J Qual Care Clin Outcomes 2019;5:145-52.
- 274. Russo V, Bottino R, Rago A, et al. Atrial fibrillation and malignancy: the clinical performance of non-vitamin K oral anticoagulants-a systematic review. Semin Thromb Hemost 2019;45:205-14.

- 275. Lee AYY. Anticoagulant therapy for venous thromboembolism in cancer. N Engl J Med 2020;382:1650-2.
- 276. Hoffmann A, Chockalingam P, Balint OH, et al. Cerebrovascular accidents in adult patients with congenital heart disease. Heart 2010;96: 1223-6.
- 277. Bouchardy J, Therrien J, Pilote L, et al. Atrial arrhythmias in adults with congenital heart disease. Circulation 2009;120:1679-86.
- Khairy P. Arrhythmias in adults with congenital heart disease: what the practicing cardiologist needs to know. Can J Cardiol 2019;35: 1698-707.
- 279. Heidendael JF, Bokma JP, de Groot JR, et al. Weighing the risks: thrombotic and bleeding events in adults with atrial arrhythmias and congenital heart disease. Int J Cardiol 2015;186:315-20.
- 280. Khairy P, Aboulhosn J, Broberg CS, et al. Thromboprophylaxis for atrial arrhythmias in congenital heart disease: a multicenter study. Int J Cardiol 2016;223:729-35.
- 281. Warnes CA, Williams RG, Bashore TM, et al. ACC/AHA 2008 guidelines for the management of adults with congenital heart disease: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (writing committee to develop guidelines on the management of adults with congenital heart disease). Circulation 2008;118:e714-833.
- 282. Khairy P, Van Hare GF, Balaji S, et al. PACES/HRS expert consensus statement on the recognition and management of arrhythmias in adult congenital heart disease: developed in partnership between the Pediatric and Congenital Electrophysiology Society (PACES) and the Heart Rhythm Society (HRS). Endorsed by the governing bodies of PACES, HRS, the American College of Cardiology (ACC), the American Heart Association (AHA), the European Heart Rhythm Association (EHRA), the Canadian Heart Rhythm Society (CHRS), and the International Society for Adult Congenital Heart Disease (ISACHD). Can J Cardiol 2014;30:e1-63.
- 283. Mongeon FP, Macle L, Beauchesne LM, et al. Non-vitamin K antagonist oral anticoagulants (NOACs) in adult congenital heart disease. Can J Cardiol 2019;35:1686-97.
- 284. Pujol C, Niesert AC, Engelhardt A, et al. Usefulness of direct oral anticoagulants in adult congenital heart disease. Am J Cardiol 2016;117:450-5.
- 285. Yang H, Bouma BJ, Dimopoulos K, et al. Non-vitamin K antagonist oral anticoagulants (NOACs) for thromboembolic prevention, are they safe in congenital heart disease? Results of a worldwide study. Int J Cardiol 2020;299:123-30.
- 286. Georgekutty J, Kazerouninia A, Wang Y, et al. Novel oral anticoagulant use in adult Fontan patients: a single center experience. Congenit Heart Dis 2018;13:541-7.
- 287. Pinto C, Samuel BP, Ratnasamy C, Vettukattil JJ. Thrombosis in Fontan patient on apixaban. Int J Cardiol 2015;182:66-7.
- Lubitz SA, Yin X, Rienstra M, et al. Long-term outcomes of secondary atrial fibrillation in the community: the Framingham Heart Study. Circulation 2015;131:1648-55.
- 289. Quon MJ, Behlouli H, Pilote L. Anticoagulant use and risk of ischemic stroke and bleeding in patients with secondary atrial fibrillation associated with acute coronary syndromes, acute pulmonary disease, or sepsis. JACC Clin Electrophysiol 2018;4:386-93.
- 290. Walkey AJ, Quinn EK, Winter MR, McManus DD, Benjamin EJ. Practice patterns and outcomes associated with use of anticoagulation among patients with atrial fibrillation during sepsis. JAMA Cardiol 2016;1:682-90.

- 291. Sogaard M, Skjoth F, Kjaeldgaard JN, Lip GYH, Larsen TB. Bleeding complications in anticoagulated patients with atrial fibrillation and sepsis: a propensity-weighted cohort study. J Am Heart Assoc 2017;6: e007453.
- 292. Stuijver DJ, van Zaane B, Romualdi E, et al. The effect of hyperthyroidism on procoagulant, anticoagulant and fibrinolytic factors: a systematic review and meta-analysis. Thromb Haemost 2012;108: 1077-88.
- 293. Presti CF, Hart RG. Thyrotoxicosis, atrial fibrillation, and embolism, revisited. Am Heart J 1989;117:976-7.
- 294. Petersen P, Hansen JM. Stroke in thyrotoxicosis with atrial fibrillation. Stroke 1988;19:15-8.
- 295. Chute JP, Ryan CP, Sladek G, Shakir KM. Exacerbation of warfarininduced anticoagulation by hyperthyroidism. Endocr Pract 1997;3: 77-9.
- 296. Guttmann OP, Rahman MS, O'Mahony C, Anastasakis A, Elliott PM. Atrial fibrillation and thromboembolism in patients with hypertrophic cardiomyopathy: systematic review. Heart 2014;100:465-72.
- 297. Guttmann OP, Pavlou M, O'Mahony C, et al. Prediction of thromboembolic risk in patients with hypertrophic cardiomyopathy (HCM Risk-CVA). Eur J Heart Fail 2015;17:837-45.
- **298.** Yang YJ, Yuan JQ, Fan CM, et al. Incidence of ischemic stroke and systemic embolism in patients with hypertrophic cardiomyopathy, nonvalvular atrial fibrillation, CHA2DS2-VASc score of ≤ 1 and without anticoagulant therapy. Heart Vessels 2016;31:1148-53.
- 299. Masri A, Kanj M, Thamilarasan M, et al. Outcomes in hypertrophic cardiomyopathy patients with and without atrial fibrillation: a survival meta-analysis. Cardiovasc Diagn Ther 2017;7:36-44.
- **300.** Noubiap JJ, Bigna JJ, Agbor VN, et al. Meta-analysis of atrial fibrillation in patients with various cardiomyopathies. Am J Cardiol 2019;124: 262-9.
- 301. Borer JS, Atar D, Marciniak T, Kim MH, Serebruany V. Atrial fibrillation and stroke in patients with hypertrophic cardiomyopathy: important new insights. Thromb Haemost 2019;119:355-7.
- **302.** Guttmann OP. Atrial fibrillation and thromboembolism in hypertrophic cardiomyopathy - an underestimated risk. Int J Cardiol 2018;273: 187-8.
- 303. Camm CF, Camm AJ. Atrial fibrillation and anticoagulation in hypertrophic cardiomyopathy. Arrhythm Electrophysiol Rev 2017;6:63-8.
- 304. Dominguez F, Climent V, Zorio E, et al. Direct oral anticoagulants in patients with hypertrophic cardiomyopathy and atrial fibrillation. Int J Cardiol 2017;248:232-8.
- 305. Jung H, Yang PS, Sung JH, et al. Hypertrophic cardiomyopathy in patients with atrial fibrillation: prevalence and associated stroke risks in a nationwide cohort study. Thromb Haemost 2019;119:285-93.
- **306.** Lee HJ, Kim HK, Jung JH, et al. Novel oral anticoagulants for primary stroke prevention in hypertrophic cardiomyopathy patients with atrial fibrillation. Stroke 2019;50:2582-6.
- 307. Feng D, Syed IS, Martinez M, et al. Intracardiac thrombosis and anticoagulation therapy in cardiac amyloidosis. Circulation 2009;119: 2490-7.
- 308. Feng D, Edwards WD, Oh JK, et al. Intracardiac thrombosis and embolism in patients with cardiac amyloidosis. Circulation 2007;116: 2420-6.

- 309. El-Am EA, Dispenzieri A, Melduni RM, et al. Direct current cardioversion of atrial arrhythmias in adults with cardiac amyloidosis. J Am Coll Cardiol 2019;73:589-97.
- Cheung CC, Roston TM, Andrade JG, Bennett MT, Davis MK. Arrhythmias in cardiac amyloidosis: challenges in risk stratification and treatment. Can J Cardiol 2020;36:416-23.
- 311. Mints YY, Doros G, Berk JL, Connors LH, Ruberg FL. Features of atrial fibrillation in wild-type transthyretin cardiac amyloidosis: a systematic review and clinical experience. ESC Heart Fail 2018;5:772-9.
- 312. Kamath S, Lip GY. Atrial fibrillation in the elderly: anticoagulation strategies and indications in the very elderly. Am J Geriatr Cardiol 2002;11:357-62. quiz 363-4.
- 313. Hylek EM, Evans-Molina C, Shea C, Henault LE, Regan S. Major hemorrhage and tolerability of warfarin in the first year of therapy among elderly patients with atrial fibrillation. Circulation 2007;115: 2689-96.
- 314. Taggar JS, Lip GY. Anticoagulation for elderly patients with atrial fibrillation: not to be neglected. Europace 2008;10:1-2.
- 315. Xing YL, Ma Q, Ma XY, et al. Characteristics of non-valvular atrial fibrillation patients who benefit most from anticoagulation treatment. Int J Clin Exp Med 2015;8:18721-8.
- 316. Chao TF, Liu CJ, Lin YJ, et al. Oral anticoagulation in very elderly patients with atrial fibrillation: a nationwide cohort study. Circulation 2018;138:37-47.
- 317. Steinberg BA, Greiner MA, Hammill BG, et al. Contraindications to anticoagulation therapy and eligibility for novel anticoagulants in older patients with atrial fibrillation. Cardiovasc Ther 2015;33:177-83.
- Ekerstad N, Karlsson T, Soderqvist S, Karlson BW. Hospitalized frail elderly patients - atrial fibrillation, anticoagulation and 12 months' outcomes. Clin Interv Aging 2018;13:749-56.
- 319. Sanchez-Barba B, Navarrete-Reyes AP, Avila-Funes JA. Are geriatric syndromes associated with reluctance to initiate oral anticoagulation therapy in elderly adults with nonvalvular atrial fibrillation? J Am Geriatr Soc 2013;61:2236-7.
- 320. Inui TS, Parina R, Chang DC, Inui TS, Coimbra R. Mortality after ground-level fall in the elderly patient taking oral anticoagulation for atrial fibrillation/flutter: a long-term analysis of risk versus benefit. J Trauma Acute Care Surg 2014;76:642-9. discussion 649-50.
- 321. Wilke T, Groth A, Pfannkuche M, et al. Real life anticoagulation treatment of patients with atrial fibrillation in Germany: extent and causes of anticoagulant under-use. J Thromb Thrombolysis 2015;40: 97-107.
- 322. Gumbinger C, Holstein T, Stock C, Rizos T, Horstmann S, Veltkamp R, et al. Reasons underlying non-adherence to and discontinuation of anticoagulation in secondary stroke prevention among patients with atrial fibrillation. Eur Neurol 2015;73:184-91.
- 323. Bahri O, Roca F, Lechani T, et al. Underuse of oral anticoagulation for individuals with atrial fibrillation in a nursing home setting in France: comparisons of resident characteristics and physician attitude. J Am Geriatr Soc 2015;63:71-6.
- 324. Palomaki A, Mustonen P, Hartikainen JE, et al. Underuse of anticoagulation in stroke patients with atrial fibrillation-the FibStroke study. Eur J Neurol 2016;23:133-9.
- **325.** Rao MP, Vinereanu D, Wojdyla DM, et al. Clinical outcomes and history of fall in patients with atrial fibrillation treated with oral anti-coagulation: insights from the ARISTOTLE trial. Am J Med 2018;131: 269-275.e262.

- 326. Pappas MA, Vijan S, Rothberg MB, Singer DE. Reducing age bias in decision analyses of anticoagulation for patients with nonvalvular atrial fibrillation - a microsimulation study. PLoS One 2018;13:e0199593.
- 327. Oqab Z, Pournazari P, Sheldon RS. What is the impact of frailty on prescription of anticoagulation in elderly patients with atrial fibrillation? A systematic review and meta-analysis. J Atr Fibrillation 2018;10:1870.
- **328.** Patti G, Pecen L, Lucerna M, et al. Net clinical benefit of non-vitamin K antagonist vs vitamin K antagonist anticoagulants in elderly patients with atrial fibrillation. Am J Med 2019;132:749-757.e745.
- 329. Okumura K, Akao M, Yoshida T, et al. Low-dose edoxaban in very elderly patients with atrial fibrillation. N Engl J Med 2020;383: 1735-45.
- 330. Eikelboom JW, Wallentin L, Connolly SJ, et al. Risk of bleeding with 2 doses of dabigatran compared with warfarin in older and younger patients with atrial fibrillation: an analysis of the randomized evaluation of long-term anticoagulant therapy (RE-LY) trial. Circulation 2011;123: 2363-72.
- 331. Halperin JL, Hankey GJ, Wojdyla DM, et al. Efficacy and safety of rivaroxaban compared with warfarin among elderly patients with nonvalvular atrial fibrillation in the Rivaroxaban Once Daily, Oral, Direct Factor Xa Inhibition Compared With Vitamin K Antagonism for Prevention of Stroke and Embolism Trial in Atrial Fibrillation (ROCKET AF). Circulation 2014;130:138-46.
- **332.** Halvorsen S, Atar D, Yang H, et al. Efficacy and safety of apixaban compared with warfarin according to age for stroke prevention in atrial fibrillation: observations from the ARISTOTLE trial. Eur Heart J 2014;35:1864-72.
- 333. Kato ET, Giugliano RP, Ruff CT, et al. Efficacy and safety of edoxaban in elderly patients with atrial fibrillation in the ENGAGE AF-TIMI 48 trial. J Am Heart Assoc 2016;5:e003432.
- 334. Sardar P, Chatterjee S, Chaudhari S, Lip GY. New oral anticoagulants in elderly adults: evidence from a meta-analysis of randomized trials. J Am Geriatr Soc 2014;62:857-64.
- 335. Boriani G, Ruff CT, Kuder JF, et al. Relationship between body mass index and outcomes in patients with atrial fibrillation treated with edoxaban or warfarin in the ENGAGE AF-TIMI 48 trial. Eur Heart J 2019;40:1541-50.
- 336. Balla SR, Cyr DD, Lokhnygina Y, et al. Relation of risk of stroke in patients with atrial fibrillation to body mass index (from Patients Treated With Rivaroxaban and Warfarin in the Rivaroxaban Once Daily Oral Direct Factor Xa Inhibition Compared with Vitamin K Antagonism for Prevention of Stroke and Embolism Trial in Atrial Fibrillation Trial). Am J Cardiol 2017;119:1989-96.
- 337. Hohnloser SH, Fudim M, Alexander JH, et al. Efficacy and safety of apixaban versus warfarin in patients with atrial fibrillation and extremes in body weight. Circulation 2019;139:2292-300.
- **338.** Malik AH, Yandrapalli S, Shetty S, et al. Impact of weight on the efficacy and safety of direct-acting oral anticoagulants in patients with non-valvular atrial fibrillation: a meta-analysis. Europace 2020;22: 361-7.
- 339. Andrade JG, Mitchell LB. Periprocedural anticoagulation for cardioversion of acute onset atrial fibrillation and flutter: evidence base for current guidelines. Can J Cardiol 2019;35:1301-10.
- 340. Roy D, Marchand E, Gagne P, Chabot M, Cartier R. Usefulness of anticoagulant therapy in the prevention of embolic complications of atrial fibrillation. Am Heart J 1986;112:1039-43.

- 341. Manning WJ, Leeman DE, Gotch PJ, Come PC. Pulsed Doppler evaluation of atrial mechanical function after electrical cardioversion of atrial fibrillation. J Am Coll Cardiol 1989;13:617-23.
- 342. Grimm RA, Stewart WJ, Maloney JD, et al. Impact of electrical cardioversion for atrial fibrillation on left atrial appendage function and spontaneous echo contrast: characterization by simultaneous transesophageal echocardiography. J Am Coll Cardiol 1993;22:1359-66.
- 343. Irani WN, Grayburn PA, Afridi I. Prevalence of thrombus, spontaneous echo contrast, and atrial stunning in patients undergoing cardioversion of atrial flutter. A prospective study using transesophageal echocardiography. Circulation 1997;95:962-6.
- 344. Manning WJ, Silverman DI, Katz SE, et al. Impaired left atrial mechanical function after cardioversion: relation to the duration of atrial fibrillation. J Am Coll Cardiol 1994;23:1535-40.
- 345. Fatkin D, Kuchar DL, Thorburn CW, Feneley MP. Transesophageal echocardiography before and during direct current cardioversion of atrial fibrillation: evidence for "atrial stunning" as a mechanism of thromboembolic complications. J Am Coll Cardiol 1994;23:307-16.
- 346. Mattioli AV, Tarabini Castellani E, et al. Restoration of atrial function after atrial fibrillation of different etiological origins. Cardiology 1996;87:205-11.
- 347. Sparks PB, Jayaprakash S, Mond HG, et al. Left atrial mechanical function after brief duration atrial fibrillation. J Am Coll Cardiol 1999;33:342-9.
- 348. McIntyre WF, Connolly SJ, Wang J, et al. Thromboembolic events around the time of cardioversion for atrial fibrillation in patients receiving antiplatelet treatment in the ACTIVE trials. Eur Heart J 2019;40:3026-32.
- 349. Klein AL, Grimm RA, Murray RD, et al. Use of transesophageal echocardiography to guide cardioversion in patients with atrial fibrillation. N Engl J Med 2001;344:1411-20.
- 350. Zhan Y, Joza J, Al Rawahi M, et al. Assessment and management of the left atrial appendage thrombus in patients with nonvalvular atrial fibrillation. Can J Cardiol 2018;34:252-61.
- 351. Cairns JA, Connolly S, McMurtry S, Stephenson M, Talajic M. Canadian Cardiovascular Society atrial fibrillation guidelines 2010: prevention of stroke and systemic thromboembolism in atrial fibrillation and flutter. Can J Cardiol 2011;27:74-90.
- **352.** Kleemann T, Becker T, Strauss M, Schneider S, Seidl K. Prevalence of left atrial thrombus and dense spontaneous echo contrast in patients with short-term atrial fibrillation < 48 hours undergoing cardioversion: value of transesophageal echocardiography to guide cardioversion. J Am Soc Echocardiogr 2009;22:1403-8.
- 353. Airaksinen KE, Gronberg T, Nuotio I, et al. Thromboembolic complications after cardioversion of acute atrial fibrillation: the FinCV (Finnish CardioVersion) study. J Am Coll Cardiol 2013;62:1187-92.
- 354. Nuotio I, Hartikainen JE, Gronberg T, Biancari F, Airaksinen KE. Time to cardioversion for acute atrial fibrillation and thromboembolic complications. JAMA 2014;312:647-9.
- 355. Gronberg T, Hartikainen JE, Nuotio I, et al. Anticoagulation, CHA2DS2VASc score, and thromboembolic risk of cardioversion of acute atrial fibrillation (from the FinCV study). Am J Cardiol 2016;117: 1294-8.
- 356. Bah A, Nuotio I, Gronberg T, et al. Sex, age, and time to cardioversion. Risk factors for cardioversion of acute atrial fibrillation from the FinCV study. Ann Med 2017;49:254-9.

- 357. Hellman T, Kiviniemi T, Nuotio I, et al. Intensity of anticoagulation and risk of thromboembolism after elective cardioversion of atrial fibrillation. Thromb Res 2017;156:163-7.
- 358. Hellman T, Kiviniemi T, Nuotio I, et al. Optimal timing for cardioversion in patients with atrial fibrillation. Clin Cardiol 2018;41:966-71.
- 359. Garg A, Khunger M, Seicean S, Chung MK, Tchou P. Incidence of thromboembolic complications within 30 days of electrical cardioversion performed within 48 hours of atrial fibrillation onset. JACC Clin Electrophysiol 2016;2:487-94.
- 360. Palomaki A, Mustonen P, Hartikainen JE, et al. Strokes after cardioversion of atrial fibrillation-the FibStroke study. Int J Cardiol 2016;203:269-73.
- 361. Hansen ML, Jepsen RM, Olesen JB, et al. Thromboembolic risk in 16 274 atrial fibrillation patients undergoing direct current cardioversion with and without oral anticoagulant therapy. Europace 2015;17:18-23.
- 362. Sjalander S, Svensson PJ, Friberg L. Atrial fibrillation patients with CHA2DS2-VASc >1 benefit from oral anticoagulation prior to cardioversion. Int J Cardiol 2016;215:360-3.
- 363. Crijns HJ, Weijs B, Fairley AM, et al. Contemporary real life cardioversion of atrial fibrillation: results from the multinational RHYTHM-AF study. Int J Cardiol 2014;172:588-94.
- 364. Atzema CL, Jackevicius CA, Chong A, et al. Prescribing of oral anticoagulants in the emergency department and subsequent long-term use by older adults with atrial fibrillation. CMAJ 2019;191:E1345-54.
- 365. Cappato R, Ezekowitz MD, Klein AL, et al. Rivaroxaban vs vitamin K antagonists for cardioversion in atrial fibrillation. Eur Heart J 2014;35: 3346-55.
- 366. Goette A, Merino JL, Ezekowitz MD, et al. Edoxaban versus enoxaparin-warfarin in patients undergoing cardioversion of atrial fibrillation (ENSURE-AF): a randomised, open-label, phase 3b trial. Lancet 2016;388:1995-2003.
- 367. Ezekowitz MD, Pollack CV Jr, Halperin JL, et al. Apixaban compared to heparin/vitamin K antagonist in patients with atrial fibrillation scheduled for cardioversion: the EMANATE trial. Eur Heart J 2018;39: 2959-71.
- 368. Kotecha D, Pollack CV Jr, De Caterina R, Renda G, Kirchhof P. Direct oral anticoagulants halve thromboembolic events after cardioversion of AF compared with warfarin. J Am Coll Cardiol 2018;72:1984-6.
- 369. Gupta S, Um KJ, Pandey A, et al. Direct oral anticoagulants versus vitamin k antagonists in patients undergoing cardioversion for atrial fibrillation: a systematic review and meta-analysis. Cardiovasc Drugs Ther 2019;33:339-52.
- 370. Nagarakanti R, Ezekowitz MD, Oldgren J, et al. Dabigatran versus warfarin in patients with atrial fibrillation: an analysis of patients undergoing cardioversion. Circulation 2011;123:131-6.
- 371. Piccini JP, Stevens SR, Lokhnygina Y, et al. Outcomes after cardioversion and atrial fibrillation ablation in patients treated with rivaroxaban and warfarin in the ROCKET AF trial. J Am Coll Cardiol 2013;61:1998-2006.
- 372. Flaker G, Lopes RD, Al-Khatib SM, et al. Efficacy and safety of apixaban in patients after cardioversion for atrial fibrillation: insights from the ARISTOTLE Trial (Apixaban for Reduction in Stroke and Other Thromboembolic Events in Atrial Fibrillation). J Am Coll Cardiol 2014;63:1082-7.
- 373. Plitt A, Ezekowitz MD, De Caterina R, et al. Cardioversion of atrial fibrillation in ENGAGE AF-TIMI 48. Clin Cardiol 2016;39:345-6.

- 374. Telles-Garcia N, Dahal K, Kocherla C, et al. Non-vitamin K antagonists oral anticoagulants are as safe and effective as warfarin for cardioversion of atrial fibrillation: a systematic review and meta-analysis. Int J Cardiol 2018;268:143-8.
- 375. Brunetti ND, Tarantino N, De Gennaro L, et al. Direct oral anticoagulants compared to vitamin-K antagonists in cardioversion of atrial fibrillation: an updated meta-analysis. J Thromb Thrombolysis 2018;45:550-6.
- 376. Di Biase L, Burkhardt JD, Santangeli P, et al. Periprocedural stroke and bleeding complications in patients undergoing catheter ablation of atrial fibrillation with different anticoagulation management: results from the Role of Coumadin in Preventing Thromboembolism in Atrial Fibrillation (AF) Patients Undergoing Catheter Ablation (COMPARE) randomized trial. Circulation 2014;129:2638-44.
- 377. Cappato R, Marchlinski FE, Hohnloser SH, et al. Uninterrupted rivaroxaban vs uninterrupted vitamin K antagonists for catheter ablation in non-valvular atrial fibrillation. Eur Heart J 2015;36:1805-11.
- Calkins H, Willems S, Gerstenfeld EP, et al. Uninterrupted dabigatran versus warfarin for ablation in atrial fibrillation. N Engl J Med 2017;376:1627-36.
- 379. Kirchhof P, Haeusler KG, Blank B, et al. Apixaban in patients at risk of stroke undergoing atrial fibrillation ablation. Eur Heart J 2018;39: 2942-55.
- 380. Reynolds MR, Allison JS, Natale A, et al. A prospective randomized trial of apixaban dosing during atrial fibrillation ablation: the AEIOU trial. JACC Clin Electrophysiol 2018;4:580-8.
- 381. Hohnloser SH, Camm J, Cappato R, et al. Uninterrupted edoxaban vs vitamin K antagonists for ablation of atrial fibrillation: the ELIMINATE-AF trial. Eur Heart J 2019;40:3013-21.
- **382.** Nogami A, Harada T, Sekiguchi Y, et al. Safety and efficacy of minimally interrupted dabigatran vs uninterrupted warfarin therapy in adults undergoing atrial fibrillation catheter ablation: a randomized clinical trial. JAMA Netw Open 2019;2:e191994.
- 383. Nakamura K, Naito S, Sasaki T, et al. Uninterrupted vs interrupted periprocedural direct oral anticoagulants for catheter ablation of atrial fibrillation: a prospective randomized single-centre study on postablation thrombo-embolic and haemorrhagic events. Europace 2019;21:259-67.
- 384. Larbig R, Dittrich R, Kochhaeuser S, et al. Influence of clinical parameters and anticoagulation on intraprocedural cerebral microembolic signals during pulmonary vein isolation. PLoS One 2016;11:e0157886.
- 385. Martin AC, Godier A, Narayanan K, Smadja DM, Marijon E. Management of intraprocedural anticoagulation in patients on non-vitamin K antagonist oral anticoagulants undergoing catheter ablation for atrial fibrillation: understanding the gaps in evidence. Circulation 2018;138: 627-33.
- 386. Wazni OM, Rossillo A, Marrouche NF, et al. Embolic events and char formation during pulmonary vein isolation in patients with atrial fibrillation: impact of different anticoagulation regimens and importance of intracardiac echo imaging. J Cardiovasc Electrophysiol 2005;16:576-81.
- 387. Oral H, Chugh A, Ozaydin M, et al. Risk of thromboembolic events after percutaneous left atrial radiofrequency ablation of atrial fibrillation. Circulation 2006;114:759-65.
- 388. Themistoclakis S, Corrado A, Marchlinski FE, et al. The risk of thromboembolism and need for oral anticoagulation after successful atrial fibrillation ablation. J Am Coll Cardiol 2010;55:735-43.

- **389.** Hunter RJ, McCready J, Diab I, et al. Maintenance of sinus rhythm with an ablation strategy in patients with atrial fibrillation is associated with a lower risk of stroke and death. Heart 2012;98:48-53.
- 390. Bunch TJ, May HT, Bair TL, et al. Atrial fibrillation ablation patients have long-term stroke rates similar to patients without atrial fibrillation regardless of CHADS2 score. Heart Rhythm 2013;10:1272-7.
- 391. Saliba W, Schliamser JE, Lavi I, et al. Catheter ablation of atrial fibrillation is associated with reduced risk of stroke and mortality: a propensity score-matched analysis. Heart Rhythm 2017;14:635-42.
- **392.** Srivatsa UN, Danielsen B, Amsterdam EA, et al. CAABL-AF (California Study of Ablation for Atrial Fibrillation): mortality and stroke, 2005 to 2013. Circ Arrhythm Electrophysiol 2018;11:e005739.
- 393. Joza J, Samuel M, Jackevicius CA, et al. Long-term risk of stroke and bleeding post-atrial fibrillation ablation. J Cardiovasc Electrophysiol 2018;29:1355-62.
- **394.** Packer DL, Mark DB, Robb RA, et al. Effect of catheter ablation vs antiarrhythmic drug therapy on mortality, stroke, bleeding, and cardiac arrest among patients with atrial fibrillation: the CABANA randomized clinical trial. JAMA 2019;321:1261-74.
- 395. Ganesan AN, Shipp NJ, Brooks AG, et al. Long-term outcomes of catheter ablation of atrial fibrillation: a systematic review and metaanalysis. J Am Heart Assoc 2013;2:e004549.
- **396.** Verma A, Ha ACT, Kirchhof P, et al. The Optimal Anti-Coagulation for Enhanced-Risk Patients Post-Catheter Ablation for Atrial Fibrillation (OCEAN) trial. Am Heart J 2018;197:124-32.
- 397. Michael KA, Redfearn DP, Baranchuk A, et al. Transesophageal echocardiography for the prevention of embolic complications after catheter ablation for atrial fibrillation. J Cardiovasc Electrophysiol 2009;20:1217-22.
- **398.** Voight J, Akkaya M, Somasundaram P, et al. Risk of new-onset atrial fibrillation and stroke after radiofrequency ablation of isolated, typical atrial flutter. Heart Rhythm 2014;11:1884-9.
- 399. Exposito V, Rodriguez-Entem F, Gonzalez-Enriquez S, et al. Stroke and systemic embolism after successful ablation of typical atrial flutter. Clin Cardiol 2016;39:347-51.
- 400. Perez FJ, Schubert CM, Parvez B, et al. Long-term outcomes after catheter ablation of cavo-tricuspid isthmus dependent atrial flutter: a meta-analysis. Circ Arrhythm Electrophysiol 2009;2:393-401.
- 401. Spyropoulos AC, Al-Badri A, Sherwood MW, Douketis JD. Periprocedural management of patients receiving a vitamin K antagonist or a direct oral anticoagulant requiring an elective procedure or surgery. J Thromb Haemost 2016;14:875-85.
- 402. Dunn AS, Spyropoulos AC, Turpie AG. Bridging therapy in patients on long-term oral anticoagulants who require surgery: the Prospective Perioperative Enoxaparin Cohort Trial (PROSPECT). J Thromb Haemost 2007;5:2211-8.
- **403.** Siegal D, Yudin J, Kaatz S, et al. Periprocedural heparin bridging in patients receiving vitamin K antagonists: systematic review and metaanalysis of bleeding and thromboembolic rates. Circulation 2012;126: 1630-9.
- 404. Douketis JD, Spyropoulos AC, Kaatz S, et al. Perioperative bridging anticoagulation in patients with atrial fibrillation. N Engl J Med 2015;373:823-33.
- 405. Spyropoulos AC, Douketis JD. How I treat anticoagulated patients undergoing an elective procedure or surgery. Blood 2012;120:2954-62.

- **406.** Beyer-Westendorf J, Gelbricht V, Forster K, et al. Peri-interventional management of novel oral anticoagulants in daily care: results from the prospective Dresden NOAC registry. Eur Heart J 2014;35:1888-96.
- **407.** Healey JS, Eikelboom J, Douketis J, et al. Periprocedural bleeding and thromboembolic events with dabigatran compared with warfarin: results from the Randomized Evaluation of Long-Term Anticoagulation Therapy (RE-LY) randomized trial. Circulation 2012;126:343-8.
- **408.** Sherwood MW, Douketis JD, Patel MR, et al. Outcomes of temporary interruption of rivaroxaban compared with warfarin in patients with nonvalvular atrial fibrillation: results from the rivaroxaban once daily, oral, direct factor Xa inhibition compared with vitamin K antagonism for prevention of stroke and embolism trial in atrial fibrillation (ROCKET AF). Circulation 2014;129:1850-9.
- 409. Garcia D, Alexander JH, Wallentin L, et al. Management and clinical outcomes in patients treated with apixaban vs warfarin undergoing procedures. Blood 2014;124:3692-8.
- 410. Douketis JD, Spyropoulos AC, Duncan J, et al. Perioperative management of patients with atrial fibrillation receiving a direct oral anticoagulant. JAMA Intern Med 2019;33:551-61.
- 411. Esteve-Pastor MA, Rivera-Caravaca JM, Shantsila A, et al. Assessing bleeding risk in atrial fibrillation patients: comparing a bleeding risk score based only on modifiable bleeding risk factors against the HAS-BLED score. The AMADEUS trial. Thromb Haemost 2017;117: 2261-6.
- **412.** Chao TF, Lip GYH, Lin YJ, et al. Major bleeding and intracranial hemorrhage risk prediction in patients with atrial fibrillation: attention to modifiable bleeding risk factors or use of a bleeding risk stratification score? A nationwide cohort study. Int J Cardiol 2018;254:157-61.
- 413. Guo Y, Zhu H, Chen Y, Lip GYH. Comparing bleeding risk assessment focused on modifiable risk factors only versus validated bleeding risk scores in atrial fibrillation. Am J Med 2018;131:185-92.
- 414. Lai KC, Lam SK, Chu KM, et al. Lansoprazole for the prevention of recurrences of ulcer complications from long-term low-dose aspirin use. N Engl J Med 2002;346:2033-8.
- 415. Yeomans N, Lanas A, Labenz J, et al. Efficacy of esomeprazole (20 mg once daily) for reducing the risk of gastroduodenal ulcers associated with continuous use of low-dose aspirin. Am J Gastroenterol 2008;103: 2465-73.
- 416. Bhatt DL, Cryer BL, Contant CF, et al. Clopidogrel with or without omeprazole in coronary artery disease. N Engl J Med 2010;363: 1909-17.
- 417. Scheiman JM, Devereaux PJ, Herlitz J, et al. Prevention of peptic ulcers with esomeprazole in patients at risk of ulcer development treated with low-dose acetylsalicylic acid: a randomised, controlled trial (OBERON). Heart 2011;97:797-802.
- **418.** Ray WA, Chung CP, Murray KT, et al. Association of oral anticoagulants and proton pump inhibitor cotherapy with hospitalization for upper gastrointestinal tract bleeding. JAMA 2018;320:2221-30.
- 419. Moayyedi P, Eikelboom JW, Bosch J, et al. Pantoprazole to prevent gastroduodenal events in patients receiving rivaroxaban and/or aspirin in a randomized, double-blind, placebo-controlled trial. Gastroenterology 2019;157. 403-12.e405.
- 420. Causada-Calo N, Germini F, Yuan Y, Eikelboom JW, Moayyedi P. Proton-pump inhibitors for the prevention of upper gastrointestinal bleeding in adults receiving antithrombotic therapy. Cochrane Database Syst Rev 2019;8:CD013415. Available at: https://doi.org/10.1002/ 14651858.CD013415.

- 421. Eikelboom JW, Quinlan DJ, van Ryn J, Weitz JI. Idarucizumab: the antidote for reversal of dabigatran. Circulation 2015;132:2412-22.
- **422.** Glund S, Stangier J, Schmohl M, et al. Idarucizumab, a specific antidote for dabigatran: immediate, complete and sustained reversal of dabigatran induced anticoagulation in elderly and renally impaired subjects [abstract 332]. Blood 2014;124:344.
- 423. Glund S, Moschetti V, Norris S, et al. A randomised study in healthy volunteers to investigate the safety, tolerability and pharmacokinetics of idarucizumab, a specific antidote to dabigatran. Thromb Haemost 2015;113:943-51.
- **424.** Glund S, Stangier J, Schmohl M, et al. Safety, tolerability, and efficacy of idarucizumab for the reversal of the anticoagulant effect of dabigatran in healthy male volunteers: a randomised, placebo-controlled, double-blind phase 1 trial. Lancet 2015;386:680-90.
- 425. Pollack CV Jr, Reilly PA, Weitz JI. Dabigatran reversal with idarucizumab. N Engl J Med 2017;377:1690-2.
- 426. Yasaka M, Yokota H, Suzuki M, et al. Idarucizumab for emergency reversal of anticoagulant effects of dabigatran: interim results of a japanese post-marketing surveillance study. Cardiol Ther 2020;9:167-88.
- 427. van der Wall SJ, van Rein N, van den Bemt B, et al. Performance of idarucizumab as antidote of dabigatran in daily clinical practice. Europace 2019;21:414-20.
- 428. Siegal DM, Curnutte JT, Connolly SJ, et al. Andexanet alfa for the reversal of Factor Xa inhibitor activity. N Engl J Med 2015;373: 2413-24.
- 429. Connolly SJ, Crowther M, Eikelboom JW, et al. Full study report of andexanet alfa for bleeding associated with Factor Xa inhibitors. N Engl J Med 2019;380:1326-35.
- 430. FDA. US Food and Drug Administration. May 3, 2018 Accellerated Approval Letter - ANDEXXA. Available at: https://www.fda.gov/media/ 113285/download. Accessed November 11, 2020.
- 431. Kaatz S, Bhansali H, Gibbs J, et al. Reversing factor Xa inhibitors clinical utility of andexanet alfa. J Blood Med 2017;8:141-9.
- **432.** Stevens VM, Trujillo T, Mueller SW, et al. Coagulation Factor Xa (recombinant), inactivated-zhzo (andexanet alfa) hemostatic outcomes and thrombotic event incidence at an academic medical center. Clin Appl Thromb Hemost 2019;25. 1076029619896619.
- **433.** Culbreth SE, Rimsans J, Sylvester K, Pallin DJ, Connors JM. Andexanet alfa-the first 150 days. Am J Hematol 2019;94:E21-4.
- 434. Hylek EM, Go AS, Chang Y, et al. Effect of intensity of oral anticoagulation on stroke severity and mortality in atrial fibrillation. N Engl J Med 2003;349:1019-26.
- 435. O'Donnell M, Oczkowski W, Fang J, et al. Preadmission antithrombotic treatment and stroke severity in patients with atrial fibrillation and acute ischaemic stroke: an observational study. Lancet Neurol 2006;5:749-54.
- 436. Macha K, Marsch A, Siedler G, et al. Cerebral ischemia in patients on direct oral anticoagulants. Stroke 2019;50:873-9.
- 437. Prabhakaran S, Silver AJ, Warrior L, McClenathan B, Lee VH. Misdiagnosis of transient ischemic attacks in the emergency room. Cerebrovasc Dis 2008;26:630-5.
- **438.** Brazzelli M, Shuler K, Quayyum Z, et al. Clinical and imaging services for TIA and minor stroke: results of two surveys of practice across the UK. BMJ Open 2013;3:e003359.
- 439. Cooperman SS, Shah AK, Rajamani K. Amyloid spells. Neurol Clin Pract 2019;9:e17-8.

- 440. Hart RG, Coull BM, Hart D. Early recurrent embolism associated with nonvalvular atrial fibrillation: a retrospective study. Stroke 1983;14: 688-93.
- 441. Paciaroni M, Agnelli G, Micheli S, Caso V. Efficacy and safety of anticoagulant treatment in acute cardioembolic stroke: a meta-analysis of randomized controlled trials. Stroke 2007;38:423-30.
- 442. Paciaroni M, Agnelli G, Corea F, et al. Early hemorrhagic transformation of brain infarction: rate, predictive factors, and influence on clinical outcome: results of a prospective multicenter study. Stroke 2008;39:2249-56.
- 443. D'Amelio M, Terruso V, Famoso G, et al. Early and late mortality of spontaneous hemorrhagic transformation of ischemic stroke. J Stroke Cerebrovasc Dis 2014;23:649-54.
- 444. Li YG, Lip GYH. Anticoagulation resumption after intracerebral hemorrhage. Curr Atheroscler Rep 2018;20:32.
- 445. Paciaroni M, Agnelli G, Falocci N, et al. Early recurrence and cerebral bleeding in patients with acute ischemic stroke and atrial fibrillation: effect of anticoagulation and its timing: the RAF study. Stroke 2015;46: 2175-82.
- 446. Wilson D, Ambler G, Banerjee G, et al. Early versus late anticoagulation for ischaemic stroke associated with atrial fibrillation: multicentre cohort study. J Neurol Neurosurg Psychiatry 2019;90:320-5.
- 447. Hong KS, Kwon SU, Lee SH, et al. Rivaroxaban vs warfarin sodium in the ultra-early period after atrial fibrillation-related mild ischemic stroke: a randomized clinical trial. JAMA Neurol 2017;74:1206-15.
- 448. Butcher KS, Ng K, Sheridan P, et al. Dabigatran treatment of acute noncardioembolic ischemic stroke. Stroke 2020;51:1190-8.
- 449. Palm F, Kraus M, Safer A, et al. Management of oral anticoagulation after cardioembolic stroke and stroke survival data from a population based stroke registry (LuSSt). BMC Neurol 2014;14:199.
- 450. Paciaroni M, Bandini F, Agnelli G, et al. Hemorrhagic transformation in patients with acute ischemic stroke and atrial fibrillation: time to initiation of oral anticoagulant therapy and outcomes. J Am Heart Assoc 2018;7:e010133.
- 451. Murthy SB, Gupta A, Merkler AE, et al. Restarting anticoagulant therapy after intracranial hemorrhage: a systematic review and metaanalysis. Stroke 2017;48:1594-600.
- **452.** Xu Y, Shoamanesh A, Schulman S, et al. Oral anticoagulant re-initiation following intracerebral hemorrhage in non-valvular atrial fibrillation: global survey of the practices of neurologists, neurosurgeons and thrombosis experts. PLoS One 2018;13:e0191137.
- **453.** Stroke Prevention in Atrial Fibrillation Study. Final results. Circulation 1991;84:527-39.
- 454. Blackshear JL, Odell JA. Appendage obliteration to reduce stroke in cardiac surgical patients with atrial fibrillation. Ann Thorac Surg 1996;61:755-9.
- 455. Holmes DR, Reddy VY, Turi ZG, et al. Percutaneous closure of the left atrial appendage versus warfarin therapy for prevention of stroke in patients with atrial fibrillation: a randomised non-inferiority trial. Lancet 2009;374:534-42.
- **456.** Holmes DR Jr, Kar S, Price MJ, et al. Prospective randomized evaluation of the Watchman Left Atrial Appendage Closure device in patients with atrial fibrillation versus long-term warfarin therapy: the PREVAIL trial. J Am Coll Cardiol 2014;64:1-12.

- **457.** Reddy VY, Doshi SK, Kar S, et al. 5-Year outcomes after left atrial appendage closure: from the PREVAIL and PROTECT AF trials. J Am Coll Cardiol 2017;70:2964-75.
- 458. Glikson M, Wolff R, Hindricks G, et al. EHRA/EAPCI expert consensus statement on catheter-based left atrial appendage occlusion an update. EuroIntervention 2020;15:1133-80.
- **459.** Toale C, Fitzmaurice GJ, Eaton D, Lyne J, Redmond KC. Outcomes of left atrial appendage occlusion using the AtriClip device: a systematic review. Interact Cardiovasc Thorac Surg 2019;29:655-62.
- 460. Almahameed ST, Khan M, Zuzek RW, et al. Left atrial appendage exclusion and the risk of thromboembolic events following mitral valve surgery. J Cardiovasc Electrophysiol 2007;18:364-6.
- 461. Yao X, Gersh BJ, Holmes DR Jr, et al. Association of surgical left atrial appendage occlusion with subsequent stroke and mortality among patients undergoing cardiac surgery. JAMA 2018;319:2116-26.
- 462. Whitlock R, Healey J, Vincent J, et al. Rationale and design of the Left Atrial Appendage Occlusion Study (LAAOS) III. Ann Cardiothorac Surg 2014;3:45-54.
- 463. Ratcliffe PJ, Wilcock GK. Cerebrovascular disease in dementia: the importance of atrial fibrillation. Postgrad Med J 1985;61:201-4.
- **464**. De Pedis G, Hedner K, Johansson BW, Steen B. Cardiac arrhythmia in geriatric patients with organic dementia. Compr Gerontol A 1987;1: 115-7.
- 465. Mead GE, Keir S. Association between cognitive impairment and atrial fibrillation: a systematic review. J Stroke Cerebrovasc Dis 2001;10: 35-43.
- 466. Udompanich S, Lip GY, Apostolakis S, Lane DA. Atrial fibrillation as a risk factor for cognitive impairment: a semi-systematic review. QJM 2013;106:795-802.
- 467. Santangeli P, Di Biase L, Bai R, et al. Atrial fibrillation and the risk of incident dementia: a meta-analysis. Heart Rhythm 2012;9:1761-8.
- 468. Kalantarian S, Stern TA, Mansour M, Ruskin JN. Cognitive impairment associated with atrial fibrillation: a meta-analysis. Ann Intern Med 2013;158:338-46.
- 469. Saglietto A, Matta M, Gaita F, et al. Stroke-independent contribution of atrial fibrillation to dementia: a meta-analysis. Open Heart 2019;6: e000984.
- 470. Liao JN, Chao TF, Liu CJ, et al. Risk and prediction of dementia in patients with atrial fibrillation–a nationwide population-based cohort study. Int J Cardiol 2015;199:25-30.
- Alonso A, Arenas de Larriva AP. Atrial fibrillation, cognitive decline and dementia. Eur Cardiol 2016;11:49-53.
- 472. Kim D, Yang PS, Yu HT, et al. Risk of dementia in stroke-free patients diagnosed with atrial fibrillation: data from a population-based cohort. Eur Heart J 2019;40:2313-23.
- 473. Dublin S, Anderson ML, Haneuse SJ, et al. Atrial fibrillation and risk of dementia: a prospective cohort study. J Am Geriatr Soc 2011;59: 1369-75.
- 474. Ihara M, Washida K. Linking atrial fibrillation with Alzheimer's disease: epidemiological, pathological, and mechanistic evidence. J Alzheimers Dis 2018;62:61-72.
- 475. Bunch TJ, Weiss JP, Crandall BG, et al. Atrial fibrillation is independently associated with senile, vascular, and Alzheimer's dementia. Heart Rhythm 2010;7:433-7.

- 476. Ott A, Breteler MM, de Bruyne MC, et al. Atrial fibrillation and dementia in a population-based study. The Rotterdam Study. Stroke 1997;28:316-21.
- 477. de Bruijn RF, Heeringa J, Wolters FJ, et al. Association between atrial fibrillation and dementia in the general population. JAMA Neurol 2015;72:1288-94.
- **478.** Chen LY, Agarwal SK, Norby FL, et al. Persistent but not paroxysmal atrial fibrillation is independently associated with lower cognitive function: ARIC study. J Am Coll Cardiol 2016;67:1379-80.
- 479. Cacciatore F, Testa G, Langellotto A, et al. Role of ventricular rate response on dementia in cognitively impaired elderly subjects with atrial fibrillation: a 10-year study. Dement Geriatr Cogn Disord 2012;34: 143-8.
- 480. Wang S, Fashanu OE, Zhao D, et al. Relation of elevated resting heart rate in mid-life to cognitive decline over 20 years (from the Atherosclerosis Risk in Communities [ARIC] study). Am J Cardiol 2019;123: 334-40.
- 481. Marzona I, O'Donnell M, Teo K, et al. Increased risk of cognitive and functional decline in patients with atrial fibrillation: results of the ONTARGET and TRANSCEND studies. CMAJ 2012;184:E329-36.
- **482.** Poggesi A, Inzitari D, Pantoni L. Atrial fibrillation and cognition: epidemiological data and possible mechanisms. Stroke 2015;46: 3316-21.
- 483. Scarsoglio S, Saglietto A, Anselmino M, Gaita F, Ridolfi L. Alteration of cerebrovascular haemodynamic patterns due to atrial fibrillation: an in silico investigation. J R Soc Interface 2017;14:20170180.
- 484. Bunch TJ, Galenko O, Graves KG, Jacobs V, May HT. Atrial fibrillation and dementia: exploring the association, defining risks and improving outcomes. Arrhythm Electrophysiol Rev 2019;8:8-12.
- 485. Diener HC, Hart RG, Koudstaal PJ, Lane DA, Lip GYH. Atrial fibrillation and cognitive function: JACC review topic of the week. J Am Coll Cardiol 2019;73:612-9.
- 486. de Leeuw FE, de Groot JC, Oudkerk M, et al. Atrial fibrillation and the risk of cerebral white matter lesions. Neurology 2000;54:1795-801.
- 487. Shao IY, Power MC, Mosley T, et al. Association of atrial fibrillation with white matter disease. Stroke 2019;50:989-91.
- 488. Wandell P, Carlsson AC, Sundquist J, Sundquist K. Antihypertensive drugs and relevant cardiovascular pharmacotherapies and the risk of incident dementia in patients with atrial fibrillation. Int J Cardiol 2018;272:149-54.
- 489. Bunch TJ, Crandall BG, Weiss JP, et al. Patients treated with catheter ablation for atrial fibrillation have long-term rates of death, stroke, and dementia similar to patients without atrial fibrillation. J Cardiovasc Electrophysiol 2011;22:839-45.
- 490. Cheng W, Liu W, Li B, Li D. Relationship of anticoagulant therapy with cognitive impairment among patients with atrial fibrillation: a meta-analysis and systematic review. J Cardiovasc Pharmacol 2018;71: 380-7.
- 491. Friberg L, Rosenqvist M. Less dementia with oral anticoagulation in atrial fibrillation. Eur Heart J 2018;39:453-60.
- 492. Friberg L, Andersson T, Rosenqvist M. Less dementia and stroke in low-risk patients with atrial fibrillation taking oral anticoagulation. Eur Heart J 2019;40:2327-35.
- 493. McGuinness B, Craig D, Bullock R, Passmore P. Statins for the prevention of dementia. Cochrane Database Syst Rev 2009;1:CD003160.

- 494. Silva R, Miranda CM, Liu T, Tse G, Roever L. Atrial fibrillation and risk of dementia: epidemiology, mechanisms, and effect of anticoagulation. Front Neurosci 2019;13:18.
- 495. Moffitt P, Lane DA, Park H, O'Connell J, Quinn TJ. Thromboprophylaxis in atrial fibrillation and association with cognitive decline: systematic review. Age Ageing 2016;45:767-75.
- **496.** Dietzel J, Haeusler KG, Endres M. Does atrial fibrillation cause cognitive decline and dementia? Europace 2018;20:408-19.
- 497. Bunch TJ, May HT, Bair TL, et al. Atrial fibrillation patients treated with long-term warfarin anticoagulation have higher rates of all dementia types compared with patients receiving long-term warfarin for other indications. J Am Heart Assoc 2016;5:e003932.
- **498.** Field TS, Weijs B, Curcio A, et al. Incident atrial fibrillation, dementia and the role of anticoagulation: a population-based cohort study. Thromb Haemost 2019;119:981-91.
- 499. Flaker GC, Pogue J, Yusuf S, et al. Cognitive function and anticoagulation control in patients with atrial fibrillation. Circ Cardiovasc Qual Outcomes 2010;3:277-83.
- 500. Jacobs V, Woller SC, Stevens S, et al. Time outside of therapeutic range in atrial fibrillation patients is associated with long-term risk of dementia. Heart Rhythm 2014;11:2206-13.
- 501. Jacobs V, May HT, Bair TL, et al. Long-term population-based cerebral ischemic event and cognitive outcomes of direct oral anticoagulants compared with warfarin among long-term anticoagulated patients for atrial fibrillation. Am J Cardiol 2016;118:210-4.
- 502. Chen N, Lutsey PL, MacLehose RF, et al. Association of oral anticoagulant type with risk of dementia among patients with nonvalvular atrial fibrillation. J Am Heart Assoc 2018;7:e009561.
- 503. Sogaard M, Skjoth F, Jensen M, et al. Nonvitamin K antagonist oral anticoagulants versus warfarin in atrial fibrillation patients and risk of dementia: a nationwide propensity-weighted cohort study. J Am Heart Assoc 2019;8:e011358.
- 504. Jin MN, Kim TH, Kang KW, et al. Atrial fibrillation catheter ablation improves 1-year follow-up cognitive function, especially in patients with impaired cognitive function. Circ Arrhythm Electrophysiol 2019;12: e007197.
- 505. Shaver CM, Chen W, Janz DR, et al. Atrial fibrillation is an independent predictor of mortality in critically ill patients. Crit Care Med 2015;43:2104-11.
- 506. Chen AY, Sokol SS, Kress JP, Lat I. New-onset atrial fibrillation is an independent predictor of mortality in medical intensive care unit patients. Ann Pharmacother 2015;49:523-7.
- 507. Yoshida T, Fujii T, Uchino S, Takinami M. Epidemiology, prevention, and treatment of new-onset atrial fibrillation in critically ill: a systematic review. J Intensive Care 2015;3:19.
- 508. Wetterslev M, Haase N, Hassager C, et al. New-onset atrial fibrillation in adult critically ill patients: a scoping review. Intensive Care Med 2019;45:928-38.
- 509. Chean CS, McAuley D, Gordon A, Welters ID. Current practice in the management of new-onset atrial fibrillation in critically ill patients: a UK-wide survey. PeerJ 2017;5:e3716.
- 510. Scheuermeyer FX, Pourvali R, Rowe BH, et al. Emergency department patients with atrial fibrillation or flutter and an acute underlying medical illness may not benefit from attempts to control rate or rhythm. Ann Emerg Med 2015;65:511-522.e512.

- 511. Stiell IG, Scheuermeyer FX, Vadeboncoeur A, et al. CAEP acute atrial fibrillation/flutter best practices checklist. CJEM 2018;20:334-42.
- 512. Wyse DG, Waldo AL, DiMarco JP, et al. A comparison of rate control and rhythm control in patients with atrial fibrillation. N Engl J Med 2002;347:1825-33.
- 513. Van Gelder IC, Hagens VE, Bosker HA, et al. A comparison of rate control and rhythm control in patients with recurrent persistent atrial fibrillation. N Engl J Med 2002;347:1834-40.
- 514. Roy D, Talajic M, Nattel S, et al. Rhythm control versus rate control for atrial fibrillation and heart failure. N Engl J Med 2008;358:2667-77.
- 515. Kirchhof P, Camm AJ, Goette A, et al. Early rhythm-control therapy in patients with atrial fibrillation. N Engl J Med 2020;383:1305-16.
- 516. Parkash R, Magee K, McMullen M, et al. The Canadian Community Utilization of Stroke Prevention Study in Atrial Fibrillation in the Emergency Department (C-CUSP ED). Ann Emerg Med 2019;73: 382-92.
- 517. Atzema CL, Fang J, Cox JL, et al. Assessment of an algorithm for prescription of oral anticoagulation for patients with atrial fibrillation in emergency departments. JAMA Netw Open 2020;3:e200306.
- 518. Atzema CL, Yu B, Schull MJ, et al. The association of follow-up care with long-term death and subsequent hospitalization in patients with atrial fibrillation who receive emergency care in the province of Ontario. Circ Arrhythm Electrophysiol 2019;12:e006498.
- 519. Hawkins NM, Scheuermeyer FX, Youngson E, et al. Impact of cardiology follow-up care on treatment and outcomes of patients with new atrial fibrillation discharged from the emergency department. Europace 2020;22:695-703.
- 520. Atzema CL, Austin PC. Rate-control with beta-blockers versus calciumchannel blockers in the emergency setting: predictors of medication class choice and associated hospitalization. Acad Emerg Med 2017;24: 1334-48.
- 521. Demircan C, Cikriklar HI, Engindeniz Z, et al. Comparison of the effectiveness of intravenous diltiazem and metoprolol in the management of rapid ventricular rate in atrial fibrillation. Emerg Med J 2005;22:411-4.
- 522. Fromm C, Suau SJ, Cohen V, et al. Diltiazem vs metoprolol in the management of atrial fibrillation or flutter with rapid ventricular rate in the emergency department. J Emerg Med 2015;49:175-82.
- 523. Martindale JL, deSouza IS, Silverberg M, Freedman J, Sinert R. β-Blockers versus calcium channel blockers for acute rate control of atrial fibrillation with rapid ventricular response: a systematic review. Eur J Emerg Med 2015;22:150-4.
- 524. Hines MC, Reed BN, Ivaturi V, et al. Diltiazem versus metoprolol for rate control in atrial fibrillation with rapid ventricular response in the emergency department. Am J Health Syst Pharm 2016;73:2068-76.
- 525. Lip GY, Apostolakis S. Atrial fibrillation (acute onset). BMJ Clin Evid 2014;2014:0210.
- 526. Jandali MB. Safety of intravenous diltiazem in reduced ejection fraction heart failure with rapid atrial fibrillation. Clin Drug Investig 2018;38: 503-8.
- 527. Hirschy R, Ackerbauer KA, Peksa GD, O'Donnell EP, DeMott JM. Metoprolol vs diltiazem in the acute management of atrial fibrillation in patients with heart failure with reduced ejection fraction. Am J Emerg Med 2019;37:80-4.
- 528. Goldenberg IF, Lewis WR, Dias VC, Heywood JT, Pedersen WR. Intravenous diltiazem for the treatment of patients with atrial fibrillation

or flutter and moderate to severe congestive heart failure. Am J Cardiol 1994;74:884-9.

- 529. Stratton T, Nasser L. BET 1: lenient or strict rate control for atrial fibrillation. Emerg Med J 2018;35:765-8.
- 530. Dorian P, Angaran P. β-Blockers and atrial fibrillation: hypertension and other medical conditions influencing their use. Can J Cardiol 2014;30:S38-41.
- 531. Michael JA, Stiell IG, Agarwal S, Mandavia DP. Cardioversion of paroxysmal atrial fibrillation in the emergency department. Ann Emerg Med 1999;33:379-87.
- 532. Slavik RS, Tisdale JE, Borzak S. Pharmacologic conversion of atrial fibrillation: a systematic review of available evidence. Prog Cardiovasc Dis 2001;44:121-52.
- 533. Burton JH, Vinson DR, Drummond K, et al. Electrical cardioversion of emergency department patients with atrial fibrillation. Ann Emerg Med 2004;44:20-30.
- 534. Coll-Vinent B, Fuenzalida C, Garcia A, Martin A, Miro O. Management of acute atrial fibrillation in the emergency department: a systematic review of recent studies. Eur J Emerg Med 2013;20:151-9.
- 535. Stiell IG, Clement CM, Rowe BH, et al. Outcomes for emergency department patients with recent-onset atrial fibrillation and flutter treated in Canadian hospitals. Ann Emerg Med 2017;69:562-571.e562.
- 536. Madrid AH, Moro C, Marin-Huerta E, et al. Comparison of flecainide and procainamide in cardioversion of atrial fibrillation. Eur Heart J 1993;14:1127-31.
- 537. Kochiadakis GE, Igoumenidis NE, Hamilos ME, et al. A comparative study of the efficacy and safety of procainamide versus propafenone versus amiodarone for the conversion of recent-onset atrial fibrillation. Am J Cardiol 2007;99:1721-5.
- 538. Stiell IG, Sivilotti MLA, Taljaard M, et al. Electrical versus pharmacological cardioversion for emergency department patients with acute atrial fibrillation (RAFF2): a partial factorial randomised trial. Lancet 2020;395:339-49.
- 539. Priori SG, Wilde AA, Horie M, et al. HRS/EHRA/APHRS expert consensus statement on the diagnosis and management of patients with inherited primary arrhythmia syndromes: document endorsed by HRS, EHRA, and APHRS in May 2013 and by ACCF, AHA, PACES, and AEPC in June 2013. Heart Rhythm 2013;10:1932-63.
- 540. Stambler BS, Wood MA, Ellenbogen KA, et al. Efficacy and safety of repeated intravenous doses of ibutilide for rapid conversion of atrial flutter or fibrillation. Ibutilide Repeat Dose Study Investigators. Circulation 1996;94:1613-21.
- 541. Volgman AS, Carberry PA, Stambler B, et al. Conversion efficacy and safety of intravenous ibutilide compared with intravenous procainamide in patients with atrial flutter or fibrillation. J Am Coll Cardiol 1998;31: 1414-9.
- 542. Abi-Mansour P, Carberry PA, McCowan RJ, et al. Conversion efficacy and safety of repeated doses of ibutilide in patients with atrial flutter and atrial fibrillation. Study Investigators. Am Heart J 1998;136:632-42.
- 543. Reisinger J, Gatterer E, Lang W, et al. Flecainide versus ibutilide for immediate cardioversion of atrial fibrillation of recent onset. Eur Heart J 2004;25:1318-24.
- 544. Simon A, Niederdoeckl J, Skyllouriotis E, et al. Vernakalant is superior to ibutilide for achieving sinus rhythm in patients with recent-onset atrial fibrillation: a randomized controlled trial at the emergency department. Europace 2017;19:233-40.

- 545. Vogiatzis I, Papavasiliou E, Dapcevitch I, Pittas S, Koulouris E. Vernakalant versus ibutilide for immediate conversion of recent-onset atrial fibrillation. Hippokratia 2017;21:67-73.
- 546. Vinson DR, Lugovskaya N, Warton EM, et al. Ibutilide effectiveness and safety in the cardioversion of atrial fibrillation and flutter in the community emergency department. Ann Emerg Med 2018;71:96-108. e102.
- 547. Zhang N, Guo JH, Zhang H, et al. Comparison of intravenous ibutilide vs propafenone for rapid termination of recent onset atrial fibrillation. Int J Clin Pract 2005;59:1395-400.
- 548. Kafkas NV, Patsilinakos SP, Mertzanos GA, et al. Conversion efficacy of intravenous ibutilide compared with intravenous amiodarone in patients with recent-onset atrial fibrillation and atrial flutter. Int J Cardiol 2007;118:321-5.
- 549. Tercius AJ, Kluger J, Coleman CI, White CM. Intravenous magnesium sulfate enhances the ability of intravenous ibutilide to successfully convert atrial fibrillation or flutter. Pacing Clin Electrophysiol 2007;30: 1331-5.
- 550. Patsilinakos S, Christou A, Kafkas N, et al. Effect of high doses of magnesium on converting ibutilide to a safe and more effective agent. Am J Cardiol 2010;106:673-6.
- 551. Stiell IG, Dickinson G, Butterfield NN, et al. Vernakalant hydrochloride: a novel atrial-selective agent for the cardioversion of recent-onset atrial fibrillation in the emergency department. Acad Emerg Med 2010;17:1175-82.
- 552. Camm AJ, Toft E, Torp-Pedersen C, et al. Efficacy and safety of vernakalant in patients with atrial flutter: a randomized, double-blind, placebo-controlled trial. Europace 2012;14:804-9.
- 553. McIntyre WF, Healey JS, Bhatnagar AK, et al. Vernakalant for cardioversion of recent-onset atrial fibrillation: a systematic review and meta-analysis. Europace 2019;21:1159-66.
- 554. Stiell IG, Roos JS, Kavanagh KM, Dickinson G. A multicenter, openlabel study of vernakalant for the conversion of atrial fibrillation to sinus rhythm. Am Heart J 2010;159:1095-101.
- 555. Pratt CM, Roy D, Torp-Pedersen C, et al. Usefulness of vernakalant hydrochloride injection for rapid conversion of atrial fibrillation. Am J Cardiol 2010;106:1277-83.
- 556. Roy D, Pratt CM, Torp-Pedersen C, et al. Vernakalant hydrochloride for rapid conversion of atrial fibrillation: a phase 3, randomized, placebo-controlled trial. Circulation 2008;117:1518-25.
- 557. Khan IA, Mehta NJ, Gowda RM. Amiodarone for pharmacological cardioversion of recent-onset atrial fibrillation. Int J Cardiol 2003;89: 239-48.
- 558. Alp NJ, Bell JA, Shahi M. Randomised double blind trial of oral versus intravenous flecainide for the cardioversion of acute atrial fibrillation. Heart 2000;84:37-40.
- 559. Boriani G, Capucci A, Lenzi T, Sanguinetti M, Magnani B. Propafenone for conversion of recent-onset atrial fibrillation. A controlled comparison between oral loading dose and intravenous administration. Chest 1995;108:355-8.
- 560. Khan IA. Single oral loading dose of propafenone for pharmacological cardioversion of recent-onset atrial fibrillation. J Am Coll Cardiol 2001;37:542-7.
- 561. Alboni P, Botto GL, Baldi N, et al. Outpatient treatment of recentonset atrial fibrillation with the "pill-in-the-pocket" approach. N Engl J Med 2004;351:2384-91.

- 562. Saborido CM, Hockenhull J, Bagust A, et al. Systematic review and cost-effectiveness evaluation of 'pill-in-the-pocket' strategy for paroxysmal atrial fibrillation compared to episodic in-hospital treatment or continuous antiarrhythmic drug therapy. Health Technol Assess 2010;14(iii-iv):1-75.
- 563. Andrade JG, MacGillivray J, Macle L, et al. Clinical effectiveness of a systematic "pill-in-the-pocket" approach for the management of paroxysmal atrial fibrillation. Heart Rhythm 2018;15:9-16.
- 564. Jacoby JL, Cesta M, Heller MB, Salen P, Reed J. Synchronized emergency department cardioversion of atrial dysrhythmias saves time, money and resources. J Emerg Med 2005;28:27-30.
- 565. Xavier Scheuermeyer F, Grafstein E, Stenstrom R, et al. Thirty-day outcomes of emergency department patients undergoing electrical cardioversion for atrial fibrillation or flutter. Acad Emerg Med 2010;17: 408-15.
- 566. Gronberg T, Nuotio I, Nikkinen M, et al. Arrhythmic complications after electrical cardioversion of acute atrial fibrillation: the FinCV study. Europace 2013;15:1432-5.
- 567. Bonfanti L, Annovi A, Sanchis-Gomar F, et al. Effectiveness and safety of electrical cardioversion for acute-onset atrial fibrillation in the emergency department: a real-world 10-year single center experience. Clin Exp Emerg Med 2019;6:64-9.
- 568. Bellone A, Etteri M, Vettorello M, et al. Cardioversion of acute atrial fibrillation in the emergency department: a prospective randomised trial. Emerg Med J 2012;29:188-91.
- 569. Ballard DW, Reed ME, Singh N, et al. Emergency department management of atrial fibrillation and flutter and patient quality of life at one month postvisit. Ann Emerg Med 2015;66:646-54.
- 570. Inacio JF, da Rosa Mdos S, Shah J, et al. Monophasic and biphasic shock for transthoracic conversion of atrial fibrillation: systematic review and network meta-analysis. Resuscitation 2016;100:66-75.
- 571. Kirkland S, Stiell I, AlShawabkeh T, et al. The efficacy of pad placement for electrical cardioversion of atrial fibrillation/flutter: a systematic review. Acad Emerg Med 2014;21:717-26.
- 572. Ramirez FD, Fiset SL, Cleland MJ, et al. Effect of applying force to selfadhesive electrodes on transthoracic impedance: implications for electrical cardioversion. Pacing Clin Electrophysiol 2016;39:1141-7.
- 573. Sadek MM, Chaugai V, Cleland MJ, et al. Association between transthoracic impedance and electrical cardioversion success with biphasic defibrillators: an analysis of 1055 shocks for atrial fibrillation and flutter. Clin Cardiol 2018;41:666-70.
- 574. Oral H, Souza JJ, Michaud GF, et al. Facilitating transthoracic cardioversion of atrial fibrillation with ibutilide pretreatment. N Engl J Med 1999;340:1849-54.
- 575. Wyse DG. Therapeutic considerations in applying rate control therapy for atrial fibrillation. J Cardiovasc Pharmacol 2008;52:11-7.
- 576. Scheuermeyer FX, Grafstein E, Stenstrom R, et al. Safety and efficiency of calcium channel blockers versus beta-blockers for rate control in patients with atrial fibrillation and no acute underlying medical illness. Acad Emerg Med 2013;20:222-30.
- 577. Segal JB, McNamara RL, Miller MR, et al. The evidence regarding the drugs used for ventricular rate control. J Fam Pract 2000;49:47-59.
- 578. Ahmad K, Dorian P. Rate control in atrial fibrillation: looking beyond the average heart rate. Curr Opin Cardiol 2006;21:88-93.
- 579. Groenveld HF, Crijns HJ, Van den Berg MP, et al. The effect of rate control on quality of life in patients with permanent atrial fibrillation:

data from the RACE II (Rate Control Efficacy in Permanent Atrial Fibrillation II) study. J Am Coll Cardiol 2011;58:1795-803.

- 580. Ulimoen SR, Enger S, Pripp AH, et al. Calcium channel blockers improve exercise capacity and reduce N-terminal pro-B-type natriuretic peptide levels compared with beta-blockers in patients with permanent atrial fibrillation. Eur Heart J 2014;35:517-24.
- 581. Ulimoen SR, Enger S, Carlson J, et al. Comparison of four single-drug regimens on ventricular rate and arrhythmia-related symptoms in patients with permanent atrial fibrillation. Am J Cardiol 2013;111: 225-30.
- 582. Aguilar M, Nattel S. Taking the pulse of atrial fibrillation: a practical approach to rate control. Can J Cardiol 2018;34:1526-30.
- 583. McAlister FA, Wiebe N, Ezekowitz JA, Leung AA, Armstrong PW. Meta-analysis: beta-blocker dose, heart rate reduction, and death in patients with heart failure. Ann Intern Med 2009;150:784-94.
- 584. Cullington D, Goode KM, Zhang J, Cleland JG, Clark AL. Is heart rate important for patients with heart failure in atrial fibrillation? JACC Heart Fail 2014;2:213-20.
- 585. Laskey WK, Alomari I, Cox M, et al. Heart rate at hospital discharge in patients with heart failure is associated with mortality and rehospitalization. J Am Heart Assoc 2015;4:e001626.
- 586. Kotecha D, Flather MD, Altman DG, et al. Heart rate and rhythm and the benefit of beta-blockers in patients with heart failure. J Am Coll Cardiol 2017;69:2885-96.
- 587. Cadrin-Tourigny J, Shohoudi A, Roy D, et al. Decreased mortality with beta-blockers in patients with heart failure and coexisting atrial fibrillation: an AF-CHF substudy. JACC Heart Fail 2017;5:99-106.
- 588. Head SJ, Milojevic M, Daemen J, et al. Mortality after coronary artery bypass grafting versus percutaneous coronary intervention with stenting for coronary artery disease: a pooled analysis of individual patient data. Lancet 2018;391:939-48.
- 589. Gjesdal K, Feyzi J, Olsson SB. Digitalis: a dangerous drug in atrial fibrillation? An analysis of the SPORTIF III and V data. Heart 2008;94: 191-6.
- 590. Shah M, Avgil Tsadok M, Jackevicius CA, et al. Relation of digoxin use in atrial fibrillation and the risk of all-cause mortality in patients ≥ 65 years of age with versus without heart failure. Am J Cardiol 2014;114: 401-6.
- 591. Eisen A, Ruff CT, Braunwald E, et al. Digoxin use and subsequent clinical outcomes in patients with atrial fibrillation with or without heart failure in the ENGAGE AF-TIMI 48 trial. J Am Heart Assoc 2017;6: e006035.
- 592. Lopes RD, Rordorf R, De Ferrari GM, et al. Digoxin and mortality in patients with atrial fibrillation. J Am Coll Cardiol 2018;71:1063-74.
- 593. Sethi NJ, Safi S, Feinberg J, et al. Digoxin versus placebo, no intervention, or other medical interventions for atrial fibrillation and atrial flutter: a protocol for a systematic review with meta-analysis and trial sequential analysis. Syst Rev 2017;6:71.
- 594. Ahmed A, Pitt B, Rahimtoola SH, et al. Effects of digoxin at low serum concentrations on mortality and hospitalization in heart failure: a propensity-matched study of the DIG trial. Int J Cardiol 2008;123: 138-46.
- 595. Hofmann R, Wimmer G, Leisch F. Intravenous amiodarone bolus immediately controls heart rate in patients with atrial fibrillation accompanied by severe congestive heart failure. Heart 2000;84:635.

- 596. Tse HF, Lam YM, Lau CP, Cheung BM, Kumana CR. Comparison of digoxin versus low-dose amiodarone for ventricular rate control in patients with chronic atrial fibrillation. Clin Exp Pharmacol Physiol 2001;28:446-50.
- 597. Connolly SJ, Camm AJ, Halperin JL, et al. Dronedarone in high-risk permanent atrial fibrillation. N Engl J Med 2011;365:2268-76.
- 598. Van Gelder IC, Groenveld HF, Crijns HJ, et al. Lenient versus strict rate control in patients with atrial fibrillation. N Engl J Med 2010;362: 1363-73.
- 599. Cooper HA, Bloomfield DA, Bush DE, et al. Relation between achieved heart rate and outcomes in patients with atrial fibrillation (from the Atrial Fibrillation Follow-up Investigation of Rhythm Management [AFFIRM] study). Am J Cardiol 2004;93:1247-53.
- 600. Wyse DG. Lenient versus strict rate control in atrial fibrillation some devils in the details. J Am Coll Cardiol 2011;58:950-2.
- 601. Steinberg BA, Kim S, Thomas L, et al. Increased heart rate is associated with higher mortality in patients with atrial fibrillation (AF): results from the Outcomes Registry for Better Informed Treatment of AF (ORBIT-AF). J Am Heart Assoc 2015;4:e002031.
- 602. Andrade JG, Roy D, Wyse DG, et al. Heart rate and adverse outcomes in patients with atrial fibrillation: a combined AFFIRM and AF-CHF substudy. Heart Rhythm 2016;13:54-61.
- 603. Lau DH, Thiyagarajah A, Willems S, et al. Device therapy for rate control: pacing, resynchronisation and AV node ablation. Heart Lung Circ 2017;26:934-40.
- 604. Kay GN, Ellenbogen KA, Giudici M, et al. The Ablate and Pace Trial: a prospective study of catheter ablation of the AV conduction system and permanent pacemaker implantation for treatment of atrial fibrillation. APT Investigators. J Interv Card Electrophysiol 1998;2:121-35.
- 605. Chatterjee NA, Upadhyay GA, Ellenbogen KA, et al. Atrioventricular nodal ablation in atrial fibrillation: a meta-analysis and systematic review. Circ Arrhythm Electrophysiol 2012;5:68-76.
- **606.** Natale A, Zimerman L, Tomassoni G, et al. AV node ablation and pacemaker implantation after withdrawal of effective rate-control medications for chronic atrial fibrillation: effect on quality of life and exercise performance. Pacing Clin Electrophysiol 1999;22:1634-9.
- 607. Chatterjee NA, Upadhyay GA, Ellenbogen KA, Hayes DL, Singh JP. Atrioventricular nodal ablation in atrial fibrillation: a meta-analysis of biventricular vs right ventricular pacing mode. Eur J Heart Fail 2012;14:661-7.
- 608. Vijayaraman P, Subzposh FA, Naperkowski A. Atrioventricular node ablation and His bundle pacing. Europace 2017;19:iv10-6.
- **609.** Yarlagadda B, Turagam MK, Dar T, et al. Safety and feasibility of leadless pacemaker in patients undergoing atrioventricular node ablation for atrial fibrillation. Heart Rhythm 2018;15:994-1000.
- **610.** Okabe T, El-Chami MF, Lloyd MS, et al. Leadless pacemaker implantation and concurrent atrioventricular junction ablation in patients with atrial fibrillation. Pacing Clin Electrophysiol 2018;41:504-10.
- 611. Curtis AB, Worley SJ, Adamson PB, et al. Biventricular pacing for atrioventricular block and systolic dysfunction. N Engl J Med 2013;368:1585-93.
- **612.** Dickstein K, Bogale N, Priori S, et al. The European cardiac resynchronization therapy survey. Eur Heart J 2009;30:2450-60.
- 613. Hayes DL, Boehmer JP, Day JD, et al. Cardiac resynchronization therapy and the relationship of percent biventricular pacing to symptoms and survival. Heart Rhythm 2011;8:1469-75.

Andrade et al. 2020 CCS/CHRS Guidelines on Management of AF

- **614.** Healey JS, Hohnloser SH, Exner DV, et al. Cardiac resynchronization therapy in patients with permanent atrial fibrillation: results from the Resynchronization for Ambulatory Heart Failure Trial (RAFT). Circ Heart Fail 2012;5:566-70.
- 615. Ganesan AN, Brooks AG, Roberts-Thomson KC, et al. Role of AV nodal ablation in cardiac resynchronization in patients with coexistent atrial fibrillation and heart failure a systematic review. J Am Coll Cardiol 2012;59:719-26.
- 616. Gasparini M, Kloppe A, Lunati M, et al. Atrioventricular junction ablation in patients with atrial fibrillation treated with cardiac resynchronization therapy: positive impact on ventricular arrhythmias, implantable cardioverter-defibrillator therapies and hospitalizations. Eur J Heart Fail 2018;20:1472-81.
- 617. Freemantle N, Lafuente-Lafuente C, Mitchell S, Eckert L, Reynolds M. Mixed treatment comparison of dronedarone, amiodarone, sotalol, flecainide, and propafenone, for the management of atrial fibrillation. Europace 2011;13:329-45.
- 618. Zimetbaum P. Antiarrhythmic drug therapy for atrial fibrillation. Circulation 2012;125:381-9.
- 619. Lafuente-Lafuente C, Valembois L, Bergmann JF, Belmin J. Antiarrhythmics for maintaining sinus rhythm after cardioversion of atrial fibrillation. Cochrane Database Syst Rev 2015;3:CD005049.
- 620. Hohnloser SH, Crijns HJ, van Eickels M, et al. Effect of dronedarone on cardiovascular events in atrial fibrillation. N Engl J Med 2009;360: 668-78.
- 621. Connolly SJ, Crijns HJ, Torp-Pedersen C, et al. Analysis of stroke in ATHENA: a placebo-controlled, double-blind, parallel-arm trial to assess the efficacy of dronedarone 400 mg BID for the prevention of cardiovascular hospitalization or death from any cause in patients with atrial fibrillation/atrial flutter. Circulation 2009;120:1174-80.
- 622. Kirchhof P, Andresen D, Bosch R, et al. Short-term versus long-term antiarrhythmic drug treatment after cardioversion of atrial fibrillation (Flec-SL): a prospective, randomised, open-label, blinded end point assessment trial. Lancet 2012;380:238-46.
- 623. Echt DS, Liebson PR, Mitchell LB, et al. Mortality and morbidity in patients receiving encainide, flecainide, or placebo. The Cardiac Arrhythmia Suppression Trial. N Engl J Med 1991;324:781-8.
- 624. Aliot E, Capucci A, Crijns HJ, Goette A, Tamargo J. Twenty-five years in the making: flecainide is safe and effective for the management of atrial fibrillation. Europace 2011;13:161-73.
- 625. Goldschlager N, Epstein AE, Naccarelli GV, et al. A practical guide for clinicians who treat patients with amiodarone: 2007. Heart Rhythm 2007;4:1250-9.
- 626. Wegener FT, Ehrlich JR, Hohnloser SH. Dronedarone: an emerging agent with rhythm- and rate-controlling effects. J Cardiovasc Electrophysiol 2006;17(suppl 2):S17-20.
- 627. Andrade JG, Champagne J, Dubuc M, et al. Cryoballoon or radiofrequency ablation for atrial fibrillation assessed by continuous monitoring: a randomized clinical trial. Circulation 2019;140:1779-88.
- **628.** Stewart MT, Haines DE, Verma A, et al. Intracardiac pulsed field ablation: proof of feasibility in a chronic porcine model. Heart Rhythm 2019;16:754-64.
- 629. Haissaguerre M, Jais P, Shah DC, et al. Spontaneous initiation of atrial fibrillation by ectopic beats originating in the pulmonary veins. N Engl J Med 1998;339:659-66.
- 630. Verma A, Jiang CY, Betts TR, et al. Approaches to catheter ablation for persistent atrial fibrillation. N Engl J Med 2015;372:1812-22.

- 631. Andrade JG, Macle L, Verma A, et al. Quality of life and health care utilization in the CIRCA-DOSE study. JACC Clin Electrophysiol 2020;6:935-44.
- **632.** Deshmukh A, Patel NJ, Pant S, et al. In-hospital complications associated with catheter ablation of atrial fibrillation in the United States between 2000 and 2010: analysis of 93 801 procedures. Circulation 2013;128:2104-12.
- 633. Cheng EP, Liu CF, Yeo I, et al. Risk of mortality following catheter ablation of atrial fibrillation. J Am Coll Cardiol 2019;74:2254-64.
- 634. Khan SU, Rahman H, Talluri S, Kaluski E. The clinical benefits and mortality reduction associated with catheter ablation in subjects with atrial fibrillation: a systematic review and meta-analysis. JACC Clin Electrophysiol 2018;4:626-35.
- 635. Asad ZUA, Yousif A, Khan MS, Al-Khatib SM, Stavrakis S. Catheter ablation versus medical therapy for atrial fibrillation: a systematic review and meta-analysis of randomized controlled trials. Circ Arrhythm Electrophysiol 2019;12:e007414.
- 636. Blomstrom-Lundqvist C, Gizurarson S, Schwieler J, et al. Effect of Catheter Ablation vs Antiarrhythmic Medication on Quality of Life in Patients With Atrial Fibrillation: the CAPTAF randomized clinical trial. JAMA 2019;321:1059-68.
- 637. Bunch TJ, May HT, Bair TL, et al. Increasing time between first diagnosis of atrial fibrillation and catheter ablation adversely affects long-term outcomes. Heart Rhythm 2013;10:1257-62.
- 638. Wazni OM, Marrouche NF, Martin DO, et al. Radiofrequency ablation vs antiarrhythmic drugs as first-line treatment of symptomatic atrial fibrillation: a randomized trial. JAMA 2005;293:2634-40.
- 639. Cosedis Nielsen J, Johannessen A, Raatikainen P, et al. Radiofrequency ablation as initial therapy in paroxysmal atrial fibrillation. N Engl J Med 2012;367:1587-95.
- 640. Morillo CA, Verma A, Connolly SJ, et al. Radiofrequency ablation vs antiarrhythmic drugs as first-line treatment of paroxysmal atrial fibrillation (RAAFT-2): a randomized trial. JAMA 2014;311:692-700.
- 641. Andrade JG, Wells GA, Deyell MW, et al; for the EARLY-AF Investigators. Cryoablation or drug therapy for initial treatment of atrial fibrillation [epub ahead of print]. N Engl J Med, https://doi.org/10.1056/NEJM0a2029980.
- 642. Anselmino M, Matta M, D'Ascenzo F, et al. Catheter ablation of atrial fibrillation in patients with left ventricular systolic dysfunction: a systematic review and meta-analysis. Circ Arrhythm Electrophysiol 2014;7: 1011-8.
- 643. Malhi N, Hawkins NM, Andrade JG, Krahn AD, Deyell MW. Catheter ablation of atrial fibrillation in heart failure with reduced ejection fraction. J Cardiovasc Electrophysiol 2018;29:1049-58.
- 644. Di Biase L, Mohanty P, Mohanty S, et al. Ablation Versus Amiodarone for Treatment of Persistent Atrial Fibrillation in Patients With Congestive Heart Failure and an Implanted Device: results from the AATAC multicenter randomized trial. Circulation 2016;133:1637-44.
- 645. Marrouche NF, Brachmann J, Andresen D, et al. Catheter ablation for atrial fibrillation with heart failure. N Engl J Med 2018;378:417-27.
- 646. Samuel M, Abrahamowicz M, Joza J, et al. Long-term effectiveness of catheter ablation in patients with atrial fibrillation and heart failure. Europace 2020;22:739-47.
- 647. Khaykin Y, Marrouche NF, Martin DO, et al. Pulmonary vein isolation for atrial fibrillation in patients with symptomatic sinus bradycardia or pauses. J Cardiovasc Electrophysiol 2004;15:784-9.

- **648.** Natale A, Newby KH, Pisano E, et al. Prospective randomized comparison of antiarrhythmic therapy versus first-line radiofrequency ablation in patients with atrial flutter. J Am Coll Cardiol 2000;35: 1898-904.
- 649. Spector P, Reynolds MR, Calkins H, et al. Meta-analysis of ablation of atrial flutter and supraventricular tachycardia. Am J Cardiol 2009;104: 671-7.
- **650.** Reithmann C, Hoffmann E, Spitzlberger G, et al. Catheter ablation of atrial flutter due to amiodarone therapy for paroxysmal atrial fibrillation. Eur Heart J 2000;21:565-72.
- 651. Budera P, Straka Z, Osmancik P, et al. Comparison of cardiac surgery with left atrial surgical ablation vs cardiac surgery without atrial ablation in patients with coronary and/or valvular heart disease plus atrial fibrillation: final results of the PRAGUE-12 randomized multicentre study. Eur Heart J 2012;33:2644-52.
- 652. Gillinov AM, Gelijns AC, Parides MK, et al. Surgical ablation of atrial fibrillation during mitral-valve surgery. N Engl J Med 2015;372: 1399-409.
- **653.** Phan K, Xie A, La Meir M, Black D, Yan TD. Surgical ablation for treatment of atrial fibrillation in cardiac surgery: a cumulative metaanalysis of randomised controlled trials. Heart 2014;100:722-30.
- **654.** Jessurun ER, van Hemel NM, Defauw JJ, et al. A randomized study of combining maze surgery for atrial fibrillation with mitral valve surgery. J Cardiovasc Surg (Torino) 2003;44:9-18.
- 655. Gillinov AM, Bhavani S, Blackstone EH, et al. Surgery for permanent atrial fibrillation: impact of patient factors and lesion set. Ann Thorac Surg 2006;82:502-13 [discussion: 513-4].
- 656. von Oppell UO, Masani N, O'Callaghan P, et al. Mitral valve surgery plus concomitant atrial fibrillation ablation is superior to mitral valve surgery alone with an intensive rhythm control strategy. Eur J Cardiothorac Surg 2009;35:641-50.
- 657. Ad N, Damiano RJ Jr, Badhwar V, et al. Expert consensus guidelines: examining surgical ablation for atrial fibrillation. J Thorac Cardiovasc Surg 2017;153:1330-1354.e1331.
- 658. Phan K, Xie A, Di Eusanio M, Yan TD. A meta-analysis of minimally invasive versus conventional sternotomy for aortic valve replacement. Ann Thorac Surg 2014;98:1499-511.
- 659. Andrade JG, Deyell MW, Lee AYK, Macle L. Sex differences in atrial fibrillation. Can J Cardiol 2018;34:429-36.
- 660. Liu XK, Jahangir A, Terzic A, et al. Age- and sex-related atrial electrophysiologic and structural changes. Am J Cardiol 2004;94:373-5.
- 661. Nattel S. Atrial fibrillation and body composition: is it fat or lean that ultimately determines the risk? J Am Coll Cardiol 2017;69:2498-501.
- **662.** Cochet H, Mouries A, Nivet H, et al. Age, atrial fibrillation, and structural heart disease are the main determinants of left atrial fibrosis detected by delayed-enhanced magnetic resonance imaging in a general cardiology population. J Cardiovasc Electrophysiol 2015;26:484-92.
- 663. Tsai WC, Chen YC, Lin YK, Chen SA, Chen YJ. Sex differences in the electrophysiological characteristics of pulmonary veins and left atrium and their clinical implication in atrial fibrillation. Circ Arrhythm Electrophysiol 2011;4:550-9.
- 664. Magnani JW, Moser CB, Murabito JM, et al. Association of sex hormones, aging, and atrial fibrillation in men: the Framingham Heart Study. Circ Arrhythm Electrophysiol 2014;7:307-12.

- 665. Nakamura H, Kurokawa J, Bai CX, et al. Progesterone regulates cardiac repolarization through a nongenomic pathway: an in vitro patch-clamp and computational modeling study. Circulation 2007;116:2913-22.
- 666. Perez MV, Wang PJ, Larson JC, et al. Effects of postmenopausal hormone therapy on incident atrial fibrillation: the Women's Health Initiative randomized controlled trials. Circ Arrhythm Electrophysiol 2012;5:1108-16.
- 667. Reynolds MR, Lavelle T, Essebag V, Cohen DJ, Zimetbaum P. Influence of age, sex, and atrial fibrillation recurrence on quality of life outcomes in a population of patients with new-onset atrial fibrillation: the Fibrillation Registry Assessing Costs, Therapies, Adverse events and Lifestyle (FRACTAL) study. Am Heart J 2006;152:1097-103.
- 668. Yao RJR, Macle L, Deyell MW, et al. Impact of female sex on clinical presentation and ablation outcomes in the CIRCA-DOSE study. JACC Clin Electrophysiol 2020;6:945-54.
- 669. Ball J, Carrington MJ, Wood KA, Stewart S; SAFETY Investigators. Women versus men with chronic atrial fibrillation: insights from the Standard versus Atrial Fibrillation spEcific managemenT studY (SAFETY). PLoS One 2013;8:e65795.
- 670. Emdin CA, Wong CX, Hsiao AJ, et al. Atrial fibrillation as risk factor for cardiovascular disease and death in women compared with men: systematic review and meta-analysis of cohort studies. BMJ 2016;532: h7013.
- 671. Appelros P, Stegmayr B, Terent A. A review on sex differences in stroke treatment and outcome. Acta Neurol Scand 2010;121:359-69.
- 672. Garcia D, Regan S, Crowther M, Hughes RA, Hylek EM. Warfarin maintenance dosing patterns in clinical practice: implications for safer anticoagulation in the elderly population. Chest 2005;127:2049-56.
- 673. Avgil Tsadok M, Jackevicius CA, Rahme E, Humphries KH, Pilote L. Sex differences in dabigatran use, safety, and effectiveness in a population-based cohort of patients with atrial fibrillation. Circ Cardiovasc Qual Outcomes 2015;8:593-9.
- 674. Olesen JB, Sorensen R, Hansen ML, et al. Non-vitamin K antagonist oral anticoagulation agents in anticoagulant naive atrial fibrillation patients: Danish nationwide descriptive data 2011-2013. Europace 2015;17:187-93.
- 675. Hsu JC, Maddox TM, Kennedy K, et al. Aspirin instead of oral anticoagulant prescription in atrial fibrillation patients at risk for stroke. J Am Coll Cardiol 2016;67:2913-23.
- 676. Avgil Tsadok M, Jackevicius CA, Rahme E, et al. Sex differences in stroke risk among older patients with recently diagnosed atrial fibrillation. JAMA 2012;307:1952-8.
- 677. Pancholy SB, Sharma PS, Pancholy DS, et al. Meta-analysis of gender differences in residual stroke risk and major bleeding in patients with nonvalvular atrial fibrillation treated with oral anticoagulants. Am J Cardiol 2014;113:485-90.
- 678. Dagres N, Nieuwlaat R, Vardas PE, et al. Gender-related differences in presentation, treatment, and outcome of patients with atrial fibrillation in Europe: a report from the Euro Heart Survey on Atrial Fibrillation. J Am Coll Cardiol 2007;49:572-7.
- 679. Lip GY, Laroche C, Boriani G, et al. Sex-related differences in presentation, treatment, and outcome of patients with atrial fibrillation in Europe: a report from the Euro Observational Research Programme Pilot survey on Atrial Fibrillation. Europace 2015;17:24-31.
- 680. Patel N, Deshmukh A, Thakkar B, et al. Gender, race, and health insurance status in patients undergoing catheter ablation for atrial fibrillation. Am J Cardiol 2016;117:1117-26.

- **681.** Schnabel RB, Pecen L, Ojeda FM, et al. Gender differences in clinical presentation and 1-year outcomes in atrial fibrillation. Heart 2017;103: 1024-30.
- **682.** Forleo GB, Tondo C, De Luca L, et al. Gender-related differences in catheter ablation of atrial fibrillation. Europace 2007;9:613-20.
- 683. Takigawa M, Kuwahara T, Takahashi A, et al. Differences in catheter ablation of paroxysmal atrial fibrillation between males and females. Int J Cardiol 2013;168:1984-91.
- 684. Boriani G, Glotzer TV, Santini M, et al. Device-detected atrial fibrillation and risk for stroke: an analysis of >10,000 patients from the SOS AF project (Stroke preventiOn Strategies based on Atrial Fibrillation information from implanted devices). Eur Heart J 2014;35:508-16.
- 685. Wasserlauf J, Passman RS. When is device-detected atrial fibrillation actionable? Card Electrophysiol Clin 2018;10:75-85.
- 686. Chen LY, Chung MK, Allen LA, et al. Atrial fibrillation burden: moving beyond atrial fibrillation as a binary entity: a scientific statement from the American Heart Association. Circulation 2018;137:e623-44.
- 687. Cheung CC, Krahn AD, Andrade JG. The emerging role of wearable technologies in detection of arrhythmia. Can J Cardiol 2018;34:1083-7.
- 688. O'Sullivan JW, Grigg S, Crawford W, et al. Accuracy of smartphone camera applications for detecting atrial fibrillation: a systematic review and meta-analysis. JAMA Netw Open 2020;3:e202064.
- 689. Jones NR, Taylor CJ, Hobbs FDR, Bowman L, Casadei B. Screening for atrial fibrillation: a call for evidence. Eur Heart J 2020;41:1075-85.
- **690.** Botto GL, Padeletti L, Santini M, et al. Presence and duration of atrial fibrillation detected by continuous monitoring: crucial implications for the risk of thromboembolic events. J Cardiovasc Electrophysiol 2009;20:241-8.
- 691. Kaplan RM, Koehler J, Ziegler PD, et al. Stroke risk as a function of atrial fibrillation duration and CHA2DS2-VASc score. Circulation 2019;140:1639-46.
- **692.** Lipshultz SE, Sleeper LA, Towbin JA, et al. The incidence of pediatric cardiomyopathy in two regions of the United States. N Engl J Med 2003;348:1647-55.
- **693.** Nugent AW, Daubeney PE, Chondros P, et al. Clinical features and outcomes of childhood hypertrophic cardiomyopathy: results from a national population-based study. Circulation 2005;112:1332-8.
- 694. Tian T, Wang Y, Sun K, et al. Clinical profile and prognostic significance of atrial fibrillation in hypertrophic cardiomyopathy. Cardiology 2013;126:258-64.
- **695.** Siontis KC, Geske JB, Ong K, et al. Atrial fibrillation in hypertrophic cardiomyopathy: prevalence, clinical correlations, and mortality in a large high-risk population. J Am Heart Assoc 2014;3:e001002.
- 696. Garg L, Gupta M, Sabzwari SRA, et al. Atrial fibrillation in hypertrophic cardiomyopathy: prevalence, clinical impact, and management. Heart Fail Rev 2019;24:189-97.
- 697. Tendera M, Wycisk A, Schneeweiss A, Polonski L, Wodniecki J. Effect of sotalol on arrhythmias and exercise tolerance in patients with hypertrophic cardiomyopathy. Cardiology 1993;82:335-42.
- **698.** Moore JC, Trager L, Anzia LE, et al. Dofetilide for suppression of atrial fibrillation in hypertrophic cardiomyopathy: a case series and literature review. Pacing Clin Electrophysiol 2018;41:396-401.
- 699. Santangeli P, Di Biase L, Themistoclakis S, et al. Catheter ablation of atrial fibrillation in hypertrophic cardiomyopathy: long-term outcomes and mechanisms of arrhythmia recurrence. Circ Arrhythm Electrophysiol 2013;6:1089-94.

- 700. Hayashi H, Hayashi M, Miyauchi Y, et al. Left atrial wall thickness and outcomes of catheter ablation for atrial fibrillation in patients with hypertrophic cardiomyopathy. J Interv Card Electrophysiol 2014;40: 153-60.
- 701. Ha HS, Wang N, Wong S, et al. Catheter ablation for atrial fibrillation in hypertrophic cardiomyopathy patients: a systematic review. J Interv Card Electrophysiol 2015;44:161-70.
- 702. Providencia R, Elliott P, Patel K, et al. Catheter ablation for atrial fibrillation in hypertrophic cardiomyopathy: a systematic review and meta-analysis. Heart 2016;102:1533-43.
- 703. Prutkin JM, Owens DS. Catheter ablation for atrial fibrillation in patients with hypertrophic cardiomyopathy. Heart 2016;102:1513-4.
- 704. Roh SY, Kim DH, Ahn J, et al. Long-term outcome of catheter ablation for atrial fibrillation in patients with apical hypertrophic cardiomyopathy. J Cardiovasc Electrophysiol 2016;27:788-95.
- 705. Zhao DS, Shen Y, Zhang Q, et al. Outcomes of catheter ablation of atrial fibrillation in patients with hypertrophic cardiomyopathy: a systematic review and meta-analysis. Europace 2016;18:508-20.
- 706. Chen X, Dong JZ, Du X, et al. Long-term outcome of catheter ablation for atrial fibrillation in patients with apical hypertrophic cardiomyopathy. J Cardiovasc Electrophysiol 2018;29:951-7.
- 707. Lapenna E, Pozzoli A, De Bonis M, et al. Mid-term outcomes of concomitant surgical ablation of atrial fibrillation in patients undergoing cardiac surgery for hypertrophic cardiomyopathydagger. Eur J Cardiothorac Surg 2017;51:1112-8.
- 708. Bogachev-Prokophiev AV, Afanasyev AV, Zheleznev SI, et al. Concomitant ablation for atrial fibrillation during septal myectomy in patients with hypertrophic obstructive cardiomyopathy. J Thorac Cardiovasc Surg 2018;155:1536-1542.e1532.
- 709. Elliott AD, Linz D, Verdicchio CV, Sanders P. Exercise and atrial fibrillation: prevention or causation? Heart Lung Circ 2018;27: 1078-85.
- Aizer A, Gaziano JM, Cook NR, et al. Relation of vigorous exercise to risk of atrial fibrillation. Am J Cardiol 2009;103:1572-7.
- 711. Ofman P, Khawaja O, Rahilly-Tierney CR, et al. Regular physical activity and risk of atrial fibrillation: a systematic review and meta-analysis. Circ Arrhythm Electrophysiol 2013;6:252-6.
- Kwok CS, Anderson SG, Myint PK, Mamas MA, Loke YK. Physical activity and incidence of atrial fibrillation: a systematic review and metaanalysis. Int J Cardiol 2014;177:467-76.
- 713. Zhu W, Shen Y, Zhou Q, et al. Association of physical fitness with the risk of atrial fibrillation: a systematic review and meta-analysis. Clin Cardiol 2016;39:421-8.
- 714. Mozaffarian D, Furberg CD, Psaty BM, Siscovick D. Physical activity and incidence of atrial fibrillation in older adults: the cardiovascular health study. Circulation 2008;118:800-7.
- 715. Ricci C, Gervasi F, Gaeta M, et al. Physical activity volume in relation to risk of atrial fibrillation. A non-linear meta-regression analysis. Eur J Prev Cardiol 2018;25:857-66.
- 716. Everett BM, Conen D, Buring JE, et al. Physical activity and the risk of incident atrial fibrillation in women. Circ Cardiovasc Qual Outcomes 2011;4:321-7.
- 717. Svedberg N, Sundstrom J, James S, et al. Long-term incidence of atrial fibrillation and stroke among cross-country skiers. Circulation 2019;140:910-20.

- 718. Raju H, Kalman JM. Management of atrial fibrillation in the athlete. Heart Lung Circ 2018;27:1086-92.
- Koopman P, Nuyens D, Garweg C, et al. Efficacy of radiofrequency catheter ablation in athletes with atrial fibrillation. Europace 2011;13: 1386-93.
- 720. Flannery MD, Kalman JM, Sanders P, La Gerche A. State of the art review: atrial fibrillation in athletes. Heart Lung Circ 2017;26:983-9.
- 721. Heidbuchel H, Anne W, Willems R, et al. Endurance sports is a risk factor for atrial fibrillation after ablation for atrial flutter. Int J Cardiol 2006;107:67-72.
- 722. Khairy P, Balaji S. Cardiac arrhythmias in congenital heart diseases. Indian Pacing Electrophysiol J 2009;9:299-317.
- 723. Somerville J. Grown-up congenital heart disease-medical demands look back, look forward 2000. Thorac Cardiovasc Surg 2001;49:21-6.
- 724. Engelfriet P, Boersma E, Oechslin E, et al. The spectrum of adult congenital heart disease in Europe: morbidity and mortality in a 5 year follow-up period. The Euro Heart Survey on adult congenital heart disease. Eur Heart J 2005;26:2325-33.
- 725. Khairy P, Dore A, Talajic M, et al. Arrhythmias in adult congenital heart disease. Expert Rev Cardiovasc Ther 2006;4:83-95.
- 726. Boineau JP, Schuessler RB, Mooney CR, et al. Natural and evoked atrial flutter due to circus movement in dogs. Role of abnormal atrial pathways, slow conduction, nonuniform refractory period distribution and premature beats. Am J Cardiol 1980;45:1167-81.
- 727. Yamashita T, Inoue H, Nozaki A, Sugimoto T. Role of anatomic architecture in sustained atrial reentry and double potentials. Am Heart J 1992;124:938-46.
- 728. Wan D, Tsui C, Kiess M, et al. Anticoagulation for thromboembolic risk reduction in adults with congenital heart disease. Can J Cardiol 2017;33:1597-603.
- 729. Waldmann V, Laredo M, Abadir S, Mondesert B, Khairy P. Atrial fibrillation in adults with congenital heart disease. Int J Cardiol 2019;287:148-54.
- 730. Khairy P. Thrombosis in congenital heart disease. Expert Rev Cardiovasc Ther 2013;11:1579-82.
- 731. Khairy P, Aboulhosn J, Gurvitz MZ, et al. Arrhythmia burden in adults with surgically repaired tetralogy of Fallot: a multi-institutional study. Circulation 2010;122:868-75.
- 732. Mandalenakis Z, Rosengren A, Lappas G, et al. Atrial fibrillation burden in young patients with congenital heart disease. Circulation 2018;137: 928-37.
- 733. Kirsh JA, Walsh EP, Triedman JK. Prevalence of and risk factors for atrial fibrillation and intra-atrial reentrant tachycardia among patients with congenital heart disease. Am J Cardiol 2002;90:338-40.
- 734. Loomba RS, Buelow MW, Aggarwal S, et al. Arrhythmias in adults with congenital heart disease: what are risk factors for specific arrhythmias? Pacing Clin Electrophysiol 2017;40:353-61.
- 735. Labombarda F, Hamilton R, Shohoudi A, et al. Increasing prevalence of atrial fibrillation and permanent atrial arrhythmias in congenital heart disease. J Am Coll Cardiol 2017;70:857-65.
- 736. Teuwen CP, Ramdjan TT, Gotte M, et al. Time course of atrial fibrillation in patients with congenital heart defects. Circ Arrhythm Electrophysiol 2015;8:1065-72.

- 737. Giamberti A, Pluchinotta FR, Chessa M, et al. Surgery for supraventricular tachycardia and congenital heart defects: long-term efficacy of the combined approach in adult patients. Europace 2017;19:1542-8.
- 738. Hoyer AW, Balaji S. The safety and efficacy of ibutilide in children and in patients with congenital heart disease. Pacing Clin Electrophysiol 2007;30:1003-8.
- 739. Rao SO, Boramanand NK, Burton DA, Perry JC. Atrial tachycardias in young adults and adolescents with congenital heart disease: conversion using single dose oral sotalol. Int J Cardiol 2009;136:253-7.
- 740. El-Assaad I, Al-Kindi SG, Abraham J, et al. Use of dofetilide in adult patients with atrial arrhythmias and congenital heart disease: a PACES collaborative study. Heart Rhythm 2016;13:2034-9.
- 741. Fish FA, Gillette PC, Benson DW Jr. Proarrhythmia, cardiac arrest and death in young patients receiving encainide and flecainide. The Pediatric Electrophysiology Group. J Am Coll Cardiol 1991;18:356-65.
- 742. Koyak Z, Kroon B, de Groot JR, et al. Efficacy of antiarrhythmic drugs in adults with congenital heart disease and supraventricular tachycardias. Am J Cardiol 2013;112:1461-7.
- 743. Thorne SA, Barnes I, Cullinan P, Somerville J. Amiodarone-associated thyroid dysfunction: risk factors in adults with congenital heart disease. Circulation 1999;100:149-54.
- 744. Stan MN, Ammash NM, Warnes CA, et al. Body mass index and the development of amiodarone-induced thyrotoxicosis in adults with congenital heart disease–a cohort study. Int J Cardiol 2013;167:821-6.
- 745. Wells R, Khairy P, Harris L, Anderson CC, Balaji S. Dofetilide for atrial arrhythmias in congenital heart disease: a multicenter study. Pacing Clin Electrophysiol 2009;32:1313-8.
- 746. Philip F, Muhammad KI, Agarwal S, Natale A, Krasuski RA. Pulmonary vein isolation for the treatment of drug-refractory atrial fibrillation in adults with congenital heart disease. Congenit Heart Dis 2012;7: 392-9.
- 747. Sohns C, Nurnberg JH, Hebe J, et al. Catheter ablation for atrial fibrillation in adults with congenital heart disease: lessons learned from more than 10 years following a sequential ablation approach. JACC Clin Electrophysiol 2018;4:733-43.
- 748. Abadir S, Waldmann V, Dyrda K, et al. Feasibility and safety of cryoballoon ablation for atrial fibrillation in patients with congenital heart disease. World J Cardiol 2019;11:149-58.
- 749. Dobrev D, Aguilar M, Heijman J, Guichard JB, Nattel S. Postoperative atrial fibrillation: mechanisms, manifestations and management. Nat Rev Cardiol 2019;16:417-36.
- 750. Wiersinga WM. Propranolol and thyroid hormone metabolism. Thyroid 1991;1:273-7.
- 751. Jabre P, Roger VL, Murad MH, et al. Mortality associated with atrial fibrillation in patients with myocardial infarction. A systematic review and meta-analysis. Circulation 2011;123:1587-93.
- 752. Walkey AJ, Wiener RS, Ghobrial JM, Curtis LH, Benjamin EJ. Incident stroke and mortality associated with new-onset atrial fibrillation in patients hospitalized with severe sepsis. JAMA 2011;306:2248-54.
- 753. Gialdini G, Nearing K, Bhave PD, et al. Perioperative atrial fibrillation and the long-term risk of ischemic stroke. JAMA 2014;312:616-22.
- 754. Matarese A, Sardu C, Shu J, Santulli G. Why is chronic obstructive pulmonary disease linked to atrial fibrillation? A systematic overview of the underlying mechanisms. Int J Cardiol 2019;276:149-51.

Andrade et al. 2020 CCS/CHRS Guidelines on Management of AF

- 755. Kuipers S, Klein Klouwenberg PM, Cremer OL. Incidence, risk factors and outcomes of new-onset atrial fibrillation in patients with sepsis: a systematic review. Crit Care 2014;18:688.
- 756. Walkey AJ, Hammill BG, Curtis LH, Benjamin EJ. Long-term outcomes following development of new-onset atrial fibrillation during sepsis. Chest 2014;146:1187-95.
- 757. Rivero-Ayerza M, Scholte Op Reimer W, Lenzen M, et al. New-onset atrial fibrillation is an independent predictor of in-hospital mortality in hospitalized heart failure patients: results of the EuroHeart Failure Survey. Eur Heart J 2008;29:1618-24.
- 758. Angeli F, Reboldi G, Garofoli M, et al. Atrial fibrillation and mortality in patients with acute myocardial infarction: a systematic overview and meta-analysis. Curr Cardiol Rep 2012;14:601-10.
- 759. Bang CN, Gislason GH, Greve AM, et al. New-onset atrial fibrillation is associated with cardiovascular events leading to death in a first time myocardial infarction population of 89,703 patients with long-term follow-up: a nationwide study. J Am Heart Assoc 2014;3:e000382.
- 760. Topaz G, Flint N, Steinvil A, et al. Long term prognosis of atrial fibrillation in ST-elevation myocardial infarction patients undergoing percutaneous coronary intervention. Int J Cardiol 2017;240:228-33.
- 761. Fuller JA, Adams GG, Buxton B. Atrial fibrillation after coronary artery bypass grafting. Is it a disorder of the elderly? J Thorac Cardiovasc Surg 1989;97:821-5.
- 762. Crosby LH, Pifalo WB, Woll KR, Burkholder JA. Risk factors for atrial fibrillation after coronary artery bypass grafting. Am J Cardiol 1990;66: 1520-2.
- 763. Leitch JW, Thomson D, Baird DK, Harris PJ. The importance of age as a predictor of atrial fibrillation and flutter after coronary artery bypass grafting. J Thorac Cardiovasc Surg 1990;100:338-42.
- 764. Hashimoto K, Ilstrup DM, Schaff HV. Influence of clinical and hemodynamic variables on risk of supraventricular tachycardia after coronary artery bypass. J Thorac Cardiovasc Surg 1991;101:56-65.
- 765. Frost L, Molgaard H, Christiansen EH, et al. Atrial fibrillation and flutter after coronary artery bypass surgery: epidemiology, risk factors and preventive trials. Int J Cardiol 1992;36:253-61.
- 766. Creswell LL, Schuessler RB, Rosenbloom M, Cox JL. Hazards of postoperative atrial arrhythmias. Ann Thorac Surg 1993;56:539-49.
- 767. Aranki SF, Shaw DP, Adams DH, et al. Predictors of atrial fibrillation after coronary artery surgery. Current trends and impact on hospital resources. Circulation 1996;94:390-7.
- 768. Mitchell LB. Incidence, timing and outcome of atrial tachyarrhythmias after cardiac surgery. In: Steinberg JS, ed. Atrial Fibrillation After Cardiac Surgery. Boston: Kluwer Academic Publishers, 2000:37-50.
- 769. Blessberger H, Lewis SR, Pritchard MW, et al. Perioperative betablockers for preventing surgery-related mortality and morbidity in adults undergoing non-cardiac surgery. Cochrane Database Syst Rev 2019;9:CD013438.
- 770. Blessberger H, Lewis SR, Pritchard MW, et al. Perioperative betablockers for preventing surgery-related mortality and morbidity in adults undergoing cardiac surgery. Cochrane Database Syst Rev 2019;9: CD013435.
- 771. Burgess DC, Kilborn MJ, Keech AC. Interventions for prevention of post-operative atrial fibrillation and its complications after cardiac surgery: a meta-analysis. Eur Heart J 2006;27:2846-57.

- 772. Blessberger H, Kammler J, Domanovits H, et al. Perioperative betablockers for preventing surgery-related mortality and morbidity. Cochrane Database Syst Rev 2018;3:CD004476.
- 773. Arsenault KA, Yusuf AM, Crystal E, et al. Interventions for preventing post-operative atrial fibrillation in patients undergoing heart surgery. Cochrane Database Syst Rev 2013;1:CD003611.
- 774. Zhu J, Wang C, Gao D, et al. Meta-analysis of amiodarone versus betablocker as a prophylactic therapy against atrial fibrillation following cardiac surgery. Intern Med J 2012;42:1078-87.
- Osmanovic E, Ostojic M, Avdic S, et al. Pharmacological prophylaxis of atrial fibrillation after surgical myocardial revascularization. Med Arch 2019;73:19-22.
- 776. Chatterjee S, Sardar P, Mukherjee D, Lichstein E, Aikat S. Timing and route of amiodarone for prevention of postoperative atrial fibrillation after cardiac surgery: a network regression meta-analysis. Pacing Clin Electrophysiol 2013;36:1017-23.
- 777. Atreya AR, Priya A, Pack QR, et al. Use and outcomes associated with perioperative amiodarone in cardiac surgery. J Am Heart Assoc 2019;8: e009892.
- Kerin NZ, Jacob S. The efficacy of sotalol in preventing postoperative atrial fibrillation: a meta-analysis. Am J Med 2011;124. 875.e871-9.
- 779. Fairley JL, Zhang L, Glassford NJ, Bellomo R. Magnesium status and magnesium therapy in cardiac surgery: a systematic review and metaanalysis focusing on arrhythmia prevention. J Crit Care 2017;42:69-77.
- 780. Solomon AJ, Berger AK, Trivedi KK, Hannan RL, Katz NM. The combination of propranolol and magnesium does not prevent postoperative atrial fibrillation. Ann Thorac Surg 2000;69:126-9.
- 781. White CM, Caron MF, Kalus JS, et al. Intravenous plus oral amiodarone, atrial septal pacing, or both strategies to prevent postcardiothoracic surgery atrial fibrillation: the Atrial Fibrillation Suppression Trial II (AFIST II). Circulation 2003;108(suppl 1):II200-6.
- 782. Lennerz C, Barman M, Tantawy M, Sopher M, Whittaker P. Colchicine for primary prevention of atrial fibrillation after open-heart surgery: systematic review and meta-analysis. Int J Cardiol 2017;249:127-37.
- 783. Gozdek M, Pawliszak W, Hagner W, et al. Systematic review and metaanalysis of randomized controlled trials assessing safety and efficacy of posterior pericardial drainage in patients undergoing heart surgery. J Thorac Cardiovasc Surg 2017;153:865-875.e812.
- 784. Abouarab AA, Leonard JR, Ohmes LB, et al. Posterior Left pericardiotomy for the prevention of postoperative Atrial fibrillation after Cardiac Surgery (PALACS): study protocol for a randomized controlled trial. Trials 2017;18:593.
- 785. Yuan X, Du J, Liu Q, Zhang L. Defining the role of perioperative statin treatment in patients after cardiac surgery: a meta-analysis and systematic review of 20 randomized controlled trials. Int J Cardiol 2017;228:958-66.
- 786. Shi R, Li ZH, Chen D, et al. Sole and combined vitamin C supplementation can prevent postoperative atrial fibrillation after cardiac surgery: a systematic review and meta-analysis of randomized controlled trials. Clin Cardiol 2018;41:871-8.
- 787. Liu XH, Xu CY, Fan GH. Efficacy of N-acetylcysteine in preventing atrial fibrillation after cardiac surgery: a meta-analysis of published randomized controlled trials. BMC Cardiovasc Disord 2014;14:52.
- 788. Gold MR, O'Gara PT, Buckley MJ, DeSanctis RW. Efficacy and safety of procainamide in preventing arrhythmias after coronary artery bypass surgery. Am J Cardiol 1996;78:975-9.

- 789. Kowey PR, Yannicelli D. Amsterdam E, COPPA-II Investigators. Effectiveness of oral propafenone for the prevention of atrial fibrillation after coronary artery bypass grafting. Am J Cardiol 2004;94:663-5.
- 790. Tagarakis GI, Aidonidis I, Daskalopoulou SS, et al. Effect of ranolazine in preventing postoperative atrial fibrillation in patients undergoing coronary revascularization surgery. Curr Vasc Pharmacol 2013;11: 988-91.
- 791. Horbach SJ, Lopes RD, da C Guaragna JC, et al. Naproxen as prophylaxis against atrial fibrillation after cardiac surgery: the NAFARM randomized trial. Am J Med 2011;124:1036-42.
- 792. Ali-Hassan-Sayegh S, Mirhosseini SJ, Zeriouh M, et al. Safety and efficacy of glucose-insulin-potassium treatment in coronary artery bypass graft surgery and percutaneous coronary intervention. Interact Cardiovasc Thorac Surg 2015;21:667-76.
- 793. Wang H, Chen J, Zhao L. N-3 polyunsaturated fatty acids for prevention of postoperative atrial fibrillation: updated meta-analysis and systematic review. J Interv Card Electrophysiol 2018;51:105-15.
- 794. Dunser MW, Bouvet O, Knotzer H, et al. Vasopressin in cardiac surgery: a meta-analysis of randomized controlled trials. J Cardiothorac Vasc Anesth 2018;32:2225-32.
- 795. Chen S, Acou WJ, Kiuchi MG, et al. Association of preoperative reninangiotensin system inhibitors with prevention of postoperative atrial fibrillation and adverse events: a systematic review and meta-analysis. JAMA Netw Open 2019;2:e194934.
- 796. Whitlock RP, Dieleman JM, Belley-Cote E, et al. The effect of steroids in patients undergoing cardiopulmonary bypass: an individual patient meta-analysis of two randomized trials. J Cardiothorac Vasc Anesth 2020;34:99-105.
- 797. Pokushalov E, Kozlov B, Romanov A, et al. Botulinum toxin injection in epicardial fat pads can prevent recurrences of atrial fibrillation after cardiac surgery: results of a randomized pilot study. J Am Coll Cardiol 2014;64:628-9.
- 798. Danias PG, Caulfield TA, Weigner MJ, Silverman DI, Manning WJ. Likelihood of spontaneous conversion of atrial fibrillation to sinus rhythm. J Am Coll Cardiol 1998;31:588-92.
- 799. Lee JK, Klein GJ, Krahn AD, et al. Rate-control versus conversion strategy in postoperative atrial fibrillation: a prospective, randomized pilot study. Am Heart J 2000;140:871-7.
- 800. Kowey PR, Stebbins D, Igidbashian L, et al. Clinical outcome of patients who develop PAF after CABG surgery. Pacing Clin Electrophysiol 2001;24:191-3.
- 801. Taylor GJ, Malik SA, Colliver JA, et al. Usefulness of atrial fibrillation as a predictor of stroke after isolated coronary artery bypass grafting. Am J Cardiol 1987;60:905-7.
- 802. Lynn GM, Stefanko K, Reed JF 3rd, Gee W, Nicholas G. Risk factors for stroke after coronary artery bypass. J Thorac Cardiovasc Surg 1992;104:1518-23.
- 803. Stanley TO, Mackensen GB, Grocott HP, et al. The impact of postoperative atrial fibrillation on neurocognitive outcome after coronary artery bypass graft surgery. Anesth Analg 2002;94:290-5. table of contents.
- 804. Meurin P, Weber H, Renaud N, et al. Evolution of the postoperative pericardial effusion after day 15: the problem of the late tamponade. Chest 2004;125:2182-7.
- 805. Gillinov AM, Bagiella E, Moskowitz AJ, et al. Rate control versus rhythm control for atrial fibrillation after cardiac surgery. N Engl J Med 2016;374:1911-21.

- 806. Ahlsson A, Fengsrud E, Bodin L, Englund A. Postoperative atrial fibrillation in patients undergoing aortocoronary bypass surgery carries an eightfold risk of future atrial fibrillation and a doubled cardiovascular mortality. Eur J Cardiothorac Surg 2010;37:1353-9.
- 807. Thoren E, Wernroth ML, Christersson C, et al. Compared with matched controls, patients with postoperative atrial fibrillation (POAF) have increased long-term AF after CABG, and POAF is further associated with increased ischemic stroke, heart failure and mortality even after adjustment for AF. Clin Res Cardiol 2020;109:1232-42.
- **808.** Phan K, Ha HS, Phan S, et al. New-onset atrial fibrillation following coronary bypass surgery predicts long-term mortality: a systematic review and meta-analysis. Eur J Cardiothorac Surg 2015;48:817-24.
- 809. Kosmidou I, Chen S, Kappetein AP, et al. New-onset atrial fibrillation after PCI or CABG for left main disease: the EXCEL trial. J Am Coll Cardiol 2018;71:739-48.
- 810. Lowres N, Mulcahy G, Jin K, et al. Incidence of postoperative atrial fibrillation recurrence in patients discharged in sinus rhythm after cardiac surgery: a systematic review and meta-analysis. Interact Cardiovasc Thorac Surg 2018;26:504-11.
- 811. Ozcan C, Strom JB, Newell JB, Mansour MC, Ruskin JN. Incidence and predictors of atrial fibrillation and its impact on long-term survival in patients with supraventricular arrhythmias. Europace 2014;16: 1508-14.
- 812. Sauer WH, Alonso C, Zado E, et al. Atrioventricular nodal reentrant tachycardia in patients referred for atrial fibrillation ablation: response to ablation that incorporates slow-pathway modification. Circulation 2006;114:191-5.
- **813.** Katritsis DG, Giazitzoglou E, Wood MA, et al. Inducible supraventricular tachycardias in patients referred for catheter ablation of atrial fibrillation. Europace 2007;9:785-9.
- 814. Sciarra L, Rebecchi M, De Ruvo E, et al. How many atrial fibrillation ablation candidates have an underlying supraventricular tachycardia previously unknown? Efficacy of isolated triggering arrhythmia ablation. Europace 2010;12:1707-12.
- Gonzalez-Zuelgaray J, Perez A. Regular supraventricular tachycardias associated with idiopathic atrial fibrillation. Am J Cardiol 2006;98: 1242-4.
- 816. Wutzler A, von Ulmenstein S, Attanasio P, et al. Where there's smoke, there's fire? Significance of atrial fibrillation in young patients. Clin Cardiol 2016;39:229-33.
- 817. Khachab H, Brembilla-Perrot B. Prevalence of atrial fibrillation in patients with history of paroxysmal supraventricular tachycardia. Int J Cardiol 2013;166:221-4.
- Johnson LS, Juhlin T, Juul-Moller S, et al. A prospective study of supraventricular activity and incidence of atrial fibrillation. Heart Rhythm 2015;12:1898-904.
- 819. Etheridge SP, Escudero CA, Blaufox AD, et al. Life-threatening event risk in children with Wolff-Parkinson-White syndrome: a multicenter international study. JACC Clin Electrophysiol 2018;4:433-44.
- 820. Munger TM, Packer DL, Hammill SC, et al. A population study of the natural history of Wolff-Parkinson-White syndrome in Olmsted County, Minnesota, 1953-1989. Circulation 1993;87:866-73.
- 821. Pappone C, Vicedomini G, Manguso F, et al. Wolff-Parkinson-White syndrome in the era of catheter ablation: insights from a registry study of 2169 patients. Circulation 2014;130:811-9.
- 822. Roth A, Elkayam I, Shapira I, et al. Effectiveness of prehospital synchronous direct-current cardioversion for supraventricular

tachyarrhythmias causing unstable hemodynamic states. Am J Cardiol 2003;91:489-91.

- 823. Neumar RW, Otto CW, Link MS, et al. Part 8: adult advanced cardiovascular life support: 2010 American Heart Association guidelines for cardiopulmonary resuscitation and emergency cardiovascular care. Circulation 2010;122:S729-67.
- 824. Sellers TD Jr, Campbell RW, Bashore TM, Gallagher JJ. Effects of procainamide and quinidine sulfate in the Wolff-Parkinson-White syndrome. Circulation 1977;55:15-22.
- 825. Glatter KA, Dorostkar PC, Yang Y, et al. Electrophysiological effects of ibutilide in patients with accessory pathways. Circulation 2001;104: 1933-9.
- 826. Sellers TD Jr, Bashore TM, Gallagher JJ. Digitalis in the pre-excitation syndrome. Analysis during atrial fibrillation. Circulation 1977;56: 260-7.
- 827. Morady F, DiCarlo LA Jr, Baerman JM, De Buitleir M. Effect of propranolol on ventricular rate during atrial fibrillation in the Wolff-Parkinson-White syndrome. Pacing Clin Electrophysiol 1987;10:492-6.
- 828. Schutzenberger W, Leisch F, Gmeiner R. Enhanced accessory pathway conduction following intravenous amiodarone in atrial fibrillation. A case report. Int J Cardiol 1987;16:93-5.
- 829. Boriani G, Biffi M, Frabetti L, et al. Ventricular fibrillation after intravenous amiodarone in Wolff-Parkinson-White syndrome with atrial fibrillation. Am Heart J 1996;131:1214-6.
- 830. Calkins H, Yong P, Miller JM, et al. Catheter ablation of accessory pathways, atrioventricular nodal reentrant tachycardia, and the atrioventricular junction: final results of a prospective, multicenter clinical trial. The Atakr Multicenter Investigators Group. Circulation 1999;99: 262-70.
- Klein GJ, Bashore TM, Sellers TD, et al. Ventricular fibrillation in the Wolff-Parkinson-White syndrome. N Engl J Med 1979;301:1080-5.
- 832. Rinne C, Klein GJ, Sharma AD, et al. Relation between clinical presentation and induced arrhythmias in the Wolff-Parkinson-White syndrome. Am J Cardiol 1987;60:576-9.
- 833. Sharma AD, Yee R, Guiraudon G, Klein GJ. Sensitivity and specificity of invasive and noninvasive testing for risk of sudden death in Wolff-Parkinson-White syndrome. J Am Coll Cardiol 1987;10:373-81.
- 834. Conen D, Tedrow UB, Koplan BA, et al. Influence of systolic and diastolic blood pressure on the risk of incident atrial fibrillation in women. Circulation 2009;119:2146-52.
- **835.** Watanabe H, Tanabe N, Watanabe T, et al. Metabolic syndrome and risk of development of atrial fibrillation: the Niigata preventive medicine study. Circulation 2008;117:1255-60.
- 836. Dublin S, Glazer NL, Smith NL, et al. Diabetes mellitus, glycemic control, and risk of atrial fibrillation. J Gen Intern Med 2010;25:853-8.
- 837. Psaty BM, Manolio TA, Kuller LH, et al. Incidence of and risk factors for atrial fibrillation in older adults. Circulation 1997;96:2455-61.
- 838. Huxley RR, Alonso A, Lopez FL, et al. Type 2 diabetes, glucose homeostasis and incident atrial fibrillation: the Atherosclerosis Risk in Communities study. Heart 2012;98:133-8.
- 839. Latini R, Staszewsky L, Sun JL, et al. Incidence of atrial fibrillation in a population with impaired glucose tolerance: the contribution of glucose metabolism and other risk factors. A post hoc analysis of the Nateglinide and Valsartan in Impaired Glucose Tolerance Outcomes Research trial. Am Heart J 2013;166:935-940.e931.

- 840. Chang SH, Wu LS, Chiou MJ, et al. Association of metformin with lower atrial fibrillation risk among patients with type 2 diabetes mellitus: a population-based dynamic cohort and in vitro studies. Cardiovasc Diabetol 2014;13:123.
- 841. Fatemi O, Yuriditsky E, Tsioufis C, et al. Impact of intensive glycemic control on the incidence of atrial fibrillation and associated cardiovascular outcomes in patients with type 2 diabetes mellitus (from the Action to Control Cardiovascular Risk in Diabetes Study). Am J Cardiol 2014;114:1217-22.
- 842. Heeringa J, Kors JA, Hofman A, van Rooij FJ, Witteman JC. Cigarette smoking and risk of atrial fibrillation: the Rotterdam Study. Am Heart J 2008;156:1163-9.
- 843. Chamberlain AM, Agarwal SK, Folsom AR, et al. Smoking and incidence of atrial fibrillation: results from the Atherosclerosis Risk in Communities (ARIC) study. Heart Rhythm 2011;8:1160-6.
- 844. Suzuki S, Otsuka T, Sagara K, et al. Association between smoking habits and the first-time appearance of atrial fibrillation in Japanese patients: evidence from the Shinken Database. J Cardiol 2015;66:73-9.
- 845. Smit MD, Maass AH, De Jong AM, et al. Role of inflammation in early atrial fibrillation recurrence. Europace 2012;14:810-7.
- 846. Hergens MP, Galanti R, Hansson J, et al. Use of Scandinavian moist smokeless tobacco (snus) and the risk of atrial fibrillation. Epidemiology 2014;25:872-6.
- 847. Liang Y, Mente A, Yusuf S, et al. Alcohol consumption and the risk of incident atrial fibrillation among people with cardiovascular disease. CMAJ 2012;184:E857-66.
- 848. Tonelo D, Providencia R, Goncalves L. Holiday heart syndrome revisited after 34 years. Arq Bras Cardiol 2013;101:183-9.
- 849. Larsson SC, Drca N, Wolk A. Alcohol consumption and risk of atrial fibrillation: a prospective study and dose-response meta-analysis. J Am Coll Cardiol 2014;64:281-9.
- 850. Mukamal KJ, Tolstrup JS, Friberg J, Jensen G, Gronbaek M. Alcohol consumption and risk of atrial fibrillation in men and women: the Copenhagen City Heart study. Circulation 2005;112:1736-42.
- 851. Ruigomez A, Johansson S, Wallander MA, Garcia Rodriguez LA. Predictors and prognosis of paroxysmal atrial fibrillation in general practice in the UK. BMC Cardiovasc Disord 2005;5:20.
- 852. Mont L, Sambola A, Brugada J, et al. Long-lasting sport practice and lone atrial fibrillation. Eur Heart J 2002;23:477-82.
- 853. Frost L, Hune LJ, Vestergaard P. Overweight and obesity as risk factors for atrial fibrillation or flutter: the Danish Diet, Cancer, and Health Study. Am J Med 2005;118:489-95.
- 854. Overvad TF, Rasmussen LH, Skjoth F, et al. Body mass index and adverse events in patients with incident atrial fibrillation. Am J Med 2013;126. 640.e9-17.
- **855.** Gami AS, Pressman G, Caples SM, et al. Association of atrial fibrillation and obstructive sleep apnea. Circulation 2004;110:364-7.
- 856. Mehra R, Benjamin EJ, Shahar E, et al. Association of nocturnal arrhythmias with sleep-disordered breathing: the Sleep Heart Health study. Am J Respir Crit Care Medi 2006;173:910-6.
- 857. Kanagala R, Murali NS, Friedman PA, et al. Obstructive sleep apnea and the recurrence of atrial fibrillation. Circulation 2003;107:2589-94.
- **858.** Naruse Y, Tada H, Satoh M, et al. Concomitant obstructive sleep apnea increases the recurrence of atrial fibrillation following radiofrequency catheter ablation of atrial fibrillation: clinical impact of continuous positive airway pressure therapy. Heart Rhythm 2013;10:331-7.

- 859. Lopez FL, Agarwal SK, Maclehose RF, et al. Blood lipid levels, lipidlowering medications, and the incidence of atrial fibrillation: the Atherosclerosis Risk in Communities study. Circ Arrhythm Electrophysiol 2012;5:155-62.
- 860. Alonso A, Yin X, Roetker NS, et al. Blood lipids and the incidence of atrial fibrillation: the Multi-Ethnic Study of Atherosclerosis and the Framingham Heart study. J Am Heart Assoc 2014;3:e001211.
- 861. Heeringa J, van der Kuip DA, Hofman A, et al. Prevalence, incidence and lifetime risk of atrial fibrillation: the Rotterdam study. Eur Heart J 2006;27:949-53.
- 862. Stortecky S, Buellesfeld L, Wenaweser P, et al. Atrial fibrillation and aortic stenosis: impact on clinical outcomes among patients undergoing transcatheter aortic valve implantation. Circ Cardiovasc Interv 2013;6: 77-84.
- 863. Diker E, Aydogdu S, Ozdemir M, et al. Prevalence and predictors of atrial fibrillation in rheumatic valvular heart disease. Am J Cardiol 1996;77:96-8.
- 864. Maisel WH, Stevenson LW. Atrial fibrillation in heart failure: epidemiology, pathophysiology, and rationale for therapy. Am J Cardiol 2003;91:2D-8D.
- 865. Sartipy U, Dahlstrom U, Fu M, Lund LH. Atrial fibrillation in heart failure with preserved, mid-range, and reduced ejection fraction. JACC Heart Fail 2017;5:565-74.
- 866. Pellicori P, Urbinati A, Kaur K, et al. Prevalence and incidence of atrial fibrillation in ambulatory patients with heart failure. Am J Cardiol 2019;124:1554-60.
- **867.** Auer J, Scheibner P, Mische T, et al. Subclinical hyperthyroidism as a risk factor for atrial fibrillation. Am Heart J 2001;142:838-42.
- 868. Selmer C, Olesen JB, Hansen ML, et al. The spectrum of thyroid disease and risk of new onset atrial fibrillation: a large population cohort study. BMJ 2012;345:e7895.
- 869. Christophersen IE, Ravn LS, Budtz-Joergensen E, et al. Familial aggregation of atrial fibrillation: a study in Danish twins. Circ Arrhythm Electrophysiol 2009;2:378-83.
- 870. Olson TM, Michels VV, Ballew JD, et al. Sodium channel mutations and susceptibility to heart failure and atrial fibrillation. JAMA 2005;293:447-54.
- 871. Chen YH, Xu SJ, Bendahhou S, et al. KCNQ1 gain-of-function mutation in familial atrial fibrillation. Science 2003;299:251-4.
- 872. Gollob MH, Jones DL, Krahn AD, et al. Somatic mutations in the connexin 40 gene (GJA5) in atrial fibrillation. N Engl J Med 2006;354: 2677-88.
- 873. Olson TM, Alekseev AE, Moreau C, et al. KATP channel mutation confers risk for vein of Marshall adrenergic atrial fibrillation. Nat Clin Pract Cardiovasc Med 2007;4:110-6.

- 874. Lubitz SA, Sinner MF, Lunetta KL, et al. Independent susceptibility markers for atrial fibrillation on chromosome 4q25. Circulation 2010;122:976-84.
- 875. Chinchilla A, Daimi H, Lozano-Velasco E, et al. PITX2 insufficiency leads to atrial electrical and structural remodeling linked to arrhythmogenesis. Circ Cardiovasc Genet 2011;4:269-79.
- 876. Brazier JE, Harper R, Jones NM, et al. Validating the SF-36 health survey questionnaire: new outcome measure for primary care. BMJ 1992;305:160-4.
- 877. Rabin R, de Charro F. EQ-5D: a measure of health status from the EuroQol Group. Ann Med 2001;33:337-43.
- 878. Spertus J, Dorian P, Bubien R, et al. Development and validation of the Atrial Fibrillation Effect on QualiTy-of-Life (AFEQT) questionnaire in patients with atrial fibrillation. Circ Arrhythm Electrophysiol 2011;4: 15-25.
- 879. Badia X, Arribas F, Ormaetxe JM, Peinado R, de Los Terreros MS. Development of a questionnaire to measure health-related quality of life (HRQoL) in patients with atrial fibrillation (AF-QoL). Health Qual Life Outcomes 2007;5:37.
- 880. Dorian P, Paquette M, Newman D, et al. Quality of life improves with treatment in the Canadian Trial of Atrial Fibrillation. Am Heart J 2002;143:984-90.
- 881. Koci F, Forbes P, Mansour MC, et al. New classification scheme for atrial fibrillation symptom severity and burden. Am J Cardiol 2014;114: 260-5.
- 882. Kirchhof P, Auricchio A, Bax J, et al. Outcome parameters for trials in atrial fibrillation: recommendations from a consensus conference organized by the German Atrial Fibrillation Competence NETwork and the European Heart Rhythm Association. Europace 2007;9:1006-23.
- 883. Mokkink LB, Prinsen CA, Bouter LM, Vet HC, Terwee CB. The COnsensus-based Standards for the selection of health Measurement INstruments (COSMIN) and how to select an outcome measurement instrument. Braz J Phys Ther 2016;20:105-13.
- 884. Guimarães HP, Lopes RD, de Barros E, Silva PGM, et al. RIVER Trial Investigators. Rivaroxaban in patients with atrial fibrillation and a bioprosthetic mitral valve. N Engl J Med 2020;383: 2117-26.
- 885. De Vriese AS, Caluwé R, Pyfferoen L, et al. Multicenter randomized controlled trial of vitamin K antagonist replacement by rivaroxaban with or without vitamin K2 in hemodialysis patients with atrial fibrillation: the Valkyrie Study. J Am Soc Nephrol 2020;31:186-96.

Supplementary Material

To access the supplementary material accompanying this article, visit the online version of the *Canadian Journal of Cardiology* at www.onlinecjc.ca and at https://doi.org/10.1016/j.cjca.2020.09.001.